

# Pre-service teacher self-efficacy and teacher efficacy in a challenged education context

by

**Carine Jonker** 

## Submitted in partial fulfilment of the requirements for the degree

PHILOSOPHIAE DOCTOR

in the Faculty of Education

at the

## **UNIVERSITY OF PRETORIA**

Supervisor: Prof. Liesel Ebersöhn Co-supervisor: Prof. Marien Graham Co-supervisor: Prof. Surette van Staden

**SEPTEMBER 2021** 



## Declaration

I, Carine Jonker (student number 04363655), hereby declare that this thesis titled: *Pre-service teacher resilience and efficacy in a challenged education context,* which I hereby submit for the degree Philosophiae Doctor in Educational Psychology at the University of Pretoria, is my own work and has not been previously submitted by me for a degree at this or any other tertiary institution. All resources, quotations and citations from literature have been acknowledged intext and referenced in full.

Carine Jonker

September 2021



## **Ethical Clearance Certificate**



RESEARCH ETHICS COMMITTEE

CLEARANCE CERTIFICATE	CLEARANCE NUMBER:	EDU089/19
DEGREE AND PROJECT	PhD	
	Pre-service teacher self-e efficacy in a challenged ed	fficacy and teacher ducation context
INVESTIGATOR	Ms Carine-Mari Jonker	
DEPARTMENT	Educational Psychology	
APPROVAL TO COMMENCE STUDY	25 November 2019	
DATE OF CLEARANCE CERTIFICATE	25 August 2021	

CHAIRPERSON OF ETHICS COMMITTEE: Prof Funke Omidire

AA

СС

Ms Thandi Mngomezulu Prof Liesel Ebersöhn Prof Marien Graham Prof Surette van Standen

This Ethics Clearance Certificate should be read in conjunction with the Integrated Declaration Form (D08) which specifies details regarding:

- Compliance with approved research protocol,
- No significant changes,
- Informed consent/assent,
- Adverse experience or undue risk,
- Registered title, and
- Data storage requirements.

---000----



## **Ethics Statement**

The author, whose name appears on the title page of this thesis, has obtained, for the research described in this work, the applicable research ethics approval. The author declares that she has observed the ethical standards required in terms of the University of Pretoria's *Code of ethics for researchers and the Policy guidelines for responsible research*.



## Dedication

AAN MY PAPPA ... MY HERO, ROLMODEL EN STANDVASTIGE ROTS



## Acknowledgements

First and foremost, I thank my heavenly Father for His provision, '*My grace is sufficient for you, for My power is made perfect in weakness*' (2 Corinthians 12:9). Nothing is possible without Him and the grace He has bestowed upon me by blessing me with an abundance of contextual resources. I, therefore, which to express my sincerest gratitude and appreciation to the following significant individuals and institutions without whom the completion of this thesis would not have been feasible:

- My supervisor, Prof Liesel Ebersöhn (Director: The Centre for the Study of Resilience, UP) for your exceptional expertise, wisdom and big-picture view. I have always stood in awe of your professional and personal capacity, which enables others. Thank you for pushing my limits further than I thought was possible. It has been such a privilege to undertake this journey with you.
- My co-supervisor and statistician, Prof. Marien Graham (Department of Science, Mathematics and Technology Education [SMTE], UP) for your invaluable input, detailed eye, and statistical passion. Your phenomenal statistical guidance, especially in the technical aspects of the analysed data, was irreplaceable. I have grown immensely in the quantitative and statistical research field due to your expertise and ever willingness to be of assistance.
- My co-supervisor, Prof. Surette van Staden (Director: The Centre for Evaluation and Assessment [CEA], UP) for your insightful guidance and extraordinary educational knowledge. Your calm composure, time, patience and undoubting belief guided me through the research minefield and meaning-making process for which I am forever grateful.
- My incredible parents, Willie and Carol, for your ceaseless support, motivation and encouragement. All my undeserved accomplishments are due to your example, belief and unconditional love. To the best sister in the world, Yvonne, brother-in-law, Dirk and the little ones (Erich, Imke and Rikus), for keeping me sane throughout this journey. Thank you for giving me perspective and having patience with my 'always working'. I can only thank God for blessing me with the most amazing family. Words cannot express my deepest gratitude for your understanding, consistent motivation and continued provision.
- My extended family, relatives, colleagues and friends for your encouragement, motivation and support throughout a difficult journey. You are invaluable in my life.
- The FIRE research project for allowing me to utilise the data generated in the original project.



- The Centre for the Study of Resilience (CSR) for permissions to use and access the resilience-related existing data generated during the FIRE project in collaboration with the CSR.
- Prof Billy Fraser (late FIRE project PI), Prof Marien Graham (statistician on the FIRE project) and Prof Ronél Ferreira (current custodian of the FIRE dataset) for permission to access the ethical application as well as the documentation of data for the FIRE project.
- Professor Caroline Mansfield (ENTREE PI) for your time and willingness to enhance the rigour of the current study by accommodating my life-changing visit at Notre Dame University (Australia).
- The Teacher Resilience Lab group, especially my PhD-colleagues (Liz-Marie Basson and Jessica Versfeld) who enriched my journey.
- The management staff at the University of Pretoria, the UP-Bursary Funding office, the Faculty of Education, Department of Educational Psychology and the Groenkloof Department of Library Service. The exceptional assistance and financial support provided enabled the completion of the thesis.
- Mrs. Anetha de Wet for your exemplary work in language and technical editing. Your expertise, time, dedication and passion for language was indispensable.
- Ms. Estelle Botha for your endless patience, assistance and magnificent attention to detail as technical editor. There can only be one 'Kaptein'. I am forever in your dept.



### Abstract

## Pre-service teacher self-efficacy and teacher efficacy in a challenged education context

Given the significant gaps in quantitative teacher resilience knowledge, specifically preservice teachers, in a challenged context the need for an evidence-based theoretical framework for teacher resilience in spaces of high challenge is evident. Thus, the purpose of this comparative secondary analysis study was to inform knowledge on teacher resilience in challenged contexts. Intrapersonal resilience-enabling pathways (i.e., self-efficacy and teacher efficacy) of pre-service teachers were compared by employing the Social Cognitive Theory as theoretical framework.

I adopted a post-positivist, quantitative research approach with a comparative case study design to compare pre-service teachers' self-efficacy and teacher efficacy beliefs as intrapersonal resilience-enabling pathways to teacher resilience. The study purposively selected extant data from the FIRE project (2015–2017). The FIRE Project included preservice teachers (N = 1,193) in their final teacher training year at the University of Pretoria. The extant data (completed FIRE Teacher Resilience Measure) was analysed to quantitatively compare the self-efficacy and teacher efficacy beliefs of pre-service teachers within a challenged context. A further objective was to analyse the self-efficacy and teacher efficacy beliefs of pre-service teachers based on demographic information. Finally, statistical procedures were performed on data to compare the relationship between self-efficacy and teacher efficacy beliefs among pre-service teachers in a challenged context. Data was analysed using SPSS by establishing reliability, validity, and statistical computing power, descriptive and inferential statistics.

The within-case and cross-case results indicated the presence of high intrapersonal resilience-enabling pathways in pre-service teachers despite contextual constraints, with a statistically significant relationship between the self-efficacy and teacher efficacy of final year pre-service teachers. Varied results were obtained for gender and enrolled pre-service teaching programmes.

This study advanced knowledge on teacher resilience in a challenged context. Selfefficacy and teacher efficacy may enable pre-service teachers to resile despite chronic and cumulative risk factors. This study (i) contributed quality, quantitatively derived teacher resilience findings from an often-under-represented Global South perspective; (ii) validated the use of a globally used teacher resilience measure in South Africa; and (iii) culminated in results that may be compared to that of others worldwide measured with the same instrument.



#### Key Words:

Teacher resilience, intrapersonal protective resources, resilience-enabling, self-efficacy, teacher efficacy, Global South, challenged context, Teacher Resilience Measurement, quantitative methodology, comparative, secondary data analysis, instrument validation.



#### Language Editor



PO Box 3172 Lyttelton South 0176 13 September 2021

#### TO WHOM IT MAY CONCERN

The thesis titled "Pre-service teacher self-efficacy and teacher efficacy in a challenged education context" by Carine Jonker has been proofread and edited for language by me.

I verify that it is ready for publication or public viewing in respect of language and style, and it has been formatted as per the institution's prescribed style.

Please note that no view is expressed regarding the document's subject-specific technical content or changes after this letter's date.

Kind regards

mdelikt

Anna M de Wet

SATI MEMBER 1003422

BA (Afrikaans, English, Classical Languages) (Cum Laude), University of Pretoria.BA Hons ((Latin) (Cum Laude), University of Pretoria.BA Hons (Psychology), University of Pretoria.

---000----



## List of Abbreviations

α	Alpha (level of significance)
ALTC	Australian Learning and Teaching Council Limited
ANA	Annual National Assessment
APA	American Psychological Association
BEd	Baccalaureus Educationis
BRiTE	Building Resilience in Teacher Education
CDE	Centre for Development and Enterprise
CFA	Confirmatory factor analysis
CSR	Centre for the Study of Resilience
DBE	Department of Basic Education
DHET	Department of Higher Education and Training
diff	Difference
ECD	Early childhood development
ENTREE	ENhancing Teacher REsilience in Europe
FET	Further education and training
FIRE	Fourth Year Intervention in Research
FIT-Choice	Factors Influencing Teaching Choice
FP	Foundation phase
GDP	Gross domestic product
GEMS	Government Employees Medical Scheme
GET	General education and training
GNP	Gross national product
Grade R	Reception year or year prior to Grade 1
H <sub>0</sub>	Null hypothesis
Ha	Alternative hypothesis
HE	Higher education
HEDA	Higher Education Data Analyzer
HEI	Higher Education Institution
HIV	Human immunodeficiency virus
IEA	International Association for the Evaluation of Educational Achievement
IMF	International Monetary Fund
IP	Intermediate phase
IQMS	Integrated Quality Management System



ISCED	International Standard Classification of Education
KW	Kruskal-Wallis test
LOC	Locus of Control
Μ	Mean
Max	Maximum
Mdn	Median
Min	Minimum
MTRS	Multidimensional Teacher Resilience Scale
MW	Mann-Whitney
n	Sample size (subsample)
Ν	Total sample size
NCES	National Center for Education Statistics
NQF	National Qualifications Framework
NSC	National Senior Certificate
OECD	Organisation for Economic Co-operation and Development
ρ	<i>p</i> -value
PGCE	Postgraduate Certificate in Education
PIRLS	Progress in International Reading Literacy Study
PISA	Programme for International Student Assessment
RCT	Relational-cultural theory
RRR	Relationship-Resourced Resilience
r <sub>s</sub>	Spearman rank-order correlation
RSA	Republic of South Africa
SA	South Africa
SACE	South African Council for Educators
SAQA	South African Qualifications Authority
SASA	South African Schools Act
SCCT	Social Cognitive Career Theory
SCT	Social Cognitive Theory
SD	Standard deviation
SDGs	Sustainable Developmental Goals
SOC	Sense of coherence
SP	Senior phase
SPSS	Statistical Package for the Social Sciences
SSA	Sub-Saharan African



Stats SA	Statistics South Africa
TALIS	Teaching and Learning International Survey
TeachEff	Teacher Efficacy
TIMSS	Trends in International Mathematics and Science Study
TR-Emot	Teacher Emotion
TR-Mot	Teacher Motivation
TR-Prof	Teacher Professionalism
TR-Soc	Teacher Social Capacity
U	Mann-Whitney test statistic
UIS	UNESCO Institute for Statistics
UN	United Nations
UNESCO	United Nations Educational, Scientific and Cultural Organization
UNICEF	United Nations Children's Fund
UP	University of Pretoria
Wits	University of the Witwatersrand
WSR	Wilcoxon signed-rank test



## **Table of Contents**

#### Page

Declaration	i
Ethical Clearance Certificate	ii
Ethics Statement	iii
Dedication	iv
Acknowledgements	v
Abstract	vii
Language Editor	ix
List of Abbreviations	x
Table of Contents	xiii
List of Figures	xxii
List of Tables	xxiv

---000----

#### Page

## Chapter 1 Introduction and Overview of the Study

1.1		. 1
1.2	INTRODUCING AND CONTEXTUALISING THE CURRENT STUDY	. 2
1.3	RATIONALE OF THE STUDY	. 4
1.3.1		. 4
1.3.2	TEACHER RESILIENCE AS AN EMERGENT RESEARCH FIELD	. 5
1.3.3	EVIDENCE-BASED RESEARCH FOR TEACHER RESILIENCE IN A CHALLENGED CONTEXT	. 7
1.4	PURPOSE OF THE STUDY	12
1.5	RESEARCH QUESTIONS AND HYPOTHESES FOR THE STUDY	14
1.6		15
<b>1.6</b> 1.6.1	CONCEPT CLARIFICATION	15 15
<b>1.6</b> 1.6.1 1.6.2	CONCEPT CLARIFICATION	15 15 15
<b>1.6</b> 1.6.1 1.6.2 1.6.3	CONCEPT CLARIFICATION	15 15 15 16
<b>1.6</b> 1.6.1 1.6.2 1.6.3 1.6.4	CONCEPT CLARIFICATION	15 15 15 16 18
<b>1.6</b> 1.6.1 1.6.2 1.6.3 1.6.4 1.6.5	CONCEPT CLARIFICATION	15 15 15 16 18 19
<ol> <li>1.6.1</li> <li>1.6.2</li> <li>1.6.3</li> <li>1.6.4</li> <li>1.6.5</li> <li>1.6.6</li> </ol>	CONCEPT CLARIFICATION	15 15 16 18 19 20



168		
1.0.0	INITIAL TEACHER EDUCATION AND SOUTH AFRICAN PROGRAMME PHASES	
1.6.9	OPERATIONALISING PRE-SERVICE TEACHERS	24
1.6.10	CHALLENGED EDUCATION CONTEXT	27
17	THEORETICAL FRAMEWORK	28
171		
1.7.1		
1.7.2	SOCIAL COGNITIVE THEORY: BASIC ASSUMPTIONS	
1.7.3	SELF-EFFICACY THEORY	30
1.8	BRIEF OVERVIEW OF PARADIGMS, METHODS, STANDARD OF RI	GOUR AND
	ETHICAL CONSIDERATIONS	33
1.9		
	SUMMART OF THE RESEARCH FINDINGS	35
1.10	OUTLINE OF CHAPTERS	
<b>1.10</b> 1.10.1	OUTLINE OF CHAPTERS	35 
<b>1.10</b> 1.10.1 1.10.2	OUTLINE OF CHAPTERS. INTRODUCTION CHAPTER 1: INTRODUCTION AND OVERVIEW OF THE STUDY	35 38 38 39
<b>1.10</b> 1.10.1 1.10.2 1.10.3	OUTLINE OF CHAPTERS. INTRODUCTION CHAPTER 1: INTRODUCTION AND OVERVIEW OF THE STUDY CHAPTER 2: LITERATURE REVIEW.	
<b>1.10</b> 1.10.1 1.10.2 1.10.3 1.10.4	OUTLINE OF CHAPTERS. INTRODUCTION CHAPTER 1: INTRODUCTION AND OVERVIEW OF THE STUDY CHAPTER 2: LITERATURE REVIEW CHAPTER 3: RESEARCH DESIGN, METHODOLOGY AND STRATEGIES.	
<b>1.10</b> 1.10.1 1.10.2 1.10.3 1.10.4 1.10.5	OUTLINE OF CHAPTERS. INTRODUCTION CHAPTER 1: INTRODUCTION AND OVERVIEW OF THE STUDY CHAPTER 2: LITERATURE REVIEW CHAPTER 3: RESEARCH DESIGN, METHODOLOGY AND STRATEGIES. CHAPTER 4: RESEARCH RESULTS OF THE STUDY	
<b>1.10</b> 1.10.1 1.10.2 1.10.3 1.10.4 1.10.5 1.10.6	OUTLINE OF CHAPTERS. INTRODUCTION CHAPTER 1: INTRODUCTION AND OVERVIEW OF THE STUDY CHAPTER 2: LITERATURE REVIEW CHAPTER 3: RESEARCH DESIGN, METHODOLOGY AND STRATEGIES CHAPTER 4: RESEARCH RESULTS OF THE STUDY CHAPTER 5: FINDINGS, CONCLUSIONS, AND RECOMMENDATIONS	

## Chapter 2 Literature Review

2.1		1
2.2	PSYCHOLOGICAL RESILIENCE	13
2.2.1		13
2.2.2		13
2.2.3	A BRIEF OVERVIEW OF THE CONCEPTUALISATION OF RESEARCH TRENDS IN RESILIENCE .	
		14
2.2.4	A RESILIENCE LENS FOR GLOBAL SOUTH CHALLENGES	51
2.3	TEACHER RESILIENCE	53
2.3.1	INTRODUCTION	53



#### Page

2.3.2		PTUALISING TEACHER RESILIENCE	54
2.3.3	TEACHE	R RESILIENCE THROUGH DIFFERENT LENSES	56
	2.3.3.1	Multidimensional, strategic, and active agent approach	57
	2.3.3.2	Multi-level Teacher Resilience Model	57
	2.3.3.3	The teacher resilience process	58
	2.3.3.4	Contextualised structural disparity lens	59
2.3.4	TRENDS	IN TEACHER RESILIENCE RESEARCH	61
	2.3.4.1	Risk factors constraining teacher resilience	62
	2.3.4.2	Expected or predicted negative outcomes: Teacher stress, burnout a	ind
		attrition	63
	2.3.4.3	Teacher resilience enablers	64
	2.3.4.4	Unexpected, unpredicted or better than expected positive outcomes	despite
		significant risks	65
2.3.5	CHALLE	NGES AND PREDICTED NEGATIVE OUTCOMES CONSTRAINING RESILIENCE I	ΝA
	GLOBAL	SOUTH EDUCATIONAL LANDSCAPE	67
	2.3.5.1	Higher education challenges pertaining to the education landscape	69
	2.3.5.2	Basic Education challenges pertaining to the education landscape	70
2.3.6	A TEACH	HER RESILIENCE LENS FOR GLOBAL SOUTH EDUCATIONAL CHALLENGES	74
2.4	SELF-E	FFICACY AS A PROTECTIVE RESOURCE ENABLING TEACHER	
	RESILI	ENCE	79
2.4.1	INTROD	UCTION	79
2.4.2		PTUALISING SELF-EFFICACY	79
2.4.3	Self-ef	FICACY IN AN EDUCATIONAL SPACE ENABLING RESILIENCE	80
2.5	TEACH	ER EFFICACY BELIEFS AS PROTECTIVE RESOURCE FOR TEAC	HER
	RESILI	ENCE	81
2.5.1	INTROD	UCTION	81
2.5.2		PTUALISING TEACHER EFFICACY BELIEFS	81
2.5.3 TRENDS IN TEACHER EFFICACY RESEARCH AS ENABLING PATH		IN TEACHER EFFICACY RESEARCH AS ENABLING PATHWAYS TO TEACHER	
2.5.3		NCE	83
2.5.3	RESILIEI		
2.5.3	RESILIEI 2.5.3.1	Protective factors associated with teacher efficacy enabling teacher	
2.5.3	RESILIEI 2.5.3.1	Protective factors associated with teacher efficacy enabling teacher resilience	83
2.5.3	RESILIEI 2.5.3.1 2.5.3.2	Protective factors associated with teacher efficacy enabling teacher resilience Risk factors associated with teacher efficacy constraining teacher res	83 silience
2.5.3	RESILIEI 2.5.3.1 2.5.3.2	Protective factors associated with teacher efficacy enabling teacher resilience Risk factors associated with teacher efficacy constraining teacher res	83 silience 85

<b>TEACHER EDUCATION</b>	 



Page

2.6.1	Introdu	JCTION	85
2.6.2	SITUATI	NG THE CURRENT STUDY WITHIN THE GLOBAL EDUCATION LANDSCAPE	86
2.6.3	CHARAC	TERISING GLOBAL INITIAL TEACHER EDUCATION	87
2.6.4	TEACHE	R RESILIENCE IN INITIAL TEACHER EDUCATION PROGRAMMES	88
2.6.5	TEACHE	R EFFICACY IN INITIAL TEACHER EDUCATION PROGRAMMES	90
2.7		E-SERVICE TEACHER WITHIN THE EDUCATION LANDSCAPE	91
2.7.1	Introdu	JCTION	91
2.7.2	CHARAC	TERISING THE PRE-SERVICE TEACHER	91
2.7.3	PRE-SE	RVICE TEACHER RESILIENCE	92
	2.7.3.1	Pre-service teacher resilience enablers	92
	2.7.3.2	Risk factors constraining pre-service teacher resilience	93
	2.7.3.3	Variations in pre-service teacher resilience demographics given the extain	∩t
		data of the current study	94
2.7.4	PRE-SER	RVICE TEACHER EFFICACY	95
	2.7.4.1	Protective factors associated with pre-service teacher efficacy enabling	
		teacher resilience	96
	2.7.4.2	Risk factors associated with pre-service teacher efficacy constraining	
		teacher resilience	97
	2.7.4.3	Variations in pre-service teacher efficacy demographics given extant data	а
		of the current study	98
2.8	THE CO	DNCEPTUAL FRAMEWORK	99
2.8.1	Introdu	JCTION	99
2.8.2	INFORM	NG KNOWLEDGE ON TEACHER RESILIENCE IN A CHALLENGED CONTEXT THROUG	GH
	A SOCIA	COGNITIVE LENS	99
2.9	CONCL	<b>USION</b> 1	103

---000----

## Chapter 3 Research Design, Methodology and Strategies

3.1		104
3.2	RESEARCH QUESTIONS AND HYPOTHESES	105
3.2.1	INTRODUCTION	105
3.2.2	PRIMARY RESEARCH QUESTION	105



3.2.3	SECONE	DARY RESEARCH QUESTIONS	106
3.2.4	HYPOTH	IESES	106
	3.2.4.1	Statistical hypothesis of the population difference	107
	3.2.4.2	Statistical hypothesis of the population correlation	107
3.3	PARAD		108
3.3.1	Introdu	JCTION	108
3.3.2	EPISTEN	NOLOGICAL PERSPECTIVES: POST-POSITIVISM AS META-THEORY	109
3.3.3	Μετηοι	DOLOGICAL PARADIGM: QUANTITATIVE RESEARCH APPROACH	112
3.4	RESEA	RCH PROCESS AND METHODOLOGICAL STRATEGIES	115
3.4.1	Introdu	JCTION	115
3.4.2	Сомрая	RATIVE CASE STUDY AS RESEARCH DESIGN	116
3.4.3	SAMPLIN	NG	119
	3.4.3.1	Introduction	119
	3.4.3.2	Sampling from FIRE project data	119
	3.4.3.3	Selection of FIRE Teacher Resilience Measure variables for an	alysis
	<b>D</b>		
3.4.4		NALYSIS AND INTERPRETATION	
	3.4.4.1	Introduction	
	3.4.4.2	Nepperemetric statistics	123
	3.4.4.3		127
	3445	Inferential statistics	
345	5.4.4.5 Standa		
0.7.0	3451	Introduction	133
	3452	Quality assurance of quantitate data: Reliability	133
	3.4.5.3	Quality assurance of quantitative data: Validity	
	3.4.5.4	Established quality assurance of the measuring instrument	
	3.4.5.5	Quality assurance of analysis and results: Secondary data analysi	s as
		empirical systematic method	137
3.4.6	ETHICAL	CONSIDERATIONS	137
	3.4.6.1	Introduction	137
	3.4.6.2	Access and permission to use extant data	138
	3.4.6.3	Privacy and extant data	138
	3.4.6.4	Ethical reporting on extant data	139
3.5	CONCL	USION	140



## Chapter 4 Research Results of the Study

4.1	INTRO		141
4.2	DEMO	GRAPHIC BACKGROUND	142
4.2.1	INTROD	UCTION	142
4.2.2	DISTRIB	UTION: COMPLETED FIRE TEACHER RESILIENCE MEASURE (QUESTIONNAIF	RES)
			143
4.2.3	Demogi	RAPHIC DISTRIBUTION: AGE	143
4.2.4	Demogi	RAPHIC DISTRIBUTION: GENDER	145
4.2.5	Demogi	RAPHIC DISTRIBUTION: LANGUAGE FLUENCY	147
4.2.6	Demogi	RAPHIC DISTRIBUTION: ENROLLED PRE-SERVICE TEACHING PROGRAMMES	150
4.3	STAND		152
4.3.1	INTROD	UCTION	152
4.3.2	Reliabi	LITY ANALYSIS: CRONBACH'S ALPHA	152
4.3.3	VALIDIT	Y ANALYSIS: CONSTRUCT VALIDITY	153
4.3.4	STATIST	ICAL POWER ANALYSIS	159
4.4	DESCR	IPTIVE STATISTICS OF THE SELF-EFFICACY AND TEACHER	
	EFFICA	CY BELIEFS OF PRE-SERVICE TEACHERS IN A CHALLENGED	
	CONTE	ХТ	160
4.4.1	INTROD	UCTION	160
4.4.2	DESCRI	PTIVE STATISTICS OF THE SELF-EFFICACY OF PRE-SERVICE TEACHERS IN A	
	CHALLEI	NGED CONTEXT (2015 TO 2017)	161
4.4.3	DESCRI	PTIVE STATISTICS OF THE TEACHER EFFICACY BELIEFS OF PRE-SERVICE	
	TEACHE	RS IN A CHALLENGED CONTEXT (2015 TO 2017)	164
4.4.4	DESCRI	PTIVE STATISTICS OF THE SELF-EFFICACY AND TEACHER EFFICACY BELIEFS	OF
	PRE-SEF	RVICE TEACHERS IN A CHALLENGED CONTEXT (2015 TO 2017)	167
4.4.5	Descrii	PTIVE RESULTS OF THE SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF	F PRE-
	SERVICE	TEACHERS BASED ON GENDER AND ENROLLED PRE-SERVICE TEACHING	
	PROGRA	MMES: MEAN VALUES FIRE TEACHER RESILIENCE MEASURE	168
	4.4.5.1	Exploring gender differences concerning levels of self-efficacy and tea	acher
		efficacy beliefs of pre-service teachers	168
	4.4.5.2	Exploring enrolled pre-service teaching programme differences conce	erning
		levels of self-efficacy and teacher efficacy beliefs	170



4.5	INFERE	INTIAL STATISTICS OF THE SELF-EFFICACY AND TEACHER
	EFFICA	CY BELIEFS OF PRE-SERVICE TEACHERS WITH A CHALLENGED
	CONTE	<b>XT</b>
4.5.1	Introdu	JCTION
4.5.2	CORREL	ATION ANALYSIS BETWEEN THE SELF-EFFICACY AND TEACHER EFFICACY BELIEFS
	OF PRE-	SERVICE TEACHERS IN A CHALLENGED CONTEXT
4.5.3	RESULT	S OF THE FIRE TEACHER RESILIENCE MEASURE CASE 1 (2015): SELF-EFFICACY
	AND TEA	CHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS (WITHIN-CASE
	COMPAR	175 ISON)
	4.5.3.1	Summary of the <i>p</i> -value for the self-efficacy and teacher efficacy beliefs of
		pre-service teachers 2015 175
	4.5.3.2	Summary of the <i>p</i> -values for gender 2015
	4.5.3.3	Summary of the <i>p</i> -values for enrolled pre-service teaching programmes
		2015
4.5.4	RESULT	S OF THE FIRE TEACHER RESILIENCE MEASURE CASE 2 (2016): SELF-EFFICACY
	AND TEA	CHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS (WITHIN-CASE
	COMPAR	178 ISON)
	4.5.4.1	Summary of the <i>p</i> -value for the self-efficacy and teacher efficacy beliefs of
		pre-service teachers in 2016
	4.5.4.2	Summary of the <i>p</i> -values for gender 2016179
	4.5.4.3	Summary of the <i>p</i> -values for enrolled pre-service teaching programmes
		2016
4.5.5	RESULT	S OF THE FIRE TEACHER RESILIENCE MEASURE CASE 3 (2017): SELF-EFFICACY
	AND TEA	CHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS (WITHIN-CASE
	COMPAF	ISON)
	4.5.5.1	Summary of the <i>p</i> -value for the self-efficacy and teacher efficacy beliefs of
		pre-service teachers 2017 181
	4.5.5.2	Summary of the <i>p</i> -values for gender 2017
	4.5.5.3	Summary of the <i>p</i> -values for enrolled pre-service teaching programmes
		2017
4.5.6	RESULT	S OF THE FIRE TEACHER RESILIENCE MEASURE CROSS-CASE COMPARISON
	(2015 T	0 2017): Self-efficacy and teacher efficacy beliefs of pre-service
	TEACHE	RS186
	4.5.6.1	Summary of the <i>p</i> -value for the self-efficacy and teacher efficacy beliefs of
		pre-service teachers (2015–2017) 186
	4.5.6.2	Summary of the <i>p</i> -values for gender (2015 to 2017)



4.5.6.3 Summary of the *p*-values for enrolled pre-service teaching programmes

Page
------

	(2015 to 2017)
4.6	<b>CONCLUSION</b>
	000
Chapte Finding	r 5 gs, Conclusions and Recommendations
5.1	INTRODUCTION
5.2	OVERVIEW OF THE PRECEDING CHAPTERS
5.2.1	INTRODUCTION
5.2.2	CHAPTER 1: INTRODUCTION AND OVERVIEW OF THE STUDY
5.2.3	CHAPTER 2: LITERATURE REVIEW
5.2.4	CHAPTER 3: RESEARCH DESIGN, METHODOLOGY AND STRATEGIES
5.2.5	CHAPTER 4: RESEARCH RESULTS OF THE STUDY
5.3	INTERPRETING RESULTS AGAINST THE BACKGROUND OF EXISTING
	<b>TEACHER RESILIENCE LITERATURE</b>
5.3.1	INTRODUCTION

5.3.2	RESULT	S THAT SUPPORT OR CONTRADICT EXISTING LITERATURE	195
	5.3.2.1	The self-efficacy and teacher efficacy beliefs of pre-service teachers	195
	5.3.2.2	Gender differences between the self-efficacy and teacher efficacy belief	s of

## 

#### 

5.3.4 NOVEL INSIGHTS INFORMING TEACHER RESILIENCE IN A CHALLENGED CONTEXT....... 199

5.4	ADDRE	SSING THE RESEARCH QUESTIONS AND HYPOTHESES	201
5.4.1	INTROD	JCTION	201
5.4.2	FINDING	S: SECONDARY RESEARCH QUESTIONS AND HYPOTHESES	201
	5.4.2.1	Findings based on the comparison of self-efficacy and teacher efficacy	
		beliefs of final year pre-service teachers	201



	5.4.2.2	Findings based on the demographic information of pre-service teach	ners'
		self-efficacy and teacher efficacy beliefs	203
	5.4.2.3	Findings based on the relationship between the self-efficacy and tea	acher
		efficacy beliefs of pre-service teachers in a challenged context	206
5.4.3	FINDING	SS: PRIMARY RESEARCH QUESTION	207
5.5	THEOF	RISING TEACHER RESILIENCE FINDINGS IN A CHALLENGED CO	NTEXT
			209
5.5.1	INTROD	UCTION	209
5.5.2	SITUATI	NG THE FINDINGS WITHIN THE CONCEPTUAL FRAMEWORK	209
5.5.3	AN EVID	DENCE-BASED FRAMEWORK FOR TEACHER RESILIENCE IN A CHALLENGED C	ONTEXT
			211
56	СНУГІ		212
561			213
562	EVTANT		213
563			213 214
564	TEACHE	ED SELE-DEDORT OUESTIONNAIDE DATA	214 214
565	RESPON		216
5.6.6	FIRE T	EACHER RESILIENCE MEASURE IN THE SOUTH AFRICAN CONTEXT	217
5.7	RECO	MMENDATIONS OF THE STUDY	217
5.7.1	INTROD	UCTION	217
5.7.2	RECOM	MENDATIONS FOR TRAINING AND DEVELOPMENT	218
5.7.3	RECOM	MENDATIONS FOR POLICY DEVELOPMENT AND PRACTICE	219
5.7.4	RECOM	MENDATIONS FOR FUTURE RESEARCH	219
	5.7.4.1	Teacher resilience-building programmes in a challenged context	219
	5.7.4.2	Pre-service resilience studies within a Global South context	219
	5.7.4.3	Longitudinal studies	220
	5.7.4.4	Questionnaire development and standardisation	220
5.8	CONCI		221
REFER	ENCES		223
APPEN	DICES		267
Append	IX A:	BACKGROUND AND OVERVIEW OF ENTREE	268
Append	IX B:	FIRE TEACHER RESILIENCE MEASURE	271
Append	IX C:	FIRE PROJECT ETHICAL APPROVAL AND EXTRACTS	277
Append	IX D:	UNIVERSITY OF PRETORIA BED PROGRAMME INFORMATION	285



## List of Figures

Page
<b>Figure 1.1</b> Flow Chart Showing the Organisation of Chapter 12
<b>Figure 1.2</b> Background to the Current Study3
<b>Figure 1.3</b> The Multi-Level Teacher Resilience Model and Challenges in a Global South Context 18
Figure 1.4 System of Triadic Reciprocal Causation Interaction in SCT
Figure 1.5 Overview of the Current Study
Figure 2.1 Outline of the Literature Review Relevant to the Current Study
<b>Figure 2.2</b> Flow Chart Showing the Organisation of Chapter 242
<b>Figure 2.3</b> Resilience as Process and Outcome
<b>Figure 2.4</b> <i>Multi-level Teacher Resilience Model</i> 58
<b>Figure 2.5</b> The Teacher Resilience Process59
Figure 2.6 Contextualised Structural Disparity Lens60
<b>Figure 2.7</b> Informing Knowledge on Teacher Resilience in a Challenged Context: An SCT Lens to Self- efficacy and Teacher Efficacy of Pre-service Teachers
<b>Figure 3.1</b> Flow Chart Showing the Organisation of Chapter 3105
<b>Figure 3.2</b> Within-case Comparison of Self-efficacy and Teacher Efficacy Beliefs of Pre-service Teachers
<b>Figure 3.3</b> Combined Cross-case Comparison of Self-efficacy and Teacher Efficacy Beliefs of Pre- service Teachers
<b>Figure 4.1</b> Flow Chart Showing the Organisation of Chapter 4141



<b>Figure 4.2</b> Age Demographic–percentage (%) Distribution of Pre-service Teachers (2015 to 2017) 144
Figure 4.3 Gender Demographic Percentage Distribution (2015 to 2017)
<b>Figure 4.4</b> Percentage (%) Language Fluency Demographic Distribution of Pre-Service Teachers 148
<b>Figure 4.5</b> Percentage (%) Enrolled Pre-service Teaching Programme Distribution for Final Year Pre- service Teachers (2015–2017)
Figure 4.6 Descriptive Results of the Self-efficacy of Pre-service Teachers
Figure 4.7 Descriptive Results of the Teacher Efficacy Beliefs of Pre-service Teachers
Figure 4.8 Comparison of the Self-efficacy and Teacher Efficacy Beliefs of Pre-service Teachers 167
<b>Figure 4.9</b> Comparison of Pre-service Teachers' Gender (Male and Female) Self-efficacy and Teacher Efficacy Beliefs
<b>Figure 5.1</b> Flow Chart Showing the Organisation of Chapter 5
<b>Figure 5.2</b> An Evidence-based Framework for Self-efficacy and Teacher Efficacy as Resilience-enabling for Teacher Resilience in a Challenged Context



## List of Tables

Pag
Table 1.1         Challenges Teachers Face in a Global South Educational Space Constraining Resilience . 1
Table 1.2         FIRE Teacher Resilience Measure: Self-Efficacy (Resilience) of Pre-Service Teachers 2
Table 1.3         FIRE Teacher Resilience Measure: Teacher Efficacy of Pre-Service Teachers
Table 1.4         Comparison of ISCED Levels with South African (SA) Teaching Phases
Table 1.5         Overview (Statistics) of Demographic Information of Extant Data
Table 1.6         Summary of Research Findings
Table 2.1         Overview of the Conceptualisation of Psychological Resilience Through Different Lenses4
Table 2.2         Protective Resources in a Global South Educational Space
Table 2.3         Teacher Demographic Landscape
Table 3.1         Paradigmatic Perspectives         10 <sup>rd</sup>
Table 3.2Key Outcome Measures Assessing Levels of Self-Efficacy and Teacher Efficacy Beliefs ofPre-Service Teachers within the Current Study
<b>Table 3.3</b> Reliability of Scales
Table 4.1         Total Completed FIRE Resilience Measure per Year (2015 to 2017)       14
Table 4.2         Mean and Standard Deviation Age Distribution (Within-case and Cross-case)
Table 4.2         Mean and Standard Deviation Age Distribution (Within-case and Cross-case)



Page
Table 4.5Number of Languages Pre-service Teachers are Fluent in
Table 4.6Enrolled Pre-service Teaching Programme (2015 to 2017)
<b>Table 4.7</b> <i>Reliability Analysis</i> 152
Table 4.8         Convergent Validity - Self Efficacy Construct
Table 4.9Convergent Validity - Teacher Efficacy Construct
Table 4.10Discriminant Validity - Self-Efficacy Vs Teacher Efficacy
Table 4.11           Descriptive Statistics of the Self-Efficacy of Pre-Service Teachers           162
Table 4.12           Descriptive Statistics of Pre-Service Teacher Efficacy Beliefs           165
Table 4.13         Correlation between Self-Efficacy Items Average and Teacher Efficacy Items Average (2015 to 2017)         172
Table 4.14Inter-Item Correlation between Self-Efficacy and Teacher Efficacy Beliefs of Pre-ServiceTeachers Items (2015 to 2017)
Table 4.15Summary of the p-value for Self-efficacy and Teacher Efficacy Beliefs (2015)
Table 4.16Summary of the p-values for Gender (2015)
Table 4.17           Summary of the p-values for Enrolled Pre-Service Teaching Programmes (2015)
Table 4.18Summary of Ad-Hoc Pairwise Test for Enrolled Pre-Service Teaching Programmes (TeacherEfficacy) (2015)178
Table 4.19Summary of the p-value for Self-efficacy and Teacher Efficacy Beliefs (2016)
Table 4.20Summary of the p-values for Gender (2016)
Table 4.21Summary of the p-values for Enrolled Pre-Service Teaching Programmes (2016)



Page
------

Table 4.22Summary of the p-value for Self-efficacy and Teacher Efficacy Beliefs (2017)
Table 4.23Summary of the p-values for Gender (2017)
Table 4.24Summary of the p-values for Enrolled Pre-Service Teaching Programmes (2017)
Table 4.25Summary of the Ad-hoc Pairwise Test for Enrolled Pre-Service Teaching Programmes (Self- efficacy) (2017)
Table 4.26Summary of Ad-hoc Pairwise Test for Enrolled Pre-service Teaching Programmes (Teacher Efficacy) (2017)
Table 4.27Summary of the p-value for Self-efficacy and Teacher Efficacy Beliefs (2015–2017)
Table 4.28Summary of the p-values for Gender (2015 to 2017)187
Table 4.29Summary of Ad-hoc Pairwise Test for Male Pre-Service Teacher Self-Efficacy (2015 to 2017)
Table 4.30Summary of the p-values for Enrolled Pre-service Teaching Programmes (2015 to 2017)189
Table 4.31         Summary of Ad-hoc Pairwise Test for Intermediate Phase (IP) Self-Efficacy of Pre-Service         Teachers         190
Table 5.1Overview of the Statistical Hypotheses Related to Assumptions Based on the PopulationDifferences Between the Self-Efficacy and Teacher Efficacy of Pre-Service Teachers 202
Table 5.2Overview of the Statistical Hypotheses Related to Assumptions Based on the DemographicDifferences Between the Self-Efficacy and Teacher Efficacy of Pre-Service Teachers in aChallenged Context204
Table 5.3           Response Rate for the Current Study



## Chapter 1 Introduction and Overview of the Study

#### 1.1 INTRODUCTION

The purpose of the current research study was to expand knowledge on teacher resilience in challenged contexts by comparing the self-efficacy and teacher efficacy beliefs (intrapersonal<sup>1</sup>, trait<sup>2</sup> resilience-enabling pathways) of final year pre-service teachers in South Africa. The current study used quantitative statistical procedures to compare (within-case and cross-case) the relationship, if any, between the self-efficacy and teacher efficacy of pre-service teachers to contribute to the knowledge base of teacher resilience in spaces where structural disparity and inequality prevail. It follows that understanding self-efficacy and teacher efficacy of preservice teachers may inform policy, theory and practice in teacher preparation and professional development within similarly challenged contexts. In addition, not only can findings from this study indicate the utility of existing teacher resilience scales (from ENhancing Teacher REsilience in Europe [ENTREE] and the Factors Influencing Teaching Choice [FIT-Choice])<sup>3</sup> to measure intrapersonal pathways to teacher resilience in a challenged context but also pave the way for comparison of these results with comparable results on the scale elsewhere worldwide.

Chapter 1 provides an overview of the current study. To this end, I introduce the contextual background of this study. I further outline the rationale for and purpose of the current study considering the research questions and hypotheses. Subsequently, I clarify key concepts, foreground the theoretical framework, and précis the paradigmatic perspective, methodological decisions and the standards of rigour employed, and the ethical considerations that informed the current study. Lastly, I briefly summarise the results and outline the succeeding chapters of the thesis. Figure 1.1 presents a flow chart showing the organisational overview of Chapter 1.

<sup>&</sup>lt;sup>1</sup> Intrapersonal factors denote constructs occurring within an individual including for example attitudes or self-concept (American Psychological Association [APA], 2020a).

<sup>&</sup>lt;sup>2</sup> Traits signify personality characteristic that governs a person's behaviour across situations (APA, 2020a).

<sup>&</sup>lt;sup>3</sup> The FIRE Teacher Resilience Measure included scales from ENTREE (Mansfield & Wosnitza, 2015; Peixoto et al., 2018; Wosnitza et al., 2014) (see Section 1.6.7 and Appendix A for ENTREE background information) including the *Teacher Resilience scale* (Morgan, 2011) and the *Teacher Efficacy scale* (Morgan, 2011; Peixoto et al., 2018) The FIRE Teacher Resilience Measure also included the Factors Influencing Teaching Choice (FIT-Choice) scale (Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012). See Appendix B for the complete FIRE Teacher Resilience Measure.



#### Figure 1.1

Flow Chart Showing the Organisation of Chapter 1



#### 1.2 INTRODUCING AND CONTEXTUALISING THE CURRENT STUDY

The current study formed part of an existing project (Centre for the Study of Resilience [CSR], University of Pretoria [UP]), namely The *Fourth Year Intervention in Research (FIRE)*<sup>4</sup> project. The CSR generates knowledge on resilience in a Global South space, such as South Africa. To contextualise the present study, I provide a brief background as depicted in Figure 1.2.

<sup>&</sup>lt;sup>4</sup> FIRE project ethical approval: University of Pretoria (UP) 14 03 01. See Appendix C for ethical approval information and additional information on the FIRE project.



Figure 1.2

Background to the Current Study



The FIRE project was implemented at the University of Pretoria between 2015 and 2017 with three final year pre-service teacher cohorts. The FIRE project was conducted with a preservice teacher in their final undergraduate teacher training year. At the University of Pretoria, Faculty of Education, the Baccalaureus Educationis (BEd) Programme is a 4-year degree with teaching practice (see Section 1.6.9) as core modules (Faculty of Education, University of Pretoria, 2016)<sup>5</sup>. The FIRE project aimed to explore how fourth-year pre-service teachers experienced the teaching practicum and the mentor role in enhancing student-teacher professional development. Pertinent to the current study is that the FIRE Teacher Resilience Measure was piloted in the FIRE project as one of the baseline data generation questionnaires. A total of 1,193 pre-service teachers, between the age ranges of 20 years to 32 years, completed the FIRE Teacher Resilience Measure from 2015 to 2017. Of the total questionnaires completed, 223 (18.7%) of the respondents included male pre-service teachers and 929 (77.9%) female pre-service teachers.

In the current study, I adopted a post-positivist (meta-theory), quantitative research approach with a comparative case study design (i.e., within-case and cross-case comparison of final year pre-service teacher cohorts) to investigate the *self-efficacy* (see Section 1.6.5) and *teacher efficacy beliefs* (see Section 1.6.6) *as intrapersonal resilience-enabling pathways to resilience* (see Section 1.6.4) *for pre-service teachers* (see Section 1.6.9) *in a challenged context* (see Section 1.6.10). I used nonparametric statistical procedures on purposively selected cross-sectional extant data (i.e., completed FIRE Teacher Resilience Measure, N = 1,193) to compare the self-efficacy and teacher efficacy of pre-service teachers in a challenged context while following ethical and rigorous research guidelines.

<sup>&</sup>lt;sup>5</sup> See Appendix C for the University of Pretoria BEd Programme Information (Faculty of Education, University of Pretoria, 2016).



#### 1.3 RATIONALE OF THE STUDY

#### 1.3.1 INTRODUCTION

Although research on resilience, in particular psychological resilience, has a rich, replete, complex and prolific history (Cassidy, 2015; Shean, 2015; Windle, 2010), *teacher resilience* as a scholarly domain is an emergent research field. Furthermore, the literature highlights *the limited empirical research* on resilience in teachers (Beltman et al., 2011, 2018; Gu, 2018; Gu & Day, 2013), markedly from a *quantitative perspective* (Ainsworth & Oldfield, 2019; Kidger et al., 2016). The lack of teacher resilience research is also evident *in low- and middle-income countries*<sup>6</sup> with severely challenged contexts (Ebersöhn, 2015, 2016, 2017; Ebersöhn et al., 2020; Theron, 2016).

Teacher resilience, including *pre-service* teacher resilience, have increasingly received attention given teacher shortages, workforce issues and quality teaching worldwide (Duffin et al., 2012; Wosnitza et al., 2014), as well as in the Global South (Ebersöhn, 2016, 2017). To quantify factors (including traits such as self-efficacy and teacher efficacy beliefs) influencing teachers' ability to resile despite the challenging profession is needed (Ainsworth & Oldfield, 2019). This need for quantitative methodology in teacher resilience research is especially needed in Global South spaces given the limited studies on resilience and pre-service teacher resilience (Beltman et al., 2011, 2018; Ebersöhn et al., 2020; Gu, 2018; Gu & Day, 2013; Mansfield et al., 2016; Morgan, 2011; Peixoto et al., 2020). The paucity of teacher resilience measures (Beltman et al., 2011; Peixoto et al., 2020) further highlights the demand for robust teacher resilience scales for comparable results.

In Section 1.6.2 and Section 2.2, I acknowledge the current conceptualisation of resilience as a socio-ecological process. I situate the current study within one resilienceenabling pathway, namely that of intrapersonal, teacher traits as enabling pathway. Therefore, the current study aimed to quantitatively compare (within-case and cross-case) the selfefficacy<sup>7</sup> and teacher efficacy<sup>8</sup> of pre-service teachers<sup>9</sup> in a challenged context<sup>10</sup> to inform knowledge on teacher resilience in a severely disparate space (i.e., Global South context).

<sup>&</sup>lt;sup>6</sup> Countries are grouped based on income (gross national product [GNP]) by the World Bank classification system (Fantom & Serajuddin, 2016).

<sup>&</sup>lt;sup>7</sup> 'Resilience' (FIRE Teacher Resilience Measure) is operationalised as pre-service teachers' confidence in recovery from setbacks in school (i.e., self-efficacy).

<sup>&</sup>lt;sup>8</sup> 'TeachEff' (FIRE Teacher Resilience Measure) is operationalised as pre-service teachers' confidence in teaching and behaviour management (i.e., teacher efficacy).

<sup>&</sup>lt;sup>9</sup> The focus of the current study is on self-efficacy (questions 114–122) and teacher efficacy (questions 123–134). Thus, pre-service teachers' self-perception regarding their confidence in recovery from setbacks in schools (Morgan, 2011), and their confidence in teaching and behaviour management (Morgan, 2011; Peixoto et al., 2018).

<sup>&</sup>lt;sup>10</sup> A challenged educational context denotes the education system in a low- and medium-income (World Bank, 2021) space with cumulative and chronic adversity – as associated with a Global South place in an emerging economy (high adversity and structural disparity due to a post-colonial history) (Dados & Connell, 2012; Ebersöhn, 2014, 2017).



This study hopes to contribute to systematic teacher resilience knowledge by identifying important intrapersonal protective traits pre-service teachers may employ in challenged settings (such as South Africa) to resile in the profession despite chronic and cumulative challenges, while recognising the current socio-ecological process discourse of resilience (Ungar, 2008, 2010, 2011, 2012a, 2012b; Ungar et al., 2013). To this extent, the next sections justify teacher resilience as an emergent research field and the need for evidence-based research on teacher resilience within the Global South context.

#### **1.3.2** TEACHER RESILIENCE AS AN EMERGENT RESEARCH FIELD

Internationally, teaching is highlighted as one of the most demanding professions and the trials teachers face daily are evident (Brunetti, 2006; Castro et al., 2010; Ebersöhn & Ferreira, 2012; Gu & Day, 2007; Kyriacou, 2001; Thieman et al., 2014; Wosnitza et al., 2014). The negative impact on teacher health (e.g., stress and burnout) and well-being because of the teaching conditions necessitate the development of teacher resilience (Day & Gu, 2013; Ebersöhn, 2014; Gu & Day, 2013). Research that reports on how teachers positively adapted, notwithstanding risks and change in the profession, is needed (Day & Gu, 2013; Ebersöhn, 2017; Peixoto et al., 2018). It, therefore, comes as no surprise that, during the last decade, teacher resilience research has escalated worldwide (Gu, 2018).

As a developing research field, teacher resilience continues to evolve (Beltman et al., 2011; Ebersöhn, 2014; Gu, 2018; Gu & Day, 2013) with a shift away from studies focusing on stress, burnout, attrition, and ineffectiveness in teaching (Coetzee, 2013; Hong, 2012; Mansfield et al., 2012; Peixoto et al., 2018; Wosnitza et al., 2014). There have been teacher resilience studies in high-income global regions, including Australia (Beltman, 2020; Beltman et al., 2011, 2018; Beltman & Mansfield, 2018; Fischer et al., 2018; Johnson & Down, 2013; Knight, 2007; Le Cornu, 2009, 2013; Mansfield et al., 2012, 2014, 2016, 2018; Nolan et al., 2014; Papatraianou & Le Cornu, 2014; Peters & Pearce, 2012); Canada (Tait, 2008); Europe (Bouillet et al., 2014; Brouskeli et al., 2018; Daniilidou et al., 2020; Fletcher & Sarkar, 2013; Lohbeck, 2018; Kunnari et al., 2018; Morgan, 2011; Olsen, 2017; O'Sullivan, 2006; Peixoto et al., 2018, 2020; Wosnitza et al., 2014, 2018; Yada et al., 2021); Singapore (Ee & Chang, 2010); United Kingdom (Ainsworth & Oldfield, 2019; Gibbs & Miller, 2014; Gu, 2014, 2018; Gu & Day, 2007, 2013; Hewitt et al., 2017; Kirk & Wall, 2010; Mackenzie, 2012; Morgan et al., 2010; Price et al., 2012) and the United States (Bobek, 2002; Brunetti, 2006; Easterly & Myers, 2018; Fleming et al., 2013; Jennings et al., 2013; Muller et al., 2014; Sosa & Gomez, 2012; J. L. Taylor, 2013<sup>11</sup>; Thieman et al., 2014; Yonezawa et al., 2011).

<sup>&</sup>lt;sup>11</sup> In-text citations of first authors with the same surname include the initials of the first author (APA, 2020b, p. 306).



Teacher resilience studies in **Iow- and medium-income countries** are limited and include, for example studies in *China* (Gu & Li, 2013; Hong, 2012) and *Turkey* (Çelik et al., 2018; Polat & İskender, 2018; Yokus, 2015), as well as **Global South countries** such as *South-Africa* (Coetzee, 2013; Coetzee et al., 2017; Ebersöhn, 2012, 2014, 2015, 2016, 2017, 2019a, 2019b; Ebersöhn & Ferreira, 2012; Ebersöhn et al., 2020; Ebersöhn & Loots, 2017; Ferreira & Ebersöhn, 2012; Mansfield et al., 2018), and *Uganda* (Wabule, 2020).

It appears that there are significant gaps in teacher resilience knowledge given a severely challenged context (Ebersöhn, 2014, 2016, 2017; Ebersöhn & Ferreira, 2012; Ebersöhn & Loots, 2017; Mansfield et al., 2018; Ngidi & Ngidi, 2019). Therefore, the significance of the current study is to gain insight into intrapersonal resilience-enabling pathways to teacher resilience (i.e., self-efficacy and teacher efficacy of pre-service teacher) in a Global South setting, including the African context. Lan et al. (2014) noted that, although increasing, Africa generates only approximately 1% of worldwide scientific knowledge.

Quantification of resilience (Ainsworth & Oldfield, 2019) may provide knowledge on how teachers adapt to adversity through capacity-based (Masten, 2018; Masten et al., 1990) or social-ecological resilience pathways (Beltman et al., 2011; Ebersöhn, 2014; Ungar, 2008, 2010, 2011, 2012a, 2012b; Ungar et al., 2013). Such insight may be important to attract, retain and support the teaching profession and enable quality teaching (Morgan, 2011; Salifu et al., 2018). By assessing teachers' level of resilience, evidence of positive adaptation, despite challenging circumstances, may be revealed (Ainsworth & Oldfield, 2019).

Teacher resilience measurement requires investigation (Beltman et al., 2018). The current study thus employed scales from an existing resilience measure. Reliability and validity of the instrument, for the current study, were established using Cronbach's alpha and construct validity, respectively. The results on the scales can be utilised for comparative studies worldwide. A variable-based view can investigate a quantifiable association between adaptation measures and intrapersonal factors (Masten, 2001).

Papatraianou and Le Cornu (2014) also argued for further research in gender differences in resilience and the development of gender nuanced elements in teacher education programmes. Peixoto et al. (2020) further identified the need for teacher resilience studies involving a greater and more diversified sample. Ebersöhn (2014) emphasised the need for teacher developmental agendas to include knowledge of teacher resilience traits as pathways to sustain adaptation processes in spaces of high challenge. *Accordingly, it is worthwhile exploring enabling intrapersonal factors, such as the self-efficacy and teacher efficacy of preservice teachers, as important indicators of why teachers resile despite hardship in, say, a Global South setting*.

Intrapersonal traits are a signature element of resources from which teachers can draw (if present) during extreme and continuous hardship to resile. Thus, notwithstanding a challenged context, individual capacity (and systemic resources) may serve as capital to



mitigate against adversity (Ebersöhn, 2014, 2017; Ebersöhn et al., 2020). The presence or absence (see Figure 2.6) of intrapersonal traits (e.g., self-efficacy and teacher efficacy) may enable or constrain pre-service teachers' resilience for the monumental teaching task given disparities in a challenged context.

Systematic knowledge on the presence or absence of such traits may guide adaptation and enhancement of pre-service teacher programmes to assist pre-service teachers with the acquisition and development of necessary resilience traits. Knowledge on teacher resilience can also create awareness for pre-service teachers regarding traits they possess and can mobilise to buffer against adversity to thrive in the teaching profession to retain quality teachers in a challenged context. The next section explores the need for evidence-based research about teacher resilience within a Global South context.

#### **1.3.3 EVIDENCE-BASED RESEARCH FOR TEACHER RESILIENCE IN A CHALLENGED CONTEXT**

This section explores South Africa as an exemplar of a Global South place to locate the current study within a challenged context. I start my discussion by highlighting the nuances between a Global South and a Global North setting, focusing on the characteristics inherent in a Global South context. I conclude the discussion by elaborating on literature depicting South Africa as a severely challenged high-risk setting and the need for evidence-based teacher resilience knowledge within a desperate placed-based context.

The historical and current expanding disparity in and between countries generated a Global North ("context of privilege") and Global South debate ("pockets of poverty") (Trefzer et al., 2014, p. 4) as an alternative notion to a global village or globalisation (De Jong, 2010). Global South can refer to regions outside Western Europe and North America (i.e., First World, industrialised countries, developed countries or Global North), such as Latin America, Oceana, South and South-East Asia, as well as Africa (i.e., Third world or developing countries) (Dados & Connell, 2012; Montiel, 2018). However, mere geographic separation cannot address the complexities of the Global North and Global South divide since these regions are at times entwined or even within one another (Dirlik, 2007; Trefzer et al., 2014). The presence of in-country socioeconomic variance highlight the possible heterogenous context within the Global South, which is an important research consideration when exploring variations within and across individuals and school settings (Fackler & Malmberg, 2016; OECD, 2019a, 2019c).

The Global South encapsulates an amalgamating and symbolic designation applied to societies with a history of colonisation, inequality, repression, social origin of risk, resource constraints, disparate low-income economic conditions, widespread psychosocial problems and cultural as well as political marginalisation (Dados & Connell, 2012; Ebersöhn, 2016; Grovogui, 2011; Montiel, 2018). In a less equal society, with a legacy of disparity,



disproportionate outcomes and adversity, socioeconomic problems and psychosocial challenges may lead to chronic (i.e., ongoing persistent) adversity over time.

As is evident in Table 1.1, cumulative (i.e., multiple stressors in various domains) risks due to a lack of accessible resources and fragmented services also contribute to skewed opportunities and adversity (Bennell et al., 2002; Dados & Connell, 2012; Ebersöhn, 2015, 2016, 2017; Mampane & Bouwer, 2006). Enablers, as well as constraints, vary in degree and scale, so in an unequal and transformative ecology, higher risks are experienced by those with limited resources due to an inequitable distribution of available sources. Consequently, on the scale of individual, family, school, institutions, communities, and society, variability exists regarding resource limitations (Ebersöhn, 2017). Individuals, families, and communities embedded in a challenged context are confronted daily with various enduring and unrelenting risk factors (Ebersöhn, 2016, 2017). These risk factors include poverty, disease, hunger, malnutrition, crime, violence, a lack of essential public services (e.g., healthcare, water, transport, sanitation, and electricity), poor service delivery as well as a dysfunctional education system (Ebersöhn, 2016; Nkoana, 2017; Statistics South Africa [Stats SA], 2020a). The risks may contribute to an incapacity to develop human and social capital leading to detrimental consequences for the entire micro-, meso-, exo- and macrosystem (i.e., individuals, families, households, schools, institutions, communities, and the wider society as well as policy development and implementation) over time (chronosystem) (Bronfenbrenner, 1979; Diale et al., 2014; Ebersöhn, 2015, 2016, 2017). Evidence from the Programme for International Student Assessment (PISA) and Teaching and Learning International Survey (TALIS)<sup>12</sup> supports recommendations to improve school and teacher quality to reduce cross-country variances, the gap in educational outcomes and differences between learners at the top and bottom of the socioeconomic distribution (Mammadov & Çimen, 2019; Nieto & Ramos, 2015).

Ebersöhn (2015) argued for place-based (e.g., Global South) research through contextual lenses to support emerging democracies (such as South Africa) that is diverse and in transition. The purpose of the research would be to change the fundamental discourse of excessive power domination, the delegitimation of non-western values and the structural inequality chronically evident in a challenged context. The concept of "place" (Ebersöhn, 2015) and geopolitical characteristics as worldviews demonstrate variability in how high-adversity and high need-barriers, as well as support and adaptation, is conceptualised and operationalised (Ebersöhn, 2015). The continuing uneven gradient of global development necessitates cognisance of pluriversality rather than universality within geopolitical variance to transpose possible assumptions of a non-normative Global South place (Ebersöhn, 2015).

<sup>&</sup>lt;sup>12</sup> The International Organisation for Economic Co-operation and Development (OECD) TALIS datasets includes representative teaching and learning data (OECD, 2019c). The data provides an opulent source of comparative international data (Fackler & Malmberg, 2016).



As an exemplar of a Global South context, South Africa's infrastructure was constructed by its ubiquitous colonial nascent (Coovadia et al., 2009; Dunn, 2020) and recent political democratisation (Montiel, 2018), resulting in accrescent systemic disparities and disruptions (Ebersöhn, 2016, 2017; Montiel, 2018). Mlachila and Moeletsi (2019) underlined that the causes of South Africa's mediocre quality of education are complex with the legacy of postcolonial outcomes playing a significant part in the problem. Given the Global South place argument, transforming post-colonial societies, such as South Africa, with vast socioeconomic, health and education challenges and poor economic, social, health, welfare and education outcomes should be anticipated (Ebersöhn, 2019a). Due to past inequalities and enduring historical privilege, individuals and communities may continue to experience challenges in accessing systemic opportunities negating policy implementation (DHET, RSA, 2018a; Ebersöhn, 2019a). Power and structural forces sustain the inequalities in resources, living standards, service distribution, high poverty levels and unemployment, among a large proportion of marginalised people (Ebersöhn, 2015, 2016, 2017, 2019a; Francis & Webster, 2019; Montiel, 2018). Unemployment rates continue to rise as economic growth stagnates with inadequate income generation, elevating poverty (Ebersöhn, 2019a; Francis & Webster, 2019) and civil unrest with violent protest action. The unemployment rate of 29.1%, determined for 2019 (Stats SA, 2020a, 2020b), perpetuate concerns regarding the poverty increase and wellbeing of South-African populations.

If a place-based (e.g., Global South context) (Ebersöhn, 2015) discourse is meaningful in gauging the relevance and intensity of chronic and cumulative barriers (see Table 1.1), poverty is indubitably a constraint in any setting (Ebersöhn, 2014, 2017). According to the International Monetary Fund (IMF, 2004), poverty encapsulated various compounding dimensions leading to insufficient resources to sustain a suitable minimum standard of living in societal structures. In a country such as South Africa, inequity and poverty are interconnected (Francis & Webster, 2019), while educational disparity amplifies other forms of systemic incongruence (McKeever, 2017). Hence, people in a challenged context may experience low quality of life, discontentment, and pessimism due to the escalating unemployment, decreased social support, negatively biased resource supply, limited health services as well as disproportionate quality education opportunities (Ebersöhn, 2016, 2017; Wissing et al., 2013). Therefore, economic welfare for all is unlikely in the foreseeable future, emphasising enabling pathways to adapt positively to existing inequality (Ebersöhn, 2017). The Global South context provides a larger place in which a challenged education space is embedded. As a result, the barriers inherent to a post-colonial, Global South context can affect the education system, teachers, learners and communities. Table 1.1 depicts an overview of the numerous systemic challenges in a Global South space due to an emerging economy, high adversity and structural disparity leading to chronic and cumulative risk factors in the daily lives of South African teachers (Ebersöhn, 2014, 2017).


#### Table 1.1

Challenges Teachers Face in a Global South Educational Space Constraining Resilience

Challenges / strains / risks / barriers / stressors / adversity / disturbances / disparities / disenabling / threats in the Global South (i.e., inhibit, constrain, disenable or challenge resilience) Macrosystem (Culture, policy, education, health, and welfare systems) Culture Government bureaucracy and corruption with imposed regulations Historical context (e.g., post-colonial society in transformation) Political context and demoralising political demands Racism and discrimination Violence Civil protest actions and strikes Policy International trends and policy influence National and governmental policy change and haphazard policy implementation Education A dysfunctional education system with poor outcomes Low public image with the education system criticised for failing Continuous changing education system including curriculum adaptation and policy reform, differentiation in expectations, standards and behaviours Socioeconomic status of the school (e.g., no-fee schools; dysfunctional schools) Insufficient or lack of education infrastructure (e.g., not enough schools) Limited health and safety at schools Poor funding application Higher and basic education access and attainment (e.g., poor pass rates and throughput rates, inadequate performance on international and national large-scale assessments Lack of status of the profession with inadequate incentives and uncompetitive salaries Adverse working conditions (e.g., personnel and/or learner strikes, inadequate school safety, heavy workload and lack of time) Inadequate teacher quality and teacher training leading to unqualified and/or under-qualified teachers Shortage of teachers and skilled personal or poor-quality teachers Teacher absenteeism Multilingual classrooms Migration and influence of traditional leaders, especially in rural and challenged contexts Health Poor public health system and health challenges (e.g., Human immunodeficiency virus [HIV]<sup>13</sup>) Deficient or a lack of health infrastructure Excessive distances to clinics (if any) Welfare Limited access to welfare Crime, specifically violent crime Socioeconomic challenges and economic disparity including intergenerational poverty and fiscal constraints Low-income households High level of unemployment Child headed households, the prevalence of orphans and teenage pregnancy Infrastructure Inadequate or lack of infrastructure and facilities (e.g., housing) Inadequate and unreliable service delivery or limited access to services (including electricity [i.e., loadshedding], sanitation and water) Poor and unreliable transport services Exosystem (Physical infrastructure, location, resources) Inadequate facilities and infrastructure (e.g., lack of classrooms, restricted teaching facilities, problematic water supply to schools and poor sanitation [e.g., 'pit latrines'], no, limited or frequent power shortages ['load-shedding'] influencing teaching and assessments Resource-constrained schools (e.g., an insufficient supply of textbooks, teaching materials and teacher aid Remote or isolated (distance) context Transportation challenges (i.e., unsatisfactory or unreliable transportation to and from school e.g., bus and taxi strikes) Overcrowded classrooms

<sup>&</sup>lt;sup>13</sup> The overall HIV prevalence rate is approximated at 13.5% among the South African population (Republic of South Africa [RSA], 2020b).



#### Mesosystem

(Relationships between learners, parents, colleagues, school and support networks)

School context Teacher

- Lack or limited access to health and support services
- Feeling unsupported or unsafe at school
- Experiencing low fiscal income or uncompetitive salaries
- Limited professional developmental opportunities
- · Heavy workload and demanding extra-mural responsibilities or duties not directly related to teaching
- High teacher turnover and attrition rates
- Unable to meet the needs of disadvantaged learners
- Poor relationships with leaders and colleagues
- Lack of relationship or scrutiny of learners, peers, parents and/or principals
- Unmet personal needs
- Stressful classroom management and curriculum delivery
- Teachers' own financial constraints and low-income household
- Learner level
- High learner-teacher ratios
- Demotivated learners and learner misbehaviour
- Ineffective classroom management and disruptive discipline challenges (e.g., learners interrupting lessons)
- Challenging, deteriorating or antisocial learner behaviour (e.g., criminal activities, intimidation or bullying, drugs or alcohol abuse, learner boycotts and drop-out)

Management and administrative level

- Inadequate management, leadership or mismanagement
- Lack of support and resources from management
- Unsupportive leadership staff
- Hostile school cultures created by school leadership

Parental/guardian and community level

- Low levels of parental education or illiteracy
- Limited or inadequate parental participation, involvement and communication
- Absence of parent support
- Lack of community support
- Curbed school-community participation and scarce community support systems

#### Microsystem

#### (Personal attributes, traits or capacities)

#### Social

- Avoid asking for assistance or difficulty in help-seeking (i.e., difficulty asking for help)
- Family pressure
- Unexpected events in personal lives

Emotional

- Overwhelmed by multiple demands, roles and responsibilities
- Burnout, stress and/or depression
- Failure to experience personal fulfilment

Professional

- Conflict between personal ideologies and implemented practices
- Lack of teaching skills and training or insufficient professional competence
- Lack of content knowledge
- Limited or no job satisfaction

Motivational

- Lack of motivation and commitment
- High extrinsic motivation (e.g., job security, career status, career change, job conditions; workload, salary, politics) with low intrinsic motivation (e.g., working with children, perceived ability, view of intrinsic value of teaching; self-efficacy, utility)

Beliefs

- Limited or negative self-beliefs or lack of confidence concerning classroom management and discipline
- · Lack of confidence or efficacy (negative self-beliefs) in teaching ability
- Experiencing a conflict between practices in the school and personal beliefs
- Behavioural dispositions (e.g., coping strategies)

• Inadequate coping strategies like unhealthy lifestyle, 'saving the world' attitude (altruism)

(Beltman et al., 2011; Bronfenbrenner, 1979; Coetzee et al., 2017; Ebersöhn, 2012, 2014, 2016, 2017;

Ebersöhn et al., 2020; Ebersöhn & Loots, 2017; Gu & Day, 2013; Harber & Mncube, 2011; Jackson & Rothmann, 2005; Mansfield et al., 2016, 2018; Maphalala & Mpofu, 2019; Milner & Khoza, 2008; Mlachila & Moeletsi, 2019; Ngidi & Ngidi, 2019; Stats SA, 2012b, 2020a, 2020b; UNESCO, 2019a; Wabule, 2020; Zuma et al., 2016)



Within a challenged education system, disparities can be characterised on each level (i.e., macro-, meso- and microsystems) as illustrated in Table 1.1. The current study acknowledges the caution needed when analysing the challenges noted in Table 1.1. Challenges, with multiple contributing factors, may not be mutually exclusive or unique to a Global South context and constraints are evident within the teaching profession worldwide (see Section 2.3.4.1) However, the concomitant need to build a collective body of research on teacher resilience (Gu & Day, 2013; Thieman et al., 2012) to enable positive education outcomes and counter structural disparity and inequality, in a post-colonial space, therefore, should be emphasised. As such, the current study provides a plausible (expanding) sphere of research within teacher resilience resilience studies in a challenged context.

Research through a resilience paradigm may offer alternative narratives, leading to evidence on enabling pathways (including traits) of well-being and innovative adaptation (i.e., succeeding despite considerable risk) needed in a Global South context rather than an emphasis on inequality, disparities and challenges which highlight a disaster perspective (Ebersöhn, 2014, 2016, 2017). Focusing on teacher resilience rather than emphasising barriers, attrition, burnout, teacher and teaching ineffectiveness seems essential (Coetzee, 2013). Given the rationale, the following section emphasises the purpose of the current study.

### 1.4 PURPOSE OF THE STUDY

The nature of this study was comparative (Babbie, 2021; L. Cohen et al., 2018; Creswell, 2014; Creswell & Creswell, 2018) since the Teacher Resilience Measure (including the Teacher Resilience [Morgan, 2011] and Teacher Efficacy scale [Morgan, 2011; Peixoto et al., 2018]) was employed at multiple cross-sectional (2015, 2016 and 2017) points with final year pre-service teachers at the University of Pretoria to see whether similar, different, or complementary results would be obtained. In countries such as Australia (Beltman et al., 2018; Mansfield & Wosnitza, 2015), the Czech Republic (Wosnitza et al., 2018), Germany (Peixoto et al., 2018; Wosnitza et al., 2018), Ireland (Morgan, 2011; Peixoto et al., 2018; Wosnitza et al., 2018), Malta (Peixoto et al., 2018; Wosnitza et al., 2018) and Portugal (Peixoto et al., 2018, 2020; Wosnitza et al., 2018) studies employed the teacher resilience (Morgan, 2011) and teacher efficacy (Morgan, 2011; Peixoto et al., 2018) scales also utilised in the current study. Given the currently available literature, the questionnaire has not been utilised with pre-service teachers on a large scale in South Africa but has been employed in a recent South African study with teachers in challenged context (Ebersöhn et al., 2020). The current study, therefore, also reported on the piloting of ENTREE scales for pre-service teachers during the FIRE project. Reporting on the piloting of the scales aimed to contribute to recommendations of the scales for the South African context.



Results from studies applying similar ENTREE<sup>14</sup> (Beltman et al., 2018; Peixoto et al., 2018; Wosnitza et al., 2018) and FIT-Choice<sup>15</sup> (Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012) scales as employed in the current study noted the importance of pre-service teachers' confidence in both recovery from setbacks and teaching and behaviour management. Morgan (2011), as well as Peixoto and colleagues (2018), found that teacher efficacy appeared to be a main predictor of teachers' confidence in recovery from setbacks in all countries. In Ireland and Germany, teacher efficacy was the only or strongest significant predictor for teacher recovery from setbacks, respectively, in comparison with social, emotional, professional and motivational capacities (Peixoto et al., 2018). A recent South African study (Ebersöhn et al., 2020), applying similar FIRE scales, also highlighted recovery from setbacks (high reliability) and teaching and behaviour management abilities (high reliability) as important resilience traits for teachers in a challenged context. Furthermore, the reliability analysis of the relevant ENTREE scales in previous international studies (Beltman et al., 2018; Peixoto et al., 2018; Wosnitza et al., 2018) was higher (ranging from high reliability to excellent reliability) than the social, emotional, professional and motivational measurement scales. Based on these findings, I opted for a focus on self-efficacy (i.e., recovery from setbacks) and teacher efficacy (i.e., teaching and behaviour management) as intrapersonal resilience-enabling pathways to teacher resilience.

Comparative research entails the identification, gathering occurrences and disconfirming occurrences of a phenomenon (i.e., teacher resilience) to reach conclusions (Dion, 2003; Esser & Vliegenthart, 2017). Comparison is therefore useful for understanding the scope and significance of psychological phenomena (Zartman, 2012) and is employed to determine what characteristics are singular or collective to objects of analysis (i.e., self-efficacy and teacher efficacy beliefs) (Dion, 2003; Esser & Vliegenthart, 2017). Such comparisons drawn from quantitative data in the current study, ensured the rigour of it and allowed knowledge generation on intrapersonal resilience-enabling pathways for pre-service teachers in a challenged context. Furthermore, a comparison may prevent researchers from over-generalising (Esser & Vliegenthart, 2017). In this case, the transferability of the findings can be limited (Hays & Singh, 2011) to contexts similar to final year pre-service teachers at the University of Pretoria. Afdal (2019) argued the importance of contextualisation within comparative research. Nonetheless, the systematic and holistic comparative element in the

<sup>&</sup>lt;sup>14</sup> "ENTREE is a project partly funded by the European Commission's Lifelong Learning Programme and is supported by an international team of experts from five European countries and Australia" (Wosnitza et al., 2014, p. 3). ENTREE aim to advance professional training modules for pre-service and in-service teachers to enable teacher resilience (Wosnitza et al., 2014). To this end survey measures were employed to explore the relationship between a global scale of resilience (Morgan, 2011) and associated capacities among different countries (Mansfield & Wosnitza, 2015; Peixoto et al., 2018). See Appendix A for an overview and background of the ENTREE project and all measures included in the latter.

<sup>&</sup>lt;sup>15</sup> The FIT-Choice questionnaire is an integrative, compressive and theoretical framework exploring the choice of teaching as profession (Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012).



current study promoted a deeper understanding with robust data and confidence in findings (Yin, 2018; Zartman, 2012) comparing the self-efficacy and teacher efficacy beliefs of preservice teachers in a challenged context. Therefore, the current study's potential significance is that within-case and cross-case comparison between the self-efficacy and teacher efficacy of pre-service teachers can be illuminated against objective measures in a challenged context.

Given the challenged Global South context of the current study (see Section 1.3.3), the intrapersonal traits (self-efficacy and teacher efficacy) are resources from which teachers can draw to resile despite spaces of severe challenge. Therefore, the current study aimed to add to the growing sphere of research on teacher resilience in a challenged context and generate global teacher resilience discourses from Africa to create new insights into the relationship between the self-efficacy and teacher efficacy of pre-service teachers within a strained Global South context. Following the current study's rationale and purpose, I pose the research questions and hypotheses for this study in the next section.

# 1.5 RESEARCH QUESTIONS AND HYPOTHESES FOR THE STUDY

In accordance with the rationale and purpose statement, the following primary research question guided the current study:

How can insight into the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged education context inform knowledge on teacher resilience?

To address the primary research question, I explored secondary research questions including:

- How do self-efficacy and teacher efficacy beliefs of final year pre-service teachers within a challenged education context (within-case and cross-case) compare?
- What are the self-efficacy and teacher efficacy beliefs of pre-service teachers within a challenged education context based on demographic information (within-case and cross-case)?
- To what extent is there a relationship between self-efficacy and teacher efficacy beliefs among pre-service teachers in a challenged education context (within-case and cross-case)?

The quantitative statistical hypotheses under consideration (Field, 2018; Pietersen & Maree, 2019d), based on the population difference, for the current study, were as follows:

- *H*<sub>0</sub>: Median<sub>dif</sub> = 0 (median population difference between self-efficacy and teacher efficacy beliefs [within-case and cross-case] does not differ significantly from zero).
- H<sub>a</sub>: Median<sub>dif</sub> ≠ 0 (median population difference between self-efficacy and teacher efficacy beliefs [within-case and cross-case] differs significantly from zero).

The statistical null hypothesis ( $H_0$ ) of the population difference stated that there was no statistically significant difference (Creswell & Creswell, 2018; López et al., 2015; Pietersen &



Maree, 2019d; Sauro & Lewis, 2016) between the self-efficacy and teacher efficacy of preservice teachers in a challenged context. In contrast, the alternative hypothesis should align with what is expected by the research and refute the null hypothesis (L. Cohen et al., 2018; Creswell & Creswell, 2018; Pietersen & Maree, 2019d; Sauro & Lewis, 2016). Thus, the alternative hypothesis indicates that not all the population medians are equal, and will differ from one another statistically (Field, 2018). In addition, the quantitative statistical hypotheses under consideration, based on the population correlation coefficient, for the current study included:

- *H*<sub>0</sub>: There is no statistically significant correlation between self-efficacy and teacher efficacy beliefs.
- *H*<sub>a</sub>: There is a statistically significant correlation between self-efficacy and teacher efficacy beliefs.

The given null hypothesis ( $H_0$ ) of the population correlation coefficient stated that there was no statistically significant correlation (Field, 2018) between the self-efficacy and teacher efficacy beliefs of final year pre-service teachers. On the other hand, the alternative hypothesis ( $H_a$ ) stated that there was a statistically significant correlation (Field, 2018) between the self-efficacy and teacher efficacy beliefs of pre-service teachers within a challenged education context. According to López et al. (2015), a hypothesis should be testable, exact, and informed by the purpose of the research. If the *p*-value is less than .05, the null hypothesis ( $H_0$ ) is rejected, and there is a statistically significant correlation (Field, 2018). On the other hand, if the *p*-value is greater than .05, the null hypothesis ( $H_0$ ) is not rejected, and there is no statistically significant correlation (Field, 2018). The research questions and hypotheses for this study are explored in detail in Section 3.2 of Chapter 3. The following section provides clarification on key concepts within the current study.

# 1.6 CONCEPT CLARIFICATION

## 1.6.1 INTRODUCTION

In this section, I describe the core concepts and terminologies within the context of my focus area to ensure a collective understanding of the key underpinnings of the current study. The concepts (and theories) defined in this section are underlying contentions of the current study. Chapter 2 of this thesis presents a more detailed discussion of the concepts listed below.

## 1.6.2 RESILIENCE

Debates and definitions on resilience appear diverse (see Section 2.2). However, researchers (Cassidy, 2015; Fletcher & Sarkar, 2013) agree that for resilience, adversity (i.e., threat, risk or challenge) and positive adaptation (i.e., resourceful and innovative enabling responses or



outcomes) must be evident. Therefore, within the current study, resilience will be conceptualised as the adaptive response pre-service teachers may demonstrate to different adversities within a challenged context (i.e., a Global South educational space).

Resilience is considered as a trait in literature (Block & Kremen, 1996; Connor & Davidson, 2003; Letzring et al., 2005; Masten, 2018; Masten et al., 1990). Within the current study, teacher resilience is measured by the FIRE Teacher Resilience Measure, as self-efficacy ('Resilience') and teacher efficacy ('TeachEff') (see Appendix B). Thus, the current study operationalised resilience measurement within trait discourses in resilience, denoting resilience as intrapersonal qualities (e.g., self-efficacy and teacher efficacy) that enable people to thrive despite severe challenges (Anthony, 1987; Block & Block, 1980; Brunetti, 2006; Connor & Davidson, 2003; Ee & Chang, 2010; Masten et al., 1990; Roth & Von Collani, 2007; Tait, 2008). Therefore, the presence of teacher resilience traits is viewed as individual or internal pathways to enable positive outcomes despite challenges (see Section 1.6.4) as depicted in Figure 2.6. Nonetheless, the current study acknowledged the process (Coetzee, 2013; Fletcher & Sarkar, 2013; Gu, 2014; Masten, 2018; Nolan et al., 2014) and socio-ecological (Ungar, 2008, 2010, 2011, 2012a, 2012b; Ungar et al., 2013) discourses within recent resilience theory and the conceptualisation of the current study as illuminated in Section 2.2.

Researchers (Beltman et al., 2018; Day & Gu, 2010; Ebersöhn, 2012; Masten & Reed, 2005; Masten & Wright, 2010; Rutter, 2006; Theron, 2011; Ungar, 2008; Ungar et al., 2013) have recently conceived resilience as a dynamic, multifaceted, multidimensional, socially constructed, developmental and interactive process between the individual and the environment. Resilience denotes positive adaptation despite high challenge which leads to unexpected, unpredicted or better than expected outcomes over time. Outcomes can include relatively good outcomes despite risk. Outcomes may also involve navigating environmental risk experiences, which suggests positive health, well-being, and fluid outcomes, even in adverse conditions (Day et al., 2006; Masten, 2001, 2011; Rutter, 2006, 2012). Therefore, given the acknowledgement of socio-ecological discourses in resilience, I view intrapersonal resilience-enabling traits as one pathway to teacher resilience within a challenged context.

### **1.6.3** TEACHER RESILIENCE AND PRE-SERVICE TEACHER RESILIENCE

Teacher resilience refers to the "process of, capacity for, and outcome of positive adaptation and ongoing professional commitment and growth in the face of challenging circumstances" (Wosnitza et al., 2014, p. 2) and recurring setbacks (Brunetti, 2006; Gu & Day, 2013; Gu & Li, 2013; Mansfield et al., 2016; Peixoto et al., 2018; Wosnitza et al., 2018). Individual and contextual characteristics dynamically interact to constrain (challenge) or enable (promote) resilience in teachers (Beltman et al., 2011; Ebersöhn, 2014; Mansfield et al., 2014). Demonstrating teacher resilience means navigating capacities, behaviour dispositions and



contextual resources to appraise and adapt to (potentially) stressful situations for better than expected outcomes (Ebersöhn, 2014; Wosnitza et al., 2014).

Teachers, drawing on enabling assets, and coping strategies not only "bounce back" but also foster job satisfaction, self-efficacy, well-being, enthusiasm, engagement, motivation, and commitment (Beltman et al., 2011; Mansfield et al., 2014, 2016; Wosnitza et al., 2014). Furthermore, teachers that resile appear to contribute positively to learner<sup>16</sup> achievement and the school community (Day, 2012; Sammons et al., 2007; Wosnitza et al., 2014). Accordingly, teacher resilience is a shared, dynamic process between a teacher and the environment with protective resources and risk factors originating in either or both the ecology and teachers themselves over time (Coetzee, 2013; Ebersöhn, 2014; Johnson & Down, 2013; Mansfield et al., 2012, 2016; Peixoto et al., 2018; Wosnitza et al., 2014). Teacher resilience enables positive unpredicted or unexpected outcomes, including high levels of well-being, job satisfaction, commitment, retention and quality education (Castro et al., 2010; Cefai & Cavioni, 2014; Ebersöhn, 2014; Mansfield et al., 2012; Peixoto et al., 2012; Peixoto et al., 2018; Wosnitza et al., 2018; Wosnitza et al., 2010; Cefai & Cavioni, 2014; Ebersöhn, 2014; Mansfield et al., 2012; Peixoto et al., 2012; Peixoto et al., 2018; Wosnitza et al., 2018; Wosnitza et al., 2018; Wosnitza et al., 2010; Cefai & Cavioni, 2014; Ebersöhn, 2014; Mansfield et al., 2012; Peixoto et al., 2018; Wosnitza et al., 2010; Cefai & Cavioni, 2014; Ebersöhn, 2014; Mansfield et al., 2012; Peixoto et al., 2018; Wosnitza et al., 2018; Wosnitza et al., 2018).

Within the current study, teacher resilience was operationalised as the ability of teachers to employ protective resources and coping strategies to facilitate positive adaptation (Beltman et al., 2011; Coetzee et al., 2017; Mansfield et al., 2016). Teacher resilience may support teachers to remain in teaching and resile despite adverse conditions (Ebersöhn, 2014). Therefore, the presence of self-efficacy (confidence in recovery from setbacks) and teacher efficacy (confidence in teaching and behaviour management) may act as individual enablers within the micro-system of pre-service teachers. Within the current study, teacher resilience is measured in terms of self-efficacy and teacher efficacy of pre-service teachers in a challenged educational context, as indicated in the FIRE Teacher Resilience Measure. Early career teachers may experience continuous and numerous challenges but also have both personal and contextual resources sustaining them during the initial teaching year(s). Contextual resources indicate the important role of relationships (including family, friends and colleagues) in the resilience process (Mansfield et al., 2014).

Pre-service teacher resilience can be defined as managing the emotional challenges and inevitable uncertainties (Gu, 2018), inherent in the realities of becoming a teacher, driven by professional commitment (Gu & Day, 2013). The current study will conceptualise pre-service teacher resilience as the confidence pre-service teachers demonstrate in dealing with setbacks in schools (i.e., self-efficacy) (Morgan, 2011) within a challenged context. The sample of pre-service teachers in the current study was  $N = 1,193^{17}$ , with 77.9% female pre-service teacher respondents and 18.7% male pre-service teacher respondents.

<sup>&</sup>lt;sup>16</sup> Within the current study, "learner" refers to any "pupil or student at an early learning site, school, further education and training institution or adult learning centre" (SACE, 2017, p.3).

<sup>&</sup>lt;sup>17</sup> According to the APA (2020b, p. 155) guidelines, "a comma is used to separate groups of three digits in most numbers of 1,000 or more".



### 1.6.4 INTRAPERSONAL RESILIENCE-ENABLING PATHWAYS

The research endeavoured to foreground the investigation of intrapersonal resilience-enabling pathways to teacher resilience in a challenged context.

### Figure 1.3

The Multi-Level Teacher Resilience Model and Challenges in a Global South Context



(Bronfenbrenner, 1979; Ebersöhn, 2014; Mansfield et al., 2018; Stats SA, 2012a, 2012b, 2020a, 2020b; Wosnitza et al., 2014)

The current study focussed on intrapersonal resilience-enabling pathways to teacher resilience (i.e., microsystem as theorised by Bronfenbrenner [1979]) in a severely challenged Global South education system. As depicted above in Figure 1.3, teachers may experience challenges such as burnout or stress, depression, avoiding asking for assistance, lack of confidence or self-efficacy, a conflict between beliefs and implemented practices, family pressure, overwhelming responsibilities as well as a lack of self-belief about classroom management on the microsystem (personal level). However, chronic poverty and the resulting compounded psychosocial problems are evident at every level of the South African ecosystem (Mansfield et al., 2018; Stats SA, 2020a). On the other hand, there are assets within the teachers' microsystem (i.e., personal protective resources) whose presence or absence could enable or constrain resilience.

These enablers may include self-efficacy, teacher efficacy beliefs, social, emotional, professional, and motivational capacities, coping behaviours and strategies, personal resources, vocational purpose and intrinsic value as well as self-care (Ainsworth & Oldfield, 2019; Beltman, 2020; Beltman et al., 2011, 2018; Cook et al., 2017; Day & Gu, 2013; Gu &



Day, 2007; Hong, 2012; Peixoto et al., 2018; Price et al., 2012; Sammons et al., 2007). A resilience lens, rather than stress and burnout, may lead to positive education outcomes through collective agency to generate knowledge on adaptive functionality despite adversity (Ebersöhn, 2017). It is, therefore, essential to attend to constructs, such as teacher resilience of pre-service teachers as important indicators of why teachers prevail despite hardship.

Self-efficacy and teacher efficacy is thus protective factors that may be mobilised as resilience-enabling intrapersonal pathways to support teachers to resile. In the current study, self-efficacy is operationalised as recovery from setbacks and teacher efficacy as confidence in teaching and behaviour management of teachers (Bobek, 2002; Castro et al., 2010; Day & Gu, 2013; Ee & Chang, 2010; Gu & Day, 2007; Peters & Pearce, 2012; Sammons et al., 2007; Tait, 2008).

### 1.6.5 SELF-EFFICACY

Self-efficacy is an intrapersonal resilience-enabling pathway to teacher resilience as discussed in Section 1.6.4. Therefore, the presence or absence of self-efficacy may enable or constrain resilience in a challenged context. As such, the concept is clarified in this section within the boundaries of the current study.

Self-efficacy is the belief the individual demonstrates to organise and achieve a required action to accomplish tasks (Bandura, 1977). Self-efficacy has correlating elements relating to required skills for performance and personal appraisal of competence to perceive a sense of control over events (Bandura, 1989a, 1989b; Schunk, 2008). As a future-oriented belief, self-efficacy influences thinking and feeling patterns which can support or hinder actions prediction of expected performance or competence (Bandura, 1997; Gavora, 2010; Snowman & McCown, 2013).

Bandura (1997) also stipulated that self-efficacy is a universal construct with a similar function across cultures. As literature (Gu & Li, 2013) postulates, the early definition of a resilient teacher as the capacity of the teacher to "bounce back from adversity" is inadequate to conceptualise the full complexity of teacher resilience. Therefore, the current study operationalised the resilience scale (Morgan, 2011) as self-efficacy. The use of extant data required that I modify goals in light of measurement constraints and definitions. Thus, self-efficacy, within the current study, is operationalised as the confidence final year pre-service teachers demonstrate in recovery from setbacks in a school setting (Morgan, 2011) as depicted by the item indicators displayed in Table 1.2. The level of measurement was ordinal and options on the Likert scale was one (absolutely not confident to seven) to seven (strongly confident).



#### Table 1.2

FIRE Teacher Resilience Measure: Self-Efficacy (Resilience) of Pre-Service Teachers

Self-efficacy (Resilience)	
Measure	Question
"How confident are pre-service teachers to deal with setbacks in school?"	Getting over setbacks in school
"How confident are pre-service teachers to deal with setbacks in school?"	Bouncing back, when things upset me
"How confident are pre-service teachers to deal with setbacks in school?"	Carrying on with my school work when things go wrong
"How confident are pre-service teachers to deal with setbacks in school?"	Carrying on in school when events upset me
"How confident are pre-service teachers to deal with setbacks in school?"	Feeling certain that things will come right even if there are serious problems in school
"How confident are pre-service teachers to deal with setbacks in school?"	Managing negative events in school when I try
"How confident are pre-service teachers to deal with setbacks in school?"	Coping with most problems on any school day
"How confident are pre-service teachers to deal with setbacks in school?"	Some negative things that have happened in school have made me better able to deal with problems
"How confident are pre-service teachers to deal with setbacks in school?"	Not getting disheartened even when children's circumstances make it difficult

Table 1.2 presents the self-efficacy (i.e., confidence in recovery from setbacks) variables measured by the Teacher Resilience scale (Morgan, 2011) from the FIRE Teacher Resilience Measure. These variables may have a likely relationship with teacher resilience and are included for analysis purposes in the current study.

### 1.6.6 TEACHER EFFICACY AND PRE-SERVICE TEACHER EFFICACY

As with self-efficacy (i.e., confidence in recovery from setbacks), teacher efficacy is an intrapersonal resilience-enabling pathway to teacher resilience as discussed in Section 1.6.4. Therefore, the presence or absence of teacher efficacy may enable or constrain resilience.

Teacher efficacy is not an objective measure of actual competence. Rather, it is the self-perceived belief a teacher demonstrates in their skills to accomplish a particular teaching task (Raath & Hay, 2016; Tschannen-Moran & Woolfolk Hoy, 2001). Therefore, teacher efficacy is a teachers' context-specific judgement or belief on how they will adapt, given the acquired skills or situation, to effectively accomplish a particular teaching task (Bandura, 1986, 1993, 1997; Tschannen-Moran et al., 1998). For the current study, teacher efficacy denoted final year pre-service teachers' confidence in teaching and behaviour management (Morgan, 2011; Peixoto et al., 2018) in a challenged context. Item indicators are displayed in Table 1.3.



#### Table 1.3

Teacher Efficacy				
	(TeachEff)			
Measure	Question			
"Teacher-efficacy for teaching"	Teaching all the subjects on the curriculum effectively			
"Teacher-efficacy for teaching"	Explaining difficult material in ways that the children will understand it			
"Teacher-efficacy for teaching"	Suggesting suitable examples when the children are having difficulty understanding			
"Teacher-efficacy for teaching"	Teaching in a way that my students <sup>18</sup> will remember important information			
"Teacher-efficacy for teaching"	Applying the new developments in the curriculum into my teaching			
"Teacher-efficacy for teaching"	Helping children focus on learning tasks and avoid distractions			
"Teacher-efficacy for behaviour management"	Managing inappropriate behaviour			
"Teacher-efficacy for behaviour management"	Encouraging students to take responsibility for their behaviour			
"Teacher-efficacy for behaviour management"	Dealing with the diverse learning needs of the students in my class			
"Teacher-efficacy for behaviour management"	Teaching students positive behaviour			
"Teacher-efficacy for behaviour management"	Providing students with clear specific behaviour expectations			
"Teacher-efficacy for behaviour management"	Communicating effectively with parents <sup>19</sup>			

FIRE Teacher Resilience Measure: Teacher Efficacy of Pre-Service Teachers

Table 1.3 presents the teacher efficacy variables (measured by the Teacher Efficacy scale [Morgan, 2011; Peixoto et al., 2018]) from the FIRE Teacher Resilience Measure. These variables, which may have a likely relationship with teacher efficacy, are included for analysis purposes in the current study. The level of measurement was ordinal and options on the Likert scale was one *(absolutely not confident)* to seven *(strongly confident)*.

Examining the teacher efficacy beliefs of pre-service teachers is important since beliefs seem pliable in the beginning teaching years (Bandura, 1977, 1997) and may be resistant to change once established (Lemon & Garvis, 2016). Hence, it is necessary to explore the efficacy beliefs of pre-service teachers while in training (Lemon & Garvis, 2016). The current study will operationalise pre-service teacher efficacy as the self-perceived confidence final year pre-service teachers report in their teaching and behaviour management (Mansfield & Wosnitza, 2015; Morgan, 2011; Peixoto et al., 2018) within a challenged context. The sample of pre-service teachers in the current study was N = 1,193, with 77.9% female pre-service teacher respondents and 18.7% male pre-service teacher respondents.

<sup>&</sup>lt;sup>18</sup> The term "student", as used in the original ENTREE resilience questionnaire (Mansfield & Wosnitza, 2015; Peixoto et al., 2018; Wosnitza et al., 2014), was depicted in the FIRE Teacher Resilience Measure despite the convention in South Africa to make reference to "learners" within the basic education field. <sup>19</sup> In South Africa, parent denotes any "natural of legally entitled custodian, parent or guardian of a learner" (SACE, 2017, p. 3).



## 1.6.7 THE FIRE TEACHER RESILIENCE MEASURE

The FIRE Teacher Resilience Measure was piloted as one of the baseline data generation questionnaires in the FIRE project. The FIRE Teacher Resilience Measure (structured, self-report pen and paper questionnaire) includes items from the measures used in the FIT-Choice scale (Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012), ENTREE project (Mansfield & Wosnitza, 2015; Peixoto et al., 2018; Wosnitza et al., 2014) and contextual resilience questions (Coetzee, 2013). The contextual resilience questions (Coetzee, 2013). The contextual resilience questions (Coetzee, 2013) were included for adaptation of the ENTREE scales given the South African context.

The FIRE Teacher Resilience Measure encompasses domains on (i) teacher professionalism (TR-Prof), (ii) teacher emotion (TR-Emot), (iii) teacher motivation (TR-Mot), (iv) teacher social capacity (TR-Soc), (v) resilience (Resilience) and (vi) teacher efficacy (TeachEff) as well as (vii) contextual resilience questions (Coetzee, 2013; Morgan, 2011; Peixoto et al., 2018; Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012)<sup>20</sup>. Respondents rate their responses to questions on a seven-point Likert scale ranging from one (i.e., *do not agree at all* or *absolutely not confident*) to seven (i.e., *strongly agree* or *strongly confident*).

## 1.6.8 INITIAL TEACHER EDUCATION AND SOUTH AFRICAN PROGRAMME PHASES<sup>21</sup>

Initial teacher education, preparation, development, or training programmes denote the procedures, policies, structures and processes established to prepare new teachers with information, behaviours and abilities required to perform teaching tasks in the classroom, school and society (DBE, RSA, 2019; OECD, 2019a; RSA, 2000; W. Taylor, 2016). Teacher education includes formal programmes that have been established by institutions for the preparation of future teachers at the early childhood development (ECD), primary/elementary or secondary school levels (W. Taylor, 2016).

The United Nations Educational, Scientific and Cultural Organization (UNESCO) developed the International Standard Classification of Education (ISCED). The ISCED (Level 1-8) is the official framework used to facilitate transnational comparisons of education systems (UNESCO Institute for Statistics (UIS), 2012; World Bank, 2020a). Within the current study, initial teacher education is operationalised as the 4-year BEd programme degree pre-service teachers are enrolled in at the University of Pretoria (public Higher Education Institution [HEI] in South Africa). The BEd teaching degree will be equivalent to the ISCED level 6 (i.e., Bachelors degree or equivalent tertiary education level) internationally and National Qualifications Framework (NQF) level 8 (Professional 4-year Bachelor degree) nationally. The fulfilment of degree requirements will lead to registration at the South African Council of

<sup>&</sup>lt;sup>20</sup> The current study focused on the Teacher Resilience scale (operationalised as self-efficacy as explained in Section 1.6.5) and the Teacher Efficacy scale on the FIRE Teacher Resilience Measure.
<sup>21</sup> Phase depicts the level of schooling or teacher specialisation in South Africa (DBE, RSA, 2019).



Educators (SACE) (DBE, RSA, 2019; RSA, 2000). In South Africa, the South African Qualifications Authority (SAQA), governed by the South African Qualifications Authority Act (No. 58 of 1995), started operations in May 1996 and was tasked with the establishment of the NQF. Level descriptors provide a category of 10 levels on the NQF achievement unit standard or qualification.

The role of the oversight body SAQA, as stipulated in the NQF Act (No. 67 of 2008), is fundamentally to advance the development and implementation of the NQF by acting as custodian of its values and quality character to ensure that standards and qualifications are comparable worldwide (SAQA, 2020). Furthermore, in South Africa, the Education Labour Relations Council (Resolution 8 of 2003) governs teacher accountability (Shalem & Hoadley, 2009). The Education Labour Relations Council introduced the Integrated Quality Management System (IQMS) (Shalem & Hoadley, 2009) regulating teachers' daily work, quarterly schedules, curriculum management systems as well as formal assessment undertakings (Shalem & Hoadley, 2009). Nationally, the Department of Basic Education (DBE) (i.e., schools from Grade R-12 and adult literacy) and the Department of Higher Education and Training (DHET) (i.e., higher education including universities and post-school training) shares the responsibility for education (DBE, RSA, 2018a; DHET, RSA, 2018b; RSA, 2019).

Enrolled pre-service teaching programmes within the BEd degree are conceptualised as the *Foundation phase/Early childhood development FP/ECD* (i.e., Grade R–3), *Intermediate phase* (IP) (i.e., Grade 4–6), *Senior phase* (SP) (i.e., Grade 7–9) or *Further education and training phase* (FET) (i.e., Grade 10–12). The sample of enrolled teaching programmes for the current study was FP/ECD n = 311, IP n = 157, SP n = 101, FET n = 436 and not specified/other category n = 136. The SA teaching phases (i.e., FP/ECD, IP, SP and FET) within the basic education sector (DBE, RSA, 2019) compare as follows to the NQF and ISCED levels:

#### Table 1.4

ISCED level	ISCED classification	SA teaching phase <sup>22</sup>	SA teaching grades	Learner enrolment 2019 <sup>23</sup>	SA teaching age <sup>24</sup>	SA NQF
	Ge	neral educatior	n and training (G	GET) band <sup>25</sup>		
ISCED 0	Early childhood education	Foundation phase (FP)	Grade R <sup>26</sup> -3	3,958,809	±5 – 9- years	NA
ISCED 1	Primary education	Foundation phase (FP)	Grade R-3			NA

Comparison of ISCED Levels with South African (SA) Teaching Phases

<sup>&</sup>lt;sup>22</sup> Phase depicts the level of schooling or teacher specialisation in South Africa (DBE, RSA, 2019).

<sup>&</sup>lt;sup>23</sup> Number of leaners by phase in public ordinary schools in South Africa (DBE, RSA, 2020).

<sup>&</sup>lt;sup>24</sup> Age ranges are clustered into phases as guidelines (DBE, RSA, 2020).

<sup>&</sup>lt;sup>25</sup> Programmes that lead to a Level 1 (NQF) which include Grades R to Grade 9 (i.e., FP, IP and SP) (DBE, RSA, 2018c).

<sup>&</sup>lt;sup>26</sup> Year preceding Grade 1 (DBE, RSA, 2018c).



ISCED level	ISCED classification	SA teaching phase <sup>22</sup>	SA teaching grades	Learner enrolment 2019 <sup>23</sup>	SA teaching age <sup>24</sup>	SA NQF
	Ge	neral educatior	n and training (G	ET) band <sup>25</sup>		
		Intermediate phase (IP)	Grade 4-6	3,127,319	±10 – 12- years	NA
ISCED 2	Lower secondary education	Senior phase (SP)	Grade 7-9	2,874,651	±13 – 15- years	NQF 1: Gr 9
Further education and training (FET) band <sup>27</sup>						
ISCED 3	Upper secondary education	Further education	Grade 10-12	2,437,719	±16 – 18- years	NQF 2: Gr 10 NQF 3: Gr 11
		and training phase (FET)				NQF 4: Gr 12

(DBE, RSA, 2018a, 2018c, 2020; UIS, 2012; World Bank, 2020a)

As indicated in Table 1.4, the GET band with a total of 9,960,779 learners had a higher learner count than the FET band. In addition, the FP had the highest proportion of learners with the succeeding levels within the schooling system, showing a proportional decrease in learners (i.e., FET phase with the lowest enrolment number) (DBE, RSA, 2018a, 2018c, 2020).

### 1.6.9 OPERATIONALISING PRE-SERVICE TEACHERS

A teacher denotes a professional individual who conveys information by assisting, intervening, and directing learners in an educational process (Fraser, 2006; OECD, 2017b). Within the South African context, a teacher is an individual registered or conditionally registered with the SACE (RSA, 2000). Pre-service teachers, also known globally as teacher candidates, novice teachers or student teachers (Le Cornu, 2009), describe students who are enrolled in initial teacher education, training or preparation programmes working towards teacher qualification or certification (DBE, RSA, 2019; RSA, 2000; W. Taylor, 2016) to become a qualified teacher.

Pre-service teachers complete supervised teaching practice (i.e., field-based teaching, field-based practical; field experience, field placement, professional experiences, school experience, student teaching or practicum), as an integral stringent component of teacher training, with the support and mentorship of the enrolled education faculty or teaching institution and cooperating or mentor teachers in a placement school (Beltman et al., 2018; Faculty of Education, University of Pretoria, 2016; Hewitt et al., 2017; IGI Global, 2020; Le Cornu, 2009; Lieberman & Darling-Hammond, 2012; Moulding et al., 2014; W. Taylor, 2016; Thieman et al., 2014).

Within the current study, the construct of a pre-service teacher will be operationalised as a final fourth-year teacher in training at the University of Pretoria, Faculty of Education, South

<sup>&</sup>lt;sup>27</sup> Programmes leading that lead to Levels 2 to 4 (NQF) which represents Grades 10 to Grade 12 at ordinary schools (DBE, RSA, 2018c).



Africa, enrolled in the BEd programme between 2015–2017 (Faculty of Education, University of Pretoria, 2016). At the University of Pretoria, the graduation numbers for the BEd programme, illustrated by Higher Education Data Analyzer (HEDA, n.d.) data, for 2015 to 2018 were n = 679 (2015); n = 751 (2016); n = 794 (2017) and n = 898 (2018) respectively. The BEd programme includes four<sup>28</sup> official teaching practice modules. The practice modules enable BEd-students to complete their work-integrated learning hours and provide practical experience in professional school environments which enables pre-service teachers to be work-ready and employable upon completion of their degree.

Teaching practice modules entail a period of work-integrated learning hours at school placements to provide practical experiences (e.g., observations, school administration, extramural activities (second year), preparation and facilitation of lessons (third and fourth year). Both second-year and third-year students spent 3 weeks at school placements, while fourthyear students spent 16 weeks (8 weeks in the second school term and 8 weeks in the third school term). Mentor teachers (school placement) mentor lecturers (university) assess the students. The teaching practice modules aim to address the fissure between theory and practical aspects of education training. In Table 1.5, I provide a summary of the pre-service teacher sample (based on information from the demographic questionnaire) that generated the extant FIRE teacher resilience data.

<sup>&</sup>lt;sup>28</sup> Teaching practice 280 is a 3-week observational placement in schools for second-year BEd-students. Teaching practice 380 is a 3-week supervised teaching placement for third-year BEd-students. Teaching practice 452 and teaching practice 453 are an 8-week teaching placement each for fourth-year BEd students.



#### Table 1.5

Overview (Statistics) of Demographic Information of Extant Data

	YEAR	2015	2016	2017	Total
DESCRIPTOF	RS				
Total question	naires completed by final year pre-	313	169	711	1,193
service teache	ers	(26.2%)	(14.2%)	(59.6%)	(100%)
Age between	(years)	21–30	20–29	20–32	20–32
Gender	Gender male (indicated)	61	21	141	223
		(19.5%)	(12.4%)	(19.8%)	(18.7%)
	Gender female (indicated)	250	140	539	929
		(79.9%)	(82.8%)	(75.8%)	(77.9%)
Languages	Afrikaans	200	98	327	625
fluent in <sup>29</sup>		(63.9%)	(58.0%)	(46.0%)	(52.4%)
	English	297	153	653	1,103
		(94.9%)	(90.5%)	(97.8%)	(92.5%)
	isiNdebele	18	5	32	55
		(5.8%)	(3.0%)	(4.5%)	(4.6%)
	isiXhosa	8	1	30	39
		(2.6%)	(0.6%)	(4.2%)	(3.3%)
	isiZulu	75	21	197	293
		(24.0%)	(12.4%)	(27.7%)	(24.6%)
	Sepedi	34	14	113	161
		(10.9%)	(8.3%)	(15.9%)	(13.5%)
	Sesotho	20	4	58	82
		(6.4%)	(2.4%)	(8.2%)	(6.9%)
	Setswana	29	8	80	117
		(9.3%)	(4.7%)	(11.3%)	(9.8%)
	SiSwati	20	3	56	79
		(6.4%)	(1.8%)	(7.9%)	(6.6%)
	Tshivenda	3	0	7	10
		(1.0%)	(0.0%)	(1.0%)	(0.8%)
	Xitsonga	5	2	18	25
		(1.6%)	(1.2%)	(2.5%)	(2.1%)
	Other	7	8	19	34
		(2.2%)	(4.7%)	(2.7%)	(2.8%)
Enrolled	Foundation phase (FP)/Early	105	45	161	311
pre-service	childhood development (ECD)	(33.5%)	(26.6%)	(22.6%)	(26.1%)
teaching	Intermediate phase (IP)	32	17	108	157
programmes		(10.2%)	(10.1%)	(15.2%)	(13.2%)
	Senior phase (SP)	30	14	57	101
		(9.6%)	(8.3%)	(8.0%)	(8.5%)
	Further education and training	110	60	266	436
	(FET) phase	(35.1%)	(35.5%)	(37.4%)	(36.5%)
	Not specified/Other	25	20	91	136
		(8.0%)	(11.8%)	(12.8%)	(11.4%)

<sup>&</sup>lt;sup>29</sup> 11 Official language distribution in South Africa: Afrikaans (12.1%), English (8.3%), isiNdebele (1.6%), isiXhosa (17.0%), isiZulu (24.6%), Sepedi (9.5%), Sesotho (8.0%), Setswana (8.8%), SiSwati (2.6%), Tshivenda (2.4%) and Xitsonga (4.2%). Languages are listed alphabetically.



As illustrated in Table 1.5, a total of 1,193 pre-service teachers, between the age ranges of 20 years to 32 years, completed the FIRE Teacher Resilience Measure from 2015 to 2017. Of the total questionnaires completed, 223 (18.7%) of the respondents included male pre-service teachers and 929 (77.9%) female pre-service teachers. The cohort (2015–2017) final year preservice teachers indicated their language of proficiency<sup>30</sup> as follows: English n = 1,103; Afrikaans n = 625; isiZulu n = 293; Sepedi n = 161; Setswana n = 117; Sesotho n = 82; siSwati n = 79; isiNdebele n = 55; isiXhosa n = 39; Xitsonga n = 25 Tshivenda n = 10; and "other" n = 34. Based on the statistics, English (n = 1,103) is seemingly indicated as the language most final year pre-service teachers feel proficient in, with Tshivenda (n = 10) being the smallest group regarding language fluency. With regards to the enrolled pre-service teaching programmes of teacher training, the largest group of respondents was the FET phase (n = 436), with the FP/ECD phase (n = 311) as the second-largest group. The IP included n = 157 respondents with the SP as the smallest group (n = 101), while 136 respondents did not indicate a specific programme phase (i.e., phase not specified/other n = 136).

### 1.6.10 CHALLENGED EDUCATION CONTEXT

In the current study, a challenged educational context denotes a space where teachers' working circumstances are complicated by systemic chronic and cumulative risk factors such as poverty, crime, health problems, poor infrastructure, inadequate teacher training, a continuously changing education system as well as inadequate service delivery. Risk encapsulates challenges, strains, barriers, and stressors that pose a substantial threat to development (Clement, 2017; Masten, 2001) and can be characterised as either internal (i.e., intrapersonal or traits) or external (i.e., interpersonal and ecology) (Ebersöhn, 2014; Mansfield et al., 2018). In a challenged education context, negative outcomes are expected or predicted. Such educational outcomes include teacher stress, teacher burnout, teacher attrition and low-quality education (Clement, 2017; Ebersöhn, 2017).

Although such circumstances and risks may be universal, the focus of a place-based discourse in the current study is on South Africa as an exemplar of a Global South context (i.e., a country outside Europe and the privileged Global North, developing or low- and middle-income countries) (Dados & Connell, 2012; Grovogui, 2011; Montiel, 2018; Trefzer et al., 2014). Challenged Global South countries share a history of colonial occupation and, as a result, had to adapt to marginalisation, disparities in living standards and limited access to resources (Dados & Connell, 2012; Ebersöhn, 2017; Ebersöhn & Loots, 2017; Grovogui, 2011;

<sup>&</sup>lt;sup>30</sup> Respondents could indicate numerous languages in the demographic question pertaining to language fluency since it was an open-ended question (see Appendix B for the FIRE Teacher Resilience Measure). South Africa has 11 official languages with most of the South African population speaking at least two of the official languages (RSA, 2019; Stats SA, 2018). General language fluency can be defined as the ability, proficiency, ease, confidence or accuracy with which communicative language (oral, reading and writing) is used (Chambers, 1997; González, 2008).



Montiel, 2018). In addition, a challenged context can encapsulate various compounding dimensions like poverty, environmental degradation, rural communities, inner-city or ghetto areas, illiteracy, inequality, gender discrimination, social dislocation or displacement, vulnerability, limited educational opportunities and poor health care systems (Ansfield, 2018; Dados & Connell, 2012; Ebersöhn, 2016; Francis & Webster, 2019; Grovogui, 2011; IMF, 2004; Montiel, 2018; Sayed & Badroodien, 2016; Stats SA, 2020a; Wilson, 2012; Wissing et al., 2013). Within the current study, South Africa is thus operationalised as a challenged education space within a Global South setting denoting an emerging economy in transition, high adversity and inequality as well as inherited structural disparity due to a post-colonial history (Dados & Connell, 2012; Ebersöhn, 2014, 2017).

## 1.7 THEORETICAL FRAMEWORK

## 1.7.1 INTRODUCTION

Given the extant data available, the current study used a Social Cognitive Theoretical (i.e., Bandura, 1986) framework to compare the self-efficacy and teacher efficacy of final year preservice teachers in a challenged context to highlight intrapersonal resilience-enabling pathways to teacher resilience. The FIRE Teacher Resilience Measure scales used in the current study are grounded in the conceptual underpinning of the self-efficacy theory of Bandura (1977). Therefore, the Social Cognitive Theory seemed well suited to understand preservice teachers' personal development and adaptation to investigate the current study's potential contribution to the phenomenon of teacher resilience. Framed within Bandura's (1986) Social Cognitive Theory, research on pre-service teachers has provided the scholarly field with various theoretical predictors (Legrain et al., 2019; McClellan, 2017; Meaney et al., 2008; Rubenstein et al., 2018). Consequently, the next sections discuss the basic assumptions of the Social Cognitive Theory and the view on self-efficacy.

## 1.7.2 SOCIAL COGNITIVE THEORY: BASIC ASSUMPTIONS

The current study embraced the Social Cognitive Theory (SCT) (Bandura, 1986), as a theoretical framework to gain insight into pre-service teachers' reality construction regarding self-efficacy and teacher efficacy as protective resources enabling teacher resilience. In this regard, pre-service teachers participating in the FIRE project, as discussed in Section 1.2, entered the FIRE project with unique determinative factors (e.g., history, background, world views, gender, and culture) (Bandura, 2002). The Social Cognitive Theory, postulated by Albert Bandura (1986, 1989a, 1989b), an eminent psychologist (Diener et al., 2014), underlined the notion that adaptation occurs in a reciprocal manner between the individual, ecology, and behaviour. (Bandura, 1986, 1989a, 1989b; Schunk, 2008) as depicted in Figure 1.4.



#### Figure 1.4

System of Triadic Reciprocal Causation Interaction in SCT



Adapted from Bandura (1986)

The interrelationship between three sets of cooperating determinants (i.e., environment, individual and behaviour) actively shape each other. The bidirectional interaction is demonstrated in Figure 1.4 (Bandura, 1986, 1989a, 1989b; Schunk, 2008). However, according to Bandura (1997), this process of triadic reciprocality is not indicative of an interrelationship strength equality since the interacting elements (i.e., environmental, interdependent behaviours and personal factors) can vary (Bandura, 1997). In this transactional stance, the relationship between the self and society, is not seen as independent but rather reciprocally determined (Bandura, 1986, 1989a, 1989b). Bandura (1997) therefore argued for a dualistic integrated causal perspective with influencing co-factors. Consequently, contributing determinants develop over time, enabling perception of the reciprocal nature (Bandura, 1997).

From the social cognitive view, human potential is directed, within biological limits, through various enactive as well as observational experiences (Bandura, 2002; Schunk, 2008) not only driven by inner forces or controlled by external stimuli (Bandura, 1986). Furthermore, previous experiences shape new experiences to create future behaviours. Behaviours providing effective outcomes are maintained while unsuccessful behavioural consequences are changed or discarded (Bandura, 2002; Schunk, 2008). Therefore, individuals are seen as pro-active, self-influencing, self-reflecting as well as self-regulating agents (Bandura, 1977, 1989a, 1989b, 1997).



Self-efficacy is a dominant component in the Social Cognitive Theory since Bandura considered the construct as the foundation of individual agency (i.e., potential control individuals have over their behaviour) and asserted that the feelings of personal efficacy are important in behaviour (Bandura, 1977, 1986, 1997; Bandura et al., 1977; Snowman & McCown, 2013; Yost, 2016). The sense of efficacy influences "whether people think optimistically or pessimistically, act in ways that are beneficial or detrimental to achieving goals, approach or avoid tasks, engage tasks with a high or low level of motivation, persevere for a short or lengthy period when tasks are difficult, and are motivated or demoralized by failure" (Snowman & McCown, 2013, pp. 188–189). Therefore, the following section explores the construct of self-efficacy as part of the Social Cognitive Theory.

### 1.7.3 SELF-EFFICACY THEORY

The self-efficacy theory (Bandura, 1977, 1986, 2001) is constructed as a central feature within the Social Cognitive Theory (SCT). Bandura in 1986 (p. 391) introduced the concept of self-efficacy as "one's capability to accomplish a given level of performance". He defined the construct as "beliefs in one's capabilities to organise and execute the courses of action required to produce given attainments" (Bandura, 1997, p. 3) or the "conviction that one can successfully execute the behaviour required to produce a given outcome" (Bandura, 2006, p. 193). This belief is based on the reciprocal nature (i.e., Figure 1.4) of individual, socio-ecological and enactive as well as vicarious courses of behavioural change (Bandura, 1986).

Within the social cognitive paradigm, self-efficacy is thus characterised as an individuals' conviction, judgment and appraisal of competence or ability to initiate, manage, perform, or employ necessary courses of actions or behaviour needed to maintain or produce an anticipated specific outcome (Bandura, 1977, 1997). Bandura (2006) did not consider self-efficacy as a static attribute, but as an active changing socio-structural belief embedded in thinking, feeling, motivational and selection states. According to Bandura (1997), self-efficacy is a universal construct with a similar function across cultures. Studies (Bandura & Adams, 1977; Bandura et al., 1977, 1980; Eastman & Marzillier, 1984) support the notion that self-efficacy ratings are predictors of behavioural competence on subsequent tasks.

When an individual accurately recognises their abilities, it may be possible to obtain a valid self-efficacy appraisal (Chesnut, 2017). Furthermore, Bandura (1986) made a distinction between self-efficacy (i.e., belief that an action can be successfully performed) and outcome expectations (i.e., that an action may lead to a positive result) that motivate action. As a result, sense of efficacy can influence goal or task selection, perseverance the expected outcome and the attribution assigned to success or failure to establish perceived competence (Bandura 1977, 1997; Gonzalez-DeHass & Willems, 2016; Lemon & Garvis, 2016; Snowman & McCown, 2013; Wang et al., 2015).



Bandura (1977, p. 191) claimed that "expectations of personal efficacy determine whether coping behaviour will be initiated, how much effort will be expended, and how long it will be sustained in the face of obstacles and aversive experiences". Therefore, according to Bandura (1986, 1993, 1996, 1997, 2006), the sense of efficacy is a powerful predictor of behaviour, regulating action and self-organisation, to guide self-reflection and to rebound from setbacks. Self-efficacy can impact performance, the ability to cope with potentially aversive events or the optimism demonstrated when faced with obstacles (Bandura, 1977, 2006; Hewitt et al., 2017).

Perceived self-efficacy, as a cognitive mechanism, enable direct or intentional control over environmental opportunities, impediments or (stressful) events that may influence action (Bandura, 1986, 1993, 1996, 1997; Lemon & Garvis, 2016). As a result, it is important to acknowledge that it is the perception that governs self-efficacy beliefs and not reality (Hewitt et al., 2017). Bandura's theory (1997) outlined self-efficacy as a personal belief in competence ability interpreted from specific sources. Individuals gain important information from these efficacy sources that influences the development of their competency convictions (Tschannen-Moran & Woolfolk Hoy, 2001). In this regard, Bandura (1977, p. 195) identified four sources of efficacy beliefs, including "performance accomplishments, vicarious experiences, verbal persuasion, and physiological states". Among the sources of efficacy, performance accomplishments or enactive mastery experiences are argued to have the strongest weight on the formation of self-efficacy as an authentic indicator of individual ability (Bandura, 1997; Pfitzner-Eden, 2016a).

Through everyday behaviours, individuals obtain information and attribute success or failure outcomes (Bandura, 1977, 1986, 1997, 2006; Lent & Brown, 2006). Successful outcomes may strengthen their self-efficacy beliefs and aid expectations in future competencies (Bandura, 1977, 1986, 1997). Efficacious individuals often expect a positive controllable outcome and tend to employ more complex skills to persevere in challenging tasks (Bandura, 1997; Snowman & McCown, 2013). On the other hand, failures are likely to reduce expectations about competency and self-efficacy (Bandura, 1977, 1986, 1997). However, weak efficacy beliefs may be negated by disconfirming experiences regarding a task or in a particular environment (Bandura, 1997).

Self-efficacy and the perceived competency may not be affected by increased stress if the individual believes that they can succeed (Gonzalez-DeHass & Willems, 2016; Klassen et al., 2011). So, even if there is adversity, self-efficacy can act as a "protective factor" or buffer (Hewitt et al., 2017) that "counteract" adversity or lead to resilience. Therefore, the Social Cognitive Theory seemed applicable to this study since it provided a lens to investigate protective factors that enable intrapersonal pathways to teacher resilience in a challenged context.



Critics (Biglan, 1987; Borkovec, 1978; Eastman & Marzillier, 1984; Kazdin, 1978; Teasdale, 1978; Tryon, 1981) however, adduced limitations in Bandura's self-efficacy theory including ambiguous definitions, methodological deficiency and inadequate evaluation or methods to substantiate conclusions. Scholars (Borkovec, 1978; Eastman & Marzillier, 1984; Kazdin, 1978; Teasdale, 1978) argued that conceptually efficacy expectations and outcome expectations, as described by Bandura (1977, p. 193) as "the conviction that one can successfully execute the behaviour to produce the outcomes" cannot be differentiated in the application of the theory. The behaviour-analytic alternative, although not denying the empirical relationships presented in support of self-efficacy, explains the association in terms of various variables (Biglan, 1987). Eastman and Marzillier (1984) noted that empirical findings might be less impressive if the circumscribed nature of the behavioural tasks is recognised, since self-efficacy theory may over-simplify the variables involved in behaviour change. As a result, the correlations between self-efficacy rating behaviour and other behaviour may be due to behavioural reinforcement (Biglan, 1987).

Tryon (1981) furthermore implied that Bandura's conclusions could likely be accounted for by social contingencies operating within highly structured behavioural approach situations, which may highlight methodological deficiencies in Bandura's centrality of self-efficacy in the process of behaviour change. Furthermore, the scale and methods for assessing self-efficacy in Bandura's experimental studies were subject to criticism (Biglan, 1987; Eastman & Marzillier, 1984) since the evidence consisted of associations between self-efficacy ratings and other behaviours which did not determine causation. Therefore, the scale seemed abstruse and could be misinterpreted based on theoretical association (Eastman & Marzillier, 1984). The literature (Eastman & Marzillier, 1984) thus caution against the elevation of selfefficacy to a grandiose status but support the construct as one likely factor in the explanation of human behaviour and the assessment of personal competence as a determinant of future behaviour.

Within the current study, the construct of self-efficacy was therefore clearly operationalised (see Section 1.6.5) given limited outcomes to pre-service teacher confidence of recovery from setbacks as well as confidence in teaching and behaviour management to address the limitations noted by scholars, given precise definitions and outcome expectations. Furthermore, the FIRE Teacher Resilience Measure, based on the conceptual premise of self-efficacy (Morgan, 2011), has been validated through numerous studies within the teaching landscape. As a result, the Social Cognitive Theory was utilised in this study to compare the self-efficacy and teacher efficacy of pre-service teachers in challenged contexts to recognise traits as possible intrapersonal resilience-enabling pathways to teacher resilience, which allow pre-service teachers to navigate a challenged context. The next section provides a brief overview of the paradigms, methodologies, standard of rigour and ethical considerations for this study.



# 1.8 BRIEF OVERVIEW OF PARADIGMS, METHODS, STANDARD OF RIGOUR AND ETHICAL CONSIDERATIONS

In this section, I present a summary of the research design, paradigmatic assumptions, methods, standards of rigour and ethical strategies relating to the current study. I substantiate methodological decisions in Chapter 3. Figure 1.5 provides an overview of the research process based on methodological decisions employed.

### Figure 1.5

Overview of the Current Study



<sup>&</sup>lt;sup>31</sup> The FIRE Teacher Resilience Measure included items from FIT-Choice (Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012), ENTREE (Mansfield & Wosnitza, 2015; Peixoto et al., 2018; Wosnitza et al., 2014) and contextual resilience questions (Coetzee, 2013).



The comparative analysis of *quantitative* extant data is framed as a *post-positivist* study (Creswell, 2014). This choice was appropriate since quantitative research abides by measurable aspects of respondents' (i.e., pre-service teachers) behaviour (Van Rensburg et al., 2010) to determine and address inquiries about the relationships between the variables under investigation (Leedy & Ormrod, 2015; Van Rensburg et al., 2010) through statistical analysis (Ary et al., 2019; Creswell & Creswell, 2018).

I used a *comparative case study design* (Babbie, 2021; Mills et al., 2006; G. Thomas, 2011; Yin, 2018; Zartman, 2012) to compare cohorts (Case 1 [2015], Case 2 [2016] and Case 3 [2017]) of final year pre-service teachers through secondary data analysis (Babbie, 2021; Johnston, 2017; Widaman et al., 2011). Extant data was purposively selected (Babbie, 2021) to perform nonparametric descriptive and inferential statistical procedures on data obtained from completed (N = 1,193) questionnaires (i.e., FIRE Teacher Resilience Measure) (Corder & Foreman, 2014; Leedy & Ormrod, 2015).

Since L. Cohen et al. (2018, p. 173) denoted that a "research design is governed by fitness for purpose" the current study employed a comparative case study design (Babbie, 2021; G. Thomas, 2011; Yin, 2018; Zartman, 2012) to analyse extant cross-sectional data. A comparative case study design allowed for a comprehensive description (L. Cohen et al., 2018; Creswell & Creswell, 2018) of the *self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context* to find what is collective and distinct in cohorts of cases (Case 1, Case 2, and Case 3) spanning over multiple cross-sectional (2015, 2016 and 2017) points. A strength of a comparative case study design is the logic of comparison approximating inferences allowing clearer hypotheses (Babbie, 2021; L. Cohen et al., 2018) to quantitatively compare the self-efficacy and teacher efficacy beliefs of pre-service teachers.

A comparative case study is also situated in natural, real-life dynamic settings and positioned to extensively compare complex real-world multiple-cases within given parameters (L. Cohen et al., 2018; J. L. Taylor, 2013; Yin, 2018). Such case studies lead to in-depth and reliable findings (Yin, 2018; Zartman, 2012). On the other hand, discontinuities in selecting an appropriate sample are a limitation of a comparative case study design. Such constraints are associated with differences in variables that may hamper the identification of a single variable's distinct association (Babbie, 2021). Due to context-dependent knowledge gained from a comparative case study design allowed for the interpretation of identified variance across variables (i.e., within-case and cross-case comparison) (G. Thomas, 2011). In addition, the rigour of the current study could not be established separately.

I utilised numerous strategies to address the quality assurance of quantitative research. I used the following standards of rigour, namely reliability analysis (Babbie, 2021; Creswell, 2014; Foxcroft & Roodt, 2013; Gravetter & Forzano, 2018; Sauro & Lewis, 2016), validity analysis (Babbie, 2021; Foxcroft & Roodt, 2013; Gravetter & Forzano, 2018; Sauro & Lewis,



2016) and statistical power analysis (L. Cohen et al., 2018; Cooper, 2018; Creswell & Creswell, 2018; Faul et al., 2007; Field, 2018; Sauro & Lewis, 2016; Simon, 2011). Further strategies I included the quality assurance of the measurement instrument (i.e., FIRE Teacher Resilience Measure) (Peixoto et al., 2018; Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012) and establishing secondary data analysis as a systematic empirical method of enquiry (Creswell, 2014; Johnston, 2017; Leedy & Ormrod, 2015; Neuman, 2014; Rubin & Babbie, 2014).

Finally, the ethical guidelines included access and permission to utilise extant data (University of Pretoria, 2015) and, the privacy of extant data (Creswell, 2014; Leedy & Ormrod, 2015). I also emphasised ethical reporting of extant data as (Babbie, 2021; Creswell, 2014; Leedy & Ormrod, 2015) as ethical consideration. The next section provides a summary of the research findings for the current study.

## 1.9 SUMMARY OF THE RESEARCH FINDINGS

In Table 1.6, a synopsis of the findings of the current study is provided. Although the results and findings are discussed in detail in Chapter 4 and Chapter 5, respectively, I provide an outline here to conclude the initial overview of the thesis.



#### Table 1.6

Summary of Research Findings

1 How do self-efficacy and teacher efficacy of final year preservice teachers within a challenged education context (within-case and cross-case) compare?

•

The *self-efficacy* (i.e., confidence in recovery from setbacks in school) and *teacher efficacy* (i.e., confidence in teaching and behaviour management) of final year pre-service teachers *in a challenged context* is high. This result reinforces the presence of intrapersonal resilience-enabling pathways (i.e., intrapersonal protective resources) identified by studies in a challenged context (Bosch, 2020; Coetzee et al., 2017; Ebersöhn, 2014; Ebersöhn et al., 2020; Mansfield et al., 2018; Wabule, 2020; Yokus, 2015). Although Ee and Chang (2010) found that pre-service teachers in Singapore were not adequately prepared to deal with inevitable setbacks, the current study's results support previous high-income countries findings (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018, 2020; Wosnitza et al., 2018) on the presence of self-efficacy and teacher efficacy as protective intrapersonal resources.

- The teacher efficacy of final year pre-service teachers is significantly • higher (within-case and cross-case) compared to final year preservice teachers' self-efficacy in a challenged context. Therefore, final year pre-service teachers are relatively more confident in teaching and behaviour management than in dealing with setbacks. This result provided insight into the confidence pre-service teachers have in recovery from setbacks and their confidence in teaching and behaviour management when facing chronic and cumulative risk factors. Literature supports the significance of teacher efficacy for teachers' recovery from setbacks and buffering against burnout in a challenged context (Bosch, 2020; Ebersöhn et al., 2020; Jackson & Rothmann, 2005) as well as in developed settings (Beltman et al., 2011, 2018; Day & Gu, 2013; Gu & Day, 2007; Gu & Li, 2013; Hong, 2012; Peixoto et al., 2018; Pendergast et al., 2011; Price et al., 2012; Sammons et al., 2007; Tournaki et al., 2009; Yada et al., 2021).
- Pre-service teachers' confidence in their teaching and behaviour management thus enables them to recover from setbacks despite high adversity (Bandura, 1986, 1993, 1996, 1997; Hewitt et al., 2017; Perkins-Gough, 2013; Waddell, 2007) in a challenged context (Bosch, 2020; Ebersöhn et al., 2020). Teacher efficacy was found to be essential for recovery from setbacks in previous high-income country studies for pre-service teachers (Morgan, 2011; Peixoto et al., 2018).



2 What are the selfefficacy and teacher efficacy of pre-service teachers within a challenged education context based on demographic information (withincase and crosscase)?

3 To what extent is there a relationship between self-efficacy and teacher efficacy among pre-service teachers in a challenged education context (within-case and cross-case)?

- Male and female pre-service teachers in the current study demonstrated high self-efficacy and teacher efficacy beliefs. The beliefs indicate that male and female final year pre-service teachers, within a challenged context, are confident to recover from setbacks and feel confident in teaching and behaviour management which support presented literature (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018; Wosnitza et al., 2018) from previous studies utilising similar scales in Europe and Australia. In the current study, female pre-service teacher self-efficacy and teacher efficacy were significantly higher than male pre-service teacher self-efficacy and teacher efficacy for Case 3. The difference confirms findings from Özdemir (2008), Üstüner (2017) and Yokus (2015) for pre-service teachers in a challenged context. However, no gender differences were noted in other cases for the current study, which confers with research in developed countries (Beltman et al., 2018; Fischer et al., 2018; Klassen & Durksen, 2014; Murshidi et al., 2006; O'Neill & Stephenson, 2012b; Pfitzner-Eden, 2016a; Tschannen-Moran & Woolfolk Hoy, 2007) and in a challenged context (Çelik et al., 2018; Ngidi & Ngidi, 2019) for gender in pre-service teachers.
- Enrolled pre-service teaching programmes for *primary and secondary phases denoted high self-efficacy and teacher efficacy for pre-service teachers* in the current study, substantiating previous global North and European research conclusions (Beltman et al., 2018; Brown et al., 2015; Moulding et al., 2014; O'Neill & Stephenson, 2012b; Peixoto et al., 2020). The *primary phase pre-service teachers in the current study demonstrated significantly higher self-efficacy and teacher efficacy* than the secondary phase pre-service teachers. Significant differences indicating higher resilience traits for primary pre-service teachers in comparison to secondary pre-service teachers are evident in literature (Bouillet et al., 2014; Kavita & Hassan, 2018; Perlman & Pearson, 2012; Woodcock, 2011) also in primary school teachers in a challenged setting (Molina et al., 2017; Zuma et al., 2016).
- Self-efficacy and teacher efficacy items on the FIRE Teacher Resilience Measure), demonstrated a significant positive correlation in a challenged context for pre-service teachers. This result indicated an interrelated relationship between final year pre-service teachers' ability to recover from setbacks (self-efficacy), confidence in teaching and behaviour management in a challenged context (teacher efficacy) supported by previous studies (Morgan, 2011; Peixoto et al., 2018) which highlight the association between the internal protective factors for pre-service teacher resilience (Beltman et al., 2018; Ee & Chang, 2010; Gu & Day, 2013; Klassen et al., 2011; Morgan, 2011; Peixoto et al., 2018, 2020; Pendergast et al., 2011; Thieman et al., 2014; Wosnitza et al., 2018).



Primary research question	How can insight into the self-efficacy and teacher efficacy of pre-service teachers in a challenged education context inform knowledge on teacher resilience?	The current study <i>contributed systematic empirical evidence to the</i> <i>emergent field of teacher resilience</i> in a challenged context given the significant gap in teacher resilience knowledge in spaces of high adversity (Ebersöhn, 2014, 2016, 2017; Ebersöhn & Ferreira, 2012; Ebersöhn & Loots, 2017; Mansfield et al., 2018; Ngidi & Ngidi, 2019). The respondents in the sample were 1,193 final year pre-service teachers at the University of Pretoria enrolled in the BEd teaching education programme (2015–2017). Results presented indicate the <i>presence of</i> <i>high intrapersonal resilience-enabling pathways to teacher resilience</i> , which is a signature element of trait resources from which pre-service teachers can draw to mitigate adversity despite chronic and cumulative risks, to provide quality teaching in a challenged context (Ebersöhn, 2017). The current study quantified intrapersonal traits that act as protective resources (i.e., presence of self-efficacy and teacher efficacy beliefs) through a post-positivist research approach with a comparative case study design. The established presence of significant internal protective resources demonstrated pre-service teachers' <i>confidence in</i> <i>recover from setbacks</i> (i.e., self-efficacy) and <i>confidence in teaching and</i> <i>behaviour management</i> (i.e., teacher efficacy) which enable pre-service teachers to resile. The positive relationship between the self-efficacy and teacher efficacy beliefs of pre-service teachers signifies high incidences of self-efficacy and teacher efficacy which may contribute to teacher resilience amongst pre-service teacher signifies as significant protective internal factors for enabling teacher resilience in spaces of high social disadvantage. Furthermore, the reliability of the current study confers with previous studies (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018, 2020; Wosnitza et al., 2018) employing similar scales also in a challenged context (Bosch, 2020; Ebersöhn et al., 2020) but provides novel insight into comparable results of th
---------------------------	--	---

The findings of the current study thus advanced knowledge on teacher resilience in a challenged context. Intrapersonal resilience-enabling pathways (i.e., self-efficacy and teacher efficacy) may enable pre-service teachers to resile despite chronic and cumulative risk factors. The current study contributed quality, quantitatively derived teacher resilience findings from a Global South perspective. This study further validated the use of a globally used teacher resilience measure in South Africa. The results may be compared to that of others worldwide measured with the same instrument. The next section outlines the structure of the thesis by providing a short overview of each chapter for the current study.

## 1.10 OUTLINE OF CHAPTERS

## 1.10.1 INTRODUCTION

The current thesis consists of five chapters. In the following sections, I provide a brief outline of these chapters to provide a synopsis of the current study.



## 1.10.2 CHAPTER 1: INTRODUCTION AND OVERVIEW OF THE STUDY

Chapter 1 provided an introduction and overview of the current study by contextualising relevant background and the current study's rationale and purpose. After formulating research questions and hypotheses and clarifying central concepts, I introduced the theoretical framework that guided this study. Furthermore, I briefly stated the paradigmatic and methodological decisions, and quality assurance measures and ethical considerations. I concluded Chapter 1 by providing a summary of the research findings.

## **1.10.3 CHAPTER 2: LITERATURE REVIEW**

Chapter 2 outlines the literature review on aspects relevant to the context of the present study. The review includes resilience, teacher resilience, self-efficacy, teacher efficacy, initial teacher education and the pre-service teacher. The literature is contextualised by reviewing challenged contexts as characterised in a Global South milieu (as depicted in Chapter 1) and explains how the latter may contribute to the understanding of teacher resilience. I conclude the chapter with the conceptual framework to ensure consistency between the purpose and the relevant theory of the topic of enquiry.

### 1.10.4 CHAPTER 3: RESEARCH DESIGN, METHODOLOGY AND STRATEGIES

Chapter 3 offers a comprehensive explanation of the selected research design and subsequent methodological strategies followed to explore the research questions and test the formulated hypotheses. The selected method of data analysis and statistical procedures employed are explained and justified. To ensure that credible research is produced, I end the chapter with a discussion on the standards of rigour and ethical guidelines followed in the current study.

## 1.10.5 CHAPTER 4: RESEARCH RESULTS OF THE STUDY

Chapter 4 presents the results obtained during this study. The chapter contains the statistical within-case and cross-case analysis of the completed FIRE Teacher Resilience Measure that was purposively selected for the current study. The results are explained in terms of descriptive and inferential numerical presentations based on nonparametric statistical procedures performed on extant data. The statistical outputs are provided and graphically represented, and the results of the analyses are discussed.



## 1.10.6 CHAPTER 5: FINDINGS, CONCLUSIONS, AND RECOMMENDATIONS

Chapter 5, the final chapter of this thesis, provides a summary of the conclusions and main findings in terms of the research questions, hypotheses, and purpose (as posed in Chapter 1). I explain the interpretation of the results I obtained in terms of the existing literature and the conceptual framework outlined in Chapter 2, based on statistically significant evidence from Chapter 4. I further highlight limitations for this study and conclude the chapter with potential recommendations.

## 1.11 CONCLUSION

The purpose of Chapter 1 was to introduce the current study and provide contextual background for the rationale and purpose of the current study. I formulated the research questions that guided me in the hypotheses I set out to test. Additionally, I clarified core concepts and introduced the chosen theoretical framework. I briefly outlined the selected paradigm, research design, methodological choices, standards or rigour and ethical considerations applicable to the current study. Lastly, I also concisely provided a summary of the research results and chapters to follow.

In Chapter 2, I conduct a thorough review of existing literature on resilience, teacher resilience, self-efficacy, teacher efficacy, initial teacher education and the pre-service teacher to contextualise the current study within a challenged context. I also discuss and explain the conceptual framework that guided the current study.

---000----



# Chapter 2 Literature Review

## 2.1 INTRODUCTION

Chapter 2 presents an overview of worldwide pivotal bodies of knowledge, focusing on highly challenged spaces typical to a Global South context. The key bodies of knowledge relevant for the current study include (i) psychological resilience and teacher resilience globally (ii) self-efficacy and teacher efficacy beliefs globally (iii) the education landscape, including initial teacher education and (iv) the pre-service teacher globally, with all these bodies viewed within highly challenged spaces. Figure 2.1 illustrates how the pivotal bodies of knowledge for the current study intersect within a challenged context.

### Figure 2.1



Outline of the Literature Review Relevant to the Current Study

This chapter explains how existing knowledge in the stated domains has meaning for the current study. I conclude the chapter with the conceptual framework for this study. Figure 2.2 presents a flow chart showing the organisational overview of Chapter 2.



# Figure 2.2

Flow Chart Showing the Organisation of Chapter 2

PSYCHOLOGICAL RESILIENCE								
Defining res	ning resilience A brief over trends in		rview of the ion of research resilience		A resilience lens for Global South challenges			
			TEACHER F	ESILIEN	ICE			
Conceptualising teacher resilience	Teacher resilience thr different len	r Trends rough resilier ises resear		s in er nce rch	Chall predic ou cor resiliend South lar	lenges and ted negative utcomes nstraining ce in a Global educational ndscape	A teacher resilience lens for Global South challenges	
SELF-EF	FICACY AS A P	ROTE	CTIVE RESC	OURCE E	NABLING	TEACHER RES	SILIENCE	
Conceptualising self-efficacy Self-efficacy in an educational space enabling resilience				I space enabling				
TEACHER E	EFFICACY BELI	EFS A	S PROTECT	IVE RES	OURCE F	OR TEACHER F	RESILIENCE	
Conceptualising teacher efficacy beliefs Trends in teacher efficacy research as enabling pathways to teacher resilience								
				L				
BRIEF OVERVIE	W OF THE GLC	BAL E	DUCATION	LANDSC	APE AND	INITIAL TEACH	ER EDUCATION	
Situating the currentCharacterising globalstudy within the globalinitial teachereducation landscapeeducation		Teacher resilience in initial teacher education programmes		nce in Tea ner ammes educ	Teacher efficacy in initial teacher lucation programmes			
THE PRE-SERVICE TEACHER WITHIN THE GLOBAL EDUCATION LANDSCAPE								
Characterising the pre-service teacher Pre-service teacher resilience Pre-service teacher efficacy								
THE CONCEPTUAL FRAMEWORK								
Informing knowledge on teacher resilience in a challenged context through a social cognitive lens								
CONCLUSION								



## 2.2 PSYCHOLOGICAL RESILIENCE

### 2.2.1 INTRODUCTION

Section 2.2 locates the current study within psychological resilience research. In this section, I delineate psychological resilience given historical and current definitions. Furthermore, I provide an overview of the conceptualisation of psychological resilience through different lenses and highlight a resilience lens for Global South challenges. I also highlight resilience discourses indicating that irrespective of context, *individual and systemic protective resources mitigate against adversity* (Ebersöhn, 2017). *Pertinent to the current study, I foreground knowledge that protective resources that may support resilience include individual traits, such as personal grit and hardiness* (Duckworth et al., 2007; Perkins-Gough, 2013; Von Culin et al., 2014), *positive emotions* (Fredrickson, 2001), *sense of coherence* (Hansson & Cederblad, 2004), *positive temperament* (Seligman, 2002) *and self-efficacy* (Bandura, 1977, 1997, 2001, 2002; Gu & Day, 2007; Hewitt et al., 2017).

### 2.2.2 DEFINING RESILIENCE

To situate the current study on *teacher resilience* within the broader context of resilience research it is necessary to define and discuss resilience given historical and current literature on psychological resilience. I use the literature as a backdrop to understanding the conceptualisation of resilience through different lenses (including grit and hardiness, a sense of coherence, positive emotions, relational resilience, adaptive coping and the socio-ecological resilience discourse.

The research proposes that a prerequisite for resilience is exposure to negative life conditions, environmental risk, threats or recurring setbacks. Resilience manifests through a positive response despite significant risk (Ee & Chang, 2010; Luthar et al., 2000; Masten, 2001). The commonly used phrase to "bounce back" is derived from the definition of resilience as the ability to quickly and efficiently recover strengths, return to normal, or rebound when faced with adversity or negative life events (Day & Gu, 2010; Masten, 2001; Price et al., 2012). By enduring challenging or adverse conditions, an individual may develop and strengthen the tools, strategies and coping mechanisms required to navigate challenges (Brunetti, 2006; Masten, 2001; Willers et al., 2013). Resilience occurs when the individual demonstrates adaptation or positive development, despite having faced significant risk (Luthar et al., 2000; Masten et al., 1990; Masten & Reed, 2005; Theron, 2011).

Walsh (2002, 2007) expounded on this notion by positing resilience as a way not to "bounce back" but to recalibrate, transform, change, and move forward. In other words, resilience implies "bouncing forward" by using assets, strengths, buffers, enabling strategies and protective systemic resources (or capital) which trigger the capacity for adaptive coping



in the presence of adversity. Windle (2010) defined resilience in people (in the current study, a pre-service teacher) as being able to effectively negotiate, adapt to, or manage significant stress by utilising protective factors (in the form of individual, life, and environmental assets). Scholars (Day & Gu, 2013; Masten, 2001) further argued that extraordinary resources are not required for resilience. Rather resilience is demonstrated in daily responses to challenges. Therefore Masten (2001) denoted resilience as *ordinary magic*, while Day and Gu (2013) call it *everyday resilience*. Consequently, the inquiry of resilience is based in the domain of positive psychology. Positive psychology highlights constructive capacities and potential within a purpose-driven, future-oriented life, rather than on negative aspects in the life and context of an individual (Seligman, 2002; Seligman et al., 2009).

Resilience has been defined as an internal trait, characteristic, outcome, or process (Fletcher & Sarkar, 2013). Initially, the construct was viewed as an individual's ability (i.e., a trait, innate quality, state, attribute, personality characteristic or positive advantage) to adapt, recover or return to equilibrium after adverse events (Anthony, 1987; Block & Block, 1980; Connor & Davidson, 2003; Ee & Chang, 2010; Masten et al., 1990; Tait, 2008). However, in recent literature, resilience is defined as a complex, oscillating and fluid construct, as opposed to being viewed solely as an innate quality, personality characteristic or trait (Beltman et al., 2018; Coetzee, 2013; Day & Gu, 2010; Ebersöhn, 2012; Masten & Reed, 2005; Masten & Wright, 2010; Nolan et al., 2014; Rutter, 2006; Theron, 2011; Ungar, 2008; Ungar et al., 2013). In this light, Luthar et al. (2000, p. 554) pronounced that "resilience should always be used when referring to the process or phenomenon of competence despite adversity, with the term resiliency used only when referring to a specific personality trait". Masten (2014, p. 10) reiterated that resilience is "the capacity of a dynamic system to adapt successfully to disturbances that threaten system function, viability and development". For this reason, it seems likely that the contexts in and the interaction through which individuals face hardship throughout their lifespan all contribute to the complex, dynamic process of resilience (Beltman et al., 2011; Beltman & Mansfield, 2018; Masten & Reed, 2005). The following section explores different theoretical lenses through which psychological resilience is conceptualised in the literature relevant to the current study.

### 2.2.3 A BRIEF OVERVIEW OF THE CONCEPTUALISATION OF RESEARCH TRENDS IN RESILIENCE

Whereas a definition (see Section 2.2.2) describes the meaning of a term, a construct is an abstract view stemming from a combination of intuition and evidence (Fletcher & Sarkar, 2013). Cassidy (2015) emphasised the debate around a global resilience construct and the conceptualisation thereof. As a result, resilience can be and is conceptualised in literature from a multitude of perspectives. This section aims to understand psychological resilience as a trait, characteristic, capacity and/or outcome.



The section locates the current study within the trait discourse while at the same time acknowledging the current socio-ecological process conceptualisation of resilience in literature. To gain such understanding, I briefly view resilience through different theoretical lenses, including grit and hardiness, a sense of coherence, positive emotions, relational resilience, adaptive coping and the socio-ecological resilience standpoint.

Resilience has been identified as a vital individual trait, protective personality factor or attribute (Ee & Chang, 2010; Roth & Von Collani, 2007). Various disciplines and perspectives, including psychiatry, child developmental psychology, cognitive psychology, personality psychology, ecological sciences and biology, conceptualised "resilient" people as being stress-resistant individuals with stable personality traits, capabilities or characteristics linked to protective factors and biological influences (Anthony, 1987; Ee & Chang, 2010; Masten, 2018; Masten et al., 1990; Masten & Reed, 2005; Tait, 2008; Windle, 2010). This conceptualisation emphasises the understanding of maladaptive reactions to adversity (e.g., socioeconomic disadvantage, maltreatment, mental disorder, violence, chronic disease, natural disasters, and traumatic experiences) (Luthar et al., 2000) with individuals differing considerably in their vulnerability to distress (Matthews, 2016).

Traits such as neuroticism and dispositional anxiety seemingly increase vulnerability, whereas hardiness, grit, and emotional intelligence support resilience (Kobasa et al., 1982; Maddi et al., 2017; Matthews, 2016; Prince-Embury, 2010). Resilience, conceived as a trait, attribute, personal capacity or quality, represents characteristics enabling adaptation, despite encountered challenges, to thrive during adversity (Connor & Davidson, 2003; Gu & Day, 2007; Masten, 2018; Masten et al., 1990). Block and Block (1980) referred to "ego resilience" as a personality parameter including traits such as resourcefulness, character strength, positive emotionality and flexibility. This belief of resilience showcased the apparent invulnerability or indestructible nature of people in acutely adverse circumstances (Anthony, 1987).

A person with high levels of ego resilience was portrayed with heightened energy, zestful, a sense of optimism, enthusiasm, ingenuity, openness to new experiences and curiosity, as well as demonstrating the ability to detach themselves from and to conceptualise problems (Block & Block, 1980; Block & Kremen, 1996). Thus, resilient people are seen as active and socially responsive to their environmental context (Ee & Chang, 2010), and they display high positive emotionality (e.g., using humour, positive dispositions or optimistic thinking) (Block & Kremen, 1996). Signified as protective characteristics, Rutter (1985, p. 600) defined it as "influences that modify, ameliorate or alter a person's response to some environmental hazard that predisposes to a maladaptive outcome".

Resilience literature has acknowledged numerous individual protective factors, including, for example, self-efficacy (see Section 2.4), hardiness and grit (Bashant, 2014; Kelly et al., 2014; Kobasa et al., 1982; Maddi et al., 2017; Perkins-Gough, 2013; Prince-Embury, 2010), sense of coherence (A. Antonovsky, 1979) and positive emotions (Fredrickson, 2001; Tugade


& Frederickson, 2004; Tugade et al., 2004). The shift towards understanding resilience as a process replaced the individual capacity as the focus within resilience conceptualisation instead emphasising aspects such as relational resilience (Jordan, 2006, 2013), adaptive coping (Skinner & Zimmer-Gembeck, 2007, 2011; Zimmer-Gembeck et al., 2018) and the socio-ecological context (Ungar, 2010, 2011, 2012a, 2012b; Ungar et al., 2013). Although the extant data limits the current study's investigation to resilience as a trait discourse (as operationalised in Chapter 1), I acknowledge the paradigm shift.

Current research no longer considers resilience solely as an innate, individual and internal capacity (Anthony, 1987; Block & Kremen, 1996; Connor & Davidson, 2003; Ee & Chang, 2010; Letzring et al., 2005; Masten et al., 1990; Tait, 2008) within the micro-system. Instead, resilience is viewed as an interactive systemic process between a person's circumstances, situation and the ecology (Ebersöhn, 2012; Rutter, 2012; Theron, 2011; Ungar, 2010, 2011; Ungar et al., 2013; Van Breda, 2018; Vanderbilt-Adriance & Shaw, 2008), as shown in Figure 2.3.

#### Figure 2.3

Resilience as Process and Outcome



Adapted from Ebersöhn (2012); Ungar (2010, 2011); Ungar et al. (2013) and Van Breda (2018)



Although resilience research initially focused on providing insights into individual capacities to forge positive adaptive pathways despite adversity, there is a growing consensus that the dynamic interplay between psychological internal personality traits (an outcome process, i.e., "resilient") and external ecologies (transactional process, i.e., resilience) that nurture resilience should be investigated (Beltman et al., 2011; Rutter, 2006; Theron et al., 2011; Ungar, 2010, 2011; Ungar et al., 2013). Ungar's (2011) urged that the individual should not be at the centre in the pursuit to understand resilience given his principle of decentrality. Instead, the context should be highlighted first, followed by how the environment interacts with the individual, and finally individual characteristics.

As illustrated in Figure 2.3, a positive resilient outcome is thus merely dependent on a person experiencing adversity. Rather, the individual's capacity to cope is enhanced with a supportive environment (dynamic interrelationships within a social system) facilitating the ability to prosper collectively despite risk (Bosch, 2020; Ebersöhn, 2012; Ungar, 2010, 2011). Regardless of the setting, environments may serve as a buffer even when resources are limited, perhaps leading to non-normative or atypical nuanced pathways (i.e., unexpected positive outcomes) to resilience (Ungar, 2011). The complexity of a "composite construct" (Gu & Li, 2013, p. 291) such as resilience requires various conceptualisations to address its ambiguous nature. To this aim, Table 2.1 provides an overview of resilience through different lenses. Section 2.2.4, afterwards, focuses on a resilience lens to Global South challenges.



### Table 2.1

Overview of the Conceptualisation of Psychological Resilience Through Different Lenses

Resilience lens	Theorists	Definition	Overview on theoretical conceptualisation related to psychological resilience
Resilience conceptualised as grit and hardiness	Grit theory (Duckworth et al., 2007; Von Culin et al., 2014)	<ul> <li>Grit is the ability to persist in the face of struggle (Duckworth et al., 2007; Perkins-Gough, 2013; Von Culin et al., 2014).</li> <li>Hardiness, as a personality trait manifesting on the cognitive, emotional as well as behavioural levels and variables regarding stress resistance (Rizeanu &amp; Vasiliu, 2016)</li> </ul>	<ul> <li>Grit and hardiness are conceptualised as helpful to people to deal effectively with challenging and stressful circumstances (Kobasa et al., 1982; Maddi et al., 2017; Prince-Embury, 2010). Hardiness and grit have been found to support resilience (Matthews, 2016).</li> <li>Grit is viewed as the quality that enables individuals to work hard and persist with their long-term passions and goals (Bashant, 2014; Perkins-Gough, 2013), to achieve success (Bashant, 2014) based on effort and interest (Von Culin et al., 2014). Resilience as part of being "gritty" entails responding resiliently to situations of failure and adversity. Grit demonstrates resilience and shows deep commitment and engagement over a prolonged period (Perkins-Gough, 2013).</li> <li>Hardiness is an individual trait of stress resistance on the cognitive, emotional, and behavioural levels (Kobasa, 1979; Kobasa et al., 1982; Maddi et al., 2017; Prince-Embury, 2010; Rizeanu &amp; Vasiliu, 2016).</li> <li>Grit and hardiness are viewed as fairly stable over time and context (Kelly et al., 2014; Von Culin et al., 2014).</li> </ul>
Resilience conceptualised as sense of coherence (SOC)	Salutogenic theory (A. Antonovsky, 1979)	• SOC is "a global orientation that expresses the extent to which one has a pervasive, enduring, though dynamic, feeling of confidence that one's internal and external environment are predictable and that there is a high probability that things will work out as well as can reasonably be expected" (A. Antonovsky, 1979, p. 132).	<ul> <li>A. Antonovsky (1979) questioned why certain people thrive optimally despite adversity, hardship, or bombarding stressors, maintain subjective well-being and keep a positive perspective on life, while others do not (H. Antonovsky &amp; Sagy, 1986; Eriksson &amp; Lindström, 2006; Viviers &amp; Cilliers, 1999).</li> <li>SOC investigates an individual's global orientation to regard daily stressors or risk factors as comprehensible, manageable, and meaningful (Flannery &amp; Flannery, 1990; Hansson &amp; Cederblad, 2004; Schnyder et al., 2000; Strauser &amp; Lustig, 2003; Ventegodt et al., 2005). It employs generalised resistance resources, protective factors or assets to achieve positive adjustment or outcomes (Almedom, 2005; Masten, 2001).</li> <li>SOC seems to be a health-promoting resource that strengthens resilience (Eriksson &amp; Lindström, 2006).</li> </ul>
Resilience conceptualised as positive emotions	Broaden-and- build theory (Fredrickson, 2001)	• This theory, postulates that positive emotions extend individuals' thought- actions. These in turn build their enduring capacities (e.g., physical, intellectual, social, and psychological resources) and increase the presence	<ul> <li>The capacity for positive emotions as an asset is central to flourishing, optimal functioning, and well-being (Fredrickson, 2001).</li> <li>Psychological resilience is viewed as the capacity to bounce back from setbacks by employing positive emotions to cope (Tugade et al., 2004).</li> <li>Cultivating positive emotions predicts an increase in resilience and life satisfaction, while negative emotions do not impede the value of positive emotions (Cohn et al., 2009).</li> </ul>



Resilience lens	Theorists	Definition	Overview on theoretical conceptualisation related to psychological resilience
		of positive affect traits (Fredrickson, 2001).	<ul> <li>Resilience mediates the association between positive emotions and life satisfaction. Well-being and health outcomes can promote the presence of positive affect traits (Fredrickson, 2001).</li> <li>In this regard, Hutchinson and Pretelt (2010) emphasise the important role that positive emotions play in human well-being and the development of resilience (i.e., cultivate resilience through an "upward spiral" of positive emotions).</li> <li>Extrapolating from the broaden-and-build theory (Fredrickson, 2001), positive emotions are active ingredients within trait resilience since positive emotions buffer against adversity, aid in building lasting resources and fuel thriving (Cohn et al., 2009; Fredrickson et al., 2003).</li> </ul>
Resilience conceptualised as relational	Relational resilience (Jordan, 2006, 2013)	Jordan's (2006) relational resilience model is founded in relational- cultural theory (RCT) with a central belief that growth occurs in relationships. RCT is a theory about basic interconnectedness, the inevitability of needing one another throughout our lives. In other words, people are wired to connect (Jordan, 2006, 2013).	<ul> <li>Relational resilience highlights "growth-fostering" connections through mutual responsive relationships, empowerment and the development of courage (Jordan, 2006, 2013; Le Cornu, 2009, 2013).</li> <li>Ebersöhn's (2012) Relationship-Resourced Resilience (RRR) framework, which draws on the collectivistic philosophies associated with indigenous African culture (e.g., Ubuntu<sup>32</sup>), theorise how resilience occurs as a communal, rather than a subjective individual, process to achieve mutually positive outcomes. The framework proposes that a collective stress response (i.e., flocking as opposed to fight, flight, freeze, faint or swarm) can be effective when individuals come together through support-seeking strategies to mobilise resources in reaction to chronic adversity (Ebersöhn, 2012, 2014, 2019a, 2019b).</li> </ul>
Resilience conceptualised as adaptive coping	Adaptive coping framework (Skinner & Zimmer- Gembeck, 2007, 2011)	A multilevel framework that foregrounds the individual's subjective sense of ability to achieve control on a stressful situation by mobilising resources. It is part of an adaptive stress response that limit adverse effects to generate positive outcomes (Skinner & Zimmer-Gembeck, 2007, 2011).	<ul> <li>Adaptive coping is seen as interactional and episodic, distinct from simply coping. Adaptive coping intercedes the influence of adversity, which gradually shapes positive adaptation and enables the resilience process (Skinner &amp; Zimmer-Gembeck, 2011).</li> <li>Adaptive coping processes are the intersection between individual capacities and the optimal mobilisation of contextual assets to navigate risk (Ebersöhn, 2014). Therefore, positive adjustment despite adversity requires a dynamic process that underlines enablement in ecologies at a given time.</li> <li>Effective coping strategies limit the influence of adversity and become more eminent for well-being than barriers (Willers et al., 2013).</li> <li>Resilience is regarded as a function of an interacting transactional adaptive system within which an individual is embedded (Skinner &amp; Zimmer-Gembeck, 2011) and shaped by cultural and contextual factors (Theron, 2016; Willers et al., 2013).</li> </ul>

<sup>&</sup>lt;sup>32</sup> *Ubuntu is* the African belief system reflecting reciprocity (a belief that understands identity as it emerges through relationships (Letseka, 2012, 2013). There is little emphasis on the individual, but rather a connection with social and emotional ties through agency and interdependence, which form part of the obligations every community member has (Ebersöhn, 2019a; Ebersöhn et al., 2014).



Resilience lens	Theorists	Definition	Overview on theoretical conceptualisation related to psychological resilience
			<ul> <li>In this sense, resilience may be characterised as involving a longitudinal, dynamic process whereby a person can achieve positive development even with exposure to substantial challenges (Ebersöhn, 2007; Masten &amp; Wright, 2010).</li> <li>Ebersöhn (2014) posited that a chain of adaptive coping strategies manifests in the resilience processes.</li> </ul>
Resilience conceptualised from an ecological and systemic process view	Socio- ecological view (Ungar, 2008, 2010, 2011, 2012a, 2012b; Ungar et al., 2013)	<ul> <li>Resilience can be viewed as ecological in nature. The context in which the individual functions are fundamental in the resilience process (Ungar, 2008).</li> <li>The socio-ecological resilience view centres on cultural moderation as valuable in informing varied resilience manifestations (Ungar et al., 2013).</li> </ul>	<ul> <li>Researchers have conceptualised resilience as dependent on the interaction between the individual and their social ecologies (Theron, 2016; Theron et al., 2011; Ungar, 2011, 2012b; Ungar et al., 2013). As such, the salience of individuals' social and physical ecology is emphasised (Ungar, 2012a, 2012b). While resilience includes an agency component, the individual's context must be able to provide health-enhancing resources accessible in culturally relevant ways to the individual (Ungar, 2011, 2012a, 2012b; Ungar et al., 2013).</li> <li>Cultural moderation entails resources influenced by a cultural meaning-making process including the value people attribute to accessible resources (Ungar et al., 2013).</li> <li>Resilience is, therefore, characterised as an effective negotiation process of navigating risk management and development notwithstanding adversity, stress or trauma within the interface between personal, cultural and contextual roles (Windle, 2010). Consequently, resilience can be viewed as socio-ecological in nature, as the context in which the individual function is intertwined in the process through which adversity is overcome (Ungar, 2008). In conclusion, research (Day &amp; Gu, 2013) suggested that the contexts in which individual works (e.g., teaching landscape within a challenged context) shape resilience by constituting barriers and assets to challenge or support resilience.</li> </ul>



## 2.2.4 A RESILIENCE LENS FOR GLOBAL SOUTH CHALLENGES

This section conceptualises contextual challenges in a Global South place through a resilience lens to locate the current study in a broader context of resilience research. In the presence of monumental adversity and structural disparity, such as South Africa as an archetype of a Global South place, resilience becomes evident (Ebersöhn, 2014). In Chapter 1, I denote South Africa as a high-risk, high-need challenged context with inequality and cultural or political marginalisation typified by an emerging economy due to a post-colonial history (Dados & Connell, 2012; Ebersöhn, 2014; Schwartz & Harris, 2017). Disparate societies imply fewer opportunities for equality, presented by substantial health, social and economic risks outcomes (Ebersöhn, 2016, 2017). Since South Africa is portrayed as a highly unequal country (Bhengu, 2019; Bisseker, 2019; Francis & Webster, 2019; McKeever, 2017), the chasm of inequality increasing (Magubane, 2019) the effects of post-colonialism as well as the numerous challenges the national government face is evident (Mansfield et al., 2018). The existing status will take many decades to amend, and extensive transformation is only pragmatic in the long term (Ebersöhn, 2017).

Researchers (Chisholm et al., 2005; Ebersöhn, 2014, 2016, 2017; Ferreira & Ebersöhn, 2012; Sayed & Badroodien, 2016; Shalem & Hoadley, 2009) urge for the use of contextual lenses to facilitate positive outcomes in an at-risk context aggravated by inequality. The context should be understood with the enabling and constraining processes to develop an accurate view of resilience (Day & Gu, 2013). Given the manifestation of multiple and chronic constraints faced in an emergent post-colonial democracy, *resilience* may act as such a lens to posit the notion of adaptive spaces to address socially-generated risk factors in various contexts (including education). Ebersöhn (2014, 2017) found that, in a space of extreme adversity, besides intrapersonal factors (i.e., intrapersonal traits), it is access to and mobilising of socio-ecological (external) resources that support resilience (see Figure 2.3).

In the field of resilience, a tendency has been noted that Western understandings of resilience may not be sensitive to a Global South context (Bosch, 2020; Ebersöhn, 2015, 2017; Masten & Wright, 2010; Saavedra & Peréz, 2018; Theron & Malindi, 2010). Accordingly, research recommended that an understanding of resilience should be developed at a local level (Shean, 2015) since context is foregrounded as a significant variable across studies of adversity and the education landscape (Beltman et al., 2011; Ebersöhn, 2017; Johnson & Down, 2013; Shean, 2015). This understanding would include an agreed insight into resilience terms so that conceptualisation of resilience is grounded in the local community (Shean, 2015) such as a Global South context. Ebersöhn (2017) denoted the plausibility that education as a buffer against adversity may differ from country to country, including between the Global South and Global North.



I apply what Ebersöhn (2015), Theron (2011), and Vanderbilt-Adriance and Shaw (2008) suggested, that researchers should be more specific in terms of the resilience-related domain, context, or place being studied and avoid "normative" assumptions. Place dissimilarities, with inherent power and structural inequality, illustrate resilience (i.e., capability required for adaptation) within the confines of a given context (Ebersöhn, 2014, 2015). Studies (Beltman et al., 2011; Ebersöhn, 2019a, 2019b; Masten & Wright, 2010; Theron & Malindi, 2010; Theron & Theron, 2014) also depicted resilience, although a universal experience, as a normative cultural process influenced by cultural practices and philosophies. Theron (2011, p. 6) noted that the "complexity of the process of resilience, the design of developmentally, contextually and culturally sensitive resilience-promoting prevention and intervention was hamstrung". There is a risk involved in applying Western conceptualisations of resilience directly to a Global South setting which may lack understanding of cultural factors that contextualise resilience in different populations or settings (Theron, 2011; Theron et al., 2011). Culture, context, and indigenous philosophies should be considered to develop a coherent body of resilience research and an understanding of well-being cognisant of contextual and cultural demographics (Beltman et al., 2011; Chun et al., 2006; Kuo, 2013). Cultural differences are important in a Global South ecology as it differs from the historically Eurocentric and Global North dominating cultures, programmes and policies. Applying a usually western positive outcome across all contexts or cultures seems unethical and obstructs meaning-making in different contexts or cultures (Averill & McRae, 2019; Ebersöhn, 2017, 2019a; Kuo, 2013; Saavedra & Pérez, 2018).

According to Ebersöhn (2014, p. 568), adapting to chronic adversity, such as is evident in the Global South, needs a "cable of nonstop vigilance". This proposes a lifeline chain of resilience, conjoining uninterrupted incidences of adaption (i.e., mini-processes of adaptation) consecutively rather than as a once-off incidental process portraying an adversity beginning and positive adaptation end. Given the circumstances of significant adversity and structural disparity in the Global South, resilience implies that adaptive responses, using protective resources or assets on all levels may lead to better than expected outcomes (Ebersöhn, 2017). Therefore, human capital (i.e., intrapersonal assets), social capital (i.e., social stability, flocking) and contextual assets or resources exist collectively (Ebersöhn, 2012, 2017, 2019a). *Contextual protective factors* can include positive institutions (Seligman et al., 2009) that provide necessary services such as schools, libraries, hospitals, clinics, faith-based organisations, and welfare resources. Echoing Jordan (2006, 2013) (see Table 2.1), Ebersöhn (2012, 2016, 2019a) also highlighted the importance of connections or relationships as a resource that nurtures resilience.



Through her honeycomb analogy, Ebersöhn (2014) revealed how collectivism<sup>33</sup> engages interpersonal resources to promote unexpected positive outcomes. In high-risk transforming ecologies where individuals and communities face chronic, cumulative stressors, relationships are used to access and mobilise resources to counter dis-enablement (Ebersöhn, 2012, 2016, 2017). The expanding resilience research, in numerous disciplines (see Section 2.2), has prepared an emergent conceptual base for research on teacher resilience (Gu, 2018). Informed by well-being lenses on a psychological and individual level, researchers (Beltman et al., 2018; Ebersöhn, 2017; Mansfield et al., 2012; Rutter, 2012; Ungar, 2012a, 2012b) deliberate how these positions may be integrated for education research. As a result, the next section highlights teacher resilience literature for the purpose of the current study.

### 2.3 TEACHER RESILIENCE

### 2.3.1 INTRODUCTION

In Section 2.3, I position the current study within the framework of teacher resilience enabling pathways. I provide an overview of the conceptualisation of teacher resilience and theoretical approaches related to teacher resilience and investigate current trends within teacher resilience research.

Teacher resilience encapsulates those teachers that withstand the challenges of the education sector to resile (i.e., choose to stay in the profession, deliver quality education<sup>34</sup> and demonstrate job satisfaction and well-being) despite adversity and structural disparity. Teacher resilience can emerge as a combination of personality traits, learnt skills and developmental facilitated and shared socio-ecological processes between the teachers and their contexts. Teacher resilience through different lenses includes, for example, a multidimensional approach, strategic approach, active agent approach, Multi-level Teacher Resilience Model as well as a contextualised structural disparity lens contemplating resilience in spaces where poverty is high. Resilience, as grounded in positive psychology, is often applied as a lens to explore teacher stress and burnout. As highlighted in Chapter 1, teacher resilience research has escalated but remains an emergent research field with limited empirical evidence regarding the phenomenon, especially in a severely challenged context. The following section explores the conceptualisation of teacher resilience worldwide.

<sup>&</sup>lt;sup>33</sup> Values, orientations, attitudes, beliefs and behaviours that binds a central unit for a meaning-making effect on individuals in a society where the group needs are prioritised (Chun et al., 2006; Kuo, 2013). <sup>34</sup> Quality education encapsulates a complex system which include quality learners, environments, content, processes and outcomes (United Nations Children's Fund [UNICEF], 2000). Quality education is also indicated as one of the United Nations' (UN) Sustainable Developmental Goals (SDGs 4) to "ensure inclusive and equitable quality education and promote lifelong learning" (UN, 2015, p. 14). Teachers are a key component of meeting SDG 4 goals (UNESCO, 2019a, 2019b).



### 2.3.2 CONCEPTUALISING TEACHER RESILIENCE

This section attempts to conceptualise teacher resilience through relevant literature and theoretical approaches relating to teacher resilience in the next sections. The challenges teachers face daily have been well documented (e.g., Castro et al., 2010; Ebersöhn, 2014; Ebersöhn & Ferreira, 2012; Gu & Day, 2007; Kyriacou, 2001), highlighting the strains and vulnerability of one of the most demanding professions worldwide. The increasingly complex exigencies under such difficult conditions may exacerbate the negative effect on teachers' health, personal life, welfare and, consequently, on their teaching (Thieman et al., 2014; Wosnitza et al., 2014). Since many teachers are exposed to unpredictability and a transitional education sector, cultivating teacher resilience has become necessary (Ebersöhn, 2014; Gu & Day, 2013). Recent years have seen a particular shift towards focusing on teacher resilience as a proactive perspective as opposed to studies highlighting a reactive perspective on stress, barriers, burnout, attrition and ineffective teaching (Coetzee, 2013; Hong, 2012; Peixoto et al., 2018; Wosnitza et al., 2014, 2018). Researchers (Fleming et al., 2013; Peixoto et al., 2018; Yonezawa et al., 2011) explore the adaptive processes and educational planning efforts that facilitate the development of teacher resilience for effective coping within the profession. A resilience focus may provide a potential alternative perspective to understand how certain teachers manage, provide quality teaching and remain motivated over time (Gu, 2014, 2018; Gu & Day, 2013; Mansfield et al., 2012).

According to Gu and Day (2013, p. 22), teachers perceived resilience as being "allied to their everyday capacity to sustain their educational purposes and successfully manage the unavoidable uncertainties which are inherent in the practice of being a teacher. Their capacity to be resilient fluctuated because of the influences of the personal, relational and organisational settings in which they worked", thus leading to "everyday resilience". The statement, echoed Masten's (2001) description of resilience as "ordinary magic" as described in Section 2.2.2. Teacher resilience thus denotes teachers who maintain motivation in their daily working lives which sustain their educational purpose to culminate in resilience-enabling outcomes such as agency, commitment, quality teaching and teacher well-being (Gu & Day, 2013; Mansfield et al., 2016). Day and Gu (2013) claimed that teacher resilience has evident characteristics highlighting the characteristics as context-specific, role-specific and a conceptualisation of more than mere bouncing back from adversity.

Firstly, teacher resilience is characterised as context-specific by Day and Gu (2013). Contextual aspects such as school leadership support, trust in the management structures and constructive feedback from management, parents as well as learners may serve as positive influences on teachers' resilience (Day & Gu, 2010; Day et al., 2006). Day and Gu (2010) referred to organisational resilience to highlight the importance of leadership in harnessing resilience in teachers. Furthermore, Day and Gu (2013) outlined resilience in teachers as role-



specific. Teacher resilience is related to the strength, purpose, and conviction of teachers' vocation to serve (Day & Gu, 2010; Sammons et al., 2007). The role enables teachers to manage challenges during uncertain and inevitable recurring setbacks (Brunetti, 2006; Gu, 2018; Gu & Day, 2013). Lastly, teacher resilience, as trait, quality or personality characteristic (i.e., outcome, see Figure 2.3) seems to be the capacity of a teacher to "bounce back" or recover as well as to develop, learn and strengthen effective coping mechanisms or strategies over time, while gaining insight, maintaining equilibrium as well as commitment and agency to teaching despite the challenges experienced in the teaching world (Brunetti, 2006; Day, 2012; Day & Gu, 2010, 2013; Gu & Day, 2013; Masten, 2001; Wosnitza et al., 2018).

Day and Gu (2013) argued that the capacity to be resilient is important for resilience in teachers, with Sammons et al. (2007) postulated that teacher qualities are relevant to account for variances in learner progress. Teacher resilience plays a vital part in teachers' experiences and, thus, by implications the educational outcomes of their learners (Day, 2012; Ebersöhn, 2014; Thieman et al., 2012). Resilient teachers tend to *respond positively in challenging settings* (Gu & Day, 2007; Stanford, 2001; Tait, 2008); *demonstrate applicable strategies and perseverance for working with demanding or struggling learners* (Gibson & Dembo, 1984; Gu & Day, 2007; Stanford, 2001; Tait, 2008); and *deliver quality teaching and learning in varied contexts* (Gu & Day, 2007; Stanford, 2001; Tait, 2008), which may result in higher *learner achievement* and *greater learner resilience* (Woolfolk Hoy & Spero, 2005).

Resilient teachers are furthermore enthusiastic about new ideas and approaches to meet their learners' needs (Guskey, 1988; Ross et al., 1996); exhibit high levels of planning and organisation (Hewitt et al., 2017); experience satisfaction in their work (Gu & Day, 2007; Stanford, 2001; Tait, 2008); have high self-efficacy, positive dispositions and are persistent despite setbacks and adversity (Gibson & Dembo, 1984) and are able to rebound after a challenging event, create meaning from it and set futuristic goals (Tait, 2008). However, teacher resilience is conceptualised as more than the ability to "bounce back" and recovering quickly, resourcefully, and efficiently from difficulties in current literature (Day & Gu, 2007, 2013; Gu & Li, 2013). Established evidence (Ainsworth & Oldfield, 2019; Beltman & Mansfield, 2018; Coetzee, 2013; Day & Gu, 2007, 2013, Ebersöhn, 2014; Gu, 2018; Johnson & Down, 2013; Price et al., 2012; Theron & Theron, 2010; Wosnitza et al., 2014) promulgate the influence of situated dimensions and contextual elements of teachers' lives, which involve more complex multifaceted components than merely internal traits, assets or characteristics determined by nature alone. Gu and Li (2013) claimed that teacher resilience is shaped by intrapersonal qualities in collaboration with dynamic conceptual embedded elements of teachers' work and lives (i.e., the interaction of personal and contextual resources). In this regard, Johnson and Down (2013, p. 703) cautioned against "the tendency towards hyperindividualisation" when conceptualising teacher resilience in isolation (i.e., reductionism) and the consequent shift of primary well-being responsibility onto teachers.



This warning resonates with Ungar's (2011) principle of decentrality in resilience research. While Ainsworth and Oldfield (2019), expounding from Price et al. (2012), found that contextual effects on teachers' well-being within the profession are crucial for adaptation and does not rest on teachers alone. Teachers who resile use adaptive coping strategies (Skinner & Zimmer-Gembeck, 2011; Willers et al., 2013) to manage (portray through a sense of coherence, [A. Antonovsky, 1979]) everyday challenges and maintain commitment (as delineated by grit, Duckworth et al., 2007) to their vocation despite adversity in the teaching context (Brunetti, 2006; Gu & Day, 2013; Wosnitza et al., 2014). Hence, teachers demonstrate positive adaptation when they can draw upon individual (trait, capacity or characteristics) and environmental protective resources and employ coping strategies to manage daily strains or risk factors (Brunetti, 2006; Coetzee et al., 2017; Gu & Day, 2013; Jackson & Rothmann, 2005; Masten, 2001; Ungar, 2010, 2011, 2012a, 2012b; Willers et al., 2013). In this sense, teacher resilience takes account of the socio-cultural setting within the teachers' landscape viewing resilience as a dynamic multidimensional process of adaptation despite severe challenges (Mansfield et al., 2016; Peixoto et al., 2018). In addition, responding to challenges may also provide opportunities for professional growth and developmental paths, enabling teachers to thrive (Beltman et al., 2011; Mansfield et al., 2014; Peixoto et al., 2018; Wosnitza et al., 2014).

Addressing challenges by using intellectual, social and organisational environmental resources (Day & Gu, 2013) that facilitate resilience can enable teachers to verve beyond merely 'bouncing back' (Gu, 2014; Gu & Day, 2013; Wosnitza et al., 2018). Teacher resilience may therefore involve the *capacity* of a teacher to harness internal and external resources, as well as the dynamic *process* whereby teacher capacities and their professional and situated dimensions interact over time, to facilitate well-being *outcomes* in their everyday world of work (Gu, 2014; Gu & Day, 2007, 2013; Mansfield et al., 2016). The next sections discuss teacher resilience through different lenses.

### 2.3.3 TEACHER RESILIENCE THROUGH DIFFERENT LENSES

In this section, I outline various perspectives relating to teacher resilience to establish an evidence-based framework for teacher resilience. Researchers have conceptualised teacher resilience through different lenses, for example, a multidimensional approach, strategic approach, active agent approach, the Multi-level Teacher Resilience Model as well as a contextualised structural disparity lens. The following section thus explores different perspectives on teacher resilience to gain an enhanced understanding of teacher resilience within an international and Global South landscape.



## 2.3.3.1 Multidimensional, strategic, and active agent approach

Gu and Day (2007) conceptualised teacher resilience as a *multidimensional approach* in which internal and external resources interact to constitute resilience in teachers. A *strategic approach was* portrayed by Patterson et al. (2004), where teacher resilience is depicted as an adaptation process where numerous strategies are utilised. Castro et al. (2010) assumed a position from both these approaches portraying teachers as "*active agents, adopting various strategies to find balance and achievement in the face of adversity, often caused by minimal resources and challenging working conditions*" (Castro et al., 2010, p. 623). Given the extant data generated through the FIRE Teacher Resilience Measure for the current study, the Multi-level Teacher Resilience Model<sup>35</sup> (Wosnitza et al., 2014) is discussed in more detail next.

# 2.3.3.2 Multi-level Teacher Resilience Model

Wosnitza et al. (2014) postulated that teacher resilience is embedded in the notion that individual behaviour is the product of multifaceted systemic interaction. Furthermore, it is assumed that resilience can be promoted through support systems and environmental factors to enhance individual resilient responses over time (Mansfield et al., 2012; Wosnitza et al., 2014).

Resilience enabling pathways require identifying available, present and unique resources such as mobilising unused resources, community building and developing partnerships (Mansfield et al., 2018). Wosnitza et al. (2014, p. 2) thus argued that "individuals, drawing on personal, professional and social resources, not only 'bounce back' but also are able to thrive professionally and personally, experience job satisfaction, positive self-beliefs, personal well-being and an ongoing commitment to the profession". As demonstrated in Figure 2.4, the dynamic, heuristic *Multi-level Teacher Resilience Model* highlight resilience resources within the micro-, meso- and macrosystem (Bronfenbrenner, 1979; Wosnitza et al., 2014).

<sup>&</sup>lt;sup>35</sup> The Multi-level Teacher Resilience Model originated from the ENTREE project (Mansfield & Wosnitza, 2015; Peixoto et al., 2018). See Appendix A for an overview of the ENTREE project.



### Figure 2.4

Multi-level Teacher Resilience Model



Adapted from Wosnitza et al. (2014)

Within the microsystem, capacities are depicted as *social* (e.g., support, relationships, social competence, interpersonal and communication skills and problem-solving), *emotional* (e.g., humour, emotional management, emotional competence, coping strategies and enjoyment), *professional* (e.g. teaching competence, organisation, reflection, flexibility, planning and time management), and *motivational* (e.g. optimism, enthusiasm, positivity, perseverance confidence, self-efficacy, intrinsic motivation and goal setting) as well as behaviour strategies and beliefs of teacher resilience (Mansfield et al., 2012, 2018; Peixoto et al., 2018; Wosnitza et al., 2014). Mansfield et al. (2018) postulated that certain teacher personal resources (e.g., optimism, perseverance, coping, problem-solving strategies) in the microsystem remain uniform regardless of context. School and personal context support are portrayed in the mesosystem, while elements such as culture and policy are relevant within the macro context (Mansfield et al., 2012; Peixoto et al., 2014). Differences at the macrosystem level may have greater variation in welfare, culture and education policies (Day & Gu, 2013; Gu & Li, 2013; Mansfield et al., 2012, 2018; Peixoto et al., 2018; Price et al., 2012; Wosnitza et al., 2014).

### 2.3.3.3 The teacher resilience process

As demonstrated in Figure 2.5, *the teacher resilience process* (Mansfield et al., 2016) is seen as multidimensional and dynamic, where resources are promoted through various strategies enabling resilience outcomes.



### Figure 2.5

The Teacher Resilience Process



Adapted from Beltman (2020) and Mansfield et al. (2016)

As indicated in Figure 2.5, through the bidirectional arrows, resilience is shaped by complex, interrelated and dynamic contextual and personal resources (Beltman, 2020; Mansfield et al., 2016). Personal resources include capacities such as emotional competence, while contextual resources encapsulate, for example, supportive relationships and mentors.

The capacity for resilience is demonstrated by the mobilisation of personal recourse as well as contextual resources using various strategies (Beltman, 2020; Beltman et al., 2018). Strategies may comprise professional learning and effective planning. This conceptualisation draws together views of resilience centring personal capacity, environmental enabling or constraining resources, adaptation processes and teacher resilience as a necessary well-being outcome (Mansfield et al., 2016). Resilience outcomes are then seen in committed teachers experiencing job satisfaction and engagement (Beltman et al., 2018).

### 2.3.3.4 Contextualised structural disparity lens

Given the Global South context of the current study, as highlighted in Chapter 1, an overview of Ebersöhn's (2014) theoretical lens for teacher resilience in a challenged context is provided and employed within the conceptual framework (see Section 2.8). Ebersöhn (2014) argued that resilience as an outcome (i.e., static resiliency or trait) and process (i.e., the dynamic and interactive interface between the individual and ecological system risk and protective factors) coexist in the South African teacher profile as represented in Figure 2.6.



#### Figure 2.6

Contextualised Structural Disparity Lens



Adapted from Ebersöhn (2014)

Ebersöhn (2014) delineated resilience in South African teachers as a transactional ecological process and an adaptive outcome or trait. She reiterated, as shown in Figure 2.6, that teacher resilience are evident in the presence of structural disparity with the teacher appraising (see Skinner & Zimmer-Gembeck, 2007, 2011 in Table 2.1) the relative significance of the challenge or risk factor based on personal perception and awareness of their context (Ebersöhn, 2014). The importance of the risk is regulated by contextual familiarity, including the physical context and psychological, social, spiritual, and cultural protective factors.

Risk, as well as protective factors, is embedded in the teacher (internal) as well as the specific context or ecology (external) (Ebersöhn, 2014; Mapfumo et al., 2012; Ungar, 2012a; Ungar et al., 2013). The teacher then evaluates available assets to adapt to the risk by engaging in specific coping behaviour (Skinner & Zimmer-Gembeck, 2007, 2011; Willers et al., 2013) to manage the stressor. A resilient teacher would be able to identify and access internal and external protective resources and employ those as mentioned earlier, in a sustained manner to mediate chronic and cumulative challenges (Ebersöhn, 2014). As a result, teachers may follow different paths to develop resilience processes to negotiate a perceived dissonance between available assets and adversity (Coetzee, 2013; Coetzee et al, 2017; Ebersöhn, 2014).

In spite of the various foci in the conceptualisation and frameworks of teacher resilience discussed, teacher resilience seems to emerge as a dynamic interaction of intrapersonal characteristics, developed skills and developmental processes (Bobek, 2002; Coetzee, 2013; Ebersöhn, 2014; Gu & Day, 2007; Johnson & Down, 2013; Knight, 2007; Mansfield et al., 2012;



Muller et al., 2014; Stanford, 2001; Yonezawa et al., 2011) delineating the interactive qualities of resilience that can buffer against adversity (Ebersöhn, 2014). The aforementioned highlights Beltman's (2020) overview of different perspectives regarding teacher resilience as *personfocused* (i.e., trait or personal capacity or agency), *process-focused* (i.e., interface between individual and contextual interaction, where teachers employ well-being strategies to resile), *context-focused* (i.e., contextual challenges and sources of support for resilience), and *system-focused* (i.e., the collective dynamic interactive nature between levels of a system).

Given the perturbing amount of teachers indicating burnout (Cefai & Cavioni, 2014; Clement, 2017; Wosnitza et al., 2014; Zuma et al., 2016), high attrition rates (Muller et al., 2014) and teachers leaving the profession, especially during the first 5 years (OECD, 2017a; Wosnitza et al., 2014), teachers need to be equipped for the demanding vocation to provide quality teaching and thrive within the teaching (Wosnitza et al., 2014). According to Wosnitza et al. (2014), a discussion on quality of teaching cannot exclude teacher resilience, with researchers (Ebersöhn, 2014; Masten, 2001; Thieman et al., 2012) echoing that one of the factors influencing learner outcomes is teacher resilience. Therefore, teacher resilience is vital, since teachers who remain and provide quality teaching may impact learners' performance (Day, 2012; Day & Gu, 2010; Thieman et al., 2012). Consequently, the following section investigates trends in teacher resilience research.

### 2.3.4 TRENDS IN TEACHER RESILIENCE RESEARCH

In recent years, research has focused on a multitude of constructs associated with or sustaining teacher resilience and factors hindering teacher resilience. Teacher resilience is a fairly novel research domain and to develop an enhanced understanding of the phenomenon, requires an inquiry into the complex relationship between the risk factors and protective resources evident in the teaching profession (Gu, 2014). Consequently, it is needed to explore and determine elements that enable teachers to choose and stay in the profession as well as the factors that disenable resilience in teachers.

The following sections explore risk factors (i.e., that constrain resilience) enabling expected or predicted negative outcomes and protective factors (i.e., facilitating resilience) enabling unexpected or unpredicted positive outcomes despite the risk given current literature trends within the teacher resilience field. I highlight risk factors constraining teacher resilience first and the enablers of resilience in subsequent sections.



### 2.3.4.1 Risk factors constraining teacher resilience

Literature identified personal and contextual risk factors which may counter the resilience in teachers, possibly leading to stress, burnout, impaired performance, absenteeism, high turnover and attrition (Cefai & Cavioni, 2014; Clement, 2017; Diale et al., 2014; Fleming et al., 2013; Gu & Li, 2013; Zuma et al., 2016).

**Personal risk factors** may typically include, for example, *lacking motivation*, *professional competence and commitment* (Fleming et al., 2013; Mansfield et al., 2016); *negative self-beliefs, confidence or efficacy* (Beltman et al., 2011); *lack of coping strategies, self-care and boundaries implementation* (Fleming et al., 2013); *ineffective classroom management beliefs* (Beltman et al., 2011; Delale-O'Connor et al., 2017; Gu & Day, 2013; Kitching et al., 2009); *difficulties in help-seeking* (Beltman et al., 2011); *limited social competence or relational resilience* (Castro et al., 2010; Coetzee et al., 2017) as well as *conflict between internal beliefs and school-prioritised expectations* (Ainsworth & Oldfield, 2019; Beltman et al., 2011). Internal barriers may affect teachers' confidence to teach and to cope with challenges (Bobek, 2002). Teacher resilience should not be simplistically embedded to enable overworked teachers to "just cope" with challenges by merely learning to "bounce back" from adversity without attempts to change the contextual discourses (Price et al., 2012). Therefore, contextual risk factors should be carefully considered.

**Contextual risk factors** include *inadequate professional preparation* (Peixoto et al., 2018); *the uncertainty of the reality of the job* (Sayed & McDonald, 2017; Thieman et al., 2014); *demoralizing political mandates* (Beltman et al., 2011), *inconsistent standards and policy-related issues* (e.g., *increased administrative and excessive policy reform*) with pressure for performance appraisal (Day & Gu, 2013; Gu, 2018; Gu & Li, 2013; Price et al., 2012); *family responsibility and relationship* (Fleming et al., 2013); *inadequate mentoring or developmental support* (Black, 2015; Hong, 2012; Le Cornu, 2009); *school reform efforts* (Fleming et al., 2013); *lack of support from school administration* (Fleming et al., 2013; Flores, 2018; Peixoto et al., 2018; Peters & Pearce, 2012; Waddell, 2007); *lack of adequate resources* (Fleming et al., 2013); *duties not directly related to teaching* (*e.g., demanding extra-mural responsibilities), role overload or expanded roles* (Castro et al., 2010); *intense and overwhelming workload* (Ainsworth & Oldfield, 2019; Castro et al., 2016); *lack of participation in school decision making* (Fleming et al., 2013); and *poor working conditions* (Ainsworth & Oldfield, 2019; Fleming et al., 2013; Price et al., 2013); *and poor working conditions* (Ainsworth & Oldfield, 2019; Fleming et al., 2013; Price et al., 2012; Zuma et al., 2012; Zuma et al., 2012; Zuma et al., 2012; Zuma et al., 2016).

Inadequate working conditions and stressors teachers may deal with include *insufficient incentives*, poor or uncompetitive salaries; work stability or education reform; burden of paperwork; overcrowded curriculum; large classes or increased class size per teacher; disruptive misbehaviour, unacceptable or challenging learner behaviour, poor learner



motivation; unmet needs that hinder teacher professional growth and development prospects; emotional demands; increased assessment, portfolios and curriculum demands; organisational challenges at schools; long hours and time constraints; with a lack of time for personal interests; and administrative responsibilities and record-keeping (Beltman et al., 2011; Castro et al., 2010; Chisholm et al., 2005; Fleming et al., 2013; Gu, 2018; Johnson & Down, 2013; Peixoto et al., 2018; Sayed & McDonald, 2017; Van Staden & Zimmerman, 2017; Waddell, 2007; Zuma et al., 2016). The presence of risk factors (or absence of protective factors) may give rise to expected or predicted negative outcomes.

# 2.3.4.2 Expected or predicted negative outcomes: Teacher stress, burnout and attrition

The teaching profession has become more demanding (Peixoto et al., 2018). The stress of the job may adversely affect teachers' well-being and preclude a suitable work-life balance (Day & Gu, 2007, 2010, 2013). Stress can be recognised through mental or psychological indicators (e.g., anxiety, negative emotions, lower concentration, and depression) as well as somatic symptoms (e.g., stomach aches, insomnia, fatigue, headaches, appetite or weight issues, and high blood pressure) (Fleming et al., 2013; Jackson & Rothmann, 2005; Selye, 1976). The negative end of the stress continuum interrelates with the notion of teacher burnout (i.e., physical, emotional, and attitudinal exhaustion resulting from an inability to cope adequately with job demands and the level of distress generated by the profession) (Cefai & Cavioni, 2014; Clement, 2017; Wosnitza et al., 2014). A (dis)stressed teacher may, therefore, be irritated, frustrated, forgetful, lack emotional availability and withdraw from relationships with learners and colleagues (Clement, 2017; Fleming et al., 2013). A lack of balance (e.g., limited time for personal, social and family life) may result in a strong work engagement and teaching responsibility but the consequences are possible stress and burnout, lower-quality interactions with learners, lowering of teacher morale and well-being leading to a likelihood of teacher attrition (Fleming et al., 2013; Jackson & Rothmann, 2005; Thieman et al., 2012).

Attention has been paid to the high teacher attrition ratio (Arends, 2011; Mansfield et al., 2018; Zuma et al., 2016), especially early career teachers within the first 5 years of teaching (Hong, 2012; OECD, 2017a; Peters & Pearce, 2012; Wosnitza et al., 2014). Debate, both nationally (Diale et al., 2014; Diko & Letseka, 2009; Ebersöhn, 2014; Mlachila & Moeletsi, 2019; Spaull, 2015) and internationally (Ee & Chang, 2010; Gu & Day, 2007; Hong, 2012; Lowe & Prout, 2019; OECD, 2019c; Peters & Pearce, 2012), in relation to quality teacher recruitment and retention remains. Frequently cited reasons why teachers consider leaving the profession included disruptive learner behaviour; lack of appropriate support adding to a sense of incompetence and creating feelings of isolation and alienation; unmanageable workloads; demanding targets; and excessive administration (Mansfield et al., 2016; Peters & Pearce, 2012).



Teacher resilience is an alternative lens to explore retention, despite challenges, in the education sector (Hong, 2012; Peixoto et al., 2018). Hong (2012) examined the variances between leaver and stayer teachers based on their resilience response processes. Indicators like values, beliefs and emotions were highlighted to appraise how leavers and stayers compared in negotiating and interpreting external contexts. The findings indicated that both groups acknowledged similar difficulties working as a teacher (e.g., curriculum delivery and classroom management) and demonstrated an intrinsic interest in working as a teacher. However, stayers displayed stronger self-efficacy beliefs and boundaries with strategies to prevent burnout (Hong, 2012). Consequently, attrition may be less likely for teachers with strong resilience (Ee & Chang, 2010; Hong, 2012). Resilience may provide teachers with the capital to cope with challenges effectively to prevent burnout and attrition (Ainsworth & Oldfield, 2019; Thieman et al., 2014).

### 2.3.4.3 Teacher resilience enablers

In contrast with risk factors, resilience studies have demonstrated the association between teacher resilience and many related constructs. Prior research has proposed several individual (i.e., person-focused, intrapersonal, traits or capacities) and contextual (i.e., context-focused) resources that may enable resilience in teachers.

Individual resilience-facilitating factors may include self-efficacy (Beltman et al., 2011, 2018; Bobek, 2002; Bosch, 2020; Day & Gu, 2013; Ee & Chang, 2010; Gu & Day, 2007; Morgan, 2011; Price et al., 2012; Sammons et al., 2007); positive self-belief (Le Cornu, 2009; Wosnitza et al., 2014); values and morals (Flores, 2018; Gu, 2018; Hong, 2012); coping skills (Morgan, 2011) or strategies (Castro et al., 2010; Connor & Davidson 2003; Tait, 2008; Willers et al., 2013); social competence (Bosch, 2020; Mansfield et al., 2012, 2018; Peixoto et al., 2018; Wosnitza et al., 2014), interpersonal skills (Gu & Day, 2007; Tait, 2008) and communication (Tait, 2008); positive emotions (Tugade & Fredrickson, 2004), emotional competence (Ainsworth & Oldfield, 2019; Beltman et al., 2011, 2018; Bosch, 2020; Cefai & Cavioni, 2014; Day & Gu, 2013; Ee & Chang, 2010; Hong, 2012; Jennings et al., 2013; Knight, 2007; Mansfield et al., 2012, 2018; Morgan, 2011; Tugade & Fredrickson, 2004) including empathy (Jennings et al., 2013; Tait, 2008), humour (Bobek, 2002) and optimism (Day, 2012; Ee & Chang, 2010; Gu & Day, 2007; Knight, 2007; Tait, 2008), professional, goals, competence and development (Beltman et al., 2011; Bosch, 2020; Day & Gu, 2007, 2010, 2013; Flores, 2018; Gu & Day, 2007; Hong, 2012; Papatraianou & Le Cornu, 2014; Peixoto et al., 2018; Tait, 2008), teacher identity formation (Beltman et al., 2011; Day et al., 2006; Hong, 2012; Johnson & Down, 2013; Pearce & Morrison, 2011; Price et al., 2012) and systematic reflection (Le Cornu, 2009; Wosnitza et al., 2014, 2018); persistence (Gu & Li, 2013; Peixoto et al., 2018), hardiness (Matthews, 2016) and grit (Matthews, 2016; Perkins-Gough, 2013);



and *motivation* (Beltman et al., 2018; Gu & Day, 2007; Lohbeck, 2018; Mackenzie, 2012; Mansfield et al., 2012, 2018; Peixoto et al., 2018; Sammons et al., 2007).

Resilience studies have, furthermore, demonstrated the association between increased teacher resilience and a *strong sense of vocation (i.e., the call to teach or serve)* (Day, 2012; Day & Gu, 2010; Hong, 2012; Sammons et al., 2007); *effective adjustment and flexibility to accommodate constant change* (Bobek, 2002; Gu & Day, 2007; Peixoto et al., 2018; Wosnitza et al., 2014); *a sensible, transient and realistic view of negative events (i.e., not take it personally)* (Papatraianou & Le Cornu, 2014); *successful management of work-life tension* (Day & Gu, 2010); and *a positive impact on learning as well as learner progress, emotion, outcomes and achievement promotion* (Day, 2012; Day & Gu, 2010; Gu & Day, 2013; Price et al., 2012; Sammons et al., 2007, Thieman et al., 2012).

Substantial attention has also been award to **contextual factors** on teacher adaptation. Contextual influence includes factors such as school culture (Ainsworth & Oldfield, 2019; Beltman et al., 2018; Day, 2012) and conditions (Gu & Li, 2013); being part of decision-making processes (Johnson & Down, 2013); supportive mutually sustainable and trusting relationships (e.g. relational resilience) (Castro et al., 2010; Cefai & Cavioni, 2014; Day & Gu, 2010, 2013; Ebersöhn, 2012, 2014; Gu & Day, 2013; Jordan, 2006, 2013) including personal and professional relationships (Le Cornu, 2013; Papatraianou & Le Cornu, 2014; Peixoto et al., 2018) with colleagues, learners, parents, communities, management or school leaders (Ainsworth & Oldfield, 2019; Day & Gu, 2010, 2013; Gu & Day, 2013; Papatraianou & Le Cornu, 2014; Peters & Pearce, 2012); support networks (Beltman et al., 2018) including social, collegial or personal support (Brunetti, 2006; Cefai & Cavioni, 2014; Day & Gu, 2010; Knight, 2007; Mansfield et al., 2012, 2014; Morgan, 2011; O'Sullivan, 2006; Peixoto et al., 2018); leadership (Ainsworth & Oldfield, 2019; Day & Gu, 2010; Peters & Pearce, 2012), community engagement (Peixoto et al., 2018) and support for teacher professional development through collaboration, networking and tangible assistance (Day & Gu, 2013; Gu & Day, 2013; Mansfield et al., 2012; Papatraianou & Le Cornu, 2014); and the development of communities that enables reflection about teaching praxes (Day & Gu, 2013; Gu & Day, 2013; Wosnitza et al., 2014, 2018). When considering the setting in which teachers work, the presence of resilience enablers may lead to unexpected, unpredicted or better than expected positive outcomes (Ebersöhn, 2014; Mansfield et al., 2016).

# 2.3.4.4 Unexpected, unpredicted or better than expected positive outcomes despite significant risks

The personal and contextual resources interact through dynamic **socio-ecological processes** enabling possible resilience outcomes such as *high levels of well-being* (Ainsworth & Oldfield, 2019; Bobek, 2002; Brouskeli et al., 2018; Cefai & Cavioni, 2014; Day & Gu, 2010, 2013; Gu & Li, 2013; Wosnitza et al., 2014); *job or career satisfaction and fulfilment* (Ainsworth &



Oldfield, 2019; Arnup & Bowles, 2016; Beltman et al., 2018; Bobek, 2002; Gu & Day, 2007; Gu & Li, 2013; Mansfield et al., 2016; Wosnitza et al., 2014); and *ongoing commitment* (Arnup & Bowles, 2016; Beltman et al., 2018; Brunetti, 2006; Chesnut, 2017; Day & Gu, 2007, 2013; Gu, 2018; Gu & Li, 2013; Mansfield et al., 2016; Morgan, 2011; Peixoto et al., 2018; Wosnitza et al., 2014) *enthusiasm* (Mansfield et al., 2016), *active engagement* (Beltman et al., 2018; Gu & Li, 2013; Hong, 2012) and *agency* (Johnson & Down, 2013; Mansfield et al., 2016; Yost, 2016) with *effective and quality teachers* (Arnup & Bowles, 2016; Bobek, 2002; Gu & Day, 2007, 2010) *who are continuously developing in the profession* (Beltman et al., 2018).

Alternatively to teacher stress, burnout and attrition (as discussed in Section 2.2.2), teacher resilience relates positively with occupational well-being (Ainsworth & Oldfield, 2019; Bobek, 2002; Brouskeli et al., 2018; Cefai & Cavioni, 2014; Day & Gu, 2013; Ebersöhn, 2014; Gu & Li, 2013; Wosnitza et al., 2014). Well-being and job satisfaction may be indicative of teacher adaptation and is vital for teacher retention within the teaching profession (Ainsworth & Oldfield, 2019; Bobek, 2002; Day, 2012; Jackson & Rothmann, 2005; Mansfield et al., 2016). Protective mechanisms, traits, factors or strategies are reported to assist teachers in maintaining a positive work-life balance to enhance teacher well-being, coping capacity and counteract stress and burnout effectively.

These mechanisms incorporate, for example *self-care habits (e.g., diet, exercise, and sleep as well as making time for themselves, leisure activities, socialisation, and practising reflection) focusing on emotional, physical and mental wellness (Beltman et al., 2011; Cook et al., 2017); optimistic thinking and mindset, positive self-talk and a positive attitude as well as humour (Beltman et al., 2011; Tait, 2008; Yost, 2016); self-esteem (Ainsworth & Oldfield, 2019); effective time management strategies or time-efficient practices (Mansfield et al., 2016; Thieman et al., 2012; Wosnitza et al., 2014); self-regulation between work and life to increase job satisfaction (Jennings et al., 2013); setting and sustaining appropriate work boundaries (<i>i.e., establishing limits in terms of workload, role, tasks and time)* (Hong, 2012); being active agents for their well-being, while accepting the transient nature of life (Fleming et al., 2013); establishing and maintaining supportive relationships within and outside the teaching landscape (Ainsworth & Oldfield, 2019; Cefai & Cavioni, 2014; Day & Gu, 2013; Olsen, 2017).

The well-being of teachers has also been shown to affect the learners in their care since satisfied teachers are likely to create an effective learning environment with more productive classrooms (Day, 2012). The affective dimension of the classroom is related to the likelihood of school completion, attendance and achievement. Hence, well-being seems important for learners, teachers, and school communities. Nonetheless, Price et al. (2012) argued that the accountability of well-being and self-care could not be the sole responsibility of the individual teacher since dominant contextual factors influence how teachers experience their role.

Teacher resilience focuses on thriving teachers who deliver high-quality and effective teaching and learning, regardless of career stage, to all learners in varied contexts (Gu & Day,



2013; Mansfield et al., 2016; Stanford, 2001). *Quality in education* can signify the positive inputs from teachers and learners (Pareek & Rathore, 2016) and holds the opportunity to change lives, especially for the youth sector (World Bank, 2018). Teachers' positive adaptation to their professional role influences their teaching performance, desire, engagement and ability to teach, their academic rigour, and cultivate learner-teacher relationships (Ainsworth & Oldfield, 2019; Day et al., 2006; Gu & Day, 2007; Guskey, 1988; Somech & Drach-Zahavy, 2000). Teacher quality is also a critical factor in effective schools where learners are educated in a conducive environment and teachers feel supported in collegial conditions (Arnup & Bowles, 2016; Beltman et al., 2011; Cefai & Cavioni, 2014; Day & Gu, 2010). These conditions, in turn, encourage teacher commitment and retention with which may result in higher learner performance. Effective schools and teachers provide learners with successful mastery experiences to encounter achievement that nurtures motivation, resilience, self-efficacy and persistence despite challenges (Arnup & Bowles, 2016; Ee & Chang, 2010; Woolfolk Hoy & Spero, 2005).

Teacher effectiveness includes planning, organisation and preparation, classroom environments as well as instruction and teaching skills (Beltman et al., 2011; Tournaki et al., 2009; Wosnitza et al., 2014). To teach effectively, teachers must maintain resilience to cope with stress (Evans-Palmer, 2010, 2016). The fact that negative experiences (even small incidents) are inevitable in the teaching landscape makes it essential for effective teachers to resile (Meyer & Turner, 2006). Therefore, effective teaching skills (Beltman et al., 2011; Tournaki et al., 2009; Wosnitza et al., 2014); as well as organisation and preparation (Tournaki et al., 2009; Wosnitza et al., 2014) is crucial to teacher resilience since it has an immediate effect on learners and impact classroom experiences that affect learning.

Aspects such as *teacher emotion and emotional regulation* (Ee & Chang, 2010; Meyer & Turner, 2006; Tait, 2008); *professional development, teacher sense of identity* (Beltman et al., 2011; Cefai & Cavioni, 2014; Day & Gu, 2010; Day et al., 2006; Hong, 2012; Price et al., 2012); *commitment and retention* (Beltman et al., 2011;Cefai & Cavioni, 2014; Day & Gu, 2007, 2010); *motivation, goal setting and teacher efficacy* (Gibson & Dembo, 1984; Meyer & Turner, 2006; Ross et al., 1996; Tournaki et al., 2009) with *responsive and differentiated personal and professional support* as well as *relationships* (Beltman et al., 2011; Cefai & Cavioni, 2014; Day & Gu, 2007, 2010, 2013) are integral to promote teaching effectiveness. Given the resilience trends discussed, the following section deliberates on a teacher resilience lens for Global South challenges within the context of the current study.

# 2.3.5 CHALLENGES AND PREDICTED NEGATIVE OUTCOMES CONSTRAINING RESILIENCE IN A GLOBAL SOUTH EDUCATIONAL LANDSCAPE

Given the far-reaching influence of South Africa as an exemplar of a Global South context, the following section highlights the barriers and predicted or expected negative outcomes



within a challenged educational space within the South African education landscape relevant to the current study.

According to the World Development Report (World Bank, 2018), there is a worldwide learning crisis, although education is still viewed as a vital factor for transformation. Worldwide the teaching profession is labelled as an emotionally and physically challenging career (Gu & Li, 2013), with health professionals characterising teaching as a vulnerable occupation with high levels of stress and burnout. Even though there are coinciding universal educational challenges in developed (e.g., Global North and Western Europe) and developing countries, the Global South educational landscape (such as South Africa) is faced with systemic risk factors that are a result of a distinctive socio-political context and emerging economy (Ebersöhn, 2014, 2015, 2017; Zuma et al., 2016). Nevertheless, Global North views and dispositions are engrained in Global South education programmes (Saavedra & Pérez, 2018), impeding the education advancement and social ascendant of indigenous people (Ebersöhn, 2019a). The knowledge systems as well as production in and views of the Global South, therefore, may be relegated in favour of Global North and Eurocentric counterparts (Saavedra & Pérez, 2018). Furthermore, there are variations in expected achievement standards and in what is taught between developing and developed countries necessitating the development of frameworks in a challenged context (Spaull et al., 2020; Van Staden & Zimmerman, 2017).

As highlighted in Chapter 1, a Global South place is characterised by chronic and cumulative barriers, which further exacerbate the adversities inherent in the education system and teaching profession within a challenged context. Ideological, socio-economic and political shifts imply incessant change and require continuous adaptation in the system (Ebersöhn, 2015). The social, cultural and political context in which an education system is embedded and in which teachers work can thus not be ignored since it influences a country's ability to retain and develop committed teachers (Gu, 2014; Tikly, 2001; Tikly & Bond, 2013). Post-colonial societies or developing countries are typified by adverse educational spaces due to structural disparity, marginalisation and inequality, predicting high vulnerability to risks and compromised opportunities for positive developmental outcomes (Ebersöhn, 2016; Nkoana, 2017; Sayed & Badroodien, 2016) with repercussions at a policy level (Howie et al., 2012; Mansfield et al., 2018), as well as consequences for the teachers' working environment (Castro et al., 2010).

Scholars (Ebersöhn, 2017; Mlachila & Moeletsi, 2019) argued that South Africa's historically low structural economic growth with a limited initiative to generate entrepreneurial income and stasis of human capital may in part be due to a decelerated transformation in the education sector with dismal outcomes. The multi-level risk factors in post-colonial South Africa may remain fundamentally unchanged which impede quality education (Ebersöhn, 2017; Harber & Mncube, 2011). Presently, the South African ecology is thus characterised by educational instability, multi-faceted systemic barriers and inequality found in place, space



and time of a Global South exemplar (Bhengu, 2019; Bisseker, 2019; Bryan, 2005; Ebersöhn, 2015; Magubane, 2019; Sayed & Badroodien, 2016). The constraints mentioned hold true for basic and higher education systems in South Africa.

# 2.3.5.1 Higher education challenges pertaining to the education landscape

Regarding the higher education sector, tertiary education attainment in South Africa is the lowest based on the OECD comparison (OECD, 2019a). Although increasing, only 7% of the population obtains a tertiary qualification demonstrating that tertiary attainment in South Africa remains low in contrast to worldwide indicators (OECD, 2019a). The DHET (DHET, RSA, 2018b) indicated that 975,837 students were enrolled in the 26 public HEIs in 2016, with around 19,214 lecturers (instructional and permanent research staff) employed. Education enrolment for students between 23–38 years at South African universities accounted for 22.0% of enrolments, which is the second-highest enrolment rate per field of study (DHET, RSA, 2018b). Furthermore, according to the DHET (DHET, RSA, 2018b), 176,986 students were enrolled in public HEIs, with 42,107 students graduating within the education field in 2016. Furthermore, related to the global teacher shortage, South Africa does not graduate enough pre-service teachers to meet the supply and demand within the teaching landscape (Sayed & McDonald, 2017; UNESCO, 2019a, 2019b). Currently, South Africa's initial teacher institutions graduate around 15,000 novice teachers per year. This number is below the benchmark (25,000-indicator) needed to sustain an effective teacher-learner ratio (DBE, RSA, 2018b; Maphalala & Mpofu, 2019; Sayed & McDonald, 2017).

National funding for public HEI is 0.6% of gross domestic product (GDP)<sup>36</sup> compared to the average of 0.9% worldwide (OECD, 2019a). Additionally, the employment rate for 25–64-year-olds South Africans with a tertiary qualification was 85%, while South Africans without a tertiary qualification seem less likely to be employed (OECD, 2019a). Nonetheless, despite the benefits higher education enables, most young South Africans leave the education system before the age of 25-years (OECD, 2019a). Moreover, responses confronting inequality structures (Ebersöhn, 2019a) give rise to movements such as *#Fees-Must-Fall*<sup>67</sup> that emerged at South African public universities from the end of 2015, encumbering HEI activities (Mavunga, 2019). However, Mlachila and Moeletsi (2019) argued that free tertiary education, given the GDP expenditure on public education discussed next, is likely to deliver disappointing results in South Africa without addressing the inadequate foundations at basic education (i.e.,

<sup>&</sup>lt;sup>36</sup> Typical government expenditure (current, capital, and transfers) on education is extracted as a percentage of GDP (OECD, 2017b, 2019c).

<sup>&</sup>lt;sup>37</sup> SA public HEI experienced student-led protest movements demanding fee-reductions, free higher education and increasing government funding, commencing in October 2015 at the University of the Witwatersrand (Wits). The aforementioned escalated to other government-funded universities including the University of Pretoria in 2016 under the #Fees-Must-Fall2016 movement which necessitated the move to online teaching and learning (Mavunga, 2019).



primary and secondary school) levels. Therefore, South African basic educational constraints, within a Global South context, are discussed on a macro-level (i.e., education expenditure, education achievement outcomes, policy and curriculum changes, language as well as access to education) and the micro-level (school, learner, and teacher) relevant to the current study.

# 2.3.5.2 Basic Education challenges pertaining to the education landscape

The average worldwide GDP expenditure for education is 5.2% (McFarland et al., 2018; OECD, 2014a). However, in 2018 the South African GDP expenditure on education was 6.16% (approximately 20% of the national budget) which is higher than the average international GDP expenditure for public education and exceeding many Sub-Saharan African (SSA) countries (OECD, 2014a; Mlachila & Moeletsi, 2019). South Africa thus spends a large part of its capital on public education funding in comparison to international standards (OECD, 2019a). Most of the funding (94%) is allocated to primary (FP/ECD, IP, SP), secondary (SP and FET) and postsecondary non-tertiary education (OECD, 2019a). However, despite considerable education and policy investments, learner outcomes (in the form of learner performance) remain inadequate (Van Staden & Zimmerman, 2017) with disappointing retention and throughput rates (Gustafsson & Nuga Deliwe, 2020; Nggakamba, 2019; Sing & Maringe, 2020; Spaull, 2015). According to the DBE, RSA (2018a), South Africa's public<sup>38</sup> school education sector had approximately 410,000 teachers working in 25,000 national schools and were responsible for educating 12.9 million learners in 2018 (DBE, RSA, 2018a; Maphalala & Mpofu, 2019). Basic education attainment in South Africa is increasing<sup>39</sup>, yet is still low compared to international indicators (OECD, 2019a, 2019c). Given the Global South backdrop, numerous attainment discrepancies and challenges are evident with limited improvement in the education structures (McKeever, 2017).

In 2018 the national pass rate for Grade 12 learners completing their South African National Senior Certificate (NSC) examinations was 78.2% with critics (Ngqakamba, 2019; Spaull, 2015) arguing that the latter is a misleading reflection of the true pass rate since almost 50% of learners commencing their schooling career in 2007 (Grade 1) did not complete Grade 12 in 2018. Although an increase was noted in the pass rate for 2019 (87.3%) and a slight decrease in 2020 (76.2%) (DBE, RSA, 2021), the drop-out rate, due to contextual factors perpetuated by a worldwide health pandemic, indicated below-expected Grade 12 outcomes (Gustafsson & Nuga Deliwe, 2020; Sing & Maringe, 2020). Similarly, the *Annual National Assessments* (ANAs)<sup>40</sup> (DBE, RSA, 2018a, 2018c) confirmed that most South African learners

<sup>&</sup>lt;sup>38</sup> Approximately 92.7% of learners attend public schools in South Africa (DBE, RSA, 2018c).

<sup>&</sup>lt;sup>39</sup> Implementation of the South African Schools Act (SASA) enforced compulsory schooling for learners aged 7 to 15 (Grade 1 – Grade 9) (RSA, 1996a).

<sup>&</sup>lt;sup>40</sup> The ANAs are standardised national assessments for languages and mathematics in the SP and IP and in literacy and numeracy for the FP (DBE, RSA, 2018c).



underperform (Van der Berg, 2015) within a national context, signifying disquieting results for education progress. Furthermore, large-scale international research studies, such as Trends in International Mathematics and Science Study (TIMSS)<sup>41</sup>, Progress in International Reading Literacy Study (PIRLS)<sup>42</sup> and PISA indicated below average results for South African learners on basic skills such as literacy, reading, science, and mathematics calculations (Van Tilburg, 2019). This conclusion especially holds true for learners attending dysfunctional (mostly nofee) schools achieving below national and international standards (Spaull, 2011). Nonetheless, given government spending, Mlachila and Moeletsi (2019) premised that insufficient funding is thus not the reason for poor-quality education, but rather how funding is applied. Ebersöhn (2017) also argued against stand-alone strategies for increased spending in education. As a result, policy and curriculum development and implementation are relevant to the discussion.

Educational policies and curriculum development have been continuously modified over the last two decades in South Africa (DBE, RSA, 2018a, 2018c; Howie et al., 2012). Therefore, on a policy and curriculum level, teachers need to keep abreast of continual adjustments to facilitate transformation (Howie et al., 2012; Plüddemann, 2015). Intrinsically, South African teachers are expected to deliver quality education against numerous policy and curricula changes, which have not always been met with keen anticipation. Educational policy and curricula reform inevitably necessitate additional time allocation to training, modification of teaching practices and further administrative tasks (Chisholm et al., 2005; Van Staden & Zimmerman, 2017). The outcomes mentioned unavoidably increase teachers' workload (Castro et al., 2010; Ebersöhn, 2014; Flores, 2018). The incessant curriculum and policy changes highlight a society in transformation and a focus on access to education (Howie et al., 2012; Plüddemann, 2015). In the South African context, another important policy factor pertains to language since language diversity affects the well-being of teachers (Olsen, 2017; Plüddemann, 2015).

Within developing countries, including Sub-Saharan Africa, many learners are not school ready when starting primary education as they do not possess the social, emotional and cognitive skills (including language) to benefit from formal instruction (Bos, 2017; Plüddemann, 2015). Education has become an influential force driving the globalisation of English and other Global North languages (Tikly, 2001; Tikly & Bond, 2013) due to the dominance of mono-lingual education orientation (Makalela, 2018; Omidire & Ayob, 2020; Plüddemann, 2015). A linguistic difference can lead to exclusion or prevent equal access (Gogolin, 2002). South Africa, as a multicultural society, views diversity (e.g., "the rainbow

<sup>&</sup>lt;sup>41</sup> Mathematics achievement indicates that numerous Grade 5 learners, in South Africa, are not able to compute basic calculations.

<sup>&</sup>lt;sup>42</sup> PIRLS, under the patronages of the International Association for the Evaluation of Educational Achievement (IEA), provides multi-level trend data and international comparisons of Grade 4 learners reading literacy (Mullis et al., 2004). Grade 4 (international benchmark) and Grade 5 (national benchmark) learners participating in PIRLS were evaluated across all official languages (Howie et al., 2012, 2016; Van Staden & Zimmerman, 2017).



nation"), including multilingualism (11-official languages with equal status) (RSA, 1996b, 2019) as an asset advocating to move away from imposed monolingual policies (Makalela, 2015; Omidire & Ayob, 2020; Plüddemann, 2015). A language comparison between the 2011 SA census (Stats SA, 2012a) and the 2016 community survey<sup>43</sup> (Stats SA, 2016) indicated that Afrikaans (-1.4%), English (-1.3%), isiNdebele (-0.5%) and Xitsonga (-0.3%) showed a decrease in home language speakers. Afrikaans, with a decrease of 1.4%, is thus the language with the largest decrease. On the other hand, isiXhosa (+1.0%), isiZulu (+1.9%), Sepedi (+0.4%), Sesotho (+0.4%), Setswana (+0.8%) and siSwati (+0.1%) demonstrated an increase in home language speakers (Stats SA, 2012a, 2016). IsiZulu (25.1%) and English (16.6%) remain the most spoken language outside the household (RSA, 2019; Stats SA, 2018). However, the value of information from large-scale international assessments (e.g., PIRLS) and national assessments (i.e., ANAs) highlighted the foundational challenges in the national education system by emphasising the difficulties of learners to master literacy and comprehension skills, regardless of the language of instruction (Van Staden & Zimmerman, 2017). Grade 4 and Grade 5 learners in South African schools demonstrated underperformance or is seen without basic reading proficiencies compared to international standards (Howie et al., 2012, 2016; Spaull et al., 2020; Van Staden & Zimmerman, 2017).

Studies (Bryan, 2005; Milner & Khoza, 2008) showed that multiple South African schools might have inadequate resources to counter challenges to learning, and teachers seem illequipped to provide quality education. Therefore, schools and teachers seem unable to address the developmental needs of learners within a challenged context which indicate the high need for improvements in developing countries (Bos, 2017; Mammadov & Çimen, 2019). As a result, there is an acute need for more effective schools with qualified and effective teachers, especially in resource-poor communities in developing countries (Bos, 2017; Jenkins, 2019). Teachers are a vital component of schools influencing learner performance (Mammadov & Çimen, 2019). In this regard, structural inequality, exacerbated by economic disparity (Ebersöhn, 2014; Ebersöhn & Loots, 2017; Schwartz & Harris, 2017), hamper the effectiveness and quality of South African schools and teachers within the basic education system. Accordingly, South Africa is struggling to provide quality education (Spaull, 2015), with approximately 62% (OECD, 2014a) of public schools in South Africa situated in resourceconstrained communities (Mansfield et al., 2018). Schools, as spaces of activism and engagement, cultural identity, connectedness, ideology, and politics, signify in many respects post-colonial societies in transition (Ebersöhn, 2015). Therefore, in the meso-system (i.e., the school context and teacher system) barriers are vast (see Table 1.1) (Ebersöhn, 2014, 2016, 2017; Ebersöhn & Loots, 2017; Milner & Khoza, 2008) which may predict negative systemic

<sup>&</sup>lt;sup>43</sup> The latest large-scale population survey conducted in South Africa to date.



educational outcomes for learning and well-being, with the government having limited success in addressing challenges in education (Mlachila & Moeletsi, 2019).

An unsurprising compounding factor is a discrepancy between teacher demand and teacher supply. An additional 6.3 million teachers are required in Sub-Saharan Africa to ensure universal primary education by 2030 (Maphalala & Mpofu, 2019; UNESCO, 2019a, 2019b). In addition, a report from the Centre for Development and Enterprise (CDE) (Simkins, 2015) specified that South Africa requires a surplus of 456,000 teachers by 2025 for quality basic education (Van Tilburg, 2019). However, between 18,000 and 22,000 teachers depart from teaching yearly in South Africa. In comparison with international statistics, these figures are higher than the number of teachers who enter the profession (Maphalala & Mpofu, 2019). As indicated in Table 2.3, the 2018 TALIS report (OECD, 2019c) indicated the average age of the South African teacher as 43-years with 32% of teachers aged 50-years and above. According to Maphalala and Mpofu (2019), these figures signify that almost half of the current teaching workforce will have to be replaced in the next decade. However, according to the DBE, RSA (2018b) the overall attrition rate due to the resignation of teachers accounts for only 1.9% of all teachers with the supply of newly qualified teachers increasing from 8,000 to 23,800 between 2012–2016. Nevertheless, the challenging working conditions and unreasonable workload of teachers may lead to absenteeism, teacher shortage and poor teacher retention hampering the quality of education in South Africa (Sayed & McDonald, 2017; Van Tilburg, 2019). In addition, according to Macupe (2018), teachers, especially in rural areas, felt that the preparation of teachers is insufficient to deal with the main challenges in schools. Although the proportion of qualified teachers in South Africa are increasing (DBE, 2018b, 2018c), this trend does not guarantee teacher competency.

Teachers may have insufficient content knowledge and teaching skills with limited professional growth opportunities (Besharati & Tsotsotso, 2015; Gardiner, 2008; Jenkins, 2019). If South Africa is unable to recruit, train and retain enough teachers, given the prediction of the Centre for Development and Enterprise (Simkins, 2015) satisfactory education to meet social and economic needs will not be possible (Maphalala & Mpofu, 2019) since the quality of teaching plays a major role in this regard (Mansfield et al., 2018). The challenges may require teachers to adopt various roles and responsibilities (Brunetti, 2006; Castro et al., 2010; Cefai & Cavioni, 2014; Ebersöhn, 2014, 2017; Ebersöhn & Ferreira, 2012; Gu & Day, 2007, 2013; Kirk & Wall, 2010; O'Sullivan, 2006; Stanford, 2001; Thieman et al., 2012) leading to possible burnout, harmful personal effects, diminished job experiences and a negative effect on institutional health (e.g. school context) (Daniilidou et al., 2015).

The arguments underline the risk factors depicted in a Global South education landscape under pressure and highlight South Africa as one of the most inefficient countries worldwide (Mlachila & Moeletsi, 2019). According to Mlachila and Moeletsi (2019), South



Africa must address the derivation of its dysfunctional education system since the inadequate quality of education handicaps equality and hinders the country's development. The latter highlights the need for individual and collective agency, combined with education policy to support agendas for well-being (Ebersöhn, 2017; Yost, 2016). Teachers in the Global South need to resile despite adversity related to structural disparity synonymous with a post-colonial society transforming towards the conditions of democracy (Bennell et al., 2002; Bhana et al., 2006). For this reason, Ebersöhn (2016, p. 1) argued that in South Africa emphasis should be placed on "enabling spaces for accessible and adaptable education practices that are fitting, with a transforming, post-colonial Global South space with teachers recognised as knowledgeable from a pluriversal stance". The next section employs a teacher resilience lens for Global South educational challenges.

### 2.3.6 A TEACHER RESILIENCE LENS FOR GLOBAL SOUTH EDUCATIONAL CHALLENGES

It is important to explore the factors contributing to and affecting teachers' ability to resile despite chronic and cumulative risk to understand the necessity of teacher resilience in challenged contexts. This study intended to contribute to teacher resilience research by comparing pre-service teachers' self-efficacy and teacher efficacy beliefs as protective factors enabling intrapersonal resilience-enabling pathways to teacher resilience within a challenged context. I, therefore, pose a teacher resilience lens for Global South educational challenges.

Since teaching is embedded in a context with inherent uncertainties, challenges and stress that influence teachers' ability to "teach and teach well over time" (Gu, 2018 p. 24) teacher resilience (Beltman, 2020; Day, 2012; Ebersöhn, 2014; Ebersöhn et al., 2020; Gu, 2014; Hong, 2012; Mansfield et al., 2012; Peixoto et al., 2018; Wosnitza et al., 2014) and teacher efficacy beliefs (Duffin et al., 2012; Klassen et al., 2011; Kleinsasser, 2014; Woodcock, 2011) are a worldwide concern. The complexity and demand of the teaching profession, as well as the consequences thereof (e.g., attrition, stress levels, burnout, supply-and-demand statistics), are well documented (Castro et al., 2010; Stanford, 2001). The global environment necessitates research on the phenomenon of teacher resilience, especially in early career teachers (Johnson & Down, 2013; Le Cornu, 2013).

Challenged systems require adaptive teacher responses to remain committed to the profession, provide quality and effective instruction, and experience job satisfaction and wellbeing. When generating teacher resilience knowledge, it is vital to include the discourses on protective resources (such as self-efficacy and teacher efficacy beliefs) involved in processes relating to resilience while acknowledging the risk factors (Ebersöhn, 2016). A teacher's ecology provides resources to navigate adaptive responses to risks (Bosch, 2020; Ebersöhn, 2019a; Ebersöhn et al., 2020). For that reason, it is essential to recognise the lifeworld of teachers as resilience is influenced through the interaction of the teachers with their ecology (Gu & Day, 2007).



In a severely challenged context (i.e., resource-constrained environment) it has become more pertinent to explore how teachers can resile despite paramount difficulties. As a result, research that focuses on teacher resilience has also escalated (see for example Bosch, 2020; Coetzee, 2013; Coetzee et al., 2017; Ebersöhn, 2012, 2014, 2015, 2016, 2017; Ebersöhn & Ferreira, 2012; Ebersöhn & Loots, 2017). As illuminated in Table 1.1, teachers in a challenged context face numerous chronic and cumulative barriers. In the national context, situated in the Global South, with its emerging economy and enduring socio-political transformation, teachers continuously face difficult working conditions which may result in an apathetic acceptance due to a lack of alternate options or in teacher resilience (Ebersöhn, 2014). By integrating a teacher resilience lens, education research may generate knowledge on how to mitigate against adversity and enhance functionality (Ebersöhn, 2017). Resilience, as defined and conceptualised in Section 2.2, indicated the continuous debate regarding the theorising of resilience as a trait, outcome, or process. Ebersöhn (2014), however, argued for the coexistence of both outcome (i.e., trait) and process (i.e., dynamic, interactive, and transactional-ecological process) indicators in the teacher resilience profile of teachers in a challenged context such as South Africa (see Figure 2.6). Vieluf et al. (2013) supported the meaningfulness of examining constructs or traits such as teacher efficacy beliefs within diverse contexts. Ebersöhn (2014, p. 569) poised teacher resilience in challenged contexts as "teachers ceaselessly adapting in a sequence of linked incidents to respond to a procession of risks". In the life chain of resilience, as postulated by Ebersöhn (2014), teachers demonstrate instances of maladaptation as well as positive outcomes and thriving. In this regard, teacher resilience is process orientated and explores how adversity and risk in one system co-exist and mobilise protective resources in aligned ecologies (Coetzee et al., 2017; Ebersöhn, 2012). In a Global South context, teacher resilience, therefore, signifies "teachers who withstand the ebbs and flows of the educational sector" and keep on teaching notwithstanding chronic and cumulative risk factors (Ebersöhn, 2014, p. 573).

Scholars (Ebersöhn, 2016; Stewart, 2014) have also noted the central role of schools in building resilience and the fact that resilience facilitates not only surviving but also thriving for learners, teachers, and the whole school community under adverse conditions. An enabling school would be a space where the risk factors are mediated by acknowledging, identifying and mobilising available capital to make the adversity context conducive to education (Ebersöhn, 2016). Resilience in schools involves a process where all role players, including teachers, principals, families, learners, and district officials, know and employ strategies that assist teachers in teaching and learning (Ebersöhn, 2017). Ebersöhn (2012, 2014, 2016, 2017) thus argued for an education enabling spaces that emphasise evidence on health, well-being and innovative adaptation (rather than a deficit focus) in a transforming, post-colonial Global-South space where accelerated progress towards equality is unlikely.



Teachers in low-resource areas or geographically remote locations identified that strong collegial relationships have great significance for teachers and view it as an opportunity to mobilise resources, assets and strengths collectively (Ebersöhn, 2012). Furthermore, Ebersöhn (2012) explained in her RRR theory (as defined in Section 2.2.3) how school reform is maintained through resources and relationships even with unfavourable educational settings. The RRR theory relates to actions, reactions and transactions between schools and communities facilitating positive adaptation (Ebersöhn, 2012), linking with the work of Jordan (2006, 2013) and, Skinner and Zimmer-Gembeck (2007, 2011) as highlighted in Section 2.2.3.

Since Ebersöhn (2012) denoted resilience as a collective experience, the capacity to "flock" is a powerful way for teachers to link and share resources for collective resilience. Teacher resilience may signify adaptive coping processes as teachers mediate the effect of barriers by engaging in specific positive adaptive coping behaviours (Skinner & Zimmer-Gembeck, 2007; Willers et al., 2013; Zimmer-Gembeck et al., 2018). To buffer risks in resource-constrained contexts (such as a post-colonial space) teachers may utilise personal traits (capacity for resilience) (Block & Block, 1980; Block & Kremen, 1996; Letzring et al., 2005; Morgan, 2011) such as self-efficacy (Bandura, 1997), grit (Duckworth et al., 2007; Perkins-Gough, 2013; Von Culin et al., 2014), positive emotions (Fredrickson, 2001; Tugade & Fredrickson, 2004), sense of coherence (A. Antonovsky, 1979) and systemic resources in dynamic processes to enable supportive spaces (Ebersöhn, 2014). In addition, education spaces that provide a positive environmental context may foster resilience (Ebersöhn et al., 2014; Masten, 2014; Theron & Theron, 2014).

The use of protective resources and strategies can buffer against the impact of disparity and mediate positive learning and development in a challenged context (Ebersöhn, 2014). Protective factors might attract, retain and support teachers to facilitate quality teaching (Morgan, 2011; Muller et al., 2014) so desperately needed in a Global South context. Literature (Beltman, 2020; Day & Gu, 2013; Ebersöhn, 2014, 2017; Gu & Li, 2013; Willers et al., 2013), showed that resilience in teachers is a process that requires teachers to utilise protective resources, personal resources as well as coping strategies to thrive in adverse conditions. Within a Global South context, education and schools may still be viewed as an avenue of hope (Ebersöhn, 2017) with 95% of South African teachers indicating that they joined teaching to influence the development of learners (Van Diemen, 2019). Nonetheless, to address the inequality and adversity in a challenged context such as South Africa, reform and action are necessary within the education system (Van Tilburg, 2019). In countries with concerns about risk factors (see Table 1.1) research underlined the importance of individual and contextual resources, as highlighted in Table 2.2, that teachers may draw on in responding to risk.



### Table 2.2

Protective Resources in a Global South Educational Space

Assets / strengths / protective factor or resources / capital / adaptive coping strategies / buffers / sustaining / enabling in a Global South context (i.e., enable, develop or support resilience)					
Macrosystem (Culture policy education health and welfere systems)					
Culture					
Diversity (including culture and language, i.e., rainbow nation)					
Ubuntu					
Freedom of expression					
Enabling policies					
Political democracy					
Education					
Medium of preparing learners for participation and fulfilling certain roles in society					
<ul> <li>School setting as a space where learners receive care and support with teachers as assets</li> <li>Selected basefits (o.g., pageion, bauging and medical aid)</li> </ul>					
• Selected benefits (e.g., pension, housing and medical aid) Health					
Hospitals and clinics					
Medical aid for government workers (e.g., Government Employees Medical Scheme (GEMS))					
Welfare					
Health and welfare resources (e.g., social grants for parents or caregivers)					
(Physical infrastructure. location. resources)					
Physical infrastructure available for utilisation					
Development and improvement of infrastructure					
Characteristics of location					
Resources (e.g., vegetable gardens, laboratories, libraries, learning materials)					
School stability / safe space for learners  Mesosystem					
(Relationships between learners, parents, colleagues, school and support networks)					
School context support					
Support from school administration, colleagues, learners, mentors, family and friend					
Responsive and differentiated scaffolding for professional development and personal learning,					
Positive attitudes of others     Coring professional communities that support learner and teacher well being and resilience					
Callegial relationships: Eacilitate emotional and practical support since school life, and personal life are					
so closely intertwined					
School culture					
Multi-lingual classrooms					
Personal context support (e.g., family and friends)					
Flocking (access and use scarce protective resources)					
Relationships or relational resilience					
<ul> <li>Support person</li> <li>Caring network of friends and family (caring network of friends and family can assist teachers in</li> </ul>					
managing the challenges of the profession					
Work-life balance: Supportive family members can also help teachers maintain work-life balance					
Microsystem					
(Personal attributes, traits or capacities)					
Social capacities					
Being personable					
Build support networks					
Build and maintain relationships					
Seek help					
Takes advise					
Iviaintain a work-life balance     Emotional capacities					
Cares for own well-being (self-kindness)					
Manages emotions					
Copes with stress					
Emotional awareness					

- Positive emotions
- Emotional intelligence



- Emotional competence (including, e.g., optimism, motivation, hope, courage, vitality, compassion, showing enthusiasm, having a positive attitude, patience) Professional capacities
- Solves problems and improvement focused
- Flexible and adaptable
- Reflective
- Taking initiative
- Ongoing learning (i.e., life-long learner)
- Being organised
- Knowing learners
- Teaching skills
- Professional goals, purpose and aspirations
- Creativity
- Motivational capacities
- Confident
- Intrinsic motivations and values
- Sense of moral purpose and vocation (e.g., making a difference or making an impact)
- Achievement and motivation
- Perseverance and persistence
- Personal grit and hardiness
- Beliefs
- Beliefs (e.g., self-efficacy and commitment)
- High expectancies
- Feeling confident and competent
- Behavioural dispositions
- Adaptive coping strategies and skills (e.g., relational support and problems solving)
- Self-care (e.g., exercise and healthy habits)

(Bronfenbrenner, 1979; Beltman et al., 2011; Chong et al., 2011; Duckworth et al., 2007; Ebersöhn, 2012, 2014, 2017; Ebersöhn et al., 2020; Fredrickson, 2001; Gu & Day, 2007; Jackson & Rothmann, 2005; Mansfield et al., 2012, 2018; Molina et al., 2017; Morgan et al., 2010; Stats SA, 2012b, 2020a, 2020b; Yost, 2016; Wabule, 2020; Willers et al., 2013).

By highlighting teacher resilience, the lens veers from *why teachers leave teaching or burnout* to *why teachers stay and cope*, but even more so; *why teachers stay and not only survive but thrive* (Beltman et al., 2011; Mansfield et al., 2014) in *challenging Global South circumstances* of extreme adversity (Ebersöhn, 2014; Stewart, 2014). Such understanding may be crucial for teacher resilience, teacher education and early career induction (Watt et al., 2012).

It seems apparent from existing knowledge that teachers, including pre-service teachers, especially in a Global South context such as South Africa, are inundated with challenges prevalent in the profession. However, given the resilience lens employed in the current study, evidence of protective resources (including internal traits such as self-efficacy and teacher efficacy beliefs) that can be mobilised to enable positive adaptation to risks is highlighted. The following section explores self-efficacy as a protective resource enabling teacher resilience.



# 2.4 SELF-EFFICACY AS A PROTECTIVE RESOURCE ENABLING TEACHER RESILIENCE

### 2.4.1 INTRODUCTION

This section localises self-efficacy as a protective resource within teacher resilience research. In addition to resilience as a scholarly research field, the construct of self-efficacy, encapsulated in the Social Cognitive Theory (Bandura, 1986, 1989a, 1989b) discussed in Chapter 1, guided my understanding regarding teacher resilience and teacher efficacy beliefs. The following sections elaborate on the definition of self-efficacy and the conceptualisation of the construct to explore self-efficacy as an enabler for teacher resilience if present and a constraint to teacher resilience if absent.

### 2.4.2 CONCEPTUALISING SELF-EFFICACY

Section 2.4.2 attempts to define self-efficacy given literature on the construct. This discussion aimed to establish this study in the broader context of self-efficacy research as enabling pathway to resilience. As discussed in Chapter 1, Bandura (1977, 1989a, 1989b, 1997, 2001, 2002) highlighted that self-efficacy functions as proximal determinants of human action. According to Bandura (1997, p. 161) "people avoid activities and environments they believe exceed their capabilities, but they readily undertake activities and pick social environments they judge themselves capable of handling". Alternatively to Bandura, Rotter's (1966) theory of the Locus of Control (LOC), within the framework of the Social Learning Theory (Rotter, 1954), highlighted the internal (e.g., personal efforts, actions, and decisions) and the external (e.g., chance, fate, circumstance and other individuals) locus of control as mediating factors in adaptation and regulation processes. The locus of control is outlined as an individual's beliefs about the extent of their control over things that happen to them (Rotter, 1966). The internal locus of control (as related to self-efficacy), as a buffer against negative events, is associated with a range of indices of physical and psychological well-being (Buddelmeyer & Powdthavee, 2016). The internal locus of control, as a personal attribute, has been related to resilience. However, multiple researchers (Cassidy, 2015; Chesnut, 2017; Duffin et al., 2012; Hewitt et al., 2017; Lemon & Garvis, 2016; Morgan et al., 2010; Moulding et al., 2014; O'Neill & Stephenson, 2012b; Pfitzner-Eden, 2016a, 2016b; Raath & Hay, 2016; Skaalvik & Skaalvik, 2010; Snowman & McCown, 2013) within the reviewed literature conform with the definition that self-efficacy is a self-perceived belief, a judgement of expectation of how capable or prepared people feel to organise or accomplish a particular task to manage prospective situations or obtain a valued outcome. Therefore, the current study is aligned with Bandura's theoretical perspective of self-efficacy, as highlighted in Chapter 1. The following section explores self-efficacy in an educational space as applicable to the current study.



### 2.4.3 SELF-EFFICACY IN AN EDUCATIONAL SPACE ENABLING RESILIENCE

In this section, I discuss self-efficacy in an educational space to frame the current study within relevant literature to lead the discussion into teacher efficacy beliefs. Morgan et al. (2010) posited that research provides an overview of why teaching is selected as a profession and the factors affecting teacher retention. However, the literature lacks information regarding the reasons that sustain teachers daily with Kunnari et al. (2018) echoing that limited research is available on how teachers can thrive in demanding profession. Morgan (2011, p. 94) argued that "the highly specific nature of efficacy suggests that the ability to recover from adverse events is itself a particular form of efficacy and is thus worthy of study in its own right". Selfefficacy, in the educational sector, has been associated with numerous positive outcomes for teachers and learners (Bandura, 1997; Lent & Brown, 2006; Pfitzner-Eden, 2016b). Selfefficacy is a dynamic, developmental process, and as a result, teacher efficacy interacts with and predicts the development of teacher resilience (Beltman et al., 2011, 2018; Day, 2012; Day & Gu, 2013; Ee & Chang, 2010; Morgan, 2011; Peixoto et al., 2018; Wosnitza et al., 2018). Major factors in the choice of staying, moving schools or leaving teaching are based on success experience in the classroom and supportive school structures (Johnson & Down, 2013). Self-efficacy beliefs govern how contextual enablers or constraints are perceived which may influence task selection, effort exerted or how long the individual will persevere when faced with obstacles (Bandura, 2006).

Multiple studies (see, for example, Cassidy, 2015; Duffin et al., 2012; Morgan et al., 2010; Vieluf et al., 2013) regarding self-efficacy have been conducted since the latter continues to be an extensively investigated construct in educational research, especially in the context of North America (Morgan et al., 2010; Vieluf et al., 2013). However, investigations have also been conducted in Africa, Australasia, Europe, the Middle East, Southeast Asia and South America (see Duffin et al., 2012). Therefore, self-efficacy in the education landscape has become a focus of interest globally (Chong et al., 2011; Woodcock, 2011). Given Banduras' (1997) self-efficacy definition, the construct as the personal belief that one can perform activities appropriately and effectively is a growing body of research on teachers' competencies and/or teacher characteristics (Raath & Hay, 2016). Self-efficacy, based on selfperceptions play an important role in behaviour, influence human functioning and teaching behaviour and is considered important for lifelong learning (Bandura, 1977; Heneman et al., 2006). Gu and Day (2007) explained that to rebound from setbacks, teachers need beliefs in their abilities as a teacher. The presence of self-efficacy within the microsystem of teachers can thus enable resilience while the absence of self-efficacy may constrain resilience. Hence, in the following section, literature regarding teacher efficacy beliefs is specifically highlighted to explore the construct as an intrapersonal resilience-enabling pathway to teacher resilience.



# 2.5 TEACHER EFFICACY BELIEFS AS PROTECTIVE RESOURCE FOR TEACHER RESILIENCE

### 2.5.1 INTRODUCTION

Within teacher resilience research, the presence of teacher efficacy beliefs has been identified as a protective factor. In Section 2.5, I provide an overview of the conceptualisation of teacher efficacy beliefs and current trends in teacher efficacy research to investigate teacher efficacy as an intrapersonal resilience-enabling pathway to teacher resilience. Given the conceptualisation of self-efficacy as a personal belief and appraisal of competence to manage, perform, or employ necessary actions, teacher efficacy pertains to the confidence teachers display in teaching and behaviour management. Furthermore, teacher efficacy is depicted as an important protective factor for teacher well-being, job satisfaction and professional practices, which can highlight unexpected positive outcomes despite diversity within teacher resilience research. Although teacher efficacy is a global research focus and an important construct for policy and practice, additional research is needed to explore teacher efficacy beliefs as an intrapersonal resilience-enabling pathway to teacher resilience in a challenged context. Therefore, the following sections provide an overview of relevant literature on teacher efficacy pertaining to intrapersonal resilience-enabling pathways.

### 2.5.2 CONCEPTUALISING TEACHER EFFICACY BELIEFS

As with teacher resilience (discussed in Section 2.3), teacher efficacy is viewed as a complex, multidimensional and multi-faceted construct (Bandura, 1997; Duffin et al., 2012; Kleinsasser, 2014; Skaalvik & Skaalvik, 2010; Woolfolk Hoy et al., 2009). Individual behaviour stems from systemic interaction between experience, capacities, knowledge, beliefs and ecology (Bandura, 1997). Hence, teachers enter the profession with particular capabilities, beliefs, proficiency, and strategies that may assist with coping with adverse events as they adapt to a new setting (Peixoto et al., 2018).

Teacher efficacy is understood as the teachers' situation-specific evaluation of how they will cope with events, given the competence they possess (Bandura, 1977, 1986, 1993, 1997). Tschannen-Moran et al. (1998, p. 207) extrapolated from Banduras' work that teachers "need efficacy for teaching" and defined teacher efficacy "as a belief in one's own capabilities to organize and execute courses of action required to accomplish a specific teaching task in a particular context successfully". Efficacy can be based on teachers' sense of effectiveness to facilitate the learning process as well as engagement and to bring about preferred learning outcomes even among challenging learners (Bandura, 1997; Moulding et al., 2014; Ross et al., 1996; Soodak & Podell, 1996). Being certain of one's ability is important to teacher agency (Vieluf et al., 2013; Yost, 2016).


The construct and measurement of teacher efficacy beliefs are grounded in the theoretical and conceptual basis of two influential psychological theories (Duffin et al., 2012; Lemon & Garvis, 2016; Moulding et al., 2014; Vieluf et al., 2013), namely Rotter's theory of the Locus of Control (see Section 2.4.2) and Bandura's Social Cognitive Theory (see Chapter 1). Rotter's (1966) theoretical perspective postulated that teacher efficacy is the competence belief a teacher demonstrates based on perceived control over the learning situation. Although the Locus of Control theory has a historical role in teacher efficacy literature, scholars (Klassen et al., 2011) have recommended a conceptualisation of teacher efficacy beliefs that align with Bandura's (1997) since self-efficacy influences human behaviour. Therefore, for the current study, given the conceptual strand of the FIRE Teacher Resilience Measure, as discussed in Chapter 1, Bandura's (1997, 2006) view on self-efficacy is adopted.

Based on Banduras' work (1977, 1997), the teachers' sense of efficacy may be created, nurtured and strengthened by four sources, including *mastery experiences* (e.g., own successful teaching; performance accomplishments, learner achievement and teaching practices); *vicarious experience* (e.g., observing behaviour of an experienced teacher; shadowing a mentor, identifying with a model, leadership observation); *verbal persuasion* (e.g., suggestions from other teachers or management) and *emotional or physiological states* (e.g., effective emotional and physical perception, interpretation and responses to challenging situations like monitoring breathing and heart rate). Furthermore, according to research (Begum et al., 2020; Heneman et al., 2006; Hewitt et al., 2017; OECD, 2019b, 2019c; Watson & Marschall, 2019), teacher efficacy comprises of different but related capacities or factors including efficacy for behaviour management, engaging learners and using different instructional strategies. Hewitt et al. (2017) argued that once teachers believe they can regulate such factors, their self-efficacy increases.

Although Bandura (1997) believed that efficacy remains stable, scholars (Ross et al., 1996; Wang et al., 2015) argued that teacher efficacy change over time given teaching experience. A non-linear relationship between work experience and teacher efficacy was found by Wolters and Daugherty (2007), who indicated that teacher efficacy increases in novice teachers until mid-career and then declines. Nonetheless, teachers may gain confidence from looking back on past successes (i.e., mastery experiences) and self-believe as teachers which necessitate investigation of factors impacting the development of teacher efficacy beliefs (Raath & Hay, 2016). The following section explores the current trends in teacher efficacy research as enabling (i.e., presence of teacher efficacy beliefs) or constraining (i.e., absence of teacher efficacy beliefs) pathways to teacher resilience.



# 2.5.3 TRENDS IN TEACHER EFFICACY RESEARCH AS ENABLING PATHWAYS TO TEACHER RESILIENCE

This section explores current literature on teacher efficacy, highlighting trends in teacher efficacy research regarding protective and risk factors associated with teacher efficacy enabling or constraining resilience.

#### 2.5.3.1 Protective factors associated with teacher efficacy enabling teacher resilience

Teacher efficacy has been a central construct in research efforts over the last few decades (e.g., Bosch, 2020; Duffin et al., 2012; Gibson & Dembo, 1984; Klassen et al., 2011; Kleinsasser, 2014; Tschannen-Moran et al., 1998; Vieluf et al., 2013; Woodcock, 2011). However, criticism (Fackler & Malmberg, 2016; Klassen et al., 2011) against current teacher efficacy research argued that most studies are single-country investigations, cross-sectional studies and did not account for classroom or school level factors with limited research in developing countries, including African countries (Bosch, 2020; Klassen et al., 2011; Moalosi, 2013; Pierre & Worrell, 2003). Nonetheless, teacher efficacy continues receiving attention globally from researchers since it is viewed as highly relevant for teaching and learning (Kleinsasser, 2014; Vieluf et al., 2013). Demonstrating the latter is the fact that teacher efficacy is included as one of the priority themes in the TALIS survey (OECD, 2019c). Therefore, teacher efficacy is a continuously growing body of knowledge about a range of educational outcomes, including teachers' competencies and behaviour.

In this regard, teacher efficacy is highlighted as an important **associated factor** for teachers' characteristics, including *resilience* (Beltman et al., 2011; Bosch, 2020; Ebersöhn et al., 2020; Pendergast et al., 2011); *well-being* (Brouwers & Tomic, 2000; Jennings et al., 2013); *job satisfaction* (Klassen et al., 2011; Skaalvik & Skaalvik, 2010; Vieluf et al., 2013); *pedagogical beliefs* (Duffin et al., 2012); *classroom management* (Woolfolk Hoy et al., 2009); *professional instructional practices* (Duffin et al., 2012; OECD, 2009; Vieluf et al., 2013) and *performance* (Begum et al., 2020); *instructional quality* (Gibson & Dembo, 1984; Moè et al., 2010; Moulding et al., 2014; Wang et al., 2015; Woolfolk Hoy et al., 2009) and *quality teaching* (Moulding et al., 2014; Raudenbush et al., 1992); *effectiveness* (Gibson & Dembo, 1984; Hewitt et al., 2017; Moè et al., 2010; Moulding et al., 2017; Moù et al., 2010; Moulding et al., 2016); ormitment (Chesnut & Burley, 2015); *retention* (Hong, 2012; O'Neill & Stephenson, 2012b); and *reduced burnout* (Brouwers & Tomic, 2000; Jackson & Rothmann, 2005; Jennings et al., 2013; Skaalvik & Skaalvik, 2010), *job tension, stress and discontent* (O'Neill & Stephenson, 2012b).

Teacher efficacy, as a result, may affect positive **learner outcomes** (Fackler & Malmberg, 2016; Woolfolk Hoy et al., 2009) such as *motivation* (Duffin et al., 2012; Fackler &



Malmberg, 2016; Gibson & Dembo, 1984; Gonzalez-DeHass & Willems, 2016; Klassen et al., 2011; Woolfolk Hoy et al., 2009); *achievement* (Duffin et al., 2012; Evans-Palmer, 2010, 2016; Fackler & Malmberg, 2016; Gavora, 2010; Gonzalez-DeHass & Willems, 2016; Klassen et al., 2011; Moulding et al., 2014; Woolfolk Hoy et al., 2009); *engagement* (Guo et al., 2011); *attitude towards school and teachers* (Gonzalez-DeHass & Willems, 2016; Woolfolk Hoy et al., 2009); and as *learners' sense of efficacy* (Gonzalez-DeHass & Willems, 2016).

Teachers with a high perception of self-efficacy appear to anticipate success in the classroom despite external conditions like a low socio-economic context, parental home or school climate (Bandura, 1997; Heneman et al., 2006; Ross et al., 1996). Teachers who possess high self-efficacy thus have the confidence to organise and manage a situation competently (Wosnitza et al., 2014). In addition, teachers with high self-efficacy beliefs may be more open to using a variety of new ideas, resources or pedagogical approaches or risk more with curriculums to support and meet the needs of their learners (Fives & Buehl, 2010; Ghaith & Yaghi, 1997; Heneman et al., 2006; Skaalvik & Skaalvik, 2010; Woolfolk Hoy et al., 2009; Woolfolk Hoy & Spero, 2005). As a result, this leads to effective teaching, autonomy-support, and a conducive classroom atmosphere (Fackler & Malmberg, 2016; Gavora, 2010; Hewitt et al., 2017; Moalosi, 2013).

A strong sense of efficacy in teachers are further likely to produce a positive impact on the behaviour, learning and achievement outcomes of learners (Duffin et al., 2012; Evans-Palmer, 2016; Fackler & Malmberg, 2016; Gavora, 2010; Gonzalez-DeHass & Willems, 2016; Klassen et al., 2011; Moulding et al., 2014; Woolfolk Hoy et al., 2009). In addition, efficacious teachers may adopt a more humanistic but strict classroom management approach by using positive and reductive strategies to achieve or maintain a desirable behaviour (Raudenbush et al., 1992; Ross et al., 1996; Woolfolk Hoy et al., 2009). Furthermore, teachers with high teacher efficacy tend to demonstrate dispositions such as *humour* (Evans-Palmer, 2010, 2016); *enthusiasm* (Hewitt et al., 2017; Woolfolk Hoy & Spero, 2005); *fairness* (Hewitt et al., 2017); *self-reflection and mindfulness* (Cook et al., 2017; Dunlosky & Rawson, 2012; Ross et al., 1996); *self-development* (Hewitt et al., 2017); *self-care* (Cook et al., 2017; Lee, 2018); and *positive connections and adaptive interaction with colleagues and learners* (Cook et al., 2017; Delale-O'Connor et al., 2017; Gibson & Dembo, 1984; Guo et al., 2011; Lemon & Garvis, 2016; Wang et al., 2015).

A positive level of self-efficacy may also enable teachers to *encourage and persist when working with struggling learners* (Gibson & Dembo, 1984; Raudenbush et al., 1992; Ross et al., 1996); *introduce peer support* (Raudenbush et al., 1992; Ross et al., 1996) *and persisting in helping all learners to reach their potential* (Pendergast et al., 2011). Lastly, teachers with high teacher efficacy seem less *critical of learners* (Tsouloupas et al., 2014) and *initiate fewer referrals to other services* (Hewitt et al., 2017). The following section explores the risk factors associated with teacher efficacy constraining teacher resilience.



#### 2.5.3.2 Risk factors associated with teacher efficacy constraining teacher resilience

Teachers with a lower sense of efficacy may be uncertain if they can cope with situations, feel overwhelmed and experience confusion, negative thinking, bodily tension and anxiety (Bandura, 1977, 1982, 1993, 1997; Klassen et al., 2011). Teachers leaving the profession showed weaker efficacy with convictions that imposed heavy burdens on themselves, possibly creating stress and emotional burnout. (Hong, 2012). Therefore, there are some areas of skills that can present challenges for even the most experienced teachers such as classroom or behaviour management.

Classroom or behaviour management is one of the key areas where all teachers benefit from ongoing skill development and preparation (Beltman et al., 2011; Brouwers & Tomic, 2000; Delale-O'Connor et al., 2017; O'Neill & Stephenson, 2012a, 2012b; Sing & Maringe, 2020). Classroom management has been associated with the need for teachers to regulate emotions and balance caring with discipline (Delale-O'Connor et al., 2017). Teacher efficacy is linked to their ideology about the teaching practice such as effective classroom management (Delale-O'Connor et al., 2017; Gibson & Dembo, 1984; Woolfolk Hoy et al., 2009). Efficacy for classroom management is thus the teacher's belief in their ability to implement the necessary actions to maintain an orderly, organised, non-distractive classroom context (Brouwers & Tomic, 2000; Delale-O'Connor et al. 2017). Delale-O'Connor et al. (2017) emphasised the need to build teachers' beliefs to inform their classroom management practices. Teachers with lower teacher efficacy also tended to support the use of verbal or physical violence. Teachers who have doubts about classroom discipline are more likely to demonstrate burnout and consider leaving the profession (Jackson & Rothmann, 2005; Tsouloupas et al., 2014). The next section situates the current study in the education landscape, focusing on the pre-service teacher given the resilience, teacher resilience, self-efficacy and teacher efficacy scholarly domains discussed above.

# 2.6 BRIEF OVERVIEW OF THE GLOBAL EDUCATION LANDSCAPE AND INITIAL TEACHER EDUCATION

#### 2.6.1 INTRODUCTION

This section locates the current study within the global education landscape. I provide an overview of the demographic profile of teachers in the global education landscape and the teaching profession pertinent to the current study. I further characterise initial teacher education and highlight intrapersonal resilience-enabling pathways to teacher resilience in teacher education programmes.



#### 2.6.2 SITUATING THE CURRENT STUDY WITHIN THE GLOBAL EDUCATION LANDSCAPE

This section includes a short description of international and national teacher demographics to contextualise the current study in the global education landscape. Within a demanding profession, given risk and protective factors, teachers worldwide must adjust to an ambiguous dynamic landscape that factors in the wider context in which countries are embedded. This argument especially holds true in a Global South milieu with chronic and cumulative risk factors. The global demographic profile of teachers (i.e., teacher age and gender profile) has remained fairly stable over time (OECD, 2009, 2019a), as displayed in Table 2.3. Table 2.3 portrays international teacher demographics illustrated by the 2018 TALIS report (OECD, 2019c) as 68% female, with 34% of teachers being 50 years and older. In South Africa, 60% of teachers were female, with 32% aged 50 years and above and with an average age of 43 years in 2018, which is comparable to the United States, Japan, and Australia (Maphalala & Mpofu, 2019; OECD, 2009, 2019a; Van Diemen, 2019).

#### Table 2.3

#### Teacher Demographic Landscape

	International average	South African average
% Teachers 50 years and younger	66%	68%
% Teachers 50 years and above	34%	32%
% Male teachers	32%	40%
% Female teachers	68%	60%
% Female primary teachers	67%	79%
% Female secondary teachers	54%	58%

(Jenkins, 2019; OECD, 2009, 2019a, 2019c; Van Diemen, 2019; World Bank, 2020b)

Although the demographic profile of teachers, as depicted above in Table 2.3, has remained relatively similar (OECD, 2009, 2019a), the world related to education in general as well as higher education, is rapidly transforming with progress in technology and information access, which influences ways of knowing and learning (Lemon & Garvis, 2016; Pareek & Rathore, 2016; Thieman et al., 2014). The education sector facilitates technological, societal, and human resource advancement (Pareek & Rathore, 2016). Therefore, teachers and teacher education must adjust to new advances in pedagogical technology, the expansion of human knowledge, and the challenge of developing a relevant and appropriate curriculum from the vast range of materials available in a twenty-first-century world of pluralistic skills, values and goals (Lemon & Garvis, 2016; W. Taylor, 2016).

Globally, teachers are required to achieve more with higher demand for learner performance while funding is decreased (Bobek, 2002; Kyriacou, 2001). All these factors may



affect how teachers are educated and trained (W. Taylor, 2016). The following section explores initial teacher education, as part of the education landscape, in more detail. I explore initial teacher education globally and intrapersonal resilience-enabling pathways to teacher resilience within initial teacher education programmes.

#### 2.6.3 CHARACTERISING GLOBAL INITIAL TEACHER EDUCATION

This section aims to characterise initial teacher education in the framework of the current study within international and national literature. I underline pathways of teacher certification and focus on teacher resilience and teacher efficacy during teacher training to highlight enablers and constraints to teacher resilience pathways. Teacher initial education, as conceptualised in Chapter 1, should equip prospective teachers to effectively execute their responsibilities (DBE, RSA, 2019; OECD, 2019b; RSA, 2000; W. Taylor, 2016). Teacher education includes formal programmes established by institutions for the preparation of future teachers at the ISCED 1 – ISCED 3 level and influences their professional development (OECD, 2009). Worldwide teacher training is tasked with novice teachers to become effective and quality teachers (Duffin et al., 2012).

Teacher education can entail pre-service and in-service training<sup>44</sup> (W. Taylor, 2016). Preservice education comprises training that precedes the novice teacher's entry to employment as a composite of pre-service education and induction (OECD, 2019b; W. Taylor, 2016). Induction, as part of in-service orientation, is presented to guide novice teachers (OECD, 2019b). Education standards depend upon the quality of its teachers, which rely on the quality of teacher education programmes preparing teachers who maintain their motivation and commitment to the teaching profession (Lowe & Prout, 2019; Mansfield et al., 2016). Therefore, pre-service teacher education should constantly be improved, reorganised or refocused to develop high calibre teachers ready for the demanding classroom (Buchanan et al., 2013; Thieman et al., 2014). However, Lowe and Prout (2019) argued that teacher education programmes are overly bureaucratic, under-resourced, and poorly delivered within the Global South landscape, leading to a decline in educational standards. Mlachila and Moeletsi (2019) concluded that improved teacher training, school management and teacher accountability may have a prominent long-term effect on educational performance.

Since the education landscape is changing at such a rapid pace, internationally the preparation of teachers can vary widely across traditional and alternative preparation programmes (Lieberman & Darling-Hammond, 2012; OECD, 2017a, 2017b; Pfitzner-Eden, 2016a; W. Taylor, 2016). In North America, and increasingly in other high-income countries,

<sup>&</sup>lt;sup>44</sup> In-service education is the training a teacher receives after commencing their teaching career (W. Taylor, 2016).



teachers are generally university graduates who commence their teaching certification after completing secondary education (W. Taylor, 2016).

In South Africa, two routes are stipulated to become a registered teacher, as required by Section 21 of the South African Council for Educators Act, No. 31 of 2000, at the South African Council for Educators (SACE), namely a Bachelor of Education degree (BEd) (4-years) or a Bachelor's degree (3 or 4-years), with a Postgraduate Certificate in Education (PGCE) (1-year) (DBE, RSA, 2019; RSA, 2000). When completed, the qualification results in classification as a profession teacher once registered at SACE (DBE, RSA, 2019). According to the TALIS report (OECD, 2019c; Van Diemen, 2019) more than half of teachers in South Africa have degrees (55.6%), with 97.7% indicating a diploma or higher as the highest level of formal education completed. About 21.5% of teachers have a national diploma, while 18.3% have an honour's degree. Only 61.7% of South African teachers felt competent to teach, for example, critical thinking skills (OECD, 2019c; Van Diemen, 2019). The statistics may emphasise Mlachila and Moeletsi's (2019) recommendations that South Africa consider implementing more intensive and contextualised teacher training to address teaching quality. Spaull (2015, p. 39) argued that teacher selection and training must be fundamentally improved to influence the education system and states that "no education system can go beyond the competencies and quality of its teachers". Ebersöhn (2014) echoed the latter by highlighting quality teachers as valuable assets to facilitate learning, especially in a highly unequal infrastructure where the education landscape is riddled with a lack of resources and poverty-saturated schools.

The quantity and quality of teachers' initial education for developed and developing countries is an important factor in shaping teachers' careers and further development (OECD, 2009; W. Taylor, 2016). Factors playing an important role in the aforementioned is, for example, the teaching practice as conceptualised in Chapter 1. Hence, worldwide training programmes are compelled to raise standards for quality teacher training and teaching practice (Duffin et al., 2012; Moulding et al., 2014). The factors are highlighted by the substantial proportion of the UNESCO budget devoted to the improvement of teacher education (W. Taylor, 2016). However, studies (Ee & Chang, 2010; Hewitt et al., 2017) have noted that preservice teachers struggle during pressured teacher training programmes due to a lack of teacher resilience and teacher efficacy. The following section explores teacher resilience in initial teacher education programmes.

#### 2.6.4 TEACHER RESILIENCE IN INITIAL TEACHER EDUCATION PROGRAMMES

The aim of this section is to situate initial teacher education within resilience research. Teacher education programmes must create awareness and prepare upcoming teachers for the expectations, likely adverse experiences and demands of their future career (Buchanan et al., 2013; Lowe & Prout, 2019; Mansfield et al., 2016; Thieman et al., 2014).



Buchanan et al. (2013, p. 115) argued "that teacher educators should be more realistic in their preparation of pre-service teachers for the rigours of teaching". The "rigours" of teaching can include working in spaces of high challenge and poverty (Ebersöhn, 2014). Consequently, there is an emphasis on the development of resilience as well as resilience-related skills during teacher training from researchers (Beltman et al., 2011, 2018; Bobek, 2002; Buchanan et al., 2013; Ebersöhn & Loots, 2017; Le Cornu, 2009; Mansfield et al., 2012).

The understanding of teacher resilience and its significance in maintaining personal capacities is vital for pre-service teacher education programmes. However, teacher educators need to engage pre-service teachers' internal (traits) and external resources to nurture teacher resilience to prepare novice teachers for teaching realities (Gu & Day, 2013; Sayed & McDonald, 2017). Therefore, the development of teacher resilience, during teacher training, should be a collective endeavour from all relevant role-players (Wosnitza et al., 2014). Preservice teacher education programmes and teacher educators have an important role in building capacity for teacher resilience (Beltman et al., 2011; Ebersöhn & Loots, 2017; Pendergast et al., 2011; Thieman et al., 2014). Ee and Chang (2010) recommended that admission processes for teacher training should identify resilient attitudes and behaviours. At the same time, Le Cornu (2009) emphasised the need for mentors in schools, training institutions and universities as well as peer support for nurturing resilience in pre-service teacher training. Ebersöhn (2014) stressed curriculum development that primes pre-service teachers for the constant adaptation required in adverse settings but equally centres the presence of assets, despite resource scarcity.

In this regard, research indicated that pre-service teacher training ought to offer systematic resilience-building activities (e.g., using scenarios, case studies, videos, action research, observations or actual classroom observations of challenging nature) to incorporate a wellness paradigm for pre-service programmes (Castro et al., 2010; Tait, 2008). The teaching of resilient strategies (Le Cornu, 2009) can assist pre-service teachers in developing coping strategies (Chong et al., 2011; Ee & Chang, 2010); emotional competence in teaching (e.g., such as self-assessment, self-regulation, emotional-regulation, motivation, empathy, and social skills) (Cefai & Cavioni, 2014; Ee & Chang, 2010; Mansfield et al., 2012); effective stress management strategies (including communication, self-help, relaxation, physical health, hobbies, time management and planning) and mindfulness training (Cefai & Cavioni, 2014; Cook et al., 2017; Jennings et al., 2013; Mansfield et al., 2016); well-being (Cefai & Cavioni, 2014; Le Cornu, 2009), self-care habits (Cook et al., 2017); reflection and reframing skills (Black, 2015; Dunlosky & Rawson, 2012; Ee & Chang, 2010; Hewitt et al., 2017; Wosnitza et al., 2018); gratitude practices (Cook et al., 2017); problem-solving skills (Day & Gu, 2013; Mansfield et al., 2016); assertiveness (Ee & Chang, 2010); motivation for teaching (Prosser, 2008); classroom management (Cefai & Cavioni, 2014; Delale-O'Connor et al., 2017); appropriate teaching strategies (Ee & Chang, 2010); enhancing teacher efficacy (Beltman et



al., 2018; Cefai & Cavioni, 2014; Ee & Chang, 2010); *support networks* (Le Cornu, 2009; Papatraianou & Le Cornu, 2014), *social competence* (Cefai & Cavioni, 2014) and *collegial collaborations* (Cefai & Cavioni, 2014; Ee & Chang, 2010); *personal and professional relationships* (Ee & Chang, 2010; Le Cornu, 2009); and *resilient behaviour, responses and ways of thinking* (Ee & Chang, 2010).

Beltman et al. (2018) suggested specifically crafted interventions during teacher training to build teacher resilience. Still, limited evidence is available on incorporating resiliencebuilding activities into teacher education programmes. Ebersöhn and Loots (2017) suggested that teacher training, especially in a context of adversity (i.e., including a challenged context such as the Global South), could benefit from incorporating foci on mapping and mobilising assets and resources. Mobilisation of assets may be achieved by using networks as well as by monitoring and managing the use of resources, according to Muller et al.'s (2014) and Windle's (2010) views on utilising protective factors (individual, life and environment assets) to promote resilience. Given the exploration of teacher efficacy as a protective factor for teacher resilience, the next section highlights teacher efficacy in initial teacher education programmes.

#### 2.6.5 TEACHER EFFICACY IN INITIAL TEACHER EDUCATION PROGRAMMES

The purpose of this section is to position initial teacher education within teacher efficacy research. Pendergast et al. (2011) stated that pre-service teacher education programmes are vital in developing neophyte teacher efficacy. Supporting the development of teacher efficacy is considered an important goal of teacher training, professional performance, identify formation and school improvement (Begum et al., 2020; Lemon & Garvis, 2016; Pendergast et al., 2011). Teachers prepared with essential content knowledge and instructional abilities feel confident enacting pedagogical practices resulting in positive educational outcomes (Duffin et al., 2012).

Quality and effective teacher training develop teacher efficacy of beginning teachers (Averill & McRae, 2019; Brown et al., 2015; Woolfolk Hoy & Spero, 2005). The malleability of teacher efficacy makes it a key intervention focus for teacher training (Al Sultan et al., 2018; Bandura, 1977, 1997; Chesnut, 2017). Therefore, establishing teacher efficacy beliefs during initial teacher preparation is important since failures may hamper self-efficacy development if they occur early on (Bandura, 1997; Pfitzner-Eden, 2016a, 2016b). Research findings (Brown et al., 2015; Cook et al., 2017; Fackler & Malmberg, 2016; Jennings et al., 2013) have shown that formal teacher training efforts affect teacher efficacy to enable fundamental self-beliefs necessary for pre-service teachers to enter the profession. In turn, teacher efficacy may positively affect teacher persistence, resilience, and higher rates of retention over time (Chesnut, 2017). Hence, exploring teacher efficacy beliefs during teacher training may generate knowledge on how pre-service teachers can flourish during and upon programme



completion (Duffin et al., 2012; Lemon & Garvis, 2016). The next section, consequently, explores pre-service teachers as a scholarly domain within the current study.

### 2.7 THE PRE-SERVICE TEACHER WITHIN THE EDUCATION LANDSCAPE

#### 2.7.1 INTRODUCTION

This section outlines the literature on the pre-service teacher as a scholarly domain to situate the pre-service teacher within the education landscape. I discuss the demographic profile of pre-service teachers and highlight intrapersonal resilience-enabling pathways to teacher resilience for pre-service teachers.

#### 2.7.2 CHARACTERISING THE PRE-SERVICE TEACHER

In this section, I review the literature on pre-service teachers within the teaching landscape by exploring the demographic profile of pre-service teachers. I further identify the characteristics of pre-service teachers to establish the risk and protective factors associated with pre-service teachers. As operationalised in Chapter 1, a pre-service teacher signifies a student enrolled in a teacher education programme working towards a teaching qualification and a completely supervised teaching practice. The dynamics, including relationships, experiences support and challenges, of being a pre-service teacher is unique (Averill & McRae, 2019; Brown et al., 2015; Woolfolk Hoy & Spero, 2005). Given the changing education landscape the current generation of pre-service teachers have access to more knowledge than ever. Nonetheless, pre-service teachers may still find it difficult to retrieve the relevant information and knowledge for integration into their practice and show concerns regarding their abilities (Thieman et al., 2014). Although beginner teachers perceive themselves as professionals with a desire to realise positive learner outcomes, they also tend to overrate their capacities (Daniels et al., 2017) since they lack classroom experience and may not yet grasp the realities of the classroom (Brown et al., 2015). Therefore, they can feel uncertainty about their fit within the profession (Thieman et al., 2014).

According to the OECD education indicators (OECD, 2014b), the median age of first graduation from university is 24.7-years worldwide. Younger students (25<sup>th</sup> percentile age distribution) graduated at age 23.2-years, while older students (75<sup>th</sup> percentile age distribution) graduate at age 27.9-years. However, international variation between countries may be high depending on structural factors (e.g., typical graduation age from secondary education and entry into tertiary education) and economic factors (e.g., lack of scholarships). In countries such as Iceland and Israel, the median age of the first graduation is approximately 27-years, while in Belgium and the United Kingdom, the median age is around 22-years. Furthermore, the Hamilton project (2017) postulated that about 80% of undergraduate students enrolled for



a 4-year degree (e.g., BEd programme degree) are between 18 and 24 years of age with about 20% of undergraduate students at 4-year institutions older than 24-years. Therefore, most final year undergraduate teachers tend to be between the age of 22 to 25 years of age (Lemon & Garvis, 2016; O'Neill & Stephenson, 2012a, 2012b).

Internationally (Jenkins, 2019; Lemon & Garvis, 2016; O'Neill & Stephenson, 2012a; World Bank, 2020b) as well as nationally (Petersen, 2014), females seem to outweigh males within the pre-service teacher demographic profile. The statistics are especially true within the FP/ECD (Petersen, 2014) and primary teaching population (O'Neill & Stephenson, 2012a; Zuma et al., 2016). However, an increase in male students entering the FP/ECD teaching phase has been noted in South Africa (Petersen, 2014). The following section introduces the scholarly domain of teacher resilience in pre-service teacher as a relevant body of knowledge within the current study.

#### 2.7.3 PRE-SERVICE TEACHER RESILIENCE

To further illustrate the importance of teacher resilience, I discuss pre-service teacher resilience, narrowing my discussion to pre-service teacher resilience enablers and constraints and the variations in pre-service teacher resilience demographics, given the available extant data of the current study.

Pre-service teacher resilience can be conceptualised, as discussed in Chapter 1, as managing the emotional challenges and inevitable vocational uncertainties driven by the commitment to the teaching profession. For novice teachers, this entails entering the teaching profession with unique views and abilities that enable effective management of professional challenges in a new context (Wosnitza et al., 2014). By exploring self-efficacy and teacher efficacy beliefs as protective resources for teacher resilience, those involved with teacher training may develop programming to promote resilience in beginner teachers. Research of self-efficacy and teacher efficacy beliefs in pre-service teachers is essential for a conceptualisation of teacher development towards effective, flourishing professionals (Thieman et al., 2014), especially in a challenged context. The following section focuses on pre-service teacher resilience enablers.

#### 2.7.3.1 Pre-service teacher resilience enablers

Thieman et al. (2014) reported that resilience and the desire of novice teachers to teach are promoted through educational roles and events within a variety of educational contexts, domains or chronological systems. Pre-service teachers, however, can differ in their entry motivation, negotiation for their career decision-making, and the strength of their commitment to teaching (Hong et al., 2018). Pre-service teachers' resilient qualities are challenged during



teaching practice, and the relationships in which they engage during this time impact their professional development and resilience (Le Cornu, 2009).

The literature highlighted that pre-service teacher resilience is associated with aspects such as *tenacity* (Çelik et al., 2018); *academic achievement* (Yokus, 2015); *acquisition of content knowledge* (Thieman et al., 2014); *motivation* and *intrinsic rewards* (Çelik et al., 2018; Hong et al., 2018; Watt & Richardson, 2008); *relational and social-support networks* (Mansfield et al., 2014; Mapfumo et al., 2012) and *reciprocal learning relationships* (Le Cornu, 2009); *mentoring from experienced teachers and training institution* (Hong, 2012; Le Cornu, 2009); *adequate* and *timely feedback* (Hong, 2012); *beliefs about competency* (Bandura, 1977, 1997; Ee & Chang, 2010); *emotional awareness, regulation, empathy,* and *optimism* (Brown et al., 2015; Ee & Chang, 2010); *commitment, persistence and perseverance* (Hong et al., 2018); *the expectations held about and assessment of the teaching profession* (Buchanan et al., 2013); and *self-knowledge, future-self and identity development* (Chong et al., 2011; Dinham et al., 2017; Pearce & Morrison, 2011; Thieman et al., 2014).

Watt and Richardson (2008) found that novice teachers identifying teaching as intrinsic reward demonstrated high levels of effort and persistence. Although pre-service teachers tend to be altruistic in their motivations (Price et al., 2012; Sayed & McDonald, 2017), they can encounter dissonance when facing the realistic nature of the profession. However, intrapersonal resilience-enabling pathways to resilience may counteract attrition in pre-service teachers transitioning into the profession. Nonetheless, additional risk factors may constrain resilience in pre-service teachers, highlighted in the next section.

#### 2.7.3.2 Risk factors constraining pre-service teacher resilience

Limited literature is available on attrition rates during initial teacher education (Pfitzner-Eden, 2016a). Research on the transition from teacher training to the profession indicate that more than 30% of final year pre-service teachers do not enter the teaching vocation (Pfitzner-Eden, 2016a; Wang et al., 2015). Thieman et al. (2014) foregrounded that the uncertainty of the teaching profession could constrain pre-service teacher resilience since a variety of educational experiences does not necessarily translate into confidence in pre-service teachers' knowledge on the daily work-lives of being a teacher (Sayed & McDonald, 2017; Thieman et al., 2014).

The literature also identified a multitude of stressors, risk factors, or complex challenges for pre-service teachers influencing their perception of the teaching profession, including *poor time management* (Knight, 2007; Mansfield et al., 2016); *limited understanding of the role emotions play in teaching* (Hewitt et al., 2017); *work stress perpetuated by limited strategies to cope with the unexpected* (Cefai & Cavioni, 2014); *supervision related matters and stressful or vulnerable teaching practices* (Le Cornu, 2009; Mapfumo et al., 2012); *impact of teacher-based unrealistic responsibility of learner well-being* (Daniels et al., 2017); *balancing personal*,



family, economic and heavy workload issues (Le Cornu, 2009; Mapfumo et al., 2012); limited resources or learning aids (Mapfumo et al., 2012); and learners' challenging behaviour and classroom management (Cefai & Cavioni, 2014; Delale-O'Connor et al., 2017; Mapfumo et al., 2012; O'Neill & Stephenson, 2012a). The following section explores the variations in preservice teacher resilience demographics based on the extant data of the current study.

# 2.7.3.3 Variations in pre-service teacher resilience demographics given the extant data of the current study

Research (Gu, 2018; Thieman et al., 2014; Wosnitza et al., 2014) emphasised that teacher resilience is vital to keep early career teachers in the teaching vocation. However, Ee and Chang (2010) found that 65.1% of pre-service teachers demonstrated below average resilience while 34.9% had average resilience. Conversely, Yokus (2015) indicated that preservice teachers' psychological resilience levels were high. Studies (Ee & Chang, 2010) found no significant difference between younger and older pre-service teacher resilience profiles, which may imply that resilience is dependent on individuals rather than being age-related. However, this finding can depend on the culture and social experience provided. Other studies (Castro et al., 2010; Gu & Day, 2007; Hong, 2012; Peixoto et al., 2018) indicated that an age difference matters for teacher resilience since resilience is a dynamic process that may fluctuate over time (Gu & Day, 2007). Studies (Hong, 2012; OECD, 2017a; Wosnitza et al., 2014) also reported the high attrition rate amongst beginning teachers. Age may thus be one factor that influences career change (Ee & Chang, 2010). However, according to Ee and Chang (2010), frequent career changes may not develop a more resilient teacher from a transience perspective. Teachers that stay in the profession do also not necessarily reflect resilient traits since endurance may indicate resistance to change or a result of restricted opportunities (Ebersöhn, 2014).

Literature (Beltman et al., 2018; Çelik et al., 2018; Fischer et al., 2018) indicated that resilience between male and female pre-service teacher profiles do not differ significantly. On the contrary, Hartman et al. (2009) found that protective factors fostering resiliency may vary across gender since male and female counterparts can rely on different individual protective factors to nurture resiliency. Comparably, Pareek and Rathore (2016), as well as Petersen (2014), highlighted a significantly higher mean value for female teachers on character strengths and virtues (including resilience). Ee and Chang (2010) noted that female pre-service teachers were more empathetic while male pre-service teachers demonstrated better impulse control which may relate to males having greater inner strengths in taking charge and control (Peglar & Reker, 2008). Similarly, Yokus (2015) observed a statistically significant difference between gender as female pre-service teachers provided more favourable responses in their self-perception, future-perception, and social resources. In addition, Bouillet et al. (2014) reported high resilience in foundation teachers. Teachers who perceived themselves as more



resilient also felt more competent in fostering resilience in their learners (Bouillet et al., 2014). While Yokus (2015) observed no statistically significant difference according to graduation type or class level variables for pre-service teachers. Nonetheless, research (Kavita & Hassan, 2018; Molina et al., 2017; Perlman & Pearson, 2012; Zuma et al., 2016) indicated a difference between primary school teachers' stress and the supportive learning context that is lower than secondary school teachers. The OECD (2018) also highlighted the different teaching and working conditions for primary and secondary teachers. The following section explores the teacher efficacy of pre-service teachers.

#### 2.7.4 PRE-SERVICE TEACHER EFFICACY

This section elaborates on the literature domain of teacher efficacy by discussing pre-service teacher efficacy concerning teacher resilience. I focus my discussion on the protective factors and risk factors associated with pre-service teacher efficacy enabling or constraining teacher resilience. Lastly, I discuss the variations in pre-service teacher efficacy demographics given the available data of the current study.

Pre-service teachers' sense of efficacy has been explored worldwide since the 1990s (Ngidi & Ngidi, 2019; O'Neill & Stephenson, 2012a), often by using a convenience sample of undergraduates enrolled at teaching institutions (Mergler & Tangen, 2010). Findings (Duffin et al., 2012; Fives & Buehl, 2010; O'Neill & Stephenson, 2012b) suggested that pre-service teachers may not yet adequately differentiate between distinct aspects of teaching efficacy (i.e., efficacy for behaviour management, engaging learners and using different instructional strategies). This finding was due to limited pedagogical knowledge and teaching experience, which warranted the use of teacher efficacy as a unidimensional construct. However, some scholars (Pfitzner-Eden, 2016a) argued that senior pre-service teachers with higher teacher efficacy seem more likely to enter the occupation on completion of the teaching experience (Swan et al., 2011). Mergler and Tangen (2010) and Pfitzner-Eden (2016a) noted that pre-service teacher efficacy changes occurred during coursework at a training institution and the supervised teaching practice at schools.

Examining pre-service teachers' efficacy beliefs is essential because these beliefs seem malleable during the commencement of teacher training (Al Sultan et al., 2018; Bandura, 1977, 1997; Duffin et al., 2012; Woolfolk et al., 2009) but can be impervious to change later (Bandura, 1997; Moè et al., 2010; Woolfolk Hoy & Spero, 2005). There is limited literature about the efficacy beliefs of pre-service teachers at the start and during training as well as the extent to which pre-service teacher efficacy influences and is influenced by other psychological factors (Pendergast et al., 2011). Given the argument that teacher efficacy beliefs may be resistant to change after the novice teaching phase, it seems imperative to explore pre-service teachers' efficacy beliefs (Lemon & Garvis, 2016).



# 2.7.4.1 Protective factors associated with pre-service teacher efficacy enabling teacher resilience

Based on Bandura's self-efficacy theory (see Chapter 1), pre-service teachers may ascertain knowledge and skills through the four principal sources of efficacy, including enactive mastery experiences, vicarious or modelling experiences, verbal or social persuasion as well as physiological and emotional states. Pre-service teachers may acquire teacher efficacy through personal experiences, verbal feedback, observation as well as physical and affective states (Bandura, 1977,1986, 2001; Moulding et al., 2014; O'Neill & Stephenson, 2012b; Pendergast et al., 2011; Pfitzner-Eden, 2016a, 2016b; Schunk, 2008; Watson & Marschall, 2019).

For pre-service teachers, factors associated with teacher efficacy have been found to include *responsibility (i.e., learner-outcome focused responsibility and teacher-based responsibility)* (Daniels et al., 2017); *motivation and realistic expectations* (Kim & Cho, 2014); *occupational commitment* (Chesnut, 2017; Chesnut & Burley, 2015; Pfitzner-Eden, 2016a); *retention (e.g., commitment to complete a teaching degree or decreased intention to quit)* (Pfitzner-Eden, 2016a); *teaching practice* (Brown et al., 2015; Clement, 2017; Moulding et al., 2014; O'Neill & Stephenson, 2012b); *identity construction* (Pendergast et al., 2011); and *learner achievement* (Moulding et al., 2014; Tschannen-Moran & Johnson, 2011).

Before any practical experience in the field, pre-service teachers appear to overestimate or miscalibrate their teacher efficacy beliefs (Pendergast et al., 2011). However, undergraduate teachers in the final semester demonstrated a lower level of teacher efficacy than the initial teacher efficacy belief (Pendergast et al., 2011). Clark (2020) echoed the decrease of teacher efficacy at the end of teacher training compared to first-year pre-service teachers. On the other hand, Klassen and Durksen (2014) found a significant linear increase in pre-service teacher efficacy throughout their teaching practice in their final year. Moulding et al. (2014) found that the pre-service teacher efficacy increased after the teacher practice regardless of the school's location or social-economic status. School placement with higher learner achievement was associated with higher pre-service teacher efficacy (Moulding et al., 2014; Tschannen-Moran & Johnson, 2011). The findings may highlight the importance of practical experience, including mastery experiences and the exposure to contextual factors during teacher training (Pendergast et al., 2011). Pfitzner-Eden (2016a) emphasised that the teacher efficacy changes differed by the stage of teacher education. What is more, research (Brown et al., 2015; Wolters & Daugherty, 2007; Woolfolk Hoy & Spero, 2005) suggested different developmental trajectory of teacher efficacy that tends to demonstrate an upward trajectory during training but decline after qualification in the first year of teaching.

Klassen et al. (2011) denoted that improving teacher efficacy is essential in ensuring a resilient profession. The literature indicated that pre-service teacher efficacy could be supported by *enabling critical reflection* (Black, 2015; Hewitt et al., 2017); *self-evaluation* (Hewitt et al.,



2017); encouragement of self-awareness (Hewitt et al., 2017); practising behaviour management (O'Neill & Stephenson, 2012b) and practical skills (Black, 2015); providing information on developmental domains and stages of teachers (Black, 2015); feelings of preparedness (Brown et al., 2015; Lieberman & Darling-Hammond, 2012); descriptive and extensive verbal and written motivational feedback (Black, 2015; Moulding et al., 2014); and support systems, supervision, and positive relationships (Hewitt et al., 2017; Lieberman & Darling-Hammond, 2012; Moulding et al., 2014; K. E. Thomas & Mucherah, 2016; Woolfolk Hoy & Spero, 2005). Immersive and support of peers and mentors and the support structures inside and outside the training institution (Hewitt et al., 2017), may enhance preservice teacher efficacy (Lieberman & Darling-Hammond, 2012; Moulding et al., 2014; Woolfolk Hoy & Spero, 2005). The next section explores risk factors associated with preservice teachers constraining teacher resilience.

# 2.7.4.2 Risk factors associated with pre-service teacher efficacy constraining teacher resilience

A lack of teacher efficacy can impact pre-service teacher outcomes and the outcomes of the learners in their care (Hewitt et al., 2017). Factors that may hamper pre-service teacher efficacy may involve the *lack of supervisors or mentor teacher approval or support* (Black, 2015; Klassen & Durksen, 2014; Moulding et al., 2014); *low motivation and confidence* (Lemon & Garvis, 2016); *limited rapport or not perceiving a positive impact with learners* (Black, 2015; Mergler & Tangen, 2010); *struggling to engage with colleagues, parents, or caregivers* (Castro et al., 2010); *instructional, curriculum planning and implementation* (Black, 2015); *inadequate exposure to mastery or practical experiences* (O'Neill & Stephenson, 2012a); *negative physical and affective states* (Bandura, 1986; O'Neill & Stephenson, 2012a); and *classroom and behaviour management* (Black, 2015; Mergler & Tangen, 2010; O'Neill & Stephenson, 2012a).

Pre-service teachers find deficient preparation for behaviour management a concern during teaching practice (Black, 2015; O'Neill & Stephenson, 2012a), demonstrating a lack of confidence in the area (Delale-O'Connor et al., 2017). Further risk factors such as *undefined roles* and *excessive responsibility* which create guilt and tension may lead to *job-related stress, burnout, decreased job satisfaction and attrition* for pre-service teachers (Clement, 2017; Fives & Buehl, 2010; Jackson & Rothmann, 2005; Pfitzner-Eden, 2016a; Zuma et al., 2016).



# 2.7.4.3 Variations in pre-service teacher efficacy demographics given extant data of the current study

Pre-service teachers seem to demonstrate a high sense of teacher efficacy, thus feeling that they can influence learning and behaviour of learners (Beltman et al., 2018; Fives & Buehl, 2010; Klassen et al.'s; 2011; Kyriakides et al., 2013; Moulding et al., 2014; Ngidi & Ngidi, 2019; O'Neill & Stephenson, 2012b; Özdemir, 2008; Swan et al., 2011). Also, it appears that preservice teacher efficacy was higher begging teachers in schools with better learner performance (Caprara et al., 2006; Moulding et al., 2014; Tschannen-Moran & Johnson, 2011) but that socio-economic status, disadvantaged communities or social background did not have a significant impact on pre-service teacher efficacy (Morgan, 2011; Moulding et al., 2014). Additionally, the age of pre-service teachers does not seem significantly related to teacher efficacy (Pendergast et al., 2011; Pfitzner-Eden, 2016a) since, according to Bandura (1997), self-efficacy is not associated with age but rather with professional teaching experience. Furthermore, literature (Beltman et al., 2018; Klassen & Durksen, 2014; Murshidi et al., 2006; Ngidi & Ngidi, 2019; O'Neill & Stephenson, 2012b; Pendergast et al., 2011; Pfitzner-Eden, 2016a; Tschannen-Moran & Woolfolk Hoy, 2007) indicated no significant difference of preservice teacher efficacy by gender. However, the association between pre-service teacher efficacy and for example, commitment seem stronger for male pre-service teachers than the relationship for female pre-service teachers (Chesnut, 2017). In comparison, female preservice teachers seemed more inclined toward mastery approaches than males (Daniels et al., 2017). On the other hand, Brandon (2000) found that female pre-service teachers had lower self-efficacy relating to teaching competencies than male pre-service teachers before teaching practice experience. In contrast, Özdemir (2008) and Üstüner (2017) reported that male preservice teachers had lower teacher efficacy beliefs than their female counterparts.

Teacher efficacy for pre-service teachers in primary (Brown et al., 2015; Moulding et al., 2014; O'Neill & Stephenson, 2012b), as well as secondary schools' (O'Neill & Stephenson, 2012b; Woolfolk Hoy & Spero, 2005) teaching education programmes, seemed high. Although Pendergast et al. (2011) noted that the mean for teacher efficacy was highest for early childhood and primary pre-service teachers, no statistically significant difference was obtained. At the same time, Woodcock (2011) reported that primary pre-service teachers held a significantly higher teacher efficacy belief than secondary pre-service teachers. Given the pivotal bodies of literature discussed in the previous sections, the following section put forth the conceptual framework for this study.



### 2.8 THE CONCEPTUAL FRAMEWORK

#### 2.8.1 INTRODUCTION

The chosen theoretical framework (i.e., Social Cognitive Theory [Bandura, 1986]) is discussed in Chapter 1 (see Section 1.7). In this section, I discuss the conceptual framework of the current comparative case study. The conceptual framework aimed to guide the process of analysis and the interpretation of results to inform systematic knowledge on intrapersonal resilienceenabling pathways (i.e., self-efficacy and teacher efficacy) to teacher resilience. Therefore, I contextualise intrapersonal resources, in a Global South educational space, to allow for a comparison of pre-service teachers' self-efficacy and teacher efficacy as intrapersonal protective pathways to teacher resilience in a challenged context.

# **2.8.2** INFORMING KNOWLEDGE ON TEACHER RESILIENCE IN A CHALLENGED CONTEXT THROUGH A SOCIAL COGNITIVE LENS

In Figure 2.7, I present a conceptual framework for this study. Figure 2.7 illustrates the conceptual framework that unites key elements to consider the self-efficacy and teacher efficacy of pre-service teachers as protective factors in a challenged context to inform intrapersonal resilience-enabling to teacher resilience.



#### Figure 2.7

Informing Knowledge on Teacher Resilience in a Challenged Context: An SCT Lens to Self-efficacy and Teacher Efficacy of Pre-service Teachers

# TO INFORM KNOWLEDGE ON TEACHER RESILIENCE IN A CHALLENGED CONTEXT VIEWED THROUGH A SCT LENS

#### The objectives were to compare:

- · the self-efficacy and teacher efficacy beliefs of pre-service teachers within a challenged context
- the self-efficacy and teacher efficacy beliefs of pre-service teachers within a challenged context based on demographic information
- · the relationship between self-efficacy and teacher efficacy beliefs among pre-service teachers in a challenged context



Adapted from Bandura (1977, 1997) and Ebersöhn (2014)



As highlighted in Section 2.2, resilience only becomes evident during monumental adversity and structural disparity (i.e., a challenged context), such as South Africa as an exemplar of a Global South place (Ebersöhn, 2014). As foregrounded in Chapter 1, South Africa is a highrisk, high-need Global South context with inequality and cultural or political marginalisation typified by an emerging economy and high adversity as well as structural disparity due to a post-colonial history (Dados & Connell, 2012; Schwartz & Harris, 2017). Ebersöhn (2017) postulated that the ecology of resilience matters when considering education and context. Research generating evidence to address challenges due to multiple and chronic disruptions can elicit valuable insights while acknowledging the risk factors. A resilience lens may promote dialogue on evidence of better than expected educational outcomes despite a high-risk ecology to counter a dis-enabling disaster perspective (Ebersöhn, 2016).

As per Section 2.2.3, I drew on trait-resilience (Block & Block, 1980; Block & Kremen, 1996; Letzring et al., 2005), grit and hardiness (Kobasa et al., 1982; Maddi et al., 2017; Prince-Embury, 2010), the salutogenic theory's sense of coherence (A. Antonovsky, 1979), Fredrickson's theory of positive emotions (Fredrickson et al., 2003), adaptive coping (Skinner & Zimmer-Gembeck, 2011; Willers et al., 2013; Zimmer-Gembeck et al., 2018), relational resilience (Ebersöhn, 2012; Jordan, 2006, 2013) as well as transactional-ecological process thinking (Ungar, 2011; Ungar et al., 2013) for the conceptualisation of psychological resilience in spaces of high challenge. Furthermore, the lenses employed to conceptualise teacher resilience included the multidimensional approach (Gu & Day, 2007), strategic approach (Patterson et al., 2004), active agent approach (Castro et al., 2010), the Multi-level Teacher Resilience Model (Wosnitza et al., 2014), the teacher resilience process (Mansfield et al., 2018) as well as a contextualised structural disparity lens (Ebersöhn, 2014).

The contextualised structural disparity lens (Ebersöhn, 2014) is specifically employed in the conceptual framework given the Global South education landscape depicted in the current study. In addition, the Social Cognitive Theory (Bandura, 1986, 1989a, 1989b), underpinning the current study as the theoretical framework (see Chapter 1), postulated the notion that adaptation occurs in a social context where pre-service teachers can actively contribute (i.e., individual as active agent) to their development within a reciprocality process over time (Bandura, 1989a, 1989b; Bandura et al., 1977; Schunk, 2008). The individual (i.e., pre-service teachers with internal traits including for example grit, hardiness, positive emotions, and efficacy), behaviour (e.g., employing adaptive coping strategies, beliefs and protective resources) and the environment (i.e., including risk [see Table 1.1] and protective factors [see Table 2.2] in a challenged context) thus continuously influence one another bidirectionally. Given the Social Cognitive Theory's assumptions, pre-service teachers may select, influence and construct their circumstances. Consequently, they are active agents (correlating with Castro et al.'s [2010] active agent view for teacher resilience), shaping themselves and their



context (Bandura, 1989a, 1989b) which include the socio-ecological environment (Ungar, 2008, 2010, 2011, 2012a, 2012b; Ungar et al., 2013).

As a central feature of the Social Cognitive Theory (1986, 1989a, 1989b), self-efficacy beliefs (Bandura, 1977, 1986; Bandura et al., 1977) can affect pre-service teachers' behaviour, perceptions, and outcomes directly or indirectly. As a result, pre-service teachers may pursue goals as well as challenges, rebound from setbacks, persist or demonstrate grit, hardiness, control as well as optimism when faced with obstacles (Bandura, 1986, 1993, 1996, 1997; Perkins-Gough, 2013). Therefore, even if there is adversity, self-efficacy can act as a protective resource if the pre-service teachers believe they can succeed and recover from setbacks (Hewitt et al., 2017). The latter may likely "counteract" adversity or enable resilience pathways. From this perspective, self-efficacy is an element of competence (Lemon & Garvis, 2016), predicting possible enabling pathways to resilience. Consequently, Bandura's theory, as discussed in Chapter 1, emphasised behaviour regulation, reflection, and active agency rather than the environmental product outcome alone (Bandura, 1977, 1997, 2006; Hewitt et al., 2017).

Perceived self-efficacy may enable control over pre-service teachers' functioning and (stressful) events that could influence their lives (Lemon & Garvis, 2016). The aforementioned may link with how pre-service teachers appraise (significant) adversity and positively adapt (Skinner & Zimmer-Gembeck, 2007, 2011) through an interacting complex transactional coping system by utilising capacities, learnt skill and the mobilisation of contextual resources to manage cumulative and chronic risk factors (Ebersöhn, 2014). Thus, the individual systems act as a pathway of adaptation resulting in unexpected positive outcomes and competencies despite significant adversity in a challenged context. Therefore, the pre-service teacher must appraise their circumstances as adverse and respond to the experience in a manner that reflects resourceful adaptation employing protective traits such as self-efficacy (i.e., recovery from setbacks) and teacher efficacy (i.e., confidence in teaching and behaviour management).

Given the latter pre-service teachers might evaluate adverse conditions in a challenged context based on its comprehensibility, manageability, and meaningfulness (A. Antonovsky & Sourani, 1988). If pre-service teachers perceive their environment as predictable and manageable, they may have the confidence to confront challenges, utilising internal resources (e.g., self-efficacy, teacher efficacy, positive emotions) and contextual assets to address barriers (A. Antonovsky & Sourani, 1988; Masten, 2001). Extrapolating from the broaden-and-build theory (Fredrickson, 2001), positive emotions are active ingredients within trait resilience (Cohn et al., 2009; Fredrickson et al., 2003) since positive emotions buffer against adversity and fuel coping, thriving as well as life satisfaction (Cohn et al., 2009; Fredrickson, 2001). Resilient pre-service teachers may thus quickly and effectively "bounce back" from challenges, and routine positive emotions can enable increased protective resources (Tugade & Fredrickson, 2004). Therefore, positive emotions (including feeling confident and competent)



assist with adaptive coping during challenges (Skinner & Zimmer-Gembeck, 2011; Tugade & Fredrickson, 2004; Willers et al., 2013; Zimmer-Gembeck et al., 2018).

Given the trait-discourse with self-efficacy and teacher efficacy postulated as protective factors enabling teacher resilience intrapersonal resilience-enabling pathways, the current study analysed the FIRE Teacher Resilience Measure to compare the self-efficacy and teacher efficacy of final year pre-service teachers (within-case and cross-case) in a challenged educating context aiming to inform knowledge on teacher resilience in a Global South setting. In Chapter 5, Section 5.5, the conceptual framework is revisited together with the hypotheses to theorise findings within the conceptual framework to highlight an evidence-based theoretical framework for teacher resilience in spaces of high risk.

#### 2.9 CONCLUSION

A main conclusion, evident from the literature, is the limited research available on *teacher resilience* in a challenged context given a Global South discourse, and particular from *quantitative studies* and with *pre-service* teachers. In Chapter 2 I situated this study within existing literature and a theoretical framework to compare the self-efficacy and teacher efficacy of pre-service teachers in a challenged context. I discussed the literature on resilience, teacher resilience, self-efficacy, teacher efficacy beliefs, teacher education, and pre-service teachers in a challenged setting. I concluded the chapter by justifying the conceptual framework that guided me in undertaking the current study.

In Chapter 3, I explain the epistemological perspective and methodological approach, together with the research design employed. I examine sample and the statistical procedures used for the current study. I conclude the chapter with a detailing the standards of rigour and the ethical guidelines followed throughout the current study.

---000----



# Chapter 3 Research Design, Methodology and Strategies

#### 3.1 INTRODUCTION

In Chapter 2, I embarked on a literature review associated with resilience, teacher resilience, self-efficacy, teacher efficacy, initial teacher education, pre-service teacher resilience and preservice teacher efficacy within a challenged context to compare the self-efficacy and teacher efficacy beliefs of pre-service teachers to gain insight into teacher resilience in a Global South education space. I further presented the theoretical framework, namely the Social Cognitive Theory (Bandura, 1986), in Chapter 1, as a lens for the conceptual framework discussed in Chapter 2.

In Chapter 3, I provide a description of the research processes followed during this study to address the research questions based on testing the formulated hypotheses. In this chapter, I also highlight the paradigmatic approach that informed the research. I further justify the choice of comparative research design and methodology, including the strengths and limitations of methodological decisions. The chapter describes the sampling and selection strategies, the statistical data analysis processes and the interpretations thereof. Based on the purpose of the current study, I conducted a non-experimental, descriptive and inferential mode of inquiry in the form of a quantitative secondary data analysis. As part of the FIRE project (Appendix C), the dataset obtained from the FIRE Teacher Resilience Measure (Appendix B) was purposefully selected to compare the self-efficacy and teacher efficacy of pre-service teachers within the context of the relevant literature. I conclude the chapter by discussing the quality criteria employed to ensure rigour and the ethical guidelines pursued throughout the current study. Figure 3.1 presents a flow chart showing the organisational overview of Chapter 3.



#### Figure 3.1

Flow Chart Showing the Organisation of Chapter 3



#### 3.2 RESEARCH QUESTIONS AND HYPOTHESES

#### 3.2.1 INTRODUCTION

According to Donnellan et al. (2011), quality research should proceed from an important, specific and interesting question grounded by theoretical concerns or knowledge gaps. Quantitative research questions investigate the relationships among variables under inquiry, while hypotheses within quantitative research, are numerical estimations about the expected outcomes of relationships (Creswell, 2014; Creswell & Creswell, 2018). Therefore, I used research questions and hypotheses to direct and steer the purpose of the current study (L. Cohen et al., 2018; Creswell, 2014; Creswell & Creswell, 2018). This section, consequently, presents the primary and secondary research questions and the hypotheses addressed by the current study.

#### 3.2.2 PRIMARY RESEARCH QUESTION

Considering the rationale and purpose of this study, depicted in Chapter 1, my inquiry was guided by the following primary research question:

How can insight into the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged education context inform knowledge on teacher resilience?



#### 3.2.3 SECONDARY RESEARCH QUESTIONS

To address the primary research question, the current study explored the subsequent secondary research questions:

- How do self-efficacy and teacher efficacy beliefs of final year pre-service teachers within a challenged education context (within-case and cross-case) compare?
- What are the self-efficacy and teacher efficacy beliefs of pre-service teachers within a challenged education context based on demographic information (within-case and cross-case)?
- To what extent is there a relationship between self-efficacy and teacher efficacy beliefs among pre-service teachers in a challenged education context (within-case and cross-case)?

#### 3.2.4 HYPOTHESES

Quantitative research is well suited for testing hypotheses (Ary et al., 2019), which contain numerical estimations the research holds about the relationship among two or more variables (i.e., self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged education context) (Creswell, 2014). A hypothesis is a tentative explanation, logical conjecture or supposition that accounts for a set of facts based on data collected from samples, and it can be tested by further investigation (Ary et al., 2019; Creswell, 2014; O'Leary, 2017). A testable statement (i.e., hypothesis) operationalise numerical procedures to draw inferences or conjectures from a given sample under investigation (L. Cohen et al., 2018; Creswell, 2014). Hypotheses cannot be proven or accepted within the post-positivist paradigm and may either be rejected or not rejected (i.e., failure to reject), as evidence is seen as fallible and imperfect (Creswell, 2014; Sauro & Lewis, 2016). Criticism opposed to the custom of hypotheses (O'Leary, 2017; Sharpe, 2013; Trafimow & Marks, 2015) contend that hypotheses have limited usefulness for conveying relevant quantitative information and may constrain social research. Nonetheless, it is still widely used in research (Sharpe, 2013) and is seen as valuable in the appropriate context (O'Leary, 2017).

The variables (constructs or units) (Ary et al., 2019; Creswell, 2014) within the current study were operationalised by using relevant scales (FIRE Teacher Resilience Measure: Teacher Resilience scale<sup>45</sup> and Teacher Efficacy scale) (see Appendix B). To address the research questions, the hypotheses were formulated to test for a statistically significant difference or correlation between variables.

<sup>&</sup>lt;sup>45</sup> Operationalised as self-efficacy as discussed in Chapter 1.



#### 3.2.4.1 Statistical hypothesis of the population difference

The statistical hypotheses related to assumptions based on the population difference are stated as follows:

- *H*<sub>0</sub>: Median<sub>dif</sub> = 0 (median population difference between self-efficacy and teacher efficacy beliefs [within-case and cross-case] does not differ significantly from zero).
- H<sub>a</sub>: Median<sub>dif</sub> ≠ 0 (median population difference between self-efficacy and teacher efficacy beliefs [within-case and cross-case] differs significantly from zero).

The null hypothesis ( $H_0$ ) state that there is no statistically significant difference (Creswell & Creswell, 2018; López et al., 2015; Pietersen & Maree, 2019d; Sauro & Lewis, 2016) between the self-efficacy and teacher efficacy beliefs of final year pre-service teachers (within-case and cross-case). Thus, the expectation of the null hypothesis is that there is no statistically significant difference in pre-service teachers' confidence in recovery from setbacks in school (i.e., self-efficacy) and their confidence in teaching and behaviour management (i.e., teacher efficacy beliefs). Conversely, the alternative hypothesis ( $H_a$ ) state that there is the expectation that there is a statistically significant difference (L. Cohen et al., 2018; Creswell & Creswell, 2018; Pietersen & Maree, 2019d; Sauro & Lewis, 2016) between the self-efficacy and teacher efficacy beliefs (within-case and cross-case) of pre-service teachers within a challenged context.

When testing a hypothesis, a *p*-value is calculated and compared to the level of significance ( $\alpha = .05$ ), with  $\alpha$  being the probability of rejecting a null hypothesis ( $H_0$ ) that is true (López et al., 2015). A 5% level of significance is typically used in behavioural studies (Cozby & Rawn, 2016). If the *p*-value is less than .05, the null hypothesis ( $H_0$ ) is rejected, and there is a statistically significant difference between the self-efficacy and teacher efficacy beliefs of preservice teachers. If the *p*-value is greater than .05, the null hypothesis ( $H_0$ ) is not rejected, and there is not a statistically significant difference between the self-efficacy and teacher efficacy and teacher efficacy of pre-service teachers (Field, 2018; Sauro & Lewis, 2016).

# 3.2.4.2 Statistical hypothesis of the population correlation

The statistical hypotheses related to assumptions based on the population correlation coefficient are stated as follows:

- *H*<sub>0</sub>: There is no statistically significant correlation between self-efficacy and teacher efficacy beliefs.
- *H*<sub>a</sub>: There is a statistically significant correlation between self-efficacy and teacher efficacy beliefs.



If the *p*-value is less than .05, the null hypothesis is rejected and there is a statistically significant correlation and, on the other hand, if the *p*-value is greater than .05, the null hypothesis is not rejected then there is no statistically significant correlation. Given the hypotheses stated, the assumption was that variations would be noted between groups of preservice teachers. It was hypothesised that within groups of pre-service teachers, their self-efficacy and teacher efficacy beliefs may differ, and the sources of variance could be different for distinct groups.

If the null hypothesis ( $H_0$ ) is not supported, a significant *p*-value does not provide any warrantable evidence about the theoretical meaningfulness or extent of the effect. Rejecting the null hypothesis ( $H_0$ ) indicates that there is a low probability of no effect or no difference, or correlation. Rejecting the null hypothesis indicates that the alternative hypothesis ( $H_a$ ) is a more likely explanation (Sharpe, 2013). The current study is not focused on determining the mediated or indirect effect on self-efficacy or teacher efficacy beliefs within a challenged context. Since causality cannot be sustained from non-experimental or cross-sectional data<sup>46</sup> (L. Cohen et al., 2018), the current study aimed to compare associations and directions of the relationship between variables (i.e., self-efficacy and teacher efficacy beliefs) (Roni et al., 2020) in a challenged context. Given the research questions and hypotheses, the next section situates the current study within a post-positivist, quantitative paradigm.

#### 3.3 PARADIGMATIC PERSPECTIVES

#### 3.3.1 INTRODUCTION

A research perspective is a set of basic beliefs applied as understanding of worldview or inferences (Babbie, 2021; Lincoln et al., 2011). A research paradigm guides how a researcher views material related to scientific inquiry (De Vos et al., 2011). A broad research approach "is thus the plan or proposal to conduct research" (Creswell & Creswell, 2018, p. 43). In this study, I employed a quantitative methodological paradigm anchored in post-positivism, as depicted in Table 3.1. Post-positivism is often associated with a quantitative approach, with quantitative researchers being primarily post-positivist (Creswell & Creswell, 2018; Johnson & Gray, 2010; Ramlo & Newman, 2011; Tuli, 2010).

<sup>&</sup>lt;sup>46</sup> Data collected at one point in time (L. Cohen et al., 2018; Creswell & Creswell, 2018). Within the current study cross-sectional data refer to the extant data collected during the FIRE project in 2015, 2016 and 2017 with final year pre-service teachers.



#### Table 3.1

Paradigmatic Perspectives

Paradigmatic perspectives		
Meta-theory	Post-positivism	
Methodological paradigm	Quantitative research approach	

#### 3.3.2 EPISTEMOLOGICAL PERSPECTIVES: POST-POSITIVISM AS META-THEORY

In Section 3.3.2, I discuss post-positivism as meta-theory for this study. I highlight the implications of the post-positivist researcher. I further note advantages and limitations of this perspective and how it was addressed in the current study.

As a scientific methodology, the post-positivist paradigm is viewed as an extension to positivism by challenging the traditional positivist worldview (Creswell & Creswell, 2018; Panhwar et al., 2017). Post-positivism postulates that the world is ambiguous and open to interpretation (O'Leary, 2017) and aims to understand phenomena holistically (e.g., teacher resilience) (Tekin & Kotaman, 2013) whilst still embracing the scientific method (L. Cohen et al., 2018). The post-positivist view holds to a realist ontology (L. Cohen et al., 2018) and therefore searches for reliable, reproducible and valid empirical evidence to explain and correspondingly verify the prediction of phenomena, laws or theories through scientific testing (Creswell & Creswell, 2018; Lincoln et al., 2011). Through this lens, the intricacies of any social occurrence and different perspectives, possibilities and points-of-view, are acknowledged while research is conducted (Lor, 2011; Tekin & Kotaman, 2013). Reality is understood probabilistically (Howell, 2013) and conceived as a subjective, mentally crafted construct within the context of influential factors and a given society (Tekin & Kotaman, 2013). Knowledge accumulated, generated, investigated and shaped through a post-positivist paradigm is grounded in clear scientific measurement, generalisation and observation of reality in the world (Creswell & Creswell, 2018; Howell, 2013; Panhwar et al., 2017). Nonetheless, an absolute truth within post-positivist research can never be proven or predicted since knowledge is assumed to be conjectural, imperfect, provisional and fallible (Creswell, 2014; Creswell & Creswell, 2018; Lincoln et al., 2011; Lor, 2011).

This reality implies that concrete universal laws, theories and facts cannot be attained as they lack context (Tekin & Kotaman, 2013). Consequently, within post-positivism, the world is without a defined truth, and knowledge may be reliant on one's ability to predict changing social phenomena (O'Leary, 2017, Tuli, 2010). Shifts in complex conclusions can accordingly occur over time given future evidence (L. Cohen et al., 2018) and are provisional within a particular society or context-bound knowledge systems (L. Cohen et al., 2018; Tekin & Kotaman, 2013). Diversity (e.g., different cultural groups) and idiographic situations are recognised (O'Leary, 2017), but also that outcomes from one context may apply to other



contexts (i.e., the purpose is generalisation) (Creswell, 2014; Lincoln et al., 2011; Lor, 2011). Individuals' subjective knowledge is shaped by rational considerations, evidence, and data and is therefore regarded as a valid form of knowledge creation (Creswell, 2014). Therefore, post-positivist research aims to develop accurate and appropriate statements that can assist in explaining multi-layered situations by adopting a distanced view to gain perspective of the data (Creswell, 2014). This paradigm predicts apparent reality by investigating the causal or underlying law-like perceptions when scientific methods are applied to understand the world (De Vos et al., 2011; Tuli, 2010).

The post-positivists researchers' goal is to maintain as much objectivity as feasible during research while acknowledging the probable influence of the researcher in the investigation (Creswell & Creswell, 2018; Ramlo & Newman, 2011). As L. Cohen et al. (2018, p. 138) denoted, "it is almost impossible for researchers to free themselves from their values and perspective in a post-positivist era, and indeed there may be unintentional breaches of ethics, researchers must be vigilant, very self-aware and reflexive". Since post-positivism believes that no universal truth (i.e., knowledge is partial, probabilistic and provisional) is possible, it leaves room for subjective perspectives, critical multiplism or pluralist view (L. Cohen et al., 2018; O'Leary, 2017; Panhwar et al., 2017). Therefore, a post-positivist researcher views objectivity as relative and open to be critiqued (O'Leary, 2017; Panhwar et al., 2017). Bird (2020, p. 83) defined the "norm of objectivity as a set of guidelines for the gathering, interpreting and reporting on research which views this reporting as intelligible, reasonable, discursive, and inherently reciprocating, public activity". Therefore, objectivity in social sciences is guided by the authentic representation of research, using measures that enable evaluation and verification to exercise accountable judgments and open dialogue (Bird, 2020). Thus, the researcher should examine methods and conclusions for bias by ensuring rigorous research (e.g., the standard of validity and reliability, as discussed in Section 3.4.5) (Creswell & Creswell, 2018), alternative elaborations and collaborative discussion (Panhwar et al., 2017). Consequently, the post-positivist researcher should remain conscious of the possible effect of research bias in all studies (Creswell, 2014; Panhwar et al., 2017) and highlight diverse viewpoints, possibilities, perspectives, and various variables that may affect results (Lor, 2011).

Post-positivist principles emphasise meaning and the interpretative creation of new knowledge (Panhwar et al., 2017). As an independent change agent, the researcher can use the scientific information to form, explain, and justify actions and policies through activism and advocacy (Howell, 2013; Lincoln et al., 2011). As a critical realist (Panhwar et al., 2017), the post-positivist researcher thus provides greater scope for human endeavour and social-educational movement (Howell, 2013). Nonetheless, when investigating the actions and behaviour of individuals or phenomena, the researcher cannot be certain about the claims of knowledge made from a post-positivist study and should scrutinise subjective assumptions



and possible bias regarding conclusions established (Creswell, 2014; Creswell & Creswell, 2018). Subsequently, I needed to remain cognisant of suppositions about conjectural conclusions drawn from the results and subsequent constraints of objectivity since an absolute truth cannot be found within a post-positivist paradigm (Creswell & Creswell, 2018; Ramlo & Newman, 2011).

A post-positivist view seemed suitable for the current study by prompting knowledge creation, quantification and creating meaning through critical realism (Lincoln et al., 2011; Lor, 2011; O'Leary, 2017; Panhwar et al., 2017). As I set out to compare the variables, self-efficacy and teacher efficacy beliefs of final year pre-service teachers within a challenged context, through research questions and hypotheses, the post-positivist paradigm appeared relevant (Creswell, 2014; Howell, 2013; Lor, 2011), from a fundamentally deterministic and reductionistic philosophy (Creswell, 2014; Creswell & Creswell, 2018; Lincoln et al., 2011). In addition, post-positivist inquiry allows for a rich, in-depth understanding of what is being studied within a more extensive social system (O'Leary, 2017; Tekin & Kotaman, 2013), which aligns with the conceptual framework employed in the current study. A post-positivist lens allowed the opportunity to compare the relationship (Creswell, 2014), if any, of interest between variables (i.e., self-efficacy and teacher efficacy beliefs). It provided amiable and unambiguous information to examine patterns across many cases (Babbie, 2021; Yin, 2018). The justification of post-positivism as a research paradigm for the current study implied the benefit of outcomes being transparent, methodical and dependable (O'Leary, 2017) given the standards of rigour the current study adopted during the research endeavour (Scotland, 2012). The post-positivist paradigm produces explicit evidence to support generated findings and conclusions based on a systematic, comparative description of a reliable extant dataset without perceived perfection and minimising the risk of bias (Panhwar et al., 2017).

Critics of post-positivism, such as phenomenologist philosophers, still question the adherence to the idea of external reality and dualist objectivity based on the interaction between "self and the world" (Howell, 2013, p. 33). Furthermore, post-positivism is usually highly structured, which can prevent the exploration of unexpected outcomes or information (Howell, 2013). As such, the conjecture is made that reality exists, but as a creation of the research respondents' construction of influential contextual factors, and can accordingly never be fully understood, resulting in a limited understanding of their perspectives (Lincoln et al., 2011; Scotland, 2012). To address the concerns, I espoused a flexible and reflexive (O'Leary, 2017; Panhwar et al., 2017) open-minded stance as this perspective is regarded as a key factor in post-positivist research.

Given that findings may be conditional and situationally embedded within a specific natural context (Tekin & Kotaman, 2013), I constantly employed a reflexive practice. This practice consisted of a generated audit trail, e.g., by stipulating an account of the research process and decisions made, the literature review, the chosen conceptual framework and my



research diary and notes, regarding the research process (Nieuwenhuis, 2016). A meticulous and organised research approach may create dependable, transparent descriptions of the research process (O'Leary, 2017). To see the entire representation, and not as the sum of its parts, a post-positivist researcher should take a distance perspective and remain openminded, self-critical, flexible and reflexive about the research (Tekin & Kotaman, 2013). A reflexive distance approach implies the capability of the researcher to evaluate the research process from the outside while unceasingly considering what is being researched and the credibility and integrity of the complete research process, despite the unavoidable practicalities (O'Leary, 2017). Even so, since knowledge within this paradigm is constructed from data, evidence and rational considerations, it is considered a valid form of knowledge (Creswell, 2014). I, thus, approached the current study from a reflexive awareness, deliberating the fact that the findings may not be co-created and fitting to the context in which the data had been collected. I furthermore continuously reflected on biases that might influence the research outcomes (Creswell & Creswell, 2018).

A further limitation of employing a post-positivist epistemology is the possibility to employ inaccurate or inappropriate statistical procedures during data analysis (e.g., performing parametric statistical procedures on data if data was not normally distributed) (Scotland, 2012). Hence, it was essential to examine the extant dataset to direct the chosen statistical procedures. This examination was conducted under the supervision of the faculty statistician and co-supervisors of this study. Therefore, based on the background, the current study employed a post-positivist worldview to provide empirically reliable data and information (Creswell & Creswell, 2018; O'Leary, 2017; Tuli, 2010) on the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context. This information was sought to gain insight into teacher resilience through quantified observations (Creswell & Creswell, 2018).

#### 3.3.3 METHODOLOGICAL PARADIGM: QUANTITATIVE RESEARCH APPROACH

In Section 3.3.3, I review the quantitative research approach as methodological for the current study. I focus on the implications for the quantitative researcher. I further consider this approach's strengths and possible caveats and how it was addressed in the current study.

Creswell (2003, p. 18) described a quantitative paradigm as "one in which the investigator primarily uses post-positivist claims for developing knowledge (i.e., cause and effect thinking, reduction to specific variables and hypotheses and questions, use of measurement and observation and the test of theories), employs strategies of inquiry such as experiments and surveys and collects data on predetermined instruments that yield statistical data". As evident from the quotation, a quantitative approach draws together accumulated data that tends to be numerical and statistical, resulting from empirical scientific methods and measures (Creswell, 2014; P. Kruger & Janeke, 2011; Ramlo & Newman, 2011; Roni et al.,



2020). A quantitative methodological paradigm seeks to quantify the social world systematically and objectively, assuming that scientific inquiry is based on general laws (L. Cohen et al., 2018; O'Leary, 2017). An objectivist approach to social science is grounded in realism, positivism, deductive determinism standardised data collection and nomothetic procedures and methods (Ary et al., 2019; L. Cohen et al., 2018). The intention is to focus on measurable aspects of behaviour or the unit of analysis through a formal systematic and controlled statistical research process (Ary et al., 2019; Creswell, 2014; Maree & Pietersen, 2019a; Van Rensburg et al., 2010). Quantitative data analysis, therefore, depends on deductive reasoning, starting with a premise (e.g., hypothesis) and then drawing logical conclusions from findings (Creswell & Creswell, 2018; Leedy & Ormrod, 2015). A quantitative approach is used to determine and address inquiries about the relationships between variables or differences between groups under investigation (Leedy & Ormrod, 2015; Van Rensburg et al., 2010) and to evaluate the social world objectively (O'Leary, 2017) to generalise, explain or make claims about the population based on the sample results or subgroup population (Creswell, 2014; Leedy & Ormrod, 2015; Maree & Pietersen, 2019c). The main objective of quantitative research is thus to portray patterns or to quantify significant relationships between variables (such as self-efficacy and teacher efficacy beliefs) by collecting numerical data using formal instruments of measure (e.g., FIRE Teacher Resilience Measure) and test quantifiable hypotheses (Ary et al., 2019; Creswell & Creswell, 2018; Durrheim & Painter, 2014).

Ontologically, the quantitative research approach assumes an objective reality external of human perception (L. Cohen et al., 2018). The quantitative researcher formulates exact research questions or poses hypotheses that can be quantified (Creswell & Creswell, 2018). Bird (2020), however, provided nuanced guidelines for the social sciences highlighting honest representation, responsible judgments and enhancing public activity as opposed to an alone standing objective reality. Epistemologically, the researcher and the respondents or object under study are independent because reality is not co-constructed (Creswell & Creswell, 2018). The researcher assumes that knowledge is independent of the investigator and remains objectively separated since a quantitative approach has a clear aim in advance and imposes an external system upon a phenomenon (Babbie, 2021; Creswell & Creswell, 2018). The researcher measures dimensions through the quantification of constructs (Babbie, 2021; Neuman, 2014). The quantitative paradigm, thus, enables a researcher to deductively use literature (Creswell & Creswell, 2018) to employ statistical methods to portray potential associations, trends or patterns in the data (Durrheim & Painter, 2014; Roni et al., 2020).

A quantitative methodological approach may resonate with the objectives of the current study. Existing numerical cross-sectional data (based on completed questionnaires generated through the FIRE project) was utilised. The data, statistical procedures employed during data analysis and the quantitative methodological approach enabled the current study to achieve the purpose and objectives. The investigation was directed at comparing the phenomenon of



intrapersonal resilience-enabling pathways within a particular context (e.g., Global South challenged educational context).

Generally, data collection through quantitative methods (e.g., surveys and questionnaires) is comparatively quick and provides precise numerical data useful for studying large numbers of people. It also provides information in breadth from a significant quantity of units to allow objective, quantifiable explanations and predictions (Ary et al., 2019). Additional advantages include the possibility of examining patterns, indicating the quantification significance of a phenomenon. Quantitative methods also serve as a control for alternative theories (Creswell & Creswell, 2018). Since data must be electronically available to facilitate statistical tests when conducting quantitative research, routine data analysis may be relatively less time consuming when using statistical software (e.g., IBM Statistical Package for the Social Sciences [SPSS] version 27, IBM Corp. [2020]) and the results are reasonably independent of the researcher protecting against researcher bias (Creswell & Creswell, 2018). However, employing a rigorous and systematic quantitative data management process can be challenging and time-intensive (O'Leary, 2017). Nonetheless, quantitative research may have higher credibility with important stakeholders and allow the replication of findings with a higher impact in terms of publication and translation (Creswell & Creswell, 2018; Roni et al., 2020). On the other hand, quantitative research measures relationships between known limited variables (e.g., self-efficacy and teacher efficacy beliefs) through a deductive process (Creswell & Creswell, 2018; Leedy & Ormrod, 2015; Neuman, 2014). Therefore, a quantitative process may not be able to reveal any new factors (Shean, 2015) that can be pertinent in, for example, the development of resilience as a process. Furthermore, the self-reported FIRE Teacher Resilience Measure used in the current study could not observe how respondents demonstrated constructs (e.g., self-efficacy or teacher efficacy beliefs) within their given environment (e.g., classroom or school setting). Likewise, biased responses due to, for example, a social desirability effect, inaccurate self-beliefs or perceived coercion are frequent concerns in self-report data (Chesnut, 2017; L. Cohen et al., 2018; Maree & Pietersen, 2019b, 2019c). Even though respondents' lack of observed demonstrated constructs is a probable limitation, a reasonable conjecture may be possible once potential associations between variables have been established (Creswell & Creswell, 2018). Such outcomes may still provide insight into noteworthy factors demonstrating intrapersonal resilience-enabling pathways in a challenged context. Bird (2020) also argued, based on his view on objective investigation and descriptions, that objectivity is a civic virtue and not only determining empirical realities.

Quantitative research may produce overtly abstract knowledge hampering specific application since a quantitative lens do not consider respondents' rich social and historical contextual construction or personal meaning-making and realities (Creswell, 2014). Consequently, the structured nature of a quantitative view may prevent a researcher from further examining unexpected outcomes and sacrifice personal significance (Babbie, 2021;



Creswell & Creswell, 2018). Thus, the researcher may miss out on phenomena occurring due to emphasis on hypothesis testing and lack the depth of qualitative research (Ary et al., 2019). Nevertheless, it can be the starting point for additional qualitative research or give rise to future inquiry (Panhwar et al., 2017). In addition, the distanced approach of the researcher may result in barriers to accommodate behaviour expressions not anticipated in the research process (Van Rensburg et al., 2010), and important research skills (e.g., the ability to interpret and find meaning) may not develop (Terre Blanche et al., 2014).

Since secondary data analysis (see Section 3.4.4.2) was employed in the current study existing clean, electronically stored, quantitative data was utilised, saving time (Donnellan et al., 2011; Eaton & Krueger, 2011; Johnston, 2017; Pienta et al., 2011; Trzesniewski et al., 2011). However, it was not possible to further consider respondent experiences during the data collection phase due to extant data utilised in the current study. Nonetheless, the quantitative paradigm enabled me to establish to what extent the respondents (i.e., final year pre-service teachers at the University of Pretoria) displayed self-efficacy and teacher efficacy beliefs but did generate universal laws applicable across a population. However, the reflexive exploration throughout the research process could contribute to the feasible generalisation of findings to all final year pre-service teachers at the University of Pretoria. As mentioned in Section 3.3.2, I employed a holistic approach while taking the dynamic and abstract nature of findings into account within the context-oriented research through a post-positivist research paradigm. I also included an inclusive explanation regarding the procedures that were used to produce findings to evaluate the credibility and possible replication of the current study (O'Leary, 2017). Therefore, a quantitative approach was followed in the current study using extant numerical teacher resilience data based on completed questionnaires. The current study intended to address the research questions and hypotheses (as depicted in Section 3.2) by analysing the nature of intrapersonal resilience-enabling pathways to teacher resilience through statistical procedures performed on data to align with the aim and process of a quantitative methodological research approach (Creswell & Creswell, 2018). The following section describes the research process and methodological strategies employed in the current study.

#### 3.4 RESEARCH PROCESS AND METHODOLOGICAL STRATEGIES

#### 3.4.1 INTRODUCTION

This section discusses the research design, research methodology, and related strategies I relied on, as displayed in Table 1.2 in Chapter 1. The choice of research design and data analysis acts as a framework and determines the research output (López et al., 2015; O'Leary, 2017), demonstrating the importance of reporting on the methodological decision-making process detail.



#### 3.4.2 COMPARATIVE CASE STUDY AS RESEARCH DESIGN

A research design provides direction for specific research inquiry, decisions and analysis (Creswell, 2014; Creswell & Creswell, 2018). The selection of a research design is informed by elements including the research problem, research assumptions, paradigmatic approach, a study's audience, data collection methods and interpretation (Creswell, 2014; Creswell & Creswell, 2018). Therefore, L. Cohen et al. (2018, p. 173) convey that a "research design is governed by fitness for purpose".

I opted for a comparative case study design (Babbie, 2021; Mills et al., 2006; G. Thomas, 2011; Yin, 2018; Zartman, 2012), which is frequently applied in the social science field (G. Thomas, 2011, to analyse extant cross-sectional data by selecting particular cases for semblance. A comparative case study design allowed for a deeper understanding (L. Cohen et al., 2018; Creswell & Creswell, 2018) of the phenomenon under study to determine what was collective and distinct in cohorts of cases (Case 1, Case 2 and Case 3) spanning over 3 years (2015–2017). A case may be classified as a single observable unit that characterises a system examined within a context (L. Cohen et al., 2018; Fraenkel et al., 2012). A case is measured through scientific observation (single or multiple forms) to present a comprehensive portrayal of an empirical inquiry (L. Cohen et al., 2018; Fraenkel et al., 2012; Yin, 2018). Figure 3.2 and Figure 3.3 depict the cohorts of multiple cases compared in the current study.

#### Figure 3.2

Within-case Comparison of Self-efficacy and Teacher Efficacy Beliefs of Pre-service Teachers





Figure 3.2 describes the within-case comparison of pre-service teachers' cohorts for 2015 (Case 1), 2016 (Case 2) and 2017 (Case 3). Case 1 included the second-largest cohort of pre-service teachers ranging from 21-years to 30-years of age, with 79.9% female teachers and the FET teaching phase (n = 110) being the largest group. Case 2, the smallest cohort of pre-service teachers, ranged from 20-years to 29-years of age with 82.8% females and the FET programme phase the largest group. Case 3, the largest cohort of pre-service teachers and again with the FET programme phase as the largest group. On the other hand, Figure 3.3 portrays the cross-case comparison combined cohort for pre-service teachers between 2015–2017 (Case 4). Case 4 included a combined total of 1,193 pre-service teachers ranging between 20-years and 32-years of age with 77.9% female students and n = 436 FET programme phase teachers.

#### Figure 3.3

Cross-case comparison (2015-2017) Case 4 (Cross-case comparison) . Cohort of final year pre-service teachers 2015-2017 . - Total: N=1193 - Age: 20-32 - Gender: Male: n=223 (18.7%) Female: n=929 (77.9%) - Programme phase: FP/ECD: n=311 IP: n=157 SP: n=101 FET: n=436 Not specified/other: n=136

Combined Cross-case Comparison of Self-efficacy and Teacher Efficacy Beliefs of Pre-service Teachers

Since a comparative case study entails identifying unique and similar occurrences of a phenomenon, it is employed to determine what characteristics are singular or collective to those occurrences (J. L. Taylor, 2013; Yin, 2018). Consequently, using a case study, and testing hypotheses (L. Cohen et al., 2018), allowed me to compare (Babbie, 2021; L. Cohen et al., 2018; Creswell & Creswell, 2018), and identify differences and similarities across intrapersonal resilience-enabling pathways for pre-service teachers in a challenged context to interpret variations or relationships (J. L. Taylor, 2013).


As demonstrated in Figure 3.2 and Figure 3.3, a comparison was first drawn within each case (i.e. per year), between the self-efficacy and teacher efficacy beliefs of preservice teachers within a challenged context, and then across cases (i.e. 2015–2017) to identify what is universal and what is particular in these cases (Yin, 2018). Fittingly, the comparison of the cases involved is an appraisal of factors in one case (i.e. within-case analysis of Case 1, Case 2 and Case 3) (see Figure 3.2) while concurrently comparing the within-case analysis with a parallel case (i.e. cross-case analysis of Case 4) (see Figure 3.3) (G. Thomas, 2011).

The strength of a comparative case study design is reflected in the logic of comparison approximating inferences, which allows a research study to attempt to define clearer hypotheses (Babbie, 2021; L. Cohen et al., 2018). Therefore, the potential in the current comparative study was to quantitatively compare the self-efficacy and teacher efficacy beliefs of pre-service teachers to gain insight into teacher resilience in a challenged context. J. L. Taylor (2013) postulates that comparative case study research is dynamic and positioned to investigate a complex real-life phenomenon. The FIRE project was conducted in the setting in which the phenomenon usually occurs (i.e., HEI), bound by time (2015-2017), place (Pretoria, South Africa) and selected individuals (final year BEd pre-service teachers participating in the FIRE project), which set boundaries for inclusion in the current study (Yin, 2018). The parameters are important to ensure that the current study remained in a reasonable scope to avoid addressing questions that are too general or have excessive objectives (Yin, 2018). Therefore, extensive information on context-specific cases can be generated (L. Cohen et al., 2018). Furthermore, opting for a comparative case study design allowed for the comparison of multiple cases (as depicted in Figure 3.2 and Figure 3.3) that may ensue a more convincing, in-depth, holistic and accurate comparison (J. L. Taylor, 2013; Yin, 2018). Comparing the findings of the cases allows for the identification of similarities and differences, giving rise to enriched reliable data (Creswell & Creswell, 2018). Therefore, a comparative case study was valuable to gain insights into pre-service teachers' self-efficacy and teacher efficacy beliefs that may promote knowledge on teacher resilience.

I, nonetheless, recognise that a comparative case study as a research design has impending limitations. The challenge of selecting an appropriate sample is the associated constraints given differences in variables (e.g., age, language and culture, causing difficulty to isolate the importance of a single variable (Babbie, 2021; Maree & Pietersen, 2019a) on self-efficacy or teacher efficacy beliefs. However, the current design is not aimed at establishing impact, and I acknowledge that self-efficacy and teacher efficacy beliefs can be associated with numerous factors such as social, cultural, personal or contextual support in a challenged setting (Castro et al., 2010; Gu & Day, 2013; Kuo, 2013). Another possible challenge of a case study design is the production of context-specific knowledge (Flyvbjerg, 2011).



The caveat that such context-dependent information signifies is that findings may not apply to larger or other populations (Babbie, 2021). This limitation implies a potential hindrance to construct theory and knowledge bases on teacher resilience. Even so, L. Cohen et al. (2018) argued that multiple case studies may interpose greater generalisability. Following the choice of comparative case study as research design, the next section elaborates on sampling within the current study given the use of secondary data.

# 3.4.3 SAMPLING

# 3.4.3.1 Introduction

Sampling entails strategies for selecting manageable units of observation related to the research method employed (Babbie, 2021; Creswell & Creswell, 2018; O'Leary, 2017). A valid sample enables generalisation so that the findings inform knowledge about a specific population in general (Maree & Pietersen, 2019a). The current study used previously collected data (McMillan & Schumacher, 2010) generated in the FIRE project (see Appendix C). Besides the FIRE Teacher Resilience Measure, the FIRE dataset also includes nominal demographic information (i.e., age, gender, teaching phase and language). Respondents in the initial FIRE project was purposely sampled with elements of convenience (University of Pretoria, 2015) based on the research design of the FIRE project. For the current study, I purposively selected (Babbie, 2021; L. Cohen et al., 2018) teacher resilience data from the FIRE dataset and analysed items related to self-efficacy and teacher efficacy beliefs. The following section elaborates on the sampling of the FIRE project and the selection of variables from the FIRE Teacher Resilience Measure for the current study.

# 3.4.3.2 Sampling from FIRE project data

As explained in Chapter 1, Section 1.2, I analysed FIRE project data, specifically the FIRE Teacher Resilience Measure, to gain insight into the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context. I selected this extant data because the target population (i.e., final year pre-service teachers in a Global South space) contained specific information required for the research purpose (Babbie, 2021; Gravetter & Forzano, 2018) insight into teacher resilience in a challenged context.

An advantage of purposively selecting existing FIRE data is the established ethical professional data collection (University of Pretoria, 2015), reliability and validity of the data (Peixoto et al., 2018, 2020; Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012). Furthermore, purposively selecting data may provide a greater depth of knowledge since cases are included based on the characteristic being sought (L. Cohen et al., 2018). The FIRE sample constitutes a total sample (i.e., a subset of the population consisting of a predetermined



number) (Maree & Pietersen, 2019a) of 1,193 final year pre-service teachers, between the age of 20-years to 32-years, who completed the FIRE Teacher Resilience Measure from 2015 to 2017 during the FIRE project. According to the FIRE project documents,<sup>47</sup> the respondents in the FIRE project (i.e., fourth-year BEd pre-service teachers at the University of Pretoria) were purposively sampled with elements of convenience (University of Pretoria, 2015) within the research design of the given project (see Appendix C for extract information on the FIRE project). Non-probability purposive sampling is employed when a specific research purpose or rationale (e.g., FIRE project aims as discussed in Chapter 1) directs the sampling selection (L. Cohen et al., 2018; Maree & Pietersen, 2019a). In contrast, non-probability convenience or opportunity sampling occurs when respondents are selected because they are conveniently available (e.g., final year pre-service teachers enrolled at the University of Pretoria), allowing easy access (L. Cohen et al., 2018; Creswell & Creswell, 2018; Roni et al., 2020).

Convenience sampling can be fast and inexpensive; there are also disadvantages. Convenience sampling but may limit the representativeness of the sample (Maree & Pietersen, 2019a). Final year pre-service teachers (2015–2017) participated in the FIRE research project as part of selected modules (see Appendix C and Appendix D). López et al. (2015) argued that using a representative sample is the only way to generalise results for a population (i.e., preservice teachers in a challenged context).

As the initial sampling strategy (i.e., for the FIRE research project<sup>48</sup>) was nonprobabilistic and purposive with elements of convenience, generalisability is not possible to a larger population of pre-service teachers (L. Cohen et al., 2018; Maree & Pietersen, 2019a). Nevertheless, Gravetter and Forzano (2018) argued that it is reasonable to presume that samples from one location can represent samples from similar settings, which leads to the assumption that extrapolation of findings may be possible in similar settings (e.g., generalisation to final year pre-service teachers at the University of Pretoria). Still, there are risks in making inferences about patterns and effects in general populations from extracted samples skewed by one or more forms of selection bias, as with purposive non-probabilistic sampling (Babbie, 2021; L. Cohen et al., 2018; Creswell & Creswell, 2018; Maree & Pietersen, 2019a). Although the sample size (N = 1,193), for the current study, was sufficient for quantitative research purposes, based on the central limit theorem (Field, 2018) and statistical power analysis (see Section 4.3.4), caution should be taken when generalising the results to pre-service teachers in South Africa or the Global South context.

<sup>&</sup>lt;sup>47</sup> FIRE project ethical approval: University of Pretoria (UP) 14 03 01.

<sup>&</sup>lt;sup>48</sup> See Appendix C for details regarding the FIRE Project sampling.



# 3.4.3.3 Selection of FIRE Teacher Resilience Measure variables for analysis

For the current study, I purposively included two variables, self-efficacy and teacher efficacy. I excluded additional FIRE Teacher Resilience Measure variables (i.e., teacher professional, teacher emotions, teacher motivation and teacher social capacities included in 2015–2017). I wanted to foreground variables denoting the confidence of pre-service teachers to both recover from setbacks, as well as in their teaching and behaviour management given a challenged context. The FIRE Teacher Resilience Measure scales used in the current study (i.e., Teacher Resilience scale and Teacher Efficacy scale (see Appendix B) originally developed by Morgan [2011]) are grounded in the self-efficacy theory of Bandura (1997) and has been validated in other countries (e.g., Australia, Germany, Ireland, Malta and Portugal) within the teaching landscape (Peixoto et al., 2020). Previous studies found that confidence in recovery from setbacks (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018, 2020; Wosnitza et al., 2018) and confidence in teaching and behaviour management (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018, 2020; Wosnitza et al., 2018) were significant or strong contributors to teacher resilience in all indicated countries in comparison with social, emotional, professional and motivational capacities that differed in significance (Peixoto et al., 2018). Thus, to quantify factors influencing teachers' ability to resile in a Global South context is needed given the emergent teacher resilience field, the lack of robust resilience measures and limited research on resilience in pre-service teachers (Beltman et al., 2011; Mansfield et al., 2016; Morgan, 2011). Insight about pre-service teachers' intrapersonal resources may be important to attract, retain and support quality pre-service teachers to resile despite challenging circumstances. In this regard, literature highlight the importance of resilience traits (i.e., self-efficacy and teacher efficacy) for teacher resilience (Beltman et al., 2018; Castro et al., 2010; Ee & Chang, 2010; Gu & Day, 2013; Klassen et al., 2011; Morgan, 2011; Peixoto et al., 2018, 2020; Pendergast et al., 2011; Thieman et al., 2014; Wosnitza et al., 2018; Yada et al., 2021) also in a Global South context (Coetzee et al., 2017; Ebersöhn, 2014, 2017; Ebersöhn et al., 2020; Mansfield et al., 2018; Raath & Hay, 2016).

Table 3.2 provides an overview of the selected variables from the FIRE Teacher Resilience Measure for the current study. Instead of using all constructs<sup>49</sup> available on the FIRE Teacher Resilience Measure (Appendix B), only a selection of variables (i.e., self-efficacy and teacher efficacy beliefs) related to the purpose of the current study were selected for analysis given the set parameters.

<sup>&</sup>lt;sup>49</sup> Teacher professionalism (TR-Prof), teacher emotion (TR-Emot), teacher motivation (TR-Mot), teacher social capacity (TR-Soc), resilience (Resilience) and teacher efficacy (TeachEff) as well as contextual resilience variables.



#### Table 3.2

Key Outcome Measures Assessing Levels of Self-Efficacy and Teacher Efficacy Beliefs of Pre-Service Teachers

within the Current Study

Variable name	Question index on the measure <sup>50</sup>	No. of items	Response type	Example item (Absolutely not confident to strongly confident)	Origin of scale	Developed by
Self- efficacy <sup>51</sup>	Questions 114–122	9	Likert scale (1–7)	Getting over setbacks in school; bouncing back when things upset met	Teacher Resilience scale	Morgan (2011)
Teacher efficacy	Questions 123–134	12	Likert scale (1–7)	Teaching all the subjects on the curriculum effectively	Teacher Efficacy scale	Morgan (2011); Peixoto et al. (2018)

The Teacher Resilience scale (Morgan, 2011) (Questions 114–122) appraises, as operationalised in the current study, self-efficacy factors (ordinal data), including "getting over setbacks in school", "bouncing back when upset", "carrying on with school work when things go wrong", "carrying on in school when upset", "feeling certain that things will come right even if there are serious problems in school", "managing negative events in school when I try", "coping with most problems on any school day", "some negative things that have happened in school have made me better able to deal with problems" and "not getting disheartened even when children's circumstances make it difficult" (see FIRE Teacher Resilience Measure, Appendix B).

The Teacher Efficacy scale (Morgan, 2011; Peixoto et al., 2018) (Questions 123–134) measures general teacher efficacy factors (ordinal data), including "teaching all the subjects on the curriculum effectively", "explaining difficult material in ways that the children will understand", "suggesting suitable examples when the children are having difficulty understanding", "teaching in a way that my students will remember important information", "applying the new developments in the curriculum into my teaching", "helping children focus on learning tasks and avoid distractions", "managing inappropriate behaviour", "encouraging students to take responsibility for their behaviour", "dealing with the diverse learning needs of the students in my class", "teaching students positive behaviour", "providing students with clear specific behaviour expectations" and "communicating effectively with parents" (see FIRE Teacher Resilience scale, Appendix B).

<sup>&</sup>lt;sup>50</sup> FIRE Teacher Resilience Measure (Appendix B).

<sup>&</sup>lt;sup>51</sup> Within the current study the variable *resilience* on the FIRE Teacher Resilience Measure is operationalised as self-efficacy as discussed in Chapter 1.



The constructs under discussion (i.e., self-efficacy and teacher efficacy beliefs) were measured using a seven-point Likert scale, producing ordinal data ranging from one (*absolutely not confident*) to seven (*strongly confident*). The Likert scale is a convenient way to measure a construct (Maree & Pietersen, 2019b). Furthermore, the ordinal data (Likert scale data) for both constructs (self-efficacy [9 items] and teacher efficacy [12 items]) were averaged using SPSS to create continuous scale variables for comparison. By including items (relevant scales) from all FIRE respondents (i.e., final year pre-service teachers), the research questions for the current study may be addressed. Hence, insight into the self-efficacy and teacher efficacy beliefs of final year pre-service teachers in a challenged context can inform knowledge on teacher resilience. The following section describes the data analysis and interpretation for this study.

# 3.4.4 DATA ANALYSIS AND INTERPRETATION

# 3.4.4.1 Introduction

The data analysis process entails the meaning-making of data by identifying associations, trends, relations or patterns (Babbie, 2021; Creswell, 2014). The purpose of analysing data is to refine the scientific understanding of the world and develop theories by testing empirical hypotheses (Donnellan et al., 2011). This section, therefore, elaborates on secondary data analysis and interpretation, the use of nonparametric statistics when data is not normally distributed and the descriptive and inferential statistics employed in the current study.

# 3.4.4.2 Secondary data analysis and interpretation

As I analysed previously collected data (extant data) (Babbie, 2021; Johnston, 2017; Logan, 2020; MacInnes, 2016) from a preceding primary research project (i.e., FIRE project), I was guided by secondary data analysis. As opposed to primary research data, the focus of the current study was on analysing rather than collecting data. Thus, I used secondary data analysis to examine numerical data derived from the FIRE Teacher Resilience Measure completed by final year pre-service teachers (N = 1,193) in one HEI. Secondary data analysis entails the use of an existing dataset collected for another primary purpose to address new or current research questions differently from the original investigation or to analyse raw materials from previous research (Johnston, 2017; Pienta et al., 2011; Widaman et al., 2011). Secondary data analysis creates the opportunity to pursue a specific social research interest (such as teacher resilience).

Secondary analysis can provide a new view on pre-existing data, analyse data not fully evaluated yet or compare extant data with newly collected data to build on previous research (Babbie, 2021; Logan, 2020; MacInnes, 2016; Neuman, 2014). As such, secondary data



analysis is an important element of scholarly research disciplines (Pienta et al., 2011). Furthermore, like studies utilising primary data, secondary data analysis is an empirical process applying rigorous research principles with steps and procedures to avoid any deception (Johnston, 2017). Awareness regarding the distinct characteristics of secondary analysis is needed so that foreseeable errors are avoided when selecting an extant dataset and performing the analysis (Babbie, 2021; Logan, 2020; MacInnes, 2016; Pienta et al., 2011). Archived data is mostly quantitative (e.g., questionnaire data), linking with the methodological paradigm of the current study. Still, increasingly, qualitative data (e.g., interview transcripts) and other non-quantitative data (e.g., video) are being stored for secondary use (i.e., available for use by people not involved in the initial data acquisition) (Donnellan et al., 2011; Pienta et al., 2011).

A researcher working with secondary data should have a thorough understanding of the original study (i.e., FIRE project), including methodology, design, sampling, measurement and theory to ensure in-depth knowledge of extant data (Babbie, 2021; Neiss et al., 2011). Moreover, secondary data analysis requires a researcher to be explicit and clear about the theory and assumptions underlining the data (Babbie, 2021; Logan, 2020; MacInnes, 2016) to ensure their purpose is compatible with the primary data (L. Cohen et al., 2018). It is essential to link original data to the goals and questions of a study (De Vos et al., 2011). Donnellan et al. (2011) further advised that researchers undertaking secondary data analysis should have a clear sense of the advantages and limitations of the approach. I, therefore, immersed myself in the available documents describing the FIRE project and publications regarding the ENTREE<sup>52</sup> project. Additionally, contact with the initial FIRE project researchers (my supervisors and other members of the FIRE project) and consultation with a Principal Investigator (i.e., Prof Caroline Mansfield) of the ENTREE project provided me with an opportunity to gain in-depth insight into the original study and measure development (Neiss et al., 2011). As such, I could address questions about the relevance of the data, data collection methods and measures utilised, parties that collected the data, and purpose of the original study (L. Cohen et al., 2018; Struwig & Stead, 2013).

Employing secondary data (i.e., an existing or pre-existing dataset that was previously collected and captured) is economical in terms of time, budget, energy, access, permission, resources and risk; making it an increasingly prevalent methodological choice (L. Cohen et al., 2018; Johnston, 2017; O'Leary, 2017; Pienta et al., 2011). Collecting a large, authentic and credible sample can be a considerable undertaking, especially for an individual or early career scholar starting in a field of research and lacking resources to undertake a primary study (L. Cohen et al., 2018; Neiss et al., 2011). Thus, having quick or even immediate access to an available quality sample is beneficial (Neiss et al., 2011). The research design can be useful

<sup>&</sup>lt;sup>52</sup> Project from which the FIRE Resilience Measure originated from.



to a researcher, the scholarly scientific field and the public by economising time and finances as unique questions or re-interpretations can extend the pool of knowledge to increase the optimum use of available raw data that is often expensive to collect (L. Cohen et al., 2018; Donnellan et al., 2011; Pienta et al., 2011; Trzesniewski et al., 2011). Time and money are thus saved since unnecessary duplication of research effort are avoided by taking advantage of existing resources (L. Cohen et al., 2018; Eaton & Krueger, 2011; Johnston, 2017). A benefit of secondary data analysis is the possibility of investigating new important research questions relating to extant data that had not been examined in previous research (Babbie, 2021), which holds true for the current study. New investigations enable exploration of the data from a different perspective with lower risks to respondents (Babbie, 2021; Logan, 2020; MacInnes, 2016) to attain a multidisciplinary or broader understanding of social concerns (De Vos et al., 2011). Consequently, secondary data analysis is associated with minimal ethical dilemmas as direct harmful or obtrusive effects to respondents are limited, and potential respondents' reactivity during data collection is avoided (L. Cohen et al., 2018). Nonetheless, the respondents in any study have the right to respect, anonymity, privacy, non-traceability, welfare, dignity and protection (L. Cohen et al., 2018; Creswell, 2014). I was cognisant of these issues during all research activities and adhered to the ethical considerations and guidelines for conducting, working and reporting on secondary data analysis as stipulated by, for example, L. Cohen et al. (2018), Leedy and Ormrod (2015), and O'Leary (2017).

I ensured that the necessary measures to protect the data, including ethical and legal implications of the analysis, were established (APA, 2020b; Babbie, 2021; Logan, 2020; MacInnes, 2016) as discussed in Section 3.4.6. An additional benefit associated with secondary data analysis is that data collection difficulty (e.g., physical access to research respondents) are excluded (O'Leary, 2017; Trzesniewski et al., 2011), and a new study does not have to be designed to collect a primary set of data (Donnellan et al., 2011). Bias or the impact of the researcher during data collection is also limited since primary data collection is not conducted, placing a buffer between the researcher and respondent (O'Leary, 2017). Nonetheless, De Vos et al. (2011) argued that possible prejudice in all extant data should be evaluated. For this reason, areas of concern may relate to the integrity, reliability and validity of the original dataset due to a lack of control over data collection in the current study (Eaton & Krueger, 2011). To this end, I investigated the reliability (see Section 3.4.5.2), validity (see Section 3.4.5.3) and quality (see Section 3.4.5.4) of the data before commencing with analysis (De Vos et al., 2011; Widaman et al., 2011). Furthermore, errors or limitations in the data might have occurred as a result of the original data collection process, over which I had no control (Babbie, 2021). For example, some data may be ambiguous or missing from the dataset, which can potentially affect statistical inferences (L. Cohen et al., 2018). Accordingly, I elected suitable quantitative methods in response to possible challenges to reduce the bias of missing data (Cooper, 2018; Trzesniewski et al., 2011). These methods included implementing the



necessary process of data cleaning (Rubin & Babbie, 2014) or subsequently deletion of selected complete cases (Kang, 2013) before data analysis to minimise errors and condense the effect of missing data.

By using quantitative secondary analysis, I could describe the phenomenon under investigation through descriptive statistics and draw inferences through inferential statistics from extant data (Leedy & Ormrod, 2015) by using SPSS. This process was also relatively time effective and enabled a quality professional representation of results with tabular and graphical output (Field, 2018). In addition, I was able to conduct various nonparametric statistical tests on the available electronic data as the data could be sorted, ordered and ranked as required. The analysis was conducted under close supervision of the statistician of the Faculty of Education at the University of Pretoria who is also a co-supervisor of the current study. However, an important potential danger of secondary analysis is the fallacy of concreteness. This fallacy is a misleading impression of accuracy given by citing statistics with more emphasis than appropriate (L. Cohen et al., 2018; Neuman, 2014). I avoided the probability by interpreting findings cautiously and being aware of the historical context in which the data was obtained.

Possible challenges may exist when employing extant data and the analysis thereof. Throughout the current study, I remained aware of these potential pitfalls when conducting secondary data analysis. The process can be more complex and deceptively time-consuming than what researchers may anticipate (De Vos et al., 2011; Donnellan et al., 2011). Willms (2011) echoed that using secondary data can be daunting and frustrating for novice researchers but rewarding. As a researcher, I remained mindful of the possibility that the purpose or defined variables for which the data was initially collected (i.e., FIRE project), may vary from the aim, operationalisation and research question of the current study (Trzesniewski et al., 2011). As detailed, the control of data collection errors was not feasible, and the analysis may be limited by the purpose of the primary research project constraining analysis or altering proposed questions (Babbie, 2021; O'Leary, 2017). Additionally, limitations in the design, methodology and measurement tools implemented in a primary study (i.e., FIRE Project) may be inevitable and impose certain restrictions (Babbie, 2021; MacInnes, 2016). There can also be no assurance that the procedures of an initial study were undertaken as described because of limited accompanying information on a study or how data was affected by potential problems (e.g., low response rate or misunderstanding of questions) (L. Cohen et al., 2018; Eaton & Krueger, 2011). Secondary data emanating from an original source (e.g., education departments or institutions), may not be neutral, causing a power imbalance and response bias in the data (L. Cohen et al., 2018). Therefore, I carefully considered and critically reflected on the existing dataset's quality, rigour and potential to address the research questions for the current study (O'Leary, 2017).



The investigation of an existing dataset can require a substantial investment of time and energy (Donnellan et al., 2011). Therefore, in conjunction with my supervisors, I carefully considered the time, effort and analytical acumen required to use the existing FIRE dataset since it can easily be underestimated. Closely related, a researcher may not have a proper understanding of the original study or problem when using secondary data analysis (De Vos et al., 2011), which further highlights strategies taken to immerse myself in the initial FIRE project information. Willms (2011) claimed that a difficult task in working with secondary data is taking the raw data provided and building a dataset that can be used for analysis.

Within the current study, the dataset was overseen with the assistance of my supervisors, one of which was the statistician of the Faculty of Education, University of Pretoria. Thus, re-using data instead of collecting primary data may forfeit contextual information and omit reflective interpretation (Babbie, 2021; L. Cohen et al., 2018). Although this possible drawback of working with extant data instead of respondents exists, quantitative secondary data analysis generates statistically significant knowledge rather than qualitatively relevant evidence (Babbie, 2021). Nonetheless, Trzesniewski et al. (2011) denoted that concern may remain regarding the likelihood that the interpretation of results through secondary data analysis may be biased if the conceptual framework does not include all quantified constructs. In response, I concentrated on the systematic process of secondary data analysis as prescribed by scholars (L. Cohen et al., 2018; Leedy & Ormrod, 2015; O'Leary, 2017) while duly considering all variables that form part of my conceptual model for congruence fit and fitness for purpose (L. Cohen et al., 2018).

I linked the epistemology of the current study with constructs in the dataset and verified the compatibility of the dataset and conceptual framework with my supervisors before commencing with data analysis (L. Cohen et al., 2018). Neiss et al. (2011) further emphasised the careful consideration of all data-related decisions, highlighting the justification of conceptual choices given congruent standard practices and adequate record-keeping of decisions and justifications. Lastly, I was aware that conclusions might only be tentative since the data utilised were not originally collected with the specific research questions (as depicted in Section 3.2) in mind and cognisant of currency (e.g. the effect of the time elapsed since data collection) of the primary data (i.e. 2015–2017) (O'Leary, 2017; Roni et al., 2020). However, the use of extant data may allow the research to be completed and findings to be produced quicker, and therefore the development and contribution of new knowledge may occur timeously despite challenges (Johnston, 2017).

## 3.4.4.3 Nonparametric statistics

Statistics in social sciences represents applied mathematical tools and techniques used to describe, summarise, and interpret the nature of numbers representing quantitative data (L. Cohen et al., 2018; Neuman, 2014; Salkind & Shaw, 2020). The statistical analysis depends



on the research design and type of data utilised (López et al., 2015). Furthermore, the extent to which data, specifically continuous data, reflect the normal distribution and the features of the data determine the statistical procedures employed during the data analysis process (Leedy & Ormrod, 2015). According to López et al. (2015), it is vital to choose the statistical procedures and test that best fits the nature of the data. The data that I analysed were nominal (e.g., gender), ordinal (e.g., Likert scale items) and continuous data (the constructs created for self-efficacy and teacher efficacy by averaging over the relevant Likert scale items). For the continuous data, I checked for normality using the Kolmogorov-Smirnov and Shapiro-Wilk tests (Field, 2018) and since the constructs were not normally distributed (see Section 4.5.1). Nonparametric or distribution-free statistics were used as they require less stringent assumptions to be made about the underlying process distribution (Chakraborti & Graham, 2019). In addition, nonparametric tests can be conducted on data that have been selected by non-probability sampling procedures, such as in the case of the current study (Chakraborti & Graham, 2019; Roni et al., 2020). The choice to use nonparametric statistics is suitable for relatively simple data analysis to determine whether constructs differ significantly (Gibbons & Chakraborti, 2010).

Nonparametric tests do not require stringent assumptions and are convenient as the associated and appropriate tests are straightforward and easy to compute (Chakraborti & Graham, 2019; Roni et al., 2020). Nonetheless, as nonparametric tests do not utilise all the information in a dataset (it, typically, uses signs and ranks), it has been argued in the literature, over the past few decades, that parametric tests are more robust and more powerful than their nonparametric counterparts (Gibbons & Chakraborti, 2010). However, since then, studies (see for example, Chakraborti & Graham, 2019) have shown that most nonparametric tests are almost as powerful as parametric tests or even similar in power. In addition, Roni et al. (2020) recently advocated using nonparametric methods over parametric methods<sup>53</sup> since nonparametric data can be a staple of education research. Therefore, nonparametric statistical analysis was performed in the current study. The facilitation of the hypotheses testing associated with the current study was supported by the availability and suitability of sufficient statistical tests as discussed in Section 3.4.4.5. Using statistical analysis, I could utilise descriptive statistics to describe the secondary data and interpret the data (i.e., draw inferences) through inferential statistics (Field, 2018; Leedy & Ormrod, 2015; Salkind & Shaw, 2020). Leedy and Ormrod (2015, p. 218) claimed that "inferential statistics involve using one or more small samples and then estimating the characteristics of the population from which each sample has been drawn", thereby permitting the testing of hypotheses for a larger population. Still, before inferences or decisions about the data could be made, I utilised descriptive statistics to summarise the data as discussed in Section 3.4.4.4.

<sup>&</sup>lt;sup>53</sup> See Roni et al., 2020 (pp. 3–4) listing the reasons why nonparametric methods are advocated over parametric methods.



Since statistical procedures form part of quantitative research (Leedy & Ormrod, 2015), I utilised descriptive and inferential statistics to address the research questions and test the hypotheses of the current study. Inferential statistics included the nonparametric Wilcoxon signed-rank (WSR) test (Wilcoxon, 1945), the nonparametric Mann-Whitney (MW) test (Mann & Whitney, 1947), the nonparametric Kruskal-Wallis (KW) test (Kruskal & Wallis, 1952) and the nonparametric Spearman correlation (Spearman, 1904). The differences between these tests and the reason for choosing them are discussed in Section 3.4.4.5. The following section discusses the descriptive procedures employed in the current study.

# 3.4.4.4 Descriptive statistics

Descriptive questions, statistics or procedures aim to describe, explore and summarise a collection of data based on basic features (Ary et al., 2019; Creswell & Creswell, 2018; O'Leary, 2017; Salkind & Shaw, 2020). Therefore, I used descriptive statistics to illustrate, organise and simplify the general nature of the raw quantitative data in a meaningful and manageable way to improve my understanding and the scope of the data properties (Field, 2018; Gravetter & Forzano, 2018; Leedy & Ormrod, 2015; O'Leary, 2017; Pietersen & Maree, 2019c). Descriptive statistics can potentially enable the researchers' exploration of data in a graphical and numerical way to improve understanding of data properties (De Vos et al., 2011; Pietersen & Maree, 2019c). Hence, I used descriptive statistics (L. Cohen et al., 2018; Field, 2018; Pietersen & Maree, 2019c) to derive values from the FIRE Teacher Resilience Measure, to determine to what extent the respondents (i.e., final year pre-service teachers in a challenged context) demonstrated self-efficacy and teacher efficacy beliefs when they completed the questionnaires.

Encapsulating quantitative variables can be complex since numerous inherent properties can be described (Pietersen & Maree, 2019c). In the current study, measures of central tendency (e.g., mean and median) and measures of spread (e.g., standard deviation and variance) (Field, 2018) were determined using SPSS. In terms of the measures of central tendency, the mean, the measure of location most used, represents the arithmetic average score of the data, and the median indicates the centre or middle value of an entire distribution, splitting the distribution (ordered from the smallest to the largest value) into equal halves (Field, 2018; Pietersen & Maree, 2019c). Regarding the measures of spread, the variance quantifies the spread of values of data around the mean, thus the average dispersions (Field, 2018; Pietersen & Maree, 2019c), while standard deviation is the square root of the variance or average difference of each value from the mean (Field, 2018; Pietersen & Maree, 2019c). In addition to relying on these strategies, I used graphical methods to represent the data included in Chapter 4. The means of the measure were derived statically to determine the aggregate of self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context.



The average levels were accomplished by statically establishing the mean scores (arithmetic average) (Field, 2018). The variables were then plotted against each other to indicate the mean scores obtained. Furthermore, researchers should investigate the reason for a possible association between variables if a correlation is identified (Leedy & Ormrod, 2015). In general, a correlation coefficient of less than .3 signifies a weak correlation, .3 to .5 is moderate and greater than .5 is strong (Heale & Twycross, 2015). However, a correlation between variables does not necessarily imply causation (Leedy & Ormrod, 2015; Sauro & Lewis, 2016) but indicate the associated direction and strength between self-efficacy and teacher efficacy beliefs (Roni et al., 2020). Still, the results do provide evidence for the justification for inferential methods. As a result, I included appropriate inferential statistical tests, including the Spearman correlation, Wilcoxon signed-rank, Mann-Whitney and the Kruskal-Wallis tests to test the hypotheses and potential correlations and differences between constructs (i.e., self-efficacy and teacher efficacy beliefs) associated with the current study.

# 3.4.4.5 Inferential statistics

Inferential questions, statistics or procedures enable researchers to generalise or draw inferences beyond descriptive analysis to reach a conclusion that encompasses the immediate dataset (Field, 2018; Pietersen & Maree, 2019d). The purposes of inferential statistics include the testing of hypotheses, comparison, correlation and estimation of population parameters or characteristics to allow generalisation by relying on probability theory (Gravetter & Forzano, 2018; O'Leary, 2017). Gravetter and Forzano (2018) argued that it is reasonable to conclude, even with non-probability sampling, that samples from one location are as representative as samples from comparable settings, which leads to the assumption that generalisation of findings may be possible when conducting research in similar settings.

While conducting hypotheses testing, a .05 level of significance ( $\alpha$ ) was set, indicating that a result occurs, on average, only 5% (or .05 probability) of the time (Field, 2018). As such, the result can also be ascribed to another factor that has influenced the data (Corder & Foreman, 2014; Leedy & Ormrod, 2015). I established if the computed *p*-value fell within the critical region by determining whether the *p*-value was less than the level of significance during quantitative data analysis (Corder & Foreman, 2014). The *p*-value evaluates the likelihood of results being more than chance and is only valid if the sample represents the population (O'Leary, 2017; Sauro & Lewis, 2016). The null hypothesis ( $H_0$ ) is rejected when it is found that the result can be attributed to something other than approximate chance difference; if not, the null hypothesis ( $H_0$ ) is not rejected (Leedy & Ormrod, 2015; Sauro & Lewis, 2016).

It is, nevertheless, possible that a Type I error (rejecting the null hypothesis when it is true, or there is actually no difference) or a Type II error (to fail to reject or retain the null hypothesis when it is false) can occur (Pietersen & Maree, 2019d; Sauro & Lewis, 2016) during data analysis. Errors are seldom due to the negligence of the researcher but can instead be



credited to the significance level selected for a particular study. The test criterion convention of  $\alpha$  = .05 provides considerable control against a Type I error, and any result with *p* < .05 is, by assumption, statistically significant while all others are not (Sauro & Lewis, 2016). When the *p*-value falls below .05, there is sufficient evidence to conclude the difference is unlikely due to chance (O'Leary, 2017; Sauro & Lewis, 2016). To analyse the research data and test the formulated hypotheses, I used the Wilcoxon signed-rank test to compare two related groups (i.e., self-efficacy and teacher efficacy of the same pre-service teachers), the Mann-Whitney test to compare two independent groups (i.e., gender) and the Kruskal-Wallis test to compare three or more independent groups (i.e., programme phases enrolled for) to determine if there is a statistically significant difference between groups. The Wilcoxon signed-rank, Mann-Whitney, and Kruskal-Wallis tests overcome the distributional limitation by ranking the data, which eliminates the effect of outliners and is suitable for data not fitting a normal distribution (Field, 2018; Pietersen & Maree, 2019a).

The Wilcoxon signed-rank test is a nonparametric statistical hypothesis test performed to compare two related samples (Field, 2018). It is warranted as an alternative to the parametric *t*-test for matched pairs (also known as the paired Student *t*-test) when data distribution cannot assume normality (Pietersen & Maree, 2019a). The Wilcoxon signed-rank test is based on ranking differences between scores and considers the direction of change and the extent of variance between two datasets and, therefore, makes full use of the data (Pietersen & Maree, 2019a). The Wilcoxon signed-rank test is typically used when data is ordinal (e.g., Likert scale data) in nature, when nonparametric statistics are used and to compare the medians of two related scores or matched samples (i.e., there are two measurements from the same entity or individual) (Field, 2018; Leedy & Ormrod, 2015). Since the same pre-service teachers completed the Teacher Resilience scale and the Teacher Efficacy scale on the FIRE Teacher Resilience Measure, the Wilcoxon signed-rank test was used. When computing the Wilcoxon signed-rank test, the test statistic and corresponding pvalue are produced. If the *p*-value is less than .05, the null hypothesis ( $H_0$ ) is rejected and there is a statistically significant difference between the results (Field, 2018) of a pre-service teacher's self-efficacy and teacher efficacy beliefs. Alternatively, if the p-value is greater than .05, it is implied that the differences between the self-efficacy and the teacher efficacy beliefs of pre-service teachers are not statistically significant (Field, 2018). Section 4.5 reports on the results of the Wilcoxon signed-rank test performed on extant data.

The Mann-Whitney test is the nonparametric counterpart to the parametric *t*-test for independent groups (Corder & Former, 2014; Roni et al., 2020). The medians between two independent (mutually exclusive) groups are compared (Roni et al., 2020) in the value of an ordinal (including Likert scale data), or a continuous variable. The null hypothesis ( $H_0$ ) (i.e., distributions are equal) and alternative hypothesis ( $H_a$ ) (i.e., distributions are not equal) are thus stated in terms of the median as opposed to the mean (independent samples *t*-test) (Pietersen



& Maree, 2019a). As a result, the Mann-Whitney test, as a nonparametric measure, was utilised to identify possible statistically significant differences between the self-efficacy and teacher efficacy beliefs of males and females (L. Cohen et al., 2018) pre-service teachers within a challenged context.

The Kruskal-Wallis test seemed to be appropriate for comparing the programme phase variable (i.e., FP/ECD; IP; SP, FET and not specified/other) since the test is used to compare medians of three or more mutually exclusive independent groups to determine any statistically significant difference (Roni et al., 2020; Utts & Heckard, 2012). The Kruskal-Wallis test is suitable when the normal distribution as the underlying process distribution of the data cannot be assumed, and the performance depends on the ranks of the measurement observations on a rating scale (Pietersen & Maree, 2019a). The nonparametric Kruskal-Wallis test is equivalent to a parametric one-way analysis of variance (ANOVA) test (L. Cohen et al., 2018; Roni et al., 2020). The null hypothesis ( $H_0$ ) and alternative hypothesis ( $H_a$ ) are compared in terms of the population medians (Kruskal-Wallis) instead of the population means (ANOVA) (Gibbons & Chakraborti, 2010). A significant Kruskal-Wallis test (p-value < .05) may indicate that at least one median differs statistically from the rest. A pairwise Mann-Whitney test with Holm corrections (Aickin & Gensler, 1996) was performed on the extant data (see Chapter 4 for results) to analyse results obtained from a significant Kruskal-Wallis test. The Wilcoxon signed-rank test, the Mann-Whitney test and the Kruskal-Wallis test, as statistically significant tests, do not imply correlation or causation between variables but test for differences among groups (Roni et al., 2020).

The correlation indicates the degree to which the variables are related; consequently, it explores the monotonic linear relationship between variables (Field, 2018; Roni et al., 2020). The Spearman correlation coefficient, denoted by  $r_s$  (an appropriate measure of similarity between two ordinal rankings of a single dataset), was employed (Field, 2018; Pietersen & Maree, 2019a) to determine whether there was a statistically significant correlation between the self-efficacy and teacher efficacy beliefs of pre-service teachers. It should be noted that correlation does not imply causation (Leedy & Ormrod, 2015; Roni et al., 2020). Correlation takes on values between (and including) -1 and +1, i.e.,  $-1 \le r_s \le 1$ .

The closer the value is to -1, the stronger the negative (linear) correlation, and the closer the value is to +1, the stronger the positive (linear) correlation (Pietersen & Maree, 2019a). Self-efficacy and teacher efficacy beliefs were correlated using the Spearman correlation coefficient ( $r_s$ ) (Field, 2018; Roni et al., 2020), which allows for the investigation of an association between two ratios, intervals, or ordinal variables (Roni et al., 2020). The results are represented in Section 4.5.2, which reports the correlation between the average scores of self-efficacy and teacher efficacy beliefs of final year pre-service teachers within a challenged context. The following section illustrates the standard of rigour that guided the current study to ensure quality criteria throughout the research process.



# 3.4.5 STANDARDS OF RIGOUR

# 3.4.5.1 Introduction

Since the current study employed a post-positivist, quantitative epistemology, I aimed to obtain reliable, valid and objective evidence in terms of the phenomena investigated (L. Cohen et al., 2018; Howell, 2013; O'Leary, 2017). Reliability and validity are ways of demonstrating and communicating the quality and rigour of the research process (Heale & Twycross, 2015). Quantitative research should provide detailed information to assess the rigour of the design, robustness of results, any knowledge claims, conclusions drawn and to allow for replication of a study (Creswell, 2014; López et al., 2015). The following sections will explore the quality assurance of quantitative data and secondary data analysis to ensure that credible research takes place.

# 3.4.5.2 Quality assurance of quantitate data: Reliability

As an overarching term, reliability denotes the dependability and stability of measures, scales, or respondent groups, across time, to achieve believable and replicable results (L. Cohen et al., 2018; Maree & Pietersen, 2019c). The reliability of a measurement instrument (such as questionnaires) can be explained as the ability of the instrument to measure a construct in a consistent (Creswell & Creswell, 2018; Gravetter & Forzano, 2018) and repeatable manner (Creswell & Creswell, 2018; Maree & Pietersen, 2019c) with the required precision (Widaman et al., 2011). As such, I determined whether the instrument (i.e., FIRE Teacher Resilience Measure) had been used in a consistent and standardised fashion (Leedy & Ormrod, 2015) by consulting the documentation (University of Pretoria, 2015) on the project describing the data collection methods and discussing the procedures with my supervisors.

I paid particular attention to the manner of administration, possible language barriers that may have existed, instrument development descriptions, and the coefficient of reliability (Cronbach's alpha). The Cronbach's alpha coefficient, the most often reported internal consistency index (Widaman et al., 2011), is based on inter-item correlations (Creswell & Creswell, 2018; Pietersen & Maree, 2019b). Therefore, if items strongly correlate with each other, a high internal consistency will be present with a Cronbach's alpha value close to one (Foxcroft & Roodt, 2013; Pietersen & Maree, 2019b). Cronbach's alpha can be interpreted in terms of excellent reliability (Cronbach's alpha > = .90), high reliability (.80 < = Cronbach's alpha < .90), moderate reliability (.70 < = Cronbach's alpha < .80), or low reliability (Cronbach's alpha < .70) (Pietersen & Maree, 2019b).

In summary, a Cronbach's alpha value of .70 or greater has been accepted as a reasonable indication of the reliability of a scale (Field, 2018). However, a Cronbach's alpha value of .60 or greater is generally accepted by researchers in the social sciences (Ghazali,



2008). It follows that reliability can be determined by examining internal consistency (i.e., item responses are consistent across constructs), but also in terms of test-retest reliability (i.e., scores are stable over time when the instrument is administered multiple times) (Creswell, 2014). However, a limitation when employing Cronbach's alpha is the possibility of overestimating internal consistency due to the number of items (López et al., 2015; Widaman et al., 2011). Thus, adding items may increase Cronbach's alpha (López et al., 2015). A scale assessing a trait construct should have high internal consistency, reliability and demonstrate high test-retest reliability (Widaman et al., 2011).

The reliability of the FIRE Teacher Resilience Measure (including, for example, the Teacher Resilience scale [Morgan, 2011] and the Teacher Efficacy scale [Morgan, 2011; Peixoto et al., 2018]) has been confirmed by previous studies (e.g., Morgan, 2011; Peixoto et al., 2018, 2020; Wosnitza et al., 2018) (see Appendix A) with a Cronbach's alpha, which ranged from .67 to .91. Table 3.3 provides an overview of previous studies' reliability analysis for the scales utilised in the current study.

#### Table 3.3

Reliability of Scales

Variable	Origin of scale		Cronbach's alpha							
		Beltman et al. (2018)	Morgan (2011)	Peixoto et al. (2018)	Peixoto et al. (2020)	Wosnitza et al. (2018)				
Self- efficacy	Teacher Resilience scale (Morgan, 2011)	.93	.91	.89	.93	.87 .92 .86 .94 90				
Teacher efficacy	Teacher Efficacy scale (Teaching) (Morgan, 2011)	.94	.88	.82	N.A.	.86 .93 .85				
	Teacher Efficacy scale (Behaviour management) (Peixoto et al., 2018)		N.A.	.81	N.A.	.93 .88				

As previously stipulated, the current study foregrounded sections of the FIRE Teacher Resilience Measure on self-efficacy<sup>54</sup> and teacher efficacy which indicated high satisfactory reliability on the original scales (i.e., Cronbach's alpha = .91 and Cronbach's alpha = .88, respectively) developed by Morgan (2011). Further studies (Beltman et al., 2018; Peixoto et al., 2018, 2020; Wosnitza et al., 2018) in numerous countries (Australia, the Czech Republic, Germany, Ireland, Malta and Portugal) have demonstrated excellent reliability (Cronbach's alpha > = .90) to high reliability (.80 < = Cronbach's alpha < .90) for the scales as displayed in Table 3.3.

<sup>&</sup>lt;sup>54</sup> Within the current study the variable *resilience* on the Teacher Resilience scale is operationalised as self-efficacy as discussed in Chapter 1.



## 3.4.5.3 Quality assurance of quantitative data: Validity

A second measure of quality assurance in quantitative research entails validity (Creswell & Creswell, 2018; Heale & Twycross, 2015). Validity may imply that a sound instrument assesses what it claims or intends to measure (L. Cohen et al., 2018; Gravetter & Forzano, 2018), which is essential when a measurement is carried out. In the current study, the validity of the FIRE Teacher Resilience Measure was derived from its intent to assess what the instrument was supposed to measure, in other words, the self-efficacy and teacher efficacy beliefs of final year pre-service teachers. The validity of an instrument is especially important in cases where a hypothetical construct is measured using an operational definition (Gravetter & Forzano, 2018). Validity is determined by establishing whether data had been collected and reported on with care and meticulousness warranting interpretations (Struwig & Stead, 2013). In quantitative research, validity often strives for controllability, replicability, consistency, predictability, generalisability, randomness, objectivity and observability to avoid threats (L. Cohen et al., 2018).

Internal validity entails the confidence to which the design (e.g., comparative case study) and data of a study allow a researcher to deduct accurate inferences about relationships of the data or outcome variables of a study (L. Cohen et al., 2018; Sauro & Lewis, 2016). In such cases, an explicit, single description applies to the correlation between variables (Gravetter & Forzano, 2018). External validity refers to the confidence with which a study's findings can be extrapolated (i.e., generalisability) beyond the context of a study (Leedy & Ormrod, 2015; Sauro & Lewis, 2016). Gravetter and Forzano (2018) caution that any aspect limiting the generalisability of findings can threaten external validity. Closely related, Leedy and Ormrod (2015) suggest that research done in a real-life setting can provide generalisability and, therefore, increase the external validity of research findings. As the extant data I analysed was collected directly from the respondents (i.e., pre-service teachers) at the University of Pretoria (i.e., a higher institution of teacher training) during the original research project, I was able to analyse data collected in the specific setting under investigation. External validity (Leedy & Ormrod, 2015) may indicate the future usefulness of exploring pre-service teachers' selfefficacy and teacher efficacy beliefs in a challenged context. However, Pietersen and Maree (2019a) cautioned against threats to validity, including reliability (i.e., if an instrument is not reliable, it cannot be valid), social desirability (e.g., agreeing to all questions or answering in an expected socially accepted manner) and item bias (e.g., culture or gender bias).

Two types of construct validity were investigated within the current study, namely convergent validity and discriminant validity (Cooper, 2018; Sauro & Lewis, 2016). Convergent and discriminant validity are subcategories of construct validity. Therefore, if there is evidence of convergent and discriminant validity, construct validity is likely (Trochim, 2006). Convergent validity shows that items that belong to the same construct are, in fact, related (Sauro & Lewis,



2016). Convergent validity can be tested by calculating correlation coefficients on the items belonging to the same constructs and checking that these correlations are high, i.e., close to - 1 or close to +1. On the other hand, discriminant validity shows that items that do not belong to the same construct are, in fact, not related (Sauro & Lewis, 2016). Discriminant validity can be tested by calculating correlation coefficients since items that do not belong to the same constructs should have lower correlations than those belonging to the same constructs.

# 3.4.5.4 Established quality assurance of the measuring instrument

In the following section, I discuss the rigour and background of the FIRE Teacher Resilience Measure selected for the current study. One criterion for demonstrating robustness is to provide evidence to show that the instrument used to measure the variables is reliable and valid, i.e., it appraises, with a degree of accuracy, what it is supposed to measure (DeVellis, 2012). Secondary analysis is generally restricted to the primary study's (i.e., FIRE project) choice of measurement tools, and limitations may be inevitable. Nonetheless, insurance must be provided regarding the properties of the instrument (MacInnes, 2016). Therefore, the validity and reliability of the data collection instrument must be considered when conducting or critiquing research (Heale & Twycross, 2015). The reliability of a measurement tool implies that an instrument consistently determines a given construct (Foxcroft & Roodt, 2013; Gravetter & Forzano, 2018). As reliability is a prerequisite for validity (Gravetter & Forzano, 2018), the accuracy of an instrument can be improved by enhancing its reliability (Leedy & Ormrod, 2015). The degree of what a researcher can learn about a phenomenon that is investigated, the extent to which meaningful conclusions can be drawn, and the probability of obtaining statistical significance are all dependent on the reliability and validity of the instruments used in a study (Leedy & Ormrod, 2015).

The FIRE Teacher Resilience Measure (completed by pre-service teachers during the FIRE project in 2015–2017) included items from the FIT-Choice Scale (Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012), ENTREE (Peixoto et al., 2018) and contextual resilience questions (Coetzee, 2013). Initially developed by Watt and Richardson (2007) and empirically validated for the Australian context with pre-service teachers (Watt & Richardson, 2007, 2008), the FIT-Choice scale has been utilised internationally in English-speaking countries such as Ireland, Norway, the United Kingdom and the United States with translations into Croatian, Dutch, Estonian, French. German, Mandarin, and Turkish (Watt & Richardson, 2012; Watt et al., 2012). Furthermore, Salifu et al. (2018) verified the structural validity of the measurement in an African context (i.e., Ghana). The scale is founded on the expectancy-value motivational theory, Social Cognitive Career Theory (SCCT) and teacher education literature providing an integrative and compressive framework exploring teaching as career choice (Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012).



The questionnaire is a sound psychometrical measure, displaying reliability and validity across cultural research and a range of contexts. (Watt & Richardson, 2012). In addition, ENTREE utilised measures including the Teacher Resilience scale (Morgan, 2011), Personal Life (Morgan, 2011), Rumination scale (Morgan, 2011), Teacher Commitment (Morgan, 2011), Teacher Efficacy scale (Morgan, 2011; Peixoto et al., 2018), Social Context (Cefai & Cavioni, 2014), School Support (Morgan, 2011), Conditions Provided by Institutions for Job Satisfaction (Silva, 2013) and Administrative and Policy Demands (Peixoto et al., 2018). Confirmatory factor analyses (CFA), from previous studies, confirmed the factor structures (CFA ranging from .935 to .999) while the reliability was confirmed by the Cronbach's alpha, which ranged from .67 to .91 (Peixoto et al., 2018).

# 3.4.5.5 Quality assurance of analysis and results: Secondary data analysis as empirical systematic method

Neuman (2014) elucidates that, even if a reliable source collected the data (i.e., FIRE project), secondary data could not be regarded as problem-free. Rubin and Babbie (2014) advise that awareness of possible limitations in reliability and validity when doing secondary data analysis is a protective factor for a researcher. The current study involved secondary data analysis; therefore, I could not regulate the reliability and validity of the measurement instruments used during data collection. Therefore, I was not able to remedy threats to internal validity (Creswell, 2014). However, I remained aware of the threats and subsequently implemented necessary measures and strategies to increase the reliability and external validity of the results. One such measure was to utilise the expertise of the Faculty of Education's statistician who is also a co-supervisor of the current study, to, for example, conduct reliability testing. However, since data can be defective, claims may be made congruently that are invalid.

Statistical conclusion validity may be threatened if inaccurate inferences are made from the data due to inadequate statistical assumptions (Creswell, 2014). To avoid this potential limitation, I took meticulous care during the data analysis and verified statistical procedures against existing literature. I correspondingly conducted data cleaning (as explained in Section 3.4.4.2) (Rubin & Babbie, 2014) to remove potential errors in the data before commencing with the data analysis. Regular contact with my supervisors further increased the possibility of an accurate data analysis and reporting of the results.

# 3.4.6 ETHICAL CONSIDERATIONS

# 3.4.6.1 Introduction

The awareness and importance of ethical concerns during the entire research process are crucial to making an argument for a study (L. Cohen et al., 2018; Gravetter & Forzano, 2018). Every step during research prompts extensive and complex ethical considerations (Creswell,



2014; Creswell & Creswell, 2018). These considerations hold when employing secondary analysis of extant data since ethics are situated to a specific situation, and secondary analysis is not absolved from ethical issues (L. Cohen et al., 2018). Therefore, ethical strategies considered for this study included permission to utilise an existing dataset, ethical concerns with mining extant data and reporting of secondary data analysis.

# 3.4.6.2 Access and permission to use extant data

The current study used extant data already accessible in the scientific domain. Permission was granted by the FIRE project's Principal Investigator (PI) (i.e., the late Professor William J. Fraser, University of Pretoria, SA) and co-researcher Professor Liesel Ebersöhn (CSR, University of Pretoria, SA) to use extant data generated during the FIRE research project from the FIRE Teacher Resilience Measure incorporated during the FIRE project. The current study could, therefore, employ data generated from the completed self-reported questionnaires of pre-service teachers (cohorts 2015–2017) during the FIRE project<sup>55</sup> (University of Pretoria, 2015). The data ownership remains with the original project and the CSR, and any restrictions regarding data use, analysis and dissemination would still apply (L. Cohen et al., 2018).

# 3.4.6.3 Privacy and extant data

As part of the broader FIRE research project, I adhered to the required ethical considerations and guidelines while conducting secondary data analysis and reporting the current study's findings. Even though, as discussed in Section 3.4.4.2, minimal ethical quandaries are associated with secondary data analysis (L. Cohen et al., 2018), the respondents in the FIRE project had the right to protection, welfare, dignity, respect, anonymity and privacy (Creswell & Creswell, 2018; Gravetter & Forzano, 2018).

During the analysis and interpretation, I ensured anonymity, data de-identification and privacy of the FIRE project respondents and did not represent data or findings in any way that may have resulted in the identification of a respondent (APA, 2017, 2020b, The Presidency, RSA, 2013). While using SPSS, no names of respondents were captured to ensure confidentiality and anonymity (Leedy & Ormrod, 2015) and adhere to the responsibility secondary researchers have to the original respondents (L. Cohen et al., 2018). I also followed prescriptions regarding the safe storage of data. I respected the guideline of data only being available to the research team to protect data privacy (The Presidency, RSA, 2013).

<sup>&</sup>lt;sup>55</sup> FIRE project ethical approval: University of Pretoria (UP) 14 03 01.



# 3.4.6.4 Ethical reporting on extant data

Ethical issues similarly apply to reporting and dissemination of the final research findings. All researchers must analyse (L. Cohen et al., 2018) and report their results ethically by including multiple perspectives and report contrary findings (Creswell & Creswell, 2018). Research has an ethical responsibility to demonstrate quality (L. Cohen et al., 2018). A rigorous, honest and precise presentation of research results, findings and challenges need to be reported without misrepresentation (Babbie, 2021; L. Cohen et al., 2018). I clarified the primary research's position (FIRE project) regarding the right to publish from secondary data and ensure that authorship on any publication that results from the secondary analysis is credited to report on the secondary analysis results (APA, 2017). The FIRE project and the original investigators will be cited in any publication or presentation resulting from the current study (Babbie, 2021).

In reporting on the results based on extant data, I did not use language or words that are biased against individuals based on age, gender, culture, race, sexual orientation or disability, and I applied appropriate language for the audience of the research (Creswell, 2014; Creswell & Creswell, 2018). I further endeavoured to remain honest with no misrepresentation, suppression, invention or falsification of the results or findings in any way (Gravetter & Forzano, 2018; Leedy & Ormrod, 2015). I strived to make the research auditable by following rigorous, methodical and systematic procedures while stipulating transparent, detailed explanations of the research study and reporting on technical failures and limitations (O'Leary, 2017; Rubin & Babbie, 2014). I also credited to all the sources or research ownership I consulted and acknowledged the use of others' words and ideas where applicable (Creswell & Creswell, 2018; Gravetter & Forzano, 2018). Throughout the entire study and during all the steps of data analysis, and when reporting on the results, I focused on acting ethically and responsibly and being mindful of my epistemology and underlying assumptions (Babbie, 2021; L. Cohen et al., 2018). In addition, I endeavoured to equip myself for the processes involved in quantitative statistical data analysis to report accurately on the results and deal with challenges that may have arisen (L. Cohen et al., 2018).

I was, furthermore, aware of the time elapsed since the collection of the data (i.e., 2015 to 2017). The questionnaires completed during the FIRE project were independently captured into Microsoft Excel. The Faculty of Education's statistician, who is also one of the current study's co-supervisors, oversaw the process. By utilising the completed questionnaires as an existing dataset in this study, I conformed to the ethical and legal guidelines concerning confidentiality and anonymity (Gravetter & Forzano, 2018) as outlined in the FIRE project (University of Pretoria, 2015) and diligently protected any sensitive information of respondents (APA, 2020b; Babbie, 2021).

I obtained ethical clearance from the Ethics Committee of the Faculty of Education at the University of Pretoria and started analysing data after obtaining permission (APA, 2020b).



Furthermore, as part of the FIRE project and the ethical guidelines of the Faculty, I adhered to the safeguarding prescriptions of the raw and interpreted data. Lastly, as per ethical guidelines, the original questionnaires (2015–2017) will be stored securely for 15 years at the CSR.

# 3.5 CONCLUSION

In this chapter, I focussed my discussion on a detailed description of the research process. I discussed the selected metatheoretical and methodological paradigm in terms of their suitability to the current study. A quantitative stance allowed me to describe and make inferences from the extant dataset from a post-positivist perspective by conducting statistical analysis. Using a comparative case study research design supported the purpose of this study as it enabled me to address the research questions and test the associated hypotheses. I reflected on the advantages and disadvantages of the chosen methodology and reported on my attempts to address the limitations. I furthermore deliberated on aspects of rigour and ethical responsibilities in this chapter.

Chapter 3, thus, aimed to provide a methodological context for the analysis of the results in the following chapter. Hence, in Chapter 4, the results of the current study are presented and discussed. For this purpose, I explain and report the results I obtained from the descriptive and inferential statistical analysis conducted.

---000----



# Chapter 4 **Research Results of the Study**

#### 4.1 INTRODUCTION

In Chapter 3, I reported on the methodological process and strategies of the current research study. I described the paradigmatic perspectives (i.e., quantitative research, situated within post-positivism) in detail. I also explained the suitability of a comparative case study as the research design with secondary data analysis. I justified the methodological decisions based on the research purpose and questions of this study, as formulated in Chapter 1. Chapter 3 also discussed the secondary data analysis of extant data and the choice of statistical techniques employed. Furthermore, the strategies to ensure rigorous and ethical research when conducting data analysis were deliberated.

In this chapter, I present the quantitative results of the current study following the quantitative analysis of the measure (i.e., FIRE Teacher Resilience Measure). I commence with demographic distribution results about the demographic information available. In addition, the reliability analysis, validity analysis and statistical power analysis for the current study are discussed. Lastly, I summarise the results obtained from the descriptive and inferential statistics conducted graphically. Figure 4.1 presents a flow chart showing the organisational overview of Chapter 4.

## Figure 4.1

ow Chart Showing the Organisation of Chapter 4						
	DEN	IOGRAPHIC BACKGR	OUND			
Distribution: Completed FIRE Teacher Resilience Measure	Demographic distribution: Age	Demographic distribution: Gender	Der dis La	mographic stribution: anguage luency <sup>56</sup>	Demographic distribution: Enrolled pre- service teaching programmes	
	S	TANDARDS OR RIGC	UR			
Reliability analysis:Validity analysis:Statistical power analysisCronbach's alphaConstruct validityStatistical power analysis						

<sup>&</sup>lt;sup>56</sup> As discussed in Section 1.6.9, general language fluency is the ability, proficiency, ease, confidence or accuracy with which communicative language (oral, reading and writing) is used.



# DESCRIPTIVE STATISTICS OF THE SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS IN A CHALLENGED CONTEXT

		Descriptive statistics	Descriptive results of
Descriptive statistics	Descriptive statistics	of the self-efficacy and	the self-efficacy and
of the self-efficacy of	of the teacher efficacy	teacher efficacy	teacher efficacy beliefs
pre-service in a	beliefs of pre-service	beliefs of pre-service	of pre-service teachers
challenged context	in a challenged	teachers in a	based on gender and
(2015 to 2017)	context (2015 to 2017)	challenged context	enrolled teaching
		(2015 to 2017)	programmes

# INFERENTIAL STATISTICS OF THE SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS IN A CHALLENGED CONTEXT

	Results of the	Results of the	Results of the	Results of the
Correlation	FIRE Teacher	FIRE Teacher	FIRE Teacher	FIRE Teacher
analysis between	Resilience	Resilience	Resilience	Resilience
the self-efficacy	Measure Case 1	Measure Case 2	Measure Case 3	Measure 2015 to
and teacher	(2015): Self-	(2016): Self-	(2017): Self-	2017: Self-
efficacy beliefs of	efficacy and	efficacy and	efficacy and	efficacy and
pre-service	teacher efficacy	teacher efficacy	teacher efficacy	teacher efficacy
teachers in a	beliefs of pre-	beliefs of pre-	beliefs of pre-	beliefs of pre-
challenged	service teachers	service teachers	service teachers	service teachers
context	(within-case	(within-case	(within-case	(cross-case
	comparison)	comparison)	comparison)	comparison)

CONCLUSION

# 4.2 DEMOGRAPHIC BACKGROUND

## 4.2.1 INTRODUCTION

Based on extant data available (i.e., *N* = 1,193; 77.9% female), I report on pre-service teachers' self-efficacy and teacher efficacy beliefs to highlight intrapersonal resilience-enabling pathways to teacher resilience in the Global South. As presented in Chapter 1, the purpose of the current study is to inform teacher resilience knowledge by statistically comparing (within-case and cross-case) teacher resilience data based on the self-efficacy and teacher efficacy beliefs of final year pre-service teachers in a challenged context. Section 4.2 provides an overview of the pre-service teachers' demographic background, relating to *age, gender, language fluency* and *enrolled pre-service teaching programmes.*, based on extant data generated from the completed FIRE Teacher Resilience Measure during the FIRE project.



# 4.2.2 DISTRIBUTION: COMPLETED FIRE TEACHER RESILIENCE MEASURE (QUESTIONNAIRES)

Table 4.1 presents an outline of the completed number (N = 1,193) and percentage (%) of questionnaires (FIRE Teacher Resilience Measure) from 2015 to 2017.

## Table 4.1

Total Completed FIRE Resilience Measure per Year (2015 to 2017)

Year	Ν	Percent (%)
Case 1 (2015)	313	26.2
Case 2 (2016)	169	14.2
Case 3 (2017)	711	59.6
Cross-case (2015–2017)	1,193	100.0

As depicted in Table 4.1, n = 313 (26.2% of the total sample [N = 1,193]) pre-service teachers completed the FIRE Teacher Resilience measure in 2015. In 2017, the largest number of questionnaires were collected, n = 711 (59.6% of the total sample). During 2016, which was also the year of the #Fees-Must-Fall2016 movement at universities in South Africa, as discussed in Chapter 1, final year pre-service teachers completed the smallest number of questionnaires, n = 169 (14.2% of the total sample).

The 2016 results should be cautiously interpreted since the sample is relatively smaller (n = 169) than 2015 (n = 313) and 2017 (n = 711). However, as discussed in Chapter 3, based on the central limit theorem (Field, 2018), the 2016 sample (n = 169) is regarded as sufficiently large so that statistical inferences can be generalised to the population (i.e., final year preservice teachers at the University of Pretoria). The following section presents an overview of the age distribution of pre-service teachers for the current study.

# 4.2.3 DEMOGRAPHIC DISTRIBUTION: AGE

Figure 4.2 gives a percentage (%) summary of the *age demographic* distribution of pre-service teachers that completed the FIRE Teacher Resilience Measure from 2015 to 2017.



### Figure 4.2





■2015 ■2016 ■2017 ■2015-2017

Figure 4.2 provides the percentage (%) overview of age demographic per year (in-case, 2015, 2016 and 2017) and cross-case (2015–2017) as indicated on the FIRE Teacher Resilience Measure. A total of N = 1,151 pre-service teachers indicated their age during the completion of the FIRE Teacher Resilience Measure from 2015 to 2017. As expected, given a 4-year BEd degree in South Africa, the largest age group was indicated as the 22-year-old frequency (n = 461; 38.6%). The 23-year-old age group was the second largest (n = 277; 23.2%) and the 21-year-old age group the third-largest (n = 180; 15.0%). As illustrated in Figure 4.2, the age distribution for within-case and cross-case comparison presents relatively similar patterns within (2015, 2016, 2017) and across the years (2015 to 2017). Table 4.2 presents the mean and standard deviation (*SD*) for the age distribution of pre-service teachers form the FIRE Teacher Resilience Measure from 2015 to 2017.



#### Table 4.2

Mean and Standard Deviation Age Distribution (Within-case and Cross-case)

Year	Mean (years)	SD
Within-case 2015	22.82	1.22
Within-case 2016	22.54	1.35
Within-case 2017	22.60	1.49
Cross-case 2015–2017	22.65	1.41

Table 4.2 shows that the overall mean age of pre-service teachers who completed the FIRE Teacher Resilience Measure from 2015 to 2017 was 22.65-years. In similar studies employing the teacher resilience (Morgan, 2011) and teacher efficacy (Morgan, 2011; Peixoto et al., 2018) scales, the mean age of pre-service teachers was higher (32.1-years in Australia [Beltman et al., 2018]; 25.2-years in Germany, Ireland, Malta and Portugal [Peixoto et al., 2018]).

Internationally, the average age of the first graduation from university is 24.7-years (OECD, 2014b), with approximately 80% of undergraduate students enrolled for a 4-year degree, being between the age of 18-years and 24-years (The Hamilton Project, 2007) with pre-service teachers tending to be 22-years to 25-years. Inferential statistics were not performed because the age biographical variable didn't vary much as respondents (i.e., final year pre-service teachers) of the same age group were sampled. Based on the international average age (24.7-years) (OECD, 2014b) distribution of graduating fourth-year students, also similar in the current study (22.7-years). The following section expounds on the gender distribution for the current study.

## 4.2.4 DEMOGRAPHIC DISTRIBUTION: GENDER

Table 4.3 indicates the *gender (male and female) demographic* distribution overview of final year pre-service teachers that completed the FIRE Teacher Resilience Measure for within-case (2015, 2016 and 2017) and cross-case (2015-2017) examples.

#### Table 4.3

	Withi	in-case	Withi	Within-case Within-case		Cross-case				
	2	015	2016		2017		2017		2015	-2017
Gender	Ν	%	Ν	%	Ν	%	Ν	%		
Male	61	19.5	21	12.4	141	19.8	223	18.7		
Female	250	79.9	140	82.8	539	75.8	929	77.9		
Total	311	99.4	161	95.3	680	95.6	1,152	96.6		
Missing	2	0.6	8	4.7	31	4.4	41	3.4		
TOTAL	313	100.0	169	100.0	711	100.0	1,193	100.0		

Gender Demographic Distribution of Completed FIRE Teacher Resilience Measure



As demonstrated in Table 4.3, the highest rate of completed female FIRE Teacher Resilience Measures was in 2016 (82.8%), while the highest rate for males was in 2017 (19.8%). Interesting to note is that the male completion rate for 2016 was the lowest (12.4%) compared to 2015 (19.5%) and 2017 (19.8%) and the cross-case comparison of 18.7%. The #Fees-Must-Fall2016 movement may have affected the response rate for 2016 since activities at HEIs were disrupted. Figure 4.3 displays the overall gender demographic distribution of pre-service that completed the FIRE Teacher Resilience Measure within a graph.

# Figure 4.3



Gender Demographic Percentage Distribution (2015 to 2017)

As illustrated in Figure 4.3, the female gender distribution (77.9%) dominates the male distribution (18.7%) for pre-service teachers in a challenged context. The distribution is higher than the international gender distribution for male and female teacher demographics (68% female vs 32% male) (OECD, 2019c), but may highlight the existing worldwide pattern that more females are still enrolling for teaching than their male counterparts (Jenkins, 2019; OECD, 2009, 2017b, 2019a, 2019c; Van Diemen, 2019; World Bank, 2020b). In similar studies employing the teacher resilience (Morgan, 2011) and teacher efficacy (Morgan, 2011; Peixoto et al., 2018) scales, female gender distribution also exceeded the male distribution (79.5%, Australia [Beltman et al., 2018]; 69.8%, Germany; 79.7%, Ireland; 80.5%, Malta; 96.5%, Portugal [Peixoto et al., 2018]).



Average female gender dominance, especially in the early education and primary phases (Jenkins, 2019; OECD, 2017b; World Bank, 2020b), is the international (Lemon & Garvis, 2016; O'Neill & Stephenson, 2012a) and national standard (OECD, 2017b; Petersen, 2014). According to the OECD (2019a), the international average for female teachers was approximately 68%, while 60% of teachers in South Africa were female. The following section considers the language distribution of pre-service teachers for the current study.

# 4.2.5 DEMOGRAPHIC DISTRIBUTION: LANGUAGE FLUENCY

Table 4.4 and Figure 4.4 indicate the *language fluency demographic* distribution of pre-service teachers that completed the FIRE Teacher Resilience Measure from 2015 to 2017. Pre-service teachers indicated their language fluency (i.e., "List of languages you are fluent in") as requested on the FIRE Teacher Resilience Measure (see Appendix B) and could, therefore, indicate more than one language of proficiency.

## Table 4.4

Language Fluency Demographic of Pre-Service Teachers

Year	Afrikaans	English	isiNdebele	isiXhosa	isiZulu	Sepedi	Sesotho	Setswana	siSwati	Tshivenda	Xitsonga	Other
Within-case 2015	200	297	18	8	75	34	20	29	17	3	5	7
Within-case 2016	98	153	5	1	21	14	4	8	2	0	2	8
Within-case 2017	327	653	32	30	197	113	58	80	54	7	18	19
Cross-case 2015–2017	625	1,103	55	39	293	161	82	117	73	10	25	34



#### Figure 4.4

Percentage (%) Language Fluency Demographic Distribution of Pre-Service Teachers



■2015 ■2016 ■2017 ■2015-2017

As depicted in Table 4.4 and Figure 4.4, English (N = 1,103; 92.5%) was specified as the main language category of proficiency by pre-service teachers for 2015 (n = 297; 94.9%), 2016 (n =153; 90.5%) and 2017 (n = 653; 91.8%), with Afrikaans (N = 625; 52.4%) and isiZulu (N = 293; 24.6%) in the second and third succession for each year (2015, 2016 and 2017). The language demographics underlines the possible emphasis placed on English as an academic language for teaching and learning in South Africa, especially after Grade 3 (McKinney & Guzula, 2019; Spaull et al., 2020). Spaull et al. (2020) denoted that 70% of FP/ECD (Grade 1 to Grade 3) learners in South Africa receive schooling in an African home language with English taught in addition. From the IP phase (Grade 4), 90% of learners are taught in English with an African language as their home language subject. Despite the national prominence (McKinney & Guzula, 2019; RSA, 1996a; Spaull et al., 2020) for home language (especially in early education, i.e., ECD), it may be interesting to note the high rate of English (cross-case comparison of 92.5%) indicated as language proficiency by final year pre-service teachers. The shortage of quality teachers within a Global South context may be exacerbated by the fact that teachers are confounded by language barriers (DHET, RSA, 2018a).



Regardless of policies, the dominance of Western languages lingers (McKinney & Guzula, 2019). Figure 4.4 demonstrates that Afrikaans is the only language declining each year (2015 = 63.9%; 2016 = 58.0%; 2017 = 46.0%), while indigenous languages, except for isiNdebele, show an increase when comparing 2015 with 2017. The decrease in Afrikaans is consistent with the comparative language results of the 2011 SA census and the 2016 community survey indicating a notable decrease in Afrikaans home language speakers in South Africa (Stats SA, 2012a, 2016). A community household survey (Stats SA, 2018) also indicated a decline for spoken languages outside the household with the exception of isiZulu and Setswana. Table 4.5 demonstrates the percentages of the number of languages preservice teachers are fluent in.

#### Table 4.5

	Within	-case	Within	-case	ase Within-case		Cross-	case
_	20	15	20	16	2017		2015–	2017
Number of languages spoken	Ν	%	Ν	%	Ν	%	Ν	%
1	48	15.3	27	16.0	126	17.7	201	16.8
2	186	59.4	107	63.3	353	49.6	646	54.1
3	41	13.1	17	10.1	102	14.3	160	13.4
4	24	7.7	5	3.0	62	8.7	91	7.6
5	9	2.9	1	0.6	20	2.8	30	2.5
6	3	1.0			10	1.4	13	1.1
7	2	0.6			4	0.6	6	0.5
8					2	0.3	2	0.2
Total			157	92.9	679	95.5	1,149	96.3
Missing			12	7.1	32	4.5	44	3.7
TOTAL	313	100	169	100	711	100	1,193	100

Number of Languages Pre-service Teachers are Fluent in

Most of the South African population can speak at least two of the official languages (RSA, 2019; Stats SA, 2018), which correlates with the given results in Table 4.5, indicating that 54.1% of pre-service teachers noted fluency in at least two languages. However, the language measure is inadequate to make further significant inferences about the possible role of multilingualism for pre-service teachers in a challenged context.

As with the age demographic, the current study did not include inferential statistics for the language distribution for further exploration. Inferential statistics were not performed for the biographical variable language because there are too many categories (since South Africa has 11 official languages), some with sparse data. Having many categories with sparse data is not ideal for comparative analysis. In addition, pre-service teachers could select more than one



language on the FIRE Teacher Resilience Measure (i.e., questionnaire) since the language demographic question asked respondents to indicate all their languages of fluency; thus, it was a multiple response question.

Multiple response questions may further complicate a comparative analysis, as it created more than 11 categories because the categories were the 11 official languages and, for example, a category for those that speak English and Afrikaans, a category for those that speak English and isiZulu, etc.; there are numerous combinations for possible categories. Given the South African context with 11 official languages and the language-based descriptive results, the current study did not perform inferential statistical procedures on the language distribution of pre-service teachers. Chapter 5 also addresses the limitation with recommendations for future questionnaire development and/or FIRE Teacher Resilience Measure amendments within a South African context given the multilingual society. The following section elaborates on the enrolled teaching programme demographic distribution of final year pre-service teachers for the current study.

# 4.2.6 DEMOGRAPHIC DISTRIBUTION: ENROLLED PRE-SERVICE TEACHING PROGRAMMES

Table 4.6 and Figure 4.4 indicate the *enrolled pre-service teaching programme demographic* (FP/ECD, IP, SP, FET) and distribution (frequency [*N*] and percentage [%]) of pre-service teachers that completed the FIRE Teacher Resilience Measure from 2015 to 2017.

#### Table 4.6

	Withir	n-case	Withi	n-case	Withir	n-case	Cross	s-case
	2015		2016		2017		2015–2017	
Enrolled pre-service teaching programme	Ν	%	Ν	%	Ν	%	Ν	%
Foundation phase (FP)/Early childhood development (ECD)	105	33.5	45	26.6	161	22.6	311	26.1
Intermediate phase (IP)	32	10.2	17	10.1	108	15.2	157	13.2
Senior phase (SP)	30	9.6	14	8.3	57	8.0	101	8.5
Further education and training phase (FET)	110	35.1	60	35.5	266	37.4	436	36.5
Enrolled pre-service teaching programmes not specified	25	8.0	20	11.8	91	12.8	136	11.4
Total	302	96.5	156	92.3	683	96.1	1,141	95.6
Missing	11	3.5	13	7.7	28	3.9	52	4.4
TOTAL	313	100.0	169	100.0	711	100.0	1,193	100.0

Enrolled Pre-service Teaching Programme (2015 to 2017)

As illustrated in Table 4.6, the pre-service teaching programme distribution within-case and cross-case present relatively similar stable within (2015, 2016, 2017) and across years (2015 to 2017) for FP/ECD, IP, SP, FET and the category of teaching programme not specified.



Figure 4.5 presents the percentage teaching programme for within-case and cross-case distribution graphically for pre-service teachers in their final year of study.

#### Figure 4.5

Percentage (%) Enrolled Pre-service Teaching Programme Distribution for Final Year Pre-service Teachers (2015–2017)



Pre-service teaching programme

■2015 ■2016 ■2017 ■2015-2017

As illustrated in Figure 4.5, the two most prevalent programmes for which pre-service teachers enrolled were the FP/ECD and FET teaching programmes. Interestingly, the FP/ECD and FET phases remained popular across the years, with the FET phase dominating the within-case (2015, 2016 and 2017) and cross-case (2015–2017) comparison. The pre-service teaching programme distribution may indicate the emphasis placed nationally and internationally on early childhood education and the importance of the final schooling years (i.e., throughput rate).

The programme with the lowest enrolment rate is the SP, with only 8.5% of pre-service teachers completing the FIRE Teacher Resilience Measure in the senior phase. The following section describes the reliability analysis, validity analysis and statistical power analysis performed on the data to verify the consistency and reliability of the extant data utilised.



# 4.3 STANDARDS OF RIGOUR

# 4.3.1 INTRODUCTION

Section 4.3 discusses the reliability (i.e., Cronbach's alpha) and the validity (i.e., construct validity) of the FIRE Teacher Resilience Measure. As part of the rigour of credible research conducted, the statistical power analysis of the current study is also described.

# 4.3.2 RELIABILITY ANALYSIS: CRONBACH'S ALPHA

As discussed in Chapter 3, the reliability of the FIRE Teacher Resilience Measure has been confirmed by previous studies (Beltman et al., 2018; Peixoto et al., 2018, 2020; Wosnitza et al., 2018). The original Teacher Resilience scale (Morgan, 2011) had a Cronbach's alpha of .91, and the Teacher Efficacy scale (Morgan, 2011; Peixoto et al., 2018) had a Cronbach's alpha of .88, indicating very satisfactory reliability. Further studies (Beltman et al., 2018; Peixoto et al., 2018, 2020; Wosnitza et al., 2018) showed excellent reliability to high reliability for the Teacher Resilience (Morgan, 2011) and Teacher Efficacy (Mansfield & Wosnitza, 2015; Morgan, 2011; Peixoto et al., 2018) scales. Reliability analysis for the scales from the current study is depicted in Table 4.7.

## Table 4.7

Reliability Analysis

FIRE Teacher Resilience Measure Scales	Cronbach's alpha
Teacher Resilience scale (Morgan, 2011)	.87
Teacher Efficacy scale (Morgan, 2011; Peixoto et al., 2018)	.91

Table 4.7 indicates that the Teacher Resilience scale (Cronbach's alpha = .87) and the Teacher Efficacy Scale (Cronbach's alpha = .91) had satisfactory reliability for the current study. This reliability analysis is in agreement with findings from previous studies (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018, 2020; Wosnitza et al., 2018). Therefore, based on the current findings displayed in Table 4.7, it may be concluded that the scales (i.e. Teacher Resilience scale and Teacher Efficacy scale) were capable of measuring pre-service teachers' confidence in recovery from setbacks (i.e., self-efficacy) and confidence in teaching and behaviour management (i.e., teacher efficacy) in a reliable and valid manner. Since reliability was established for the current study, the following section describes the validity analysis conducted.



# 4.3.3 VALIDITY ANALYSIS: CONSTRUCT VALIDITY

As discussed in Chapter 3, validity implies that a sound instrument assesses what it claims or intends to measure (L. Cohen et al., 2018; Foxcroft & Roodt, 2013; Gravetter & Forzano, 2018; Maree & Pietersen, 2019c; Pietersen & Maree, 2019b). Therefore, in the current study, the validity of the measure (i.e., FIRE Teacher Resilience Measure) was derived from the ability to assess the self-efficacy and teacher efficacy beliefs of final year pre-service teachers. Convergent validity shows that items that belong to the same construct are related (Sauro & Lewis, 2016). This assumption can be tested by calculating correlation coefficients on the items belonging to the same constructs and checking that these correlations are high, i.e., close to - 1 or close to +1. All the correlations were statistically significant (i.e., all *p*-values < .001) for the self-efficacy construct, with the weakest correlation being .316 and the strongest being .661, as depicted in Table 4.8.


Convergent Validity - Self Efficacy Construct

	Bouncing back, when things upset me	Carrying on with my school work when things go wrong	Carrying on in school when events upset me	Feeling certain that things will come right even if there are serious problems in school	Managing negative events in school when I try	Coping with most problems on any school day	Some negative things that have happened in school have made me better able to deal with problems	Not getting disheartened even when children's circumstances make it difficult
Getting over setbacks in school	.577	.462	.398	.348	.430	.421	.316	.356
Bouncing back, when things upset me		.493	.416	.383	.435	.416	.344	.359
Carrying on with my school work when things go wrong			.661	.396	.431	.410	.365	.368
Carrying on in school when events upset me				.444	.492	.460	.371	.371
Feeling certain that things will come right even if there are serious problems in school					.528	.428	.321	.329
Managing negative events in school when I try						.592	.441	.417
Coping with most problems on any school day							.469	.415
Some negative things that have happened in school have made me better able to deal with problems								.468



For the teacher efficacy construct, all the correlations were statistically significant (all *p*-values < .001), with the weakest correlation being .269 and the strongest being .628, as portrayed in Table 4.9. These coefficients provide evidence for convergent validity.

#### Table 4.9

Convergent Validity - Teacher Efficacy Construct

	Explaining difficult material in ways that the children will understand it	Suggesting suitable examples when the children are having difficulty understanding	Teaching in a way that my students will remember important information	Applying the new developments in the curriculum into my teaching	Helping children focus on learning tasks and avoid distractions	Managing inappropriate behaviour	Encouraging students to take responsibility for their behaviour	Dealing with the diverse learning needs of the students in my class	Teaching students positive behaviour	Providing students with clear specific behaviour expectations	Communicating effectively with parents
Teaching all the subjects on the curriculum effectively	.527	.410	.406	.450	.406	.327	.362	.384	.356	.396	.308
Explaining difficult material in ways that the children will understand it		.628	.589	.458	.445	.353	.375	.409	.395	.397	.307
Suggesting suitable examples when the children are having difficulty understanding			.622	.480	.421	.310	.391	.431	.432	.416	.269
Teaching in a way that my students will remember important information				.531	.477	.363	.431	.433	.457	.441	.326
Applying the new developments in the curriculum into my teaching					.531	.358	.417	.447	.439	.436	.386



	Explaining difficult material in ways that the children will understand it	Suggesting suitable examples when the children are having difficulty understanding	Teaching in a way that my students will remember important information	Applying the new developments in the curriculum into my teaching	Helping children focus on learning tasks and avoid distractions	Managing inappropriate behaviour	Encouraging students to take responsibility for their behaviour	Dealing with the diverse learning needs of the students in my class	Teaching students positive behaviour	Providing students with clear specific behaviour expectations	Communicating effectively with parents
Helping children focus on learning tasks and avoid distractions						.462	.489	.456	.460	.454	.395
Managing inappropriate behaviour							.540	.409	.442	.454	.362
Encouraging students to take responsibility for their behaviour								.505	.540	.508	.367
Dealing with the diverse learning needs of the students in my class									.561	.525	.416
Teaching students positive behaviour										.612	.409
Providing students with clear specific behaviour expectations											.430

In addition, discriminant validity shows that items that do not belong to the same construct are not statistically related. This discrimination can be tested by calculating correlation coefficients since items that do not belong to the same constructs should have lower correlations than those belonging to the same constructs. Table 4.10 demonstrates the discriminant validity analysis between self-efficacy and teacher efficacy constructs.



Discriminant Validity - Self-Efficacy Vs Teacher Efficacy

	Teaching all the subjects on the curriculum effectively	Explaining difficult material in ways that the children will understand it	Suggesting suitable examples when the children are having difficulty understanding	Teaching in a way that my students will remember important information	Applying the new developments in the curriculum into my teaching	Helping children focus on learning tasks and avoid distractions	Managing inappropriate behaviour	Encouraging students to take responsibility for their behaviour	Dealing with the diverse learning needs of the students in my class	Teaching students positive behaviour	Providing students with clear specific behaviour expectations	Communicating effectively with parents
Getting over setbacks in school	.288	.305	.309	.275	.343	.292	.276	.295	.341	.314	.318	.228
Bouncing back, when things upset me	.215	.310	.270	.238	.262	.273	.250	.299	.296	.288	.268	.171
Carrying on with my school work when things go wrong	.321	.308	.299	.281	.308	.314	.261	.321	.323	.338	.310	.236
Carrying on in school when events upset me	.297	.288	.265	.262	.235	.301	.258	.283	.275	.283	.276	.223
Feeling certain that things will come right even if there are serious problems in school	.248	.289	.258	.262	.266	.307	.242	.295	.286	.294	.295	.227
Managing negative events in school when I try	.334	.355	.316	.286	.329	.374	.339	.330	.320	.340	.340	.296





When correlating the self-efficacy construct against the teacher efficacy construct, all the correlations were statistically significant (all *p*-values < .001). The weakest correlation was .171, and the strongest was .374, as shown in Table 4.10. This finding indicates that the correlations of items that do not belong to the same constructs are lower than those belonging to the same constructs. Therefore, these coefficients provide evidence for discriminant validity. Thus, from the evidence depicted in Table 4.8, Table 4.9 and Table 4.10, the likeliness of construct validity is reasonably established. The following section explores the statistical power analysis for data analysis.

### 4.3.4 STATISTICAL POWER ANALYSIS

Statistical power, or the power of a hypothesis test, is the probability of correctly rejecting false null hypotheses or the ability of a selected statistical test to detect a true effect, significant association or difference if it exists to separate it from random chance (Faul et al., 2007; Field, 2018; Simon, 2011). Statistical power is a vital element of high-quality, rigorous quantitative education research (Cooper, 2018; Creswell & Creswell, 2018).

There are two types of power analyses: a priori and observed power analyses (Chen & Liu, 2019). A priori power analysis is done during the planning phase of a study, and it enables researchers to establish the minimum sample size for obtaining a high statistical power (Cooper, 2018). In other words, it is used to obtain the minimum sample size requirement for a given power (typically, a power of at least .8 is desired), for a specified level of significance ( $\alpha = .05$ ) and effect size (the software G\*Power [Faul et al., 2007] can compute this, or the researcher can use J. Cohen's [1992] classification of small, medium or large effect size as thresholds). Although the SPSS version 27 makes provision for the computation of a power analysis, the option is only available for parametric statistics. Therefore, I utilised the G\*Power software to compute the achieved power for the current study.

Since the current study employed secondary data analysis, the data has already been collected and, accordingly, an a priori power analysis is not relevant here. Instead, an observed power analysis is used to calculate the achieved power since the data from the sample has already been collected. To compute the achieved power, one needs the sample size, effect size, and significance level (Faul et al., 2007; Field, 2018). Taking a conservative approach for the current study, the smallest sample size was used to compute the achieved power utilising G\*Power. For a level of significance of .05, a sample size of  $n_1 = 14$  for group 1 and  $n_2 = 17$  (smallest pairwise-comparison sample sizes) and a large effect size of .8 (J. Cohen, 1992), the achieved power for the current study was .757.

Although this power estimate is smaller than the desired minimum achieved power of .8, it is close to it (J. Cohen, 1988). Therefore, a conservative approach was employed where the smallest pairwise comparison sample sizes were used, i.e., for the other pairwise comparisons,



the achieved power was higher than .757 for each comparison, respectively. Based on the established reliability, validity and acceptable achieved statistical power for the current study, the following section provides an overview of the descriptive statistics of final year pre-service teachers' self-efficacy and teacher efficacy beliefs in a challenged context.

# 4.4 DESCRIPTIVE STATISTICS OF THE SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS IN A CHALLENGED CONTEXT

# 4.4.1 INTRODUCTION

This section contains an exposition on descriptive statistics, which served to summarise and aid understanding of the data's important characteristics obtained from the FIRE Teacher Resilience Measure. As depicted in Chapter 3, descriptive statistics aim to describe and summarise the observations in a dataset (Ary et al., 2019; Salkind & Shaw, 2020). I used descriptive statistics to derive values from the FIRE Teacher Resilience Measure to determine pre-service teachers' self-efficacy and teacher efficacy beliefs within a challenged context. More specifically, the questionnaire aimed at collecting data on confidence in recovery from setbacks (operationalised as self-efficacy in the current study) and confidence in teaching and behaviour management (operationalised as teacher efficacy beliefs in this study).

Since the early definition of teacher resilience as the capacity of teachers to "bounce back or recover from adversity" is inadequate to conceptualise the full complexity of teacher resilience, the current study operationalised the scale of resilience (Morgan, 2011) as *selfefficacy*. Therefore, the current study focused on intrapersonal resilience-enabling pathways (i.e., traits on the microsystem) to teacher resilience in a severely challenged Global South education system. These pathways are assets within the microsystem (i.e., personal protective resources) of teachers whose presence or absence can enable or constrain resilience. These enablers may include self-efficacy, teacher efficacy beliefs, social, emotional, professional, and motivational capacities, coping behaviours and strategies, personal resources, vocational purpose and intrinsic value, as well as self-care. Thus, the current study operationalised resilience measurement within trait discourses in resilience, denoting resilience as personal qualities (e.g., self-efficacy and teacher efficacy) that enable people to flourish despite adversity.

The domains on self-efficacy and teacher efficacy beliefs (i.e., on the FIRE Teacher Resilience Measure) that formed the current study's focus were completed by N = 1,193 respondents. The means of the constructs were derived to determine the average level of self-efficacy and teacher efficacy beliefs. The means were obtained by statistically establishing the mean or the aggregate of the scores (arithmetic average) (Field, 2018). Both constructs (i.e., self-efficacy [items 1–9] and teacher efficacy [items 1–12]) were averaged using SPSS to



create continuous scale variables for comparison. The constructs were then plotted against each other to explain the overall average scores attained by final year pre-service teachers.

# 4.4.2 DESCRIPTIVE STATISTICS OF THE SELF-EFFICACY OF PRE-SERVICE TEACHERS IN A CHALLENGED CONTEXT (2015 TO 2017)

Table 4.11 and Figure 4.6 provide an overview of the descriptive statistics on pre-service teachers' self-efficacy from the FIRE Teacher Resilience Measure (questions 114–122). Questionnaire items that were included in this scale are:

- "getting over setbacks in school";
- "bouncing back when things upset me";
- "carrying on with school work when things go wrong";
- "carrying on in school when events upset me";
- "feeling certain that things will come right even if there are serious problems in school";
- "managing negative events in school when I try";
- "coping with most problems on any school day";
- "some negative things that have happened in school assisted in dealing better with problems" and
- "not getting disheartened even when children's circumstances make it difficult".

As discussed in Chapter 3, the FIRE Teacher Resilience Measure employed a seven-point Likert scale ranging from one (*absolutely not confident*) to seven (*strongly confident*)<sup>57</sup>.

<sup>&</sup>lt;sup>57</sup> Note that, as a seven-point Likert scale was used, if the mean and median were above the midpoint of four (neutral), the respondents tended to respond in agreement with the statement (as opposed to indicating disagreement with the statement).



# Descriptive Statistics of the Self-Efficacy of Pre-Service Teachers

Variable		Within-cas	е		Within-cas	е		Within-cas	е		Cross-cas	se
		2015			2016			2017			2015–201	7
	Z	W	SD	z	W	SD	Z	W	SD	Z	W	SD
Getting over setbacks in school	311	5.41	1.03	168	5.64	0.92	702	5.49	0.95	1,181	5.49	0.97
Bouncing back, when things upset me	307	5.49	1.22	167	5.69	0.95	697	5.66	1.07	1,171	5.62	1.10
Carrying on with my school work when things go wrong	310	5.76	1.01	167	5.94	0.85	700	5.72	1.04	1,177	5.76	1.01
Carrying on in school when events upset me	311	5.73	1.07	168	5.89	0.83	704	5.67	1.04	1,183	5.72	1.02
Feeling certain that things will come right even if there are serious problems in school	311	5.67	1.07	168	5.80	0.86	703	5.71	1.05	1,182	5.71	1.03
Managing negative events in school when I try	310	5.50	1.02	167	5.68	0.88	701	5.57	0.93	1,178	5.57	0.95
Coping with most problems on any school day	311	5.67	1.05	168	5.85	0.90	701	5.77	0.92	1,180	5.76	0.96
Some negative things that have happened in school have made me better able to deal with problems	308	6.13	1.06	168	6.15	0.85	703	6.01	0.97	1,179	6.06	0.98
Not getting disheartened even when children's circumstances make it difficult	309	5.51	1.16	168	5.73	1.02	702	5.59	1.09	1,179	5.59	1.10
Self-efficacy items average	311	5.65	0.79	168	5.82	0.61	706	5.69	0.70	1,185	5.70	0.72

*Note*. Scale minimum = 1.00; scale maximum = 7.00. The median value for all questions was = 6.00.



As illustrated in Table 4.11, the within-case and cross-case mean comparison for the selfefficacy items demonstrate a consistently high positive response by pre-service teachers. *Preservice teachers in a challenged context, therefore, feel confident to deal with setbacks in school* based on the mean data available for 2015 (5.65); 2016 (5.82); 2017 (5.69) and for 2015 to 2017 (5.70). However, the mean data and the median value (6.00 for all questions) may indicate a response bias skewed to the left relating to the positive response effect or overly positive reporting of teachers (i.e., the Dunning-Kruger effect discussed as a possible limitation of the current study in Chapter 5). Three items that consistently (within-case and cross-case) scored higher than the mean average per year for the self-efficacy items, included "carrying on with my school work when things go wrong"; "coping with most problems on any school day" and "some negative things that have happened in school have made me better able to deal with problems". The item "getting over setbacks in school", also the first item on the scale, consistently scored the lowest per year (within-case) and cross-case (2015–2017). Figure 4.6 represents the descriptive results of the self-efficacy items of pre-service teachers in a visual format.

#### Figure 4.6



Descriptive Results of the Self-efficacy of Pre-service Teachers

As illustrated by Figure 4.6, in 2016, the mean comparison of the self-efficacy items for preservice teachers is substantially higher than those reported in 2015 and 2017.



# 4.4.3 DESCRIPTIVE STATISTICS OF THE TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS IN A CHALLENGED CONTEXT (2015 TO 2017)

Table 4.12 and Figure 4.7 summarise the descriptive statistics on the teacher efficacy beliefs about pre-service from the FIRE Teacher Resilience Measure (questions 123–134). Questionnaire items included:

- "teaching all the subjects on the curriculum effectively";
- "explaining difficult material in ways that the children will understand it";
- "suggesting suitable examples when the children are having difficulty understanding";
- "teaching in a way that my students will remember important information";
- "applying the new developments in the curriculum into my teaching";
- "helping children focus on learning tasks and avoid distractions";
- "managing inappropriate behaviour";
- "encouraging students to take responsibility for their behaviour";
- "dealing with the diverse learning needs of the students in my class";
- "teaching students positive behaviour";
- "providing students with clear specific behaviour expectations" and
- "communicating effectively with parents".

Pre-service teachers indicated their responses on a seven-point Likert scale ranging from one (*absolutely not confident*) to seven (*strongly confident*).



# Descriptive Statistics of Pre-Service Teacher Efficacy Beliefs

Variable		Within-case	9		Within-case	)		Within-case	)	(	Cross-case	
		2015			2016			2017		:	2015–2017	
	z	W	SD	z	N	SD	z	N	SD	Z	N	SD
Teaching all the subjects on the curriculum effectively	310	5.59	1.27	167	5.86	1.01	698	5.64	1.24	1,175	5.66	1.22
Explaining difficult material in ways that the children will understand it	307	5.99	1.14	168	6.05	0.87	693	6.00	0.96	1,168	6.01	0.99
Suggesting suitable examples when the children are having difficulty understanding	310	6.02	1.10	168	6.15	0.86	704	6.02	0.94	1,182	6.04	0.97
Teaching in a way that my students will remember important information	311	6.08	1.09	168	6.26	0.75	703	6.15	0.90	1,182	6.15	0.94
Applying the new developments in the curriculum into my teaching	311	5.68	1.09	168	5.95	0.89	702	5.80	0.91	1,181	5.79	0.96
Helping children focus on learning tasks and avoid distractions	311	5.64	1.06	168	6.04	0.80	702	5.81	0.94	1,181	5.80	0.96
Managing inappropriate behaviour	309	5.67	1.21	167	5.76	1.18	699	5.75	1.03	1,175	5.73	1.10
Encouraging students to take responsibility for their behaviour	309	5.91	1.13	168	6.18	0.82	702	6.01	0.93	1,179	6.01	0.97
Dealing with the diverse learning needs of the students in my class	311	5.86	1.08	167	6.17	0.88	702	5.86	0.98	1,180	5.90	1.00
Teaching students positive behaviour	310	6.16	1.06	165	6.35	0.78	699	6.16	0.92	1,174	6.19	0.95
Providing students with clear specific behaviour expectations	310	5.96	1.12	168	6.20	0.75	706	6.03	0.91	1,184	6.03	0.95
Communicating effectively with parents	311	5.68	1.29	162	5.70	1.30	694	5.42	1.38	1,167	5.53	1.35
Teacher efficacy items average	311	5.86	0.89	168	6.06	0.57	706	5.88	0.70	1,185	5.90	0.74

*Note*. Scale minimum = 1.00; scale maximum = 7.00. The median value for all questions was = 6.00.



As demonstrated in Table 4.12, the within-case and cross-case mean comparison for the teacher efficacy items also reveals a consistently positive final year pre-service teachers' response. *Pre-service teachers consequently feel confident regarding teaching and behaviour management* given the mean data available for 2015 (5.86); 2016 (6.06); 2017 (5.88), and 2015 to 2017 (5.90). Items that constantly (within-case and cross-case) scored higher than the mean average per year for the teacher efficacy items included "suggesting suitable examples when the children are having difficulty understanding"; "teaching in a way that my students will remember important information"; "encouraging students to take responsibility for their behaviour" and "providing students with clear specific behaviour expectations".

The item "teaching students positive behaviour" was also the highest-scoring mean for 2015 (6.16), 2016 (6.35), 2017 (6.16) and 2015 to 2017 (6.19), indicating the possible need for pre-service teachers to teach learners positive behaviour. However, the item with the lowest mean for 2016 (5.70); 2017 (5.42) and 2015 to 2017 (5.53), namely "communicating effectively with parents", may signify pre-service teachers' limited experiences and skills in this regard. In 2015, the lowest mean, (5.59), was denoted by the item "teaching all the subjects on the curriculum effectively", indicating a possible lower teaching efficacy for skills across curriculum subjects. Figure 4.7 graphically represents the descriptive results of the teacher efficacy items of pre-service teachers.



#### Figure 4.7

Descriptive Results of the Teacher Efficacy Beliefs of Pre-service Teachers

Figure 4.7 elucidates that the mean comparison of the teacher efficacy items for pre-service teachers is higher in 2016 than in 2015 and 2017.

FIRE Teacher Resilience Measure Teacher Efficacy Questions 2015 2016 2017 2015-2017



# 4.4.4 DESCRIPTIVE STATISTICS OF THE SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS IN A CHALLENGED CONTEXT (2015 TO 2017)

This section presents an overview of descriptive statistics of pre-service teachers' self-efficacy, and teacher efficacy beliefs compared within-case and cross-case, as depicted in Figure 4.8.

## Figure 4.8

Comparison of the Self-efficacy and Teacher Efficacy Beliefs of Pre-service Teachers



As illustrated in Figure 4.8, the *teacher efficacy is higher than the self-efficacy for pre-service teachers in a challenged context for within-case and cross-case comparison.* Furthermore, 2016 self-efficacy (5.82) and teacher efficacy (6.06) item means are higher than the 2015 and 2017 item means for both constructs (i.e., self-efficacy and teacher efficacy beliefs). However, as previously indicated, the higher means should be cautiously interpreted due to the relatively smaller 2016 sample given the #Fees-Must-Fall2016 movement at South African universities. Thus, overall (within-case and cross-case comparison), the self-efficacy (i.e., confidence in recovery from setbacks) and teacher efficacy (i.e., confidence in teaching and behaviour management), based on the mean values, of final year pre-service teachers in a challenged context is high. Furthermore, pre-service teachers' self-efficacy is slightly lower than the teacher efficacy beliefs of final year pre-service teachers in a challenged setting. See Section 4.5 for the results of inferential statistics indicating the statistical significance for pre-service teachers' self-efficacy and teacher efficacy beliefs (within-case and cross-case comparison).



# 4.4.5 DESCRIPTIVE RESULTS OF THE SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS BASED ON GENDER AND ENROLLED PRE-SERVICE TEACHING PROGRAMMES: MEAN VALUES FIRE TEACHER RESILIENCE MEASURE

This section presents an integration of descriptive statistics of self-efficacy and teacher efficacy beliefs of final year pre-service teachers based on the mean values of gender (Figure 4.9) and programme enrolment (Figure 4.10).

# 4.4.5.1 Exploring gender differences concerning levels of self-efficacy and teacher efficacy beliefs of pre-service teachers

Figure 4.9 compares the mean values for male and female self-efficacy and teacher efficacy beliefs of final year pre-service teachers with a graphical presentation.



#### Figure 4.9



Comparison of Pre-service Teachers' Gender (Male and Female) Self-efficacy and Teacher Efficacy Beliefs



Comparison of female self-efficacy and teacher efficacy

Female Self-Efficacy Items Average

Female Teacher Efficacy Items Average

As indicated in Figure 4.9, the mean for teacher efficacy was consistently higher than the mean for self-efficacy for both male and female pre-service teachers, except for 2016, where the mean for self-efficacy (6.04) was higher than teacher efficacy (5.93) for males. The year 2016 also produced the highest mean (6.06) for self-efficacy for female pre-service teachers in comparison with 2015 (5.86) and 2017 (5.94). See Section 4.5 for the results of inferential statistics indicating statistical significance for gender (within-case and cross-case comparison).



# 4.4.5.2 Exploring enrolled pre-service teaching programme differences concerning levels of self-efficacy and teacher efficacy beliefs

Figure 4.10 provides a visual comparison of the mean values for the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context enrolled in a teaching programme.

#### Figure 4.10

Comparison of Pre-Service Teachers' Self-Efficacy and Teacher Efficacy Beliefs for Enrolled Teaching Programmes



■ FP/ECD Self-Efficacy Items Average ■ FP/ECD Teacher Efficacy Items Average



SP Self-Efficacy Items Average SP Teacher Efficacy Items Average

Comparison of self-efficay and teacher efficacy beliefs for IP



IP Self-Efficacy Items Average

■ IP Teacher Efficacy Items Average



FET Self-Efficacy Items Average FET Teacher Efficacy Items Average



As illustrated in Figure 4.10, the mean for teacher efficacy was consistently higher than the mean for self-efficacy for all teaching programmes. The mean self-efficacy values for FP/EC 2015 (5.72) and 2017 (5.89) is the highest compared to other enrolled teaching programmes, except for 2016, where the IP has the highest self-efficacy mean (5.92). In addition, the mean teacher efficacy values for the FP/EC rank the highest for all years (see Section 4.5 for the results of inferential statistics indicating statistical significance for enrolled teaching programmes [within-case and cross-case comparison]). Given the confirmation of the reliability analysis (Section 4.3) and the descriptive results (Section 4.4) between the self-efficacy and teacher efficacy of pre-service teachers, inferential statistics were performed on data for further investigation of the relationship between self-efficacy and teacher efficacy beliefs of final year pre-service teachers in a challenged context as presented in the following section.

# 4.5 INFERENTIAL STATISTICS OF THE SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS WITH A CHALLENGED CONTEXT

# 4.5.1 INTRODUCTION

In this section, I present the correlation analysis and the results (within-case and cross-case) of the inferential statistics for the FIRE Teacher Resilience Measure to identify potentially significant differences and similarities across possible intrapersonal resilience-enabling pathways concerning the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context. Before conducting inferential statistics, the continuous variables were tested for normality because parametric methods can be used if they are normally distributed. If the continuous variables are not normally distributed, nonparametric methods must be used. Note that the continuous constructs referred to here were created by averaging the items relating to each construct. These continuous variables were tested for normality, and since the p-values of the Kolmogorov-Smirnov and Shapiro-Wilk test, for both constructs, were less than .05 (all p-values < .001), it was found that the continuous variables were not normally distributed and, accordingly, nonparametric statistics had to be used.

The inferences drawn based on the sample data (Pietersen & Maree, 2019d) aimed to address the hypotheses as formulated in Chapter 3. If the *p*-value is less than .05, the null hypothesis (i.e.,  $H_0$ : Medians do not differ significantly from each other) is rejected, and the medians differ significantly from each other (i.e.,  $H_a$ : Medians differ significantly from each other). Alternately, if the *p*-value is greater than .05, the null hypothesis ( $H_0$ ) is not rejected, and the medians do not differ significantly from each other.



# 4.5.2 CORRELATION ANALYSIS BETWEEN THE SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS IN A CHALLENGED CONTEXT

As described in Chapter 3, the correlation between variables indicates that the degree to which the variables are correlated is linearly related (Field, 2018; Roni et al., 2020). To determine if there is a significant correlation between the self-efficacy and teacher efficacy beliefs of preservice teachers, the Spearman correlation coefficient, denoted by  $r_s$  (an appropriate measure of similarity between two ordinal rankings of a single dataset), was employed (Field, 2018; Roni et al., 2020). It should be noted that correlation does not imply causation (Leedy & Ormrod, 2015; Sauro & Lewis, 2016).

Correlation takes on values between (and including) -1 and +1, i.e.,  $-1 \le r_s \le 1$ . Thus, the closer the value is to +1, the stronger the positive correlation, and the closer the value is to -1, the stronger the negative correlation (Pietersen & Maree, 2019a). In Table 4.13, the correlation between the averages of the self-efficacy and teacher efficacy belief items of preservice teacher items from the FIRE Teacher Resilience Measure are displayed.

#### Table 4.13

Correlation between Self-Efficacy Items Average and Teacher Efficacy Items Average (2015 to 2017)

Variables	Spearman's rho (r <sub>s</sub> )	Ν	p
Self-efficacy items average and Teacher efficacy items average	.580	1,185	.000

Based on the Spearman correlation coefficient (.580) with *p*-value < .001, *self-efficacy and teacher efficacy* beliefs of pre-service teacher items have a statistically significant positive correlation as depicted in Table 4.13. In Table 4.14, a comprehensive outline is provided depicting the correlation between all self-efficacy and teacher efficacy beliefs of pre-service teachers for the FIRE Teacher Resilience Measure (2015 to 2017).



Inter-Item Correlation between Self-Efficacy and Teacher Efficacy Beliefs of Pre-Service Teachers Items (2015 to 2017)

Self-efficacy Items (SE)		Teaching all the subjects on the curriculum effectively	Explaining difficult material in ways that the children will understand it	Suggesting suitable examples when the children are having difficulty understanding	Teaching in a way that my students will remember important information	Applying the new developments in the curriculum into my teaching	Helping children focus on learning tasks and avoid distractions	Managing inappropriate behaviour	Encouraging students to take responsibility for their behaviour	Dealing with the diverse learning needs of the students in my class	Teaching students positive behaviour	Providing students with clear specific behaviour expectations	Communicating effectively with parents
Getting over setbacks in school	r <sub>s</sub>	.288	.305	.309	.275	.343	.292	.276	.295	.341	.314	.318	.228
	Ν	1,171	1,164	1,178	1,178	1,177	1,177	1,171	1,175	1,176	1,170	1,180	1,163
Bouncing back. when things upset me	rs	.215	.310	.270	.238	.262	.273	.250	.299	.296	.288	.268	.171
	Ν	1,161	1,156	1,169	1,168	1,168	1,167	1,163	1,165	1,167	1,161	1,170	1,153
Carrying on with my school work when	rs	.321	.308	.299	.281	.308	.314	.261	.321	.323	.338	.310	.236
	Ν	1,168	1,162	1,175	1,174	1,173	1,173	1,168	1,172	1,172	1,166	1,176	1,160
Carrying on in school when events upset me	rs	.297	.288	.265	.262	.235	.301	.258	.283	.275	.283	.276	.223
	Ν	1,173	1,166	1,180	1,180	1,179	1,179	1,173	1,177	1,178	1,172	1,182	1,166
Feeling certain that things will come right	rs	.248	.289	.258	.262	.266	.307	.242	.295	.286	.294	.295	.227
even if there are serious problems in school	Ν	1,173	1,166	1,180	1,180	1,178	1,178	1,172	1,176	1,177	1,171	1,181	1,165
	rs	.334	.355	.316	.286	.329	.374	.339	.330	.320	.340	.340	.296



						Teac	her Efficacy	ltems (TE)						•
Self-efficacy Items (SE)		Teaching all the subjects on the curriculum effectively	Explaining difficult material in ways that the children will understand it	Suggesting suitable examples when the children are having difficulty understanding	Teaching in a way that my students will remember important information	Applying the new developments in the curriculum into my teaching	Helping children focus on learning tasks and avoid distractions	Managing inappropriate behaviour	Encouraging students to take responsibility for their behaviour	Dealing with the diverse learning needs of the students in my class	Teaching students positive behaviour	Providing students with clear specific behaviour expectations	Communicating effectively with parents	
Managing negative events in school when I try	Ν	1,169	1,162	1,176	1,175	1,174	1,175	1,168	1,173	1,173	1,167	1,177	1,160	
Coping with most problems on any school	rs	.359	.365	.321	.299	.332	.365	.347	.342	.360	.348	.346	.260	
day	Ν	1,172	1,164	1,178	1,179	1,177	1,177	1,172	1,174	1,175	1,169	1,179	1,162	
Some negative things that have happened in	rs	.298	.321	.308	.265	.316	.298	.267	.346	.311	.296	.315	.207	
school have made me better able to deal with problems	Ν	1,170	1,164	1,177	1,177	1,175	1,175	1,170	1,173	1,174	1,168	1,178	1,161	
Not getting disheartened even when children's circumstances make it difficult	rs	.331	.352	.320	.308	.371	.352	.288	.340	.326	.311	.308	.285	
children's circumstances make it diricult	Ν	1,171	1,164	1,178	1,177	1,175	1,175	1,170	1,174	1,174	1,168	1,178	1,161	



According to Spearman's correlation coefficients and the corresponding *p*-values (p < .001), all items for self-efficacy and teacher efficacy beliefs of pre-service teachers have a significant positive correlation, as Table 4.14 demonstrates. Since there was a significant positive correlation, the null hypothesis, as depicted in Section 3.2.4.2, was rejected. The correlation, as explained in Chapter 3, however, does not imply that one variable causes the other but instead highlights the association between the variables (i.e., self-efficacy and teacher efficacy). The following section elaborates on the statistically significant results obtained for within-case and cross-case analysis of final year-pre-service teachers' self-efficacy and teacher efficacy beliefs.

# 4.5.3 RESULTS OF THE FIRE TEACHER RESILIENCE MEASURE CASE 1 (2015): SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS (WITHIN-CASE COMPARISON)

# 4.5.3.1 Summary of the *p*-value for the self-efficacy and teacher efficacy beliefs of preservice teachers 2015

Table 4.15 presents the results of the Wilcoxon signed-rank (WSR) test (Z) performed on withincase data (for the year 2015) to compare the self-efficacy and teacher efficacy beliefs of preservice teachers. The nonparametric Wilcoxon signed-rank test was used since it is the appropriate statistical test when comparing two related groups for Likert scale data where normality cannot be assumed.

#### Table 4.15

Summary of the p-value for Self-efficacy and Teacher Efficacy Beliefs (2015)

Pairwise comparison	Ζ	р	Significant diff	Conclusion
Self-efficacy vs Teacher efficacy	-6.007	.000	Yes	The median for teacher efficacy of 6.00 is statistically significantly higher than the median of self-efficacy ( $Mdn = 5.78$ )

As demonstrated in Table 4.15, there was a statistically significant difference (diff) between preservice teachers' self-efficacy and teacher efficacy beliefs in 2015. The median (5.78) for preservice teachers' self-efficacy was statistically significantly lower than the median for teacher efficacy beliefs (*Mdn* = 6.00). The Wilcoxon signed-rank test statistics (z = -6.007, p < .001) indicated that the difference was statistically significant, and therefore, the null hypothesis (i.e.,  $H_0$ : Medians do not differ significantly from each other) was rejected.

# 4.5.3.2 Summary of the *p*-values for gender 2015

Table 4.16 presents the results of the Mann-Whitney test conducted on gender (2015) to compare the self-efficacy and teacher efficacy beliefs of male and female pre-service teachers.



The nonparametric statistical procedure (i.e., Mann-Whitney test) was performed on the data (i.e., gender) since it is the appropriate statistical test when comparing two independent groups for Likert scale data (i.e., data that is not normally distributed).

#### Table 4.16

Summary of the p-values for Gender (2015)

		Μ	ale			Fen	nale				
	Z	Mdn	Μ	SD	Ζ	Mdn	Μ	SD	D	ď	Significant diff
Self-efficacy items average	60	5.78	5.65	0.64	249	5.78	5.66	0.81	7012.00	.460	No
Teacher efficacy items average	60	6.00	5.90	0.60	249	6.00	5.86	0.91	7176.50	.636	No

As indicated in Table 4.16, there were no statistically significant differences between the selfefficacy and teacher efficacy beliefs of pre-service teachers' items for male and female preservice teachers in 2015.

# 4.5.3.3 Summary of the *p*-values for enrolled pre-service teaching programmes 2015

Table 4.17 presents the results of the Kruskal-Wallis test conducted on the enrolled pre-service teaching programmes (2015) to compare the self-efficacy and teacher efficacy beliefs of pre-service teachers based on the teaching programme they were enrolled in. The nonparametric Kruskal-Wallis test was performed on the data (i.e., the enrolled pre-service teaching programme data). It is the appropriate statistical test when comparing three or more independent groups for Likert scale data.



Summary of the p-values for Enrolled Pre-Service Teaching Programmes (2015)

	Foundation phase (FP)/Early Int childhood development (ECD)		Inte	ermediate	e phase	(IP)	ŝ	Senior ph	nase (SF	?)	Fu	rther edu training	ucation a g (FET)	Ind	E tead	nrolled ching pr spe	pre-serv ogramm cified	ice e not					
	Z	Mdn	W	SD	Z	Mdn	W	SD	Z	Mdn	M	SD	Z	Mdn	Μ	SD	Z	Mdn	W	SD	κw	đ	Significant diff
Self-efficacy items average	105	5.78	5.72	0.76	32	5.56	5.30	1.11	30	5.61	5.68	0.60	108	5.89	5.70	0.78	25	5.78	5.60	0.65	6.24	.182	No
Teacher efficacy items average	105	6.17	6.03	0.84	32	5.79	5.48	1.36	30	5.88	5.85	0.83	108	6.00	5.84	0.79	25	6.00	5.89	0.55	10.23	.037	Yes



There was a statistically significant difference for teacher efficacy items for the enrolled preservice teaching programmes in 2015, for ad-hoc pairwise tests (i.e., Mann-Whitney tests), as indicated in Table 4.17. Table 4.18 also includes the Holm corrections with the indicated adjusted *p*-values. As discussed in Chapter 3, the Holm corrections, as a method of choice, are conducted for multiple comparisons (e.g., the Kruskal-Wallis test) to calculate multiple test adjustments (Aickin & Gensler, 1996).

#### Table 4.18

Summary of Ad-Hoc Pairwise Test for Enrolled Pre-Service Teaching Programmes (Teacher Efficacy) (2015)

Pairwise comparison	U	р	Adjusted <i>p</i> -value	Conclusion
FP/ECD vs IP	1159.00	.008	.081	Although the unadjusted <i>p</i> -value showed a statistically significant difference, the adjusted <i>p</i> -value showed no statistically significant difference
IP vs SP	409.00	.316	1.000	No statistically significant difference
SP vs FET	1601.50	.924	1.000	No statistically significant difference
FET vs UP Not specified	1331.00	.913	1.000	No statistically significant difference
FP/ECD vs SP	1318.50	.174	1.000	No statistically significant difference
IP vs FET	1492.00	.241	1.000	No statistically significant difference
SP vs UP Not specified	357.50	.767	1.000	No statistically significant difference
FP/ECD vs FET	4629.0	.020	.180	Although the unadjusted <i>p</i> -value showed a statistically significant difference, the adjusted <i>p</i> -value showed no statistically significant difference
IP vs Not specified	353.50	.454	1.000	No statistically significant difference
FP/ECD vs Not specified	1011.50	.075	.600	No statistically significant difference

As indicated in Table 4.18, there was no statistically significant difference for teacher efficacy items for the enrolled pre-service teaching programmes in 2015.

# 4.5.4 RESULTS OF THE FIRE TEACHER RESILIENCE MEASURE CASE 2 (2016): SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS (WITHIN-CASE COMPARISON)

# 4.5.4.1 Summary of the *p*-value for the self-efficacy and teacher efficacy beliefs of preservice teachers in 2016

Table 4.19 displays the results of the Wilcoxon signed-rank test used on within-case data (2016) to compare the self-efficacy and teacher efficacy beliefs of pre-service teachers.



Summary of the p-value for Self-efficacy and Teacher Efficacy Beliefs (2016)

Pairwise comparison	Ζ	р	Significant diff	Conclusion
Self-efficacy vs Teacher efficacy	-5.089	.000	Yes	The median for teacher efficacy of 6.08 is statistically significantly higher than the median of self-efficacy ( $Mdn = 5.89$ )

Table 4.19 indicated that the median in 2016 for pre-service teachers' self-efficacy equalled 5.89, and the median for teacher efficacy beliefs equalled 6.08. From the Wilcoxon signed-rank statistics (Z = -5.089, p < .001), it can be seen that the difference is statistically significant. Therefore, the null hypothesis, as depicted in Section 3.2.4.1, was rejected.

# 4.5.4.2 Summary of the *p*-values for gender 2016

Table 4.20 presents the results of the Mann-Whitney test conducted on gender in 2016 to compare the self-efficacy and teacher efficacy beliefs of male and female pre-service teachers.

#### Table 4.20

Summary of the p-values for Gender (2016)

		N	lale			Fe	male		-		
	Z	Mdn	W	SD	Z	Mdn	W	SD	n	d	Significant diff
Self-efficacy items average	21	5.89	6.04	0.46	139	5.89	5.77	0.63	1152.50	.120	No
Teacher efficacy items average	21	5.83	5.93	0.55	139	6.08	6.06	0.57	1251.00	.292	No

As highlighted in Table 4.20, there was no statistically significant difference between the selfefficacy and teacher efficacy items for male and female pre-service teachers in 2016.



# 4.5.4.3 Summary of the *p*-values for enrolled pre-service teaching programmes 2016

Table 4.21 presents the results of the Kruskal-Wallis test conducted to compare the self-efficacy and teacher efficacy beliefs of pre-service teachers based on the enrolled pre-service teaching programmes in 2016.

#### Table 4.21

Summary of the p-values for Enrolled Pre-Service Teaching Programmes (2016)

	Four de	ndation Early cl velopm	phase hildhoo lent (E0	(FP)/ d CD)	Inter	Intermediate phase (IP)		S	Senior phase (SP)			Further education and training (FET)				Enrolled pre-service teaching programme not specified							
	Z	Mdn	W	SD	Z	Mdn	W	SD	Z	Mdn	W	SD	Z	Mdn	W	SD	Z	Mdn	W	SD	KW	đ	Significant diff
Self-efficacy items average	45	5.89	5.86	0.63	17	6.00	5.92	0.54	14	5.88	5.83	0.52	60	5.78	5.76	0.66	20	5.78	5.68	0.60	2.44	.656	No
Teacher efficacy items average	45	6.17	6.20	0.52	17	6.08	6.04	0.38	14	5.92	5.95	0.56	60	6.00	5.96	0.58	20	6.08	5.96	0.74	4.41	.353	No



As indicated in Table 4.21, there were no statistically significant differences between the selfefficacy and teacher efficacy beliefs of pre-service teachers by teaching programme enrolled in 2016.

# 4.5.5 RESULTS OF THE FIRE TEACHER RESILIENCE MEASURE CASE 3 (2017): SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS (WITHIN-CASE COMPARISON)

# 4.5.5.1 Summary of the *p*-value for the self-efficacy and teacher efficacy beliefs of preservice teachers 2017

Table 4.22 portrays the results of the Wilcoxon signed-rank test conducted to compare the selfefficacy and teacher efficacy beliefs of pre-service teachers in 2017.

# Table 4.22

Summary of the p-value for Self-efficacy and Teacher Efficacy Beliefs (2017)

Pairwise comparison	Ζ	p	Significant diff	Conclusion
Self-efficacy vs Teacher efficacy	-8.092	.000	Yes	The median for teacher efficacy of 5.92 is statistically significantly higher than the median of self-efficacy ( $Mdn = 5.78$ )

As indicated in Table 4.22, there was a statistically significant difference between the self-efficacy (Mdn = 5.78) and teacher efficacy (Mdn = 5.92) beliefs of pre-service teachers within-case comparison during 2017. Based on the Wilcoxon signed-rank test statistics (Z = -8.092, p < .001), consequently, the null hypothesis, as depicted in Section 3.2.4.1, was rejected.

# 4.5.5.2 Summary of the *p*-values for gender 2017

Table 4.23 presents the results of the Mann-Whitney test conducted on gender in 2017 to compare the self-efficacy and teacher efficacy beliefs of male and female pre-service teachers.



Summary of the p-values for Gender (2017)

		М	ale			Fei	male				
	Z	Mdn	Μ	SD	Z	Mdn	Μ	SD	C	d	Significant diff
Self-efficacy items average	141	5.56	5.56	0.76	535	5.78	5.73	0.66	33087.00	.025	Yes
Teacher- efficacy items average	141	5.75	5.71	0.76	535	6.00	5.94	0.66	30969.50	.001	Yes

As indicated in, Table 4.23 the median for females of 5.78 was statistically significantly higher than the median for males of 5.56 for the self-efficacy items in 2017. In addition, the median for females of 6.00 was also statistically significantly higher than the median for males of 5.75 for teacher efficacy items in 2017.

# 4.5.5.3 Summary of the *p*-values for enrolled pre-service teaching programmes 2017

Table 4.24 presents the results of the Kruskal-Wallis test conducted on enrolled pre-service teaching programmes (2017) to compare the self-efficacy and teacher efficacy beliefs of pre-service teachers based on the teaching programme they were enrolled in.



Summary of the p-values for Enrolled Pre-Service Teaching Programmes (2017)

	Four de	ndation Early c velopm	phase hildhoo nent (E0	(FP)/ d CD)	Inte	Intermediate phase (IP)			Senior phase (SP)				Further education and training (FET)				Enrolled pre-service teaching programme not specified						
	Z	Mdn	М	SD	Z	Mdn	М	SD	Z	Mdn	М	SD	Z	Mdn	М	SD	Z	Mdn	М	SD	KW	ď	Significant diff
Self-efficacy items average	160	5.89	5.89	0.54	107	5.88	5.77	0.70	56	5.83	5.67	0.73	266	5.67	5.62	0.69	89	5.78	5.49	0.80	18.06	.001	Yes
Teacher efficacy items average	160	6.17	6.09	0.58	107	6.08	6.01	0.70	56	5.95	5.87	0.71	266	5.79	5.77	0.70	89	5.75	5.75	0.72	31.26	.000	Yes



As highlighted in Table 4.24, there was a statistically significant difference between pre-service teachers' self-efficacy and teacher efficacy beliefs' items for enrolled pre-service teaching programmes for 2017. Since there was a statistically significant difference for self-efficacy items for the teaching programme pre-service teachers were enrolled for, an ad-hoc test (i.e., Mann-Whitney tests) was run to test pairwise for enrolled pre-service teaching programmes as indicated in Table 4.25. Table 4.25 also includes the Holm corrections (Aickin & Gensler, 1996) with the indicated adjusted *p*-values.

#### Table 4.25

Pairwise comparison	U	p	Adjusted <i>p</i> -value	Conclusion
FP/ECD vs IP	7963.00	.333	1.000	No statistically significant difference between the medians
IP vs SP	2823.50	.546	1.000	No statistically significant difference between the medians
SP vs FET	6952.00	.433	1.000	No statistically significant difference between the medians
FET vs UP Not specified	10919.00	.273	1.000	No statistically significant difference between the medians
FP/ECD vs SP	3886.00	.139	.834	No statistically significant difference between the medians
IP vs FET	12485.00	.063	.441	No statistically significant difference between the medians
SP vs UP Not specified	2172.00	.193	.965	No statistically significant difference between the medians
FP/ECD vs FET	16762.00	.000	.000	The median for FP/ECD of 5.89 is statistically significantly higher than the median of FET ( <i>Mdn</i> = 5.67)
IP vs Not specified	3884.50	.026	.208	Although the unadjusted <i>p</i> -value showed a statistically significant difference, the adjusted <i>p</i> -value showed no statistically significant difference
FP/ECD vs Not specified	5360.00	.001	.009	The median for FP/ECD of 5.89 is statistically significantly higher than the median of Not specified ( <i>Mdn</i> = 5.78)

Summary of the Ad-hoc Pairwise Test for Enrolled Pre-Service Teaching Programmes (Self-efficacy) (2017)



As specified in Table 4.25, based on the adjusted *p*-values for the ad-hoc pairwise test for enrolled teaching programme (self-efficacy) (2017), the median for the Foundation phase (FP)/Early childhood development (ECD) of 5.89 was statistically significantly higher than the median for the Further education and training phase of 5.67. In addition, the median for Foundation phase (FP)/Early childhood development (ECD) of 5.89 was statistically significantly higher than the median of Not Specified of 5.78. Furthermore, as demonstrated in Table 4.24, there was also a statistically significant difference for teacher efficacy items for the teaching programme pre-service teachers were enrolled for; ad-hoc tests (i.e., Mann-Whitney tests) was run to test pairwise for UP teaching programme enrolled for as indicated in Table 4.26. Table 4.26 also includes the Holm corrections (Aickin & Gensler, 1996) with the indicated adjusted *p*-values.

#### Table 4.26

Pairwise comparison	U	p	Adjusted <i>p</i> -value	Conclusion
FP/ECD vs IP	8150.00	.507	1.000	No statistically significant difference between the medians
IP vs SP	2648.50	.224	1.000	No statistically significant difference between the medians
SP vs FET	6697.50	.236	1.000	No statistically significant difference between the medians
FET vs Not specified	11479.00	.669	1.000	No statistically significant difference between the medians
FP/ECD vs SP	3760.50	.073	.438	No statistically significant difference between the medians
IP vs FET	11081.00	.001	.008	The median for IP of 6.08 is statistically significantly higher than the median of FET ( $Mdn = 5.79$ )
SP vs Not specified	2223.50	.275	1.000	No statistically significant difference between the medians
FP/ECD vs FET	15275.00	.000	.000	The median for FP/ECD of 6.17 is statistically significantly higher than the median of FET ( $Mdn = 5.79$ )
IP vs Not specified	3657.50	.005	.035	The median for IP of 6.08 is statistically significantly higher than the median of Not specified ( $Mdn = 5.75$ )
FP/ECD vs Not specified	5055.50	.000	.000	The median for FP/ECD of 6.17 is statistically significantly higher than the median of Not specified ( $Mdn = 5.75$ )

Summary of Ad-hoc Pairwise Test for Enrolled Pre-service Teaching Programmes (Teacher Efficacy) (2017)



# 4.5.6 RESULTS OF THE FIRE TEACHER RESILIENCE MEASURE CROSS-CASE COMPARISON (2015 TO 2017): SELF-EFFICACY AND TEACHER EFFICACY BELIEFS OF PRE-SERVICE TEACHERS

# 4.5.6.1 Summary of the *p*-value for the self-efficacy and teacher efficacy beliefs of preservice teachers (2015–2017)

Table 4.27 shows the results of the Wilcoxon signed-rank test performed on cross-case data (2015–2017) to compare the self-efficacy and teacher efficacy beliefs of pre-service teachers. A nonparametric statistical procedure (i.e., Wilcoxon signed-rank test) was used since it was the appropriate statistical test when comparing two related groups for Likert scale data where normality cannot be assumed.

# Table 4.27

Pairwise comparison	Ζ	р	Significant diff	Conclusion
Self-efficacy vs Teacher efficacy	-5.089	.000	Yes	The median for teacher efficacy of 6.00 is statistically significantly higher than the median of self-efficacy ( <i>Mdn</i> = 5.78)

Summary of the p-value for Self-efficacy and Teacher Efficacy Beliefs (2015–2017)

Table 4.27 demonstrated the difference between the self-efficacy and teacher efficacy beliefs of pre-service teachers' cross-cases (2015–2017). The median for teacher efficacy equalling 6.00 was statistically significantly higher than the median of self-efficacy equalling 5.78. Given the Wilcoxon signed-rank test statistics (Z = -5.089, p < .001), the difference was shown to be statistically significant and, therefore, the null hypothesis, as depicted in Section 3.2.4.1, was rejected.

# 4.5.6.2 Summary of the *p*-values for gender (2015 to 2017)

Table 4.28 presents the results of the Kruskal-Wallis test conducted for a cross-case analysis between 2015 to 2017 to compare the self-efficacy and teacher efficacy beliefs of male and female pre-service teachers.



Summary of the p-values for Gender (2015 to 2017)

		20	15			20	16		2017						
	Z	Mdn	W	SD	Z	Mdn	W	SD	Ζ	Mdn	W	SD	KW	đ	Significant diff
								Male							
Self-Efficacy Items Average	60	5.78	5.65	0.64	21	5.89	6.04	0.46	141	5.56	5.56	0.76	7.83	.020	Yes
Teacher Efficacy Items Average	60	6.00	5.90	0.60	21	5.83	5.93	0.55	141	5.75	5.71	0.76	3.41	.181	No
								Female							
Self-Efficacy Items Average	249	5.78	5.66	0.81	139	5.89	5.77	0.63	535	5.78	5.73	0.66	0.64	.725	No
Teacher Efficacy Items Average	249	6.00	5.86	0.91	139	6.08	6.06	0.57	535	6.00	5.94	0.66	3.18	.204	No



As indicated in Table 4.28, there was no statistically significant difference between the selfefficacy items or teacher efficacy items for female pre-service teachers for 2015 to 2017. Furthermore, there is no statistically significant difference between the teacher efficacy items for male pre-service teachers for 2015 to 2017. However, there was a statistically significant difference between the self-efficacy items for male pre-service teachers for 2015 to 2017. Given the *statistically significant difference for male pre-service teacher self-efficacy items*, an ad-hoc test (i.e., Mann-Whitney test) was run to test pairwise for self-efficacy indicated in Table 4.29. Table 4.29 also includes the Holm correction (Aickin & Gensler, 1996) with the indicated adjusted p-values.

#### Table 4.29

Pairwise comparison	U	p	Adjusted <i>p</i> -value	Conclusion
2015 vs 2016	415.50	.020	.040	The median of 5.89 for 2016 is statistically significantly higher than the median of 5.78 for 2015
2015 vs 2017	4007.00	.553	.553	No statistically significant difference
2016 vs 2017	929.00	.006	.018	The median of 5.89 for 2016 is statistically significantly higher than the median of 5.56 for 2017

Summary of Ad-hoc Pairwise Test for Male Pre-Service Teacher Self-Efficacy (2015 to 2017)

As illustrated in Table 4.29, the median for male self-efficacy in 2016 (5.89) was statistically significantly higher than the median for 2015 (5.78) when comparing 2015 with 2016, while the median for 2016 (5.89) was again statistically significantly higher than the median for 2017 (5.56) when comparing 2016 with 2017. Therefore, the results indicate that 2016 had a significantly higher median for the self-efficacy of male pre-service teachers.

# 4.5.6.3 Summary of the *p*-values for enrolled pre-service teaching programmes (2015 to 2017)

Table 4.30 presents the results of the Kruskal-Wallis test conducted on enrolled pre-service teaching programmes (2015 to 2017) to compare the self-efficacy and teacher efficacy beliefs of pre-service teachers based on the teaching programme they were enrolled in.



Summary of the p-values for Enrolled Pre-service Teaching Programmes (2015 to 2017)

	2015				2016				2017						
	Z	Mdn	W	SD	Z	Mdn	W	SD	z	Mdn	W	SD	KW	đ	Significant diff
	Foundation phase (FP)/Early childhood development (ECD)														
Self-Efficacy Items Average	105	5.78	5.72	0.76	45	5.89	5.86	0.63	160	5.89	5.89	0.54	2.11	.349	No
Teacher Efficacy Items Average	105	6.17	6.03	0.84	45	6.17	6.20	0.52	160	6.17	6.09	0.58	0.82	.663	No
	Intermediate phase (IP)														
Self-Efficacy Items Average	32	5.56	5.30	1.11	17	6.00	5.92	0.54	107	5.88	5.77	0.70	7.43	.024	Yes
Teacher Efficacy Items Average	32	5.79	5.48	1.36	17	6.08	6.04	0.38	107	6.08	6.01	0.70	4.52	.105	No
		Senior phase (SP)													
Self-Efficacy Items Average	30	5.61	5.68	0.60	14	5.88	5.83	0.52	56	5.83	5.67	0.73	0.23	.891	No
Teacher Efficacy Items Average	30	5.88	5.85	0.83	14	5.92	5.95	0.56	56	5.95	5.87	0.71	0.06	.972	No
	Further education and training (FET)														
Self-Efficacy Items Average	108	5.89	5.70	0.78	60	5.78	5.76	0.66	266	5.67	5.62	0.69	2.98	.225	No
Teacher Efficacy Items Average	108	6.00	5.84	0.79	60	6.00	5.96	0.58	266	5.79	5.77	0.70	4.19	.123	No
				Enrolle	d pre-sei	vice teach	ing progra	amme not s	specified						
Self-Efficacy Items Average	25	5.78	5.60	0.65	20	5.78	5.68	0.60	89	5.78	5.49	0.80	0.55	.759	No
Teacher Efficacy Items Average	25	6.00	5.89	0.55	20	6.08	5.96	0.74	89	5.75	5.75	0.72	2.22	.329	No


Table 4.30 indicates that there was only a statistically significant difference between the self-efficacy items of pre-service teachers for the Intermediate phase (IP) for 2015 to 2017. Since there was a statistically significant difference for the self-efficacy items for the IP teaching programme enrolled for, an ad-hoc test (i.e., Mann-Whitney tests) was run to test pairwise for the IP phase as indicated in Table 4.31. Table 4.31 also includes the Holm correction (Aickin & Gensler, 1996) with the indicated adjusted *p*-values.

#### Table 4.31

Summary of Ad-hoc Pairwise Test for Intermediate Phase (IP) Self-Efficacy of Pre-Service Teachers

Pairwise	U	р	Adjusted	Conclusion
comparison			<i>p</i> -value	
2015 vs 2016	150.00	.010	.030	The median of 6.00 for 2016 is statistically
				significantly higher than the median of 5.56
				for 2015
2015 vs 2017	1251.00	.021	.042	The median of 5.88 for 2017 is statistically
				significantly higher than the median of 5.56
				for 2015
2016 vs 2017	791.50	.390	.390	No statistically significant difference

As depicted in Table 4.31, the median for IP self-efficacy in 2016 (6.00) was statistically significantly higher than the median for 2015 (5.56) when comparing 2015 vs 2016, while the median for 2017 (5.88) was statistically significantly higher than the median for 2015 (5.56) when comparing 2015 vs 2017. Therefore, the results in Table 4.31 implied that preservice teachers in the IP phase had a significantly lower median for self-efficacy in 2015.

#### 4.6 CONCLUSION

In Chapter 4, I presented the results of the current study based on a statistical analysis of the data given the establishment of reliability and validity of the extant data available. I relied on the scores obtained from descriptive statistics and the Wilcoxon signed-rank test (i.e., comparison of self-efficacy and teacher efficacy beliefs), the Mann-Whitney tests (i.e., comparisons of gender) and the Kruskal-Wallis tests (i.e., comparison of pre-service teaching programme), to provide insight into the self-efficacy (i.e., confidence in recovery from setbacks) and teacher efficacy beliefs (i.e., confidence in teaching and behaviour management) of pre-service teachers within a challenged context.

I represented the within-case and cross-case results graphically and discussed the representations based on the main results. Below the main results in this chapter are listed:

• high self-efficacy (i.e., recovery from setbacks) and teacher efficacy beliefs (i.e., confidence in teaching and behaviour management) for pre-service teachers in a



challenged context, with teacher efficacy beliefs significantly higher (in-case and cross-case) than self-efficacy;

- female pre-service teachers' self-efficacy and teacher efficacy beliefs for Case 3 (2017) was significantly higher than male pre-service teacher self-efficacy and teacher efficacy beliefs, while male pre-service teachers' cross-case comparison demonstrated significantly higher self-efficacy during Case 2 (2016) compared to Case 1 (2015) and Case 3 (2017);
- pre-service teachers in the Foundation phase (FP)/Early childhood development (ECD) demonstrated significantly higher self-efficacy than the Further education and training (FET) pre-service teachers; Foundation phase (FP)/Early childhood development (ECD) and Intermediate phase (IP) pre-service teachers demonstrated significantly higher teacher efficacy beliefs than the Further education and training (FET) pre-service teachers (in-case comparison); Intermediate phase (IP) pre-service teachers demonstrated a significant lower self-efficacy in 2015.
- a significant positive correlation between the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context highlighting the possible interrelated relationship between the constructs.

In the final chapter, I revisit existing literature, as discussed in Chapter 2, to interpret the results given the chosen theoretical framework and determine findings to address research questions. Furthermore, I highlight the current study's potential contribution to the teacher resilience knowledge base and reflect on the current study's limitations. Finally, I craft recommendations for future research, training and development, and practice of preservice teachers and well-being agendas for educational policy.

---000----



# Chapter 5 Findings, Conclusions and Recommendations

### 5.1 INTRODUCTION

Through a post-positivist lens, the current quantitative study aimed to inform knowledge on teacher resilience in a challenged context by comparing the self-efficacy and teacher efficacy beliefs of final year pre-service teachers. To this end, the previous chapter presented the results of the within-case and cross-case secondary analysis of the FIRE teacher resilience data.

In this final chapter of the thesis, I interpret the results against the background of existing literature. In addition, I present conclusions by addressing the primary and secondary research questions and the hypotheses that guided the current study. Next, I revisit the conceptual framework, as discussed in Chapter 2, and highlight insights into teacher resilience related to pre-service teachers in a challenged context, in relation to the findings. I then reflect on the challenges of the current study. Finally, I propose suggestions for training and development, policy development and practice, and future research on teacher resilience in a challenged context. Figure 5.1 presents a flow chart showing the organisational overview of Chapter 5, after which I commence the chapter by providing an overview of what has been presented in this thesis.



#### Figure 5.1

Flow Chart Showing the Organisation of Chapter 5



# 5.2 OVERVIEW OF THE PRECEDING CHAPTERS

#### 5.2.1 INTRODUCTION

In Section 5.2, I provide an overview of what has been presented in the current study. The overview aims to provide a synopsis of the preceding chapters.



### 5.2.2 CHAPTER 1: INTRODUCTION AND OVERVIEW OF THE STUDY

The introduction to this study contained the background, rationale and the purpose for the current study titled, "*Pre-service teacher self-efficacy and teacher efficacy in a challenged education context*". Chapter 1, furthermore, introduced the formulated research questions and hypotheses which guided the current study. Subsequently, I clarified key concepts and presented the theoretical framework (i.e., Social Cognitive Theory). I also briefly described the paradigmatic perspectives and the methodological choices incorporated, standards or rigour employed and provided an overview of ethical considerations. An outline of the thesis chapters and results obtained concluded Chapter 1.

## 5.2.3 CHAPTER 2: LITERATURE REVIEW

Limited research is available on *quantitative methodology research* for *pre-service teacher resilience* in a challenged context given a Global South discourse. Therefore, in Chapter 2, I discussed the existing literature consulted, focusing on resilience, teacher resilience, self-efficacy, teacher efficacy, teacher education and pre-service teachers within a challenged setting given that teacher resilience, as a scholarly domain, is an emerging field with limited empirical research on resilience in teachers. Therefore, this study aimed to generate knowledge by quantitatively comparing pre-service teachers' self-efficacy and teacher efficacy beliefs in a severely challenged context to inform the teacher resilience field in a Global South space. I concluded Chapter 2 by explaining the conceptual framework employed in the current study.

#### 5.2.4 CHAPTER 3: RESEARCH DESIGN, METHODOLOGY AND STRATEGIES

In Chapter 3, I described the empirical investigation I conducted. I discussed and justified my decisions to take a post-positivist, quantitative research stance and utilise a comparative case study design given extant data within the teacher resilience context. Throughout, I related my choices to the research questions, hypotheses and purpose of the current study. Subsequently, I included a description of data selection based on the availability of extant data obtained from the FIRE Research Project. For data analysis, I specifically focused on explaining which descriptive and inferential statistical procedures were performed on extant data obtained from the FIRE Teacher Resilience Measure. I concluded the chapter by considering the standards of rigour employed and ethical considerations I adhered to for quality assurance.



## 5.2.5 CHAPTER 4: RESEARCH RESULTS OF THE STUDY

In Chapter 4, I posed the results I obtained as background to the discussion in this chapter. The statistical results derived from quantitative secondary data analysis were presented and explained for the purpose of this study. Descriptive and inferential statistics were employed to compare final year pre-service teachers' self-efficacy and teacher efficacy beliefs in a challenged context. The results serve as background to the findings and conclusions put forward in this chapter against the framework of existing literature and the conceptual framework that guided the investigation.

# 5.3 INTERPRETING RESULTS AGAINST THE BACKGROUND OF EXISTING TEACHER RESILIENCE LITERATURE

#### 5.3.1 INTRODUCTION

Section 5.3 focuses on a discussion of the results derived from the current study compared to existing literature and previous related studies in the education research field. Therefore, the section reflects on the results, as reported in Chapter 4, to provide findings. I discuss the implications of the results in terms of supporting or contracting existing literature. I also attempt to highlight silences in the data and present possible novel insights based on the results. Even though I presented the results, based on the secondary data analysis obtained from the FIRE Teacher Resilience Measure in Chapter 4, I reflect on the findings in this chapter to provide a comprehensive description.

# 5.3.2 RESULTS THAT SUPPORT OR CONTRADICT EXISTING LITERATURE

#### 5.3.2.1 The self-efficacy and teacher efficacy beliefs of pre-service teachers

The current study found that the high indication (within-case and cross-case) of pre-service teachers' self-efficacy and teacher efficacy beliefs contribute to their teacher resilience in a challenged context. Final year pre-service teachers in the current study were thus able to employ specific internal traits as a protective resource. Other studies similarly found that confidence in recovery from setbacks in Australia and Europe (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018, 2020; Wosnitza et al., 2018) and confidence in teaching and behaviour management (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018) were needed for teacher resilience.

Yokus (2015) likewise found that resilient traits of pre-service teachers were high in Turkey while Ee and Chang (2010) noted that pre-service teachers were not adequality prepared to deal with inevitable setbacks in Singapore. In addition, supporting findings of previous research in a challenged context (Bosch, 2020; Coetzee et al., 2017; Ebersöhn,



2014; Ebersöhn et al., 2020; Mansfield et al., 2018) indicated that teachers seem able to identify, mobilise and navigate intrapersonal resilience-enabling pathways (i.e., protective traits such as self-efficacy and teacher efficacy) to resile despite severe adversity and an unstable education system.

The confidence in recovery from setbacks, pre-service teachers reported in the current study (within-case and cross-case), is supported by literature (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018, 2020; Wosnitza et al., 2018) in Australia and Europe. The current study (within-case and cross-case) also denoted the presence of teacher efficacy beliefs as an important protective factor for teacher resilience supporting previous teacher resilience research in high-income countries (Beltman et al., 2011, 2018; Day & Gu, 2013; Gu & Day, 2007; Peixoto et al., 2018; Pendergast et al., 2011; Price et al., 2012; Sammons et al., 2007; Yada et al., 2021) and also in lower-income contexts (Bosch, 2020; Ebersöhn et al., 2020; Gu & Li, 2013; Hong, 2012; Mansfield et al., 2018).

Beltman et al. (2018) found that pre-service teachers in Australia reported similar levels of confidence in recovery from setbacks and confidence in teaching and behaviour management. On the other hand, Peixoto et al. (2018) indicated relatively higher confidence in recovery from setbacks than confidence in teaching and behaviour management of pre-service teachers in Germany, Ireland, Malta and Portugal. In the current study, the teacher efficacy of pre-service teachers is significantly higher (within-case and cross-case) than pre-service teachers' self-efficacy in a challenged context. Therefore, final year pre-service teachers are relatively more confident in teaching and behaviour management than in dealing with setbacks in schools.

The current study consequently provides novel data on the presence of significantly higher teacher efficacy for pre-service teachers in a challenged context, as discussed in Section 5.3.4. The significant result in the current study is supported by previous literature indicating the significance of teacher efficacy for pre-service teachers' confidence in recovery from setbacks in high-income countries (Morgan, 2011; Peixoto et al., 2018).

# 5.3.2.2 Gender differences between the self-efficacy and teacher efficacy beliefs of pre-service teachers

The current study (within-case and cross-case) reported high self-efficacy and teacher efficacy for gender (i.e., male and female pre-service teachers in a challenged context) supporting earlier studies in Australia and Europe (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018; Wosnitza et al., 2018). Case 1, Case 2 and cross-case comparison found no statistically significant difference for the recovery from setbacks for gender. These results support previous pre-service teachers' research in Australia (Beltman et al., 2018; Fischer et al., 2018) and Turkey (Çelik et al., 2018), indicating the resilient traits for male



and female teachers did not differ significantly. On the other hand, the current study (Case 3) found that female pre-service teachers' confidence in recovery from setbacks was significantly higher than males, corresponding with findings from another study in Turkey (Yokus, 2015). Male self-efficacy (cross-case comparison) was significantly higher in 2016 (Case 1) compared to 2015 (Case 1) and 2017 (Case 3). Pre-service male teachers' confidence in recovery from setbacks was highest during extreme systemic disruptions such as the #Fees-Must-Fall movement in a challenged context. Literature (Ee & Chang; 2010; Hartman et al., 2009; Pareek & Rathore, 2016; Peglar & Reker, 2008; Petersen, 2014; Yokus, 2015) substantiate that protective factors enabling resilience may vary across gender.

As with self-efficacy, teacher efficacy beliefs for pre-service teachers indicated no statistically significant difference for Case 1, Case 2 and cross-case comparison for gender. These results are consistent with literature for a challenged context (Murshidi et al., 2006; Ngidi & Ngidi, 2019) and are also highlighted in high-income contexts (Beltman et al., 2018; Klassen & Durksen, 2014; O'Neill & Stephenson, 2012b; Pfitzner-Eden, 2016a; Tschannen-Moran & Woolfolk Hoy, 2007) implying that gender does not influence the teacher efficacy of pre-service teachers. Contradictory, the current study (Case 3) reported a significantly higher teacher efficacy for female teachers compared to their male counterparts for Case 3. Özdemir (2008) and Üstüner (2017) also reported higher teacher efficacy beliefs for female pre-service teachers compared to male pre-service teachers in Turkey.

# 5.3.2.3 Enrolled pre-service teacher programme differences between the selfefficacy and teacher efficacy beliefs of pre-service teachers

The current study (within-case and cross-case) reported high self-efficacy and teacher efficacy for all enrolled pre-service teacher programmes (i.e., FP/ECD, IP, SP and FET) which support existing literature in privileged settings (Beltman et al., 2018; Brown et al., 2015; Moulding et al., 2014; O'Neill & Stephenson, 2012b; Peixoto et al., 2020; Woolfolk Hoy & Spero, 2005). Pre-service teachers' confidence in recovery from setbacks, however, was significantly higher for primary in comparison to secondary pre-service teachers for Case 3 verifying results from previous studies in developed (Bouillet et al., 2014; Perlman & Pearson, 2012) and developing countries (Kavita & Hassan, 2018) on teacher resilience differences between teachers at the primary and secondary level.

Although the teacher efficacy for all enrolled pre-service teacher programmes was high, the primary pre-service teachers' teacher efficacy was significantly higher than the secondary teacher efficacy for Case 3. The significant result is in agreement with Woodcock (2011) reporting significantly lower teacher efficacy beliefs for secondary pre-



service teachers in comparison with their primary counterparts in Australia. In addition, although Pendergast et al. (2011) obtained no statistically significant difference between primary and secondary levels in Australia, the mean for pre-service primary teacher efficacy was highest.

# 5.3.2.4 The relationship between the self-efficacy and teacher efficacy beliefs of pre-service teachers

The current study found a statistically significant correlation between self-efficacy and teacher efficacy for pre-service teachers in a challenged context. Similar results have been noted for the relationship in countries such as Australia, Germany, Ireland, Malta and Portugal in studies with pre-service teachers while employing the scales utilised in the current study (Morgan, 2011; Peixoto et al., 2018).

These results indicated that teacher efficacy is a main predictor of teachers' confidence in recovery from setbacks. In Ireland and Germany, teacher efficacy was the strongest significant predictor for teacher recovery from setbacks (Peixoto et al., 2018). Additional literature from high-income countries (Beltman et al., 2018; Ee & Chang, 2010; Gu & Day, 2013; Klassen et al., 2011; Peixoto et al., 2018, 2020; Pendergast et al., 2011; Thieman et al., 2014; Wosnitza et al., 2018) also highlighted the association between the internal enablers for pre-service teacher resilience.

#### 5.3.3 SILENCES IN THE DATA

Some silences were evident in the data of the current study. As indicated in Chapter 2, current literature emphasises that resilience is not regarded as a stable or singular individual capacity alone, but as a dynamic, interactive process across and between systems that are successfully adapting to risk (Gu, 2014; Masten, 2018). Resilience is thus depicted as adaptation enabling individuals to respond stoutly to risk leading to developmental outcomes, despite monumental threats to adaptation in the system (Masten, 2001, 2011, 2014, 2018). Accordingly, resilience as a complex process across an individual's life (Fletcher & Sarkar, 2013; Masten & Reed, 2005; Masten & Wright, 2010; Vanderbilt-Adriance & Shaw, 2008). The personal capacity to cope with a supportive environment; thus, a dynamic interrelationship within a social system, facilitate the ability to flourish collectively despite adversity (Ebersöhn, 2012; Ungar, 2010, 2011). A resilient disposition tends to be informed by intrapersonal resources and socio-ecological interaction with risk and protective (present or absent) mechanisms (Ungar, 2010; Ungar et al., 2013).



Regardless of the context, social ecologies may serve as protection even when assets are sparse, perhaps leading to non-normative or atypical nuanced pathways (i.e., unexpected positive outcomes) to resilience (Ungar, 2011). Previous literature on the influence of, for example, learner achievement (Caprara et al., 2006; Moulding et al., 2014; Tschannen-Moran & Johnson, 2011) and socioeconomic status or social background on pre-service teachers (Morgan, 2011; Moulding et al., 2014) was also not found in the results of the current study.

The current study did not present results on the exploration of systemic and contextual factors enabling or constraining resilience for pre-service teachers. Therefore, the current study could not contribute to teacher resilience knowledge by describing contextual resources. Furthermore, due to the cross-sectional nature of extant data available, any developmental trajectory regarding pre-service teachers' self-efficacy or teacher efficacy enabling teacher resilience in a challenged context was not evident in the results. Scholars (Brown et al., 2015; Clark, 2020; Klassen & Durksen, 2014; Mergler & Tangen, 2010; Pfitzner-Eden, 2016a; Swan et al., 2011) have noted differentiated developmental trajectories for pre-service teachers during initial teacher education as well as during the first year of teaching. Therefore, further research for development trajectories relating to teacher resilience is needed in a challenged context (see Section 5.7).

#### 5.3.4 NOVEL INSIGHTS INFORMING TEACHER RESILIENCE IN A CHALLENGED CONTEXT

Section 5.3.4 discusses novel insights based on results obtained from this study. The results of the current study provided novel insights into the self-efficacy and teacher efficacy beliefs of pre-service teachers related to teacher resilience in a challenged context.

The measure utilised within the current study (i.e., FIRE Teacher Resilience Measure) has been employed in countries such as Australia (Beltman et al., 2018), the Czech Republic (Wosnitza et al., 2018), Germany (Peixoto et al., 2018; Wosnitza et al., 2018), Ireland (Morgan, 2011; Peixoto et al., 2018; Wosnitza et al., 2018), Malta (Peixoto et al., 2018; Wosnitza et al., 2018) and Portugal (Peixoto et al., 2018, 2020; Wosnitza et al., 2018). Ebersöhn et al. (2020) found that teachers in South Africa demonstrated confidence in recovery from setbacks and, teaching and behaviour management in a challenged context. However, up to date, the FIRE Teacher Resilience Measure has not been used with pre-service teachers in a Global South place such as South Africa on a large scale. Research within a Global South challenged education space, such as South Africa, thus provided a much-needed lens outside developed countries by delivering evidence on the use of existing teacher resilience scales and yielding comparative results for future studies.



The presence of intrapersonal resilience-enabling pathways such as self-efficacy and teacher efficacy in a challenged context, therefore, inform knowledge in the Global South of how teachers resile in the profession despite chronic and cumulative risk factors. The current study's results also found that the teacher efficacy of final year pre-service teachers in a challenged context is significantly higher than their self-efficacy. Thus, pre-service teachers feel more confident in their teaching and behaviour management than in their recovery from setbacks in schools. In addition, the description of final year pre-service teachers' confidence in recovery from setbacks in schools and confidence in teaching and behaviour management provided insights into the current self-perceived professional status of pre-service teachers in a challenged context. Although literature (Pendergast et al., 2011) indicated the pre-service teachers are prone to overestimate their abilities, it is crucial to explore the beliefs of beginning teachers entering the profession (Caprara et al., 2006; Tschannen-Moran & Woolfolk Hoy, 2007).

The varied results on gender differences in the current study may indicate that further research is needed regarding protective factors fostering resilience for male and female pre-service teachers in a challenged context.

Limited research is also available about significant differences for enrolled teaching programmes (i.e., FP/ECD, IP, SP and FET) for pre-service teachers during initial teacher training. Although literature (Kavita & Hassan, 2018; OECD, 2018; Perlman & Pearson, 2012) underlines differences between primary and secondary school teachers, the current study provided novel insight regarding the significant difference between pre-service teacher self-efficacy and teacher efficacy that is lower for secondary school pre-service teachers in a challenged context. The significant differences may highlight the need for differentiated teacher policy settings to effectively address specific challenges and needs of primary and secondary school teachers (OECD, 2018) in spaces of high adversity.

The result provides important insights to guide the development of teacher education training to respond adequately to the development of pre-service teachers in a challenged context. The identification and presence of intrapersonal resilience-enabling pathways (i.e., self-efficacy and teacher efficacy beliefs) can be further developed during initial teacher education programmes. Therefore, the results may guide international teacher training for developing countries.

The current study also contributed to teacher resilience knowledge by establishing the reliability and validity of the Teacher Resilience scale and Teacher Efficacy scale, of the FIRE Teacher Resilience Measure, for pre-service teachers in a challenged context. The results provide an opportunity to improve or amend the questionnaire to render it better suited to a South African context, as discussed in Section 5.7.4.4.

Both male and female self-efficacy and teacher efficacy scored higher during 2016 (Case 2). During 2016 the #Fees-Must-Fall movement disrupted activities at the University



of Pretoria. Since 2016 denoted the smallest sample size, caution should be taken when interpreting results. Nonetheless, the higher self-efficacy and teacher efficacy during this time, although not significant, may indicate how pre-service teachers positively respond and adapt to high risk. In addition, this positive response may also be indicative of a system (including HEI and staff) that rallied to provide additional support to students (including pre-service teachers) during adversity. Finally, novel insights are proposed as an evidence-based theoretical framework for teacher resilience within a challenged context, as discussed in Section 5.5.

# 5.4 ADDRESSING THE RESEARCH QUESTIONS AND HYPOTHESES

#### 5.4.1 INTRODUCTION

In this section, I address the research questions and hypotheses posed for the current study. Creswell and Creswell (2018, p. 220) stated that "an interpretation in quantitative research means that the researcher draws conclusions from the results for the research questions, hypotheses, and the larger meaning of the results". Therefore, the secondary research questions are addressed and conclusions drawn in terms of the formulated hypotheses I set out to test. These conclusions are followed by a discussion of the primary research question based on the findings I obtained before linking the results to the conceptual framework that guided me in undertaking the current study (see Section 5.5).

#### 5.4.2 FINDINGS: SECONDARY RESEARCH QUESTIONS AND HYPOTHESES

In the following sections, I revisit the secondary research questions and related hypotheses. In this process, I attempt to draw conclusions, which in turn, addresses the primary research question.

# 5.4.2.1 Findings based on the comparison of self-efficacy and teacher efficacy beliefs of final year pre-service teachers

#### Secondary research question 1

How do self-efficacy and teacher efficacy beliefs of final year pre-service teachers within a challenged education context (within-case and cross-case) compare?

To compare the self-efficacy and teacher efficacy beliefs of pre-service teachers within a challenged context, descriptive and inferential statistics for each of the scales (i.e., Teacher Resilience scale [Morgan, 2011] and Teacher Efficacy scale [Morgan, 2011; Peixoto et al., 2018]) from the FIRE Teacher Resilience Measure were calculated.



Final year pre-service teachers *consistently demonstrate high self-efficacy and teacher efficacy beliefs* for within-case (Case 1 [2015], Case 2 [2016] and Case 3 [2017]) and cross-case (2015–2017). The results indicated that final year pre-service teachers, within a challenged context, are confident of their recovery from setbacks in schools (i.e., operationalised as self-efficacy in the current study) and feel confident in their teaching and behaviour management (i.e., operationalised as teacher efficacy beliefs in this study). Ee and Chang (2010) found that pre-service teachers did not feel sufficiently equipped to deal with inevitable setbacks in Singapore. While other high-income country findings (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018, 2020; Wosnitza et al., 2018) confirm the current results indicating high resilience traits in pre-service teachers. This result was also found in studies given a challenged context (Bosch, 2020; Coetzee et al., 2017; Ebersöhn, 2014; Ebersöhn et al., 2020; Mansfield et al., 2018; Yokus, 2015).

The formulated statistical hypotheses of the population difference for the current study included:

- H<sub>0</sub>: Median<sub>dif</sub> = 0 (median population difference between self-efficacy and teacher efficacy beliefs [within-case and cross-case] does not differ significantly from zero).
- H<sub>a</sub>: Median<sub>dif</sub> ≠ 0 (median population difference between self-efficacy and teacher efficacy beliefs [within-case and cross-case] differs significantly from zero).

Table 5.1 presents an overview of the statistical hypotheses related to assumptions based on the population difference between the self-efficacy and teacher efficacy of pre-service teachers in a challenged context based on the hypothesis testing performed on data.

#### Table 5.1

Overview of the Statistical Hypotheses Related to Assumptions Based on the Population Differences Between the Self-Efficacy and Teacher Efficacy of Pre-Service Teachers

	Year	Significant difference between Self-Efficacy and Teacher Efficacy of pre-service teachers	$H_0$ vs $H_a$
Within- case	2015 ( <i>N</i> = 313)	Yes: Teacher efficacy beliefs are significantly higher than self-efficacy	$H_0$ is rejected
	2016 ( <i>N</i> = 169)	Yes: Teacher efficacy beliefs are significantly higher than self-efficacy	$H_0$ is rejected
	2017 ( <i>N</i> = 711)	Yes: Teacher efficacy beliefs are significantly higher than self-efficacy	$H_0$ is rejected
Cross- case	2015–2017 ( <i>N</i> = 1,193)	Yes: Teacher efficacy beliefs are significantly higher than self-efficacy	$H_0$ is rejected



As indicated in Table 5.1, the findings showed a significant difference between preservice teachers' self-efficacy and teacher efficacy beliefs (within-case and cross-case) in a challenged context. *Teacher efficacy beliefs*, of final year pre-service teachers, were *significantly higher* than pre-service teachers' *self-efficacy when evaluating within-case and cross-case results*. Pre-service teachers in a challenged context may thus feel more confident in teaching and behaviour management (teacher efficacy) than in recovery from setbacks in schools (self-efficacy) which is supported by previous studies in Ireland and Europe (Morgan, 2011; Peixoto et al., 2018).

# 5.4.2.2 Findings based on the demographic information of pre-service teachers' self-efficacy and teacher efficacy beliefs

## Secondary research question 2

What are the self-efficacy and teacher efficacy beliefs of pre-service teachers within a challenged education context based on demographic information (within-case and cross-case)?

For all biographical variables<sup>58</sup>, descriptive statistics such as means, medians, standard deviations, frequencies and percentages were computed (see Chapter 4). Inferential statistics were performed for all biographical variables except for age and language. For a detailed discussion on the exclusion of age and language, see Section 4.2.3 and Section 4.2.5, respectively. Therefore, to address the second research question, descriptive and inferential statistics for each of the scales were calculated for gender (i.e., male and female) and enrolled pre-service teaching programmes (i.e., FP/ECD, IP, SP, FET and not specified/other) and reported along with an examination of the differences between groups.

Like results for previous research in developed countries (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018; Wosnitza et al., 2018), the findings demonstrated selfefficacy and teacher efficacy beliefs of both *male and female final year pre-service teachers were consistently high*. The high self-efficacy and teacher efficacy beliefs indicated that both male and female pre-service teachers, in a challenged context, is confident to recover from setbacks and feel confident in teaching and behaviour management.

Table 5.2 provides an overview of the statistical hypotheses related to assumptions based on the demographic difference between pre-service teachers' self-efficacy and teacher efficacy beliefs.

<sup>&</sup>lt;sup>58</sup> Demographic information (as depicted in Section 4.2) available from the extant data, based on the completed FIRE Teacher Resilience Measure (see Table 1.5), included final year pre-service teachers' age, gender, language and enrolled pre-service teaching programmes.



#### Table 5.2

Overview of the Statistical Hypotheses Related to Assumptions Based on the Demographic Differences Between the Self-Efficacy and Teacher Efficacy of Pre-Service Teachers in a

#### Challenged Context

	Year	p	Gender			Enrolled pre-service teaching programmes				
			Self-Efficacy	$H_0$ vs $H_a$	Teacher Efficacy	$H_0$ vs $H_a$	Self-Efficacy	$H_0$ vs $H_a$	Teacher Efficacy	$H_0$ vs $H_a$
Within- case	2015 ( <i>N</i> = 313)	Adjusted <i>p</i> -values	No statistically significant difference	$H_0$ is not rejected	No statistically significant difference	H <sub>0</sub> is not rejected	No statistically significant difference	<i>H</i> ₀ is not rejected	No statistically significant difference	<i>H</i> ₀ is not rejected
	2016 ( <i>N</i> = 169)	Original <i>p</i> -values	No statistically significant difference	$H_0$ is not rejected	No statistically significant difference	<i>H</i> <sup>0</sup> is not rejected	No statistically significant difference	<i>H</i> ₀ is not rejected	No statistically significant difference	<i>H₀</i> is not rejected
	2017 ( <i>N</i> = 711)	Adjusted <i>p</i> -values	Yes • Females higher than males ↑♀	<i>H₀</i> is rejected	Yes • Females higher than males ↑♀	<i>H₀</i> is rejected	Yes • ↑FP/ECD vs FET • ↑FP/ECD vs Not enrolled	<i>H</i> ₀ is rejected	Yes • ↑FP/ECD vs FET • ↑FP/ECD vs Not enrolled • ↑IP vs FET • ↑IP vs Not enrolled	<i>H</i> ₀ is rejected
Cross- case	2015–2017 ( <i>N</i> = 1,193)	Adjusted <i>p</i> -values	<ul> <li>Yes: Male</li> <li>2016 higher than 2015</li> <li>2016 higher than 2017</li> <li>= 2016 ↑</li> </ul>	<i>H₀</i> is rejected	No statistically significant difference (Male and Female)	<i>H</i> <sub>0</sub> is not rejected	Yes: IP • IP - 2016 higher than 2015 - 2017 higher than 2015 = 2015 ↓	<i>H</i> ₀ is rejected	No statistically significant difference	<i>H</i> <sub>0</sub> is not rejected



As indicated in Table 5.2, a significant statistical difference was found for the cross-case analysis (2015–2017) for male self-efficacy, which suggests that *pre-service male teacher self-efficacy was significantly higher in 2016* compared to 2015 and 2017. Although the result should be interpreted with caution due to the smaller sample of male pre-service teachers, the difference may be indicative of male pre-service teachers relying on different individual protective factors, such as inner strengths in taking charge and control to nature resilience during significant disruptions (e.g., the #Fees-Must-Fall movement) (Hartman et al., 2009; Peglar & Reker, 2008).

Statistically significant differences for Case 3 indicated that 2017 female pre-service teacher self-efficacy and teacher efficacy beliefs are higher than male pre-service teacher self-efficacy and teacher efficacy. This result supports existing studies for pre-service teachers indicating higher female resilience traits in other developing countries (Özdemir, 2008; Üstüner, 2017; Yokus, 2015).

The literature in developed (Ee & Chang, 2010; Hartman et al., 2009) and developing (Pareek & Rathore, 2016; Petersen, 2014; Yokus, 2015) countries emphasised that female and male teachers can utilise different intrapersonal resilience-enabling pathways to foster resilience. Additional inquiry about the result for a challenged context would have been interesting without the limitation of extant data.

The differential result found on *no gender differences* (Case 1 and Case 2) is also highlighted in high-income countries (Beltman et al., 2018; Fischer et al., 2018; Klassen & Durksen, 2014; O'Neill & Stephenson, 2012b; Pfitzner-Eden, 2016a; Tschannen-Moran & Woolfolk Hoy, 2007). Lower-income countries such as Turkey (Çelik et al., 2018), Malaysia (Murshidi et al., 2006) and South Africa (Ngidi & Ngidi, 2019; Petersen, 2014) also found no gender differences.

The within-case and cross-case comparison for *enrolled pre-service teaching programmes also denoted high self-efficacy and teacher efficacy beliefs* for final year preservice teachers within a challenged context. Therefore, all enrolled pre-service teaching programmes had a high recovery from setbacks and confidence in teaching and behaviour management.

Within-case comparison for Case 1 (2015) and Case 2 (2016) resulted in *no statistically significant differences* for enrolled pre-service teaching programmes. However, for Case 3 (2017), a *statistically significant difference* between the self-efficacy and teacher efficacy beliefs of pre-service teachers was indicated between primary and secondary pre-service teachers. Secondary school teachers may have higher stress and lower levels of supportive learning context (Kavita & Hassan, 2018; Molina et al., 2017; Perlman & Pearson, 2012; Zuma et al., 2016).

Results highlighting higher intrapersonal resilience-enabling factors for pre-service teachers enrolled in the primary phase is seen in the literature for developed countries (Bouillet



et al., 2014; Perlman & Pearson, 2012; Woodcock, 2011) and in a challenged context (Kavita & Hassan, 2018; Molina et al., 2017; Zuma et al., 2016). For the cross-case comparison (2015–2017), the only statistically significant difference for enrolled pre-service teaching programmes resulted from the IP phase. *IP pre-service teachers demonstrated a significantly lower recovery from setbacks* (i.e., self-efficacy) in 2015 compared to 2016 and 2017. Given the constraints of the available secondary data, further exploration of this result was not possible.

In summary, significant results for pre-service teacher demographic information were found for the self-efficacy and teacher efficacy for gender (Case 3), male self-efficacy (crosscase comparison), self-efficacy and teacher efficacy for enrolled pre-service teaching programmes (Case 3) and self-efficacy for IP enrolled programme phase (cross-case comparison). Conclusions in terms of formulated hypotheses are highlighted in Section 5.4.3.

# 5.4.2.3 Findings based on the relationship between the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context

#### Secondary research question 3

To what extent is there a relationship between self-efficacy and teacher efficacy beliefs among pre-service teachers in a challenged education context (within-case and cross-case)?

The association between self-efficacy and teacher efficacy beliefs responses were analysed to examine the relationship between the self-efficacy and teacher efficacy of pre-service teachers as measured by two scales to address the third secondary research question. The following statistical hypotheses of the population correlation coefficient were formulated testing for the relationship between the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context:

- *H*<sub>0</sub>: There is no statistically significant correlation between self-efficacy and teacher efficacy beliefs.
- *H*<sub>a</sub>: There is a statistically significant correlation between self-efficacy and teacher efficacy beliefs.

The findings indicate a significant positive correlation between the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context. The conclusion was based on the results discussed in Chapter 4 (see Section 4.5). The correlation indicated a significant positive relationship between final year pre-service teachers, within a challenged context, for self-efficacy and teacher efficacy beliefs. As discussed in Chapter 3, correlation does not imply causation (Field, 2018; Roni et al., 2020), but is intended to reflect the degree to which self-efficacy and teacher efficacy beliefs are associated. The results, therefore, imply that preservice teachers' confidence to recover from setbacks are significantly related to their confidence in teaching and behaviour management in a challenged context.



The association between the self-efficacy and teacher efficacy beliefs for teachers has been established in previous studies. Studies in Australia (Beltman et al., 2018); Ireland (Morgan, 2011), Europe (Peixoto et al., 2018; Wosnitza et al., 2018) and South Africa (Ebersöhn et al., 2020) also employed similar scales with comparable outcomes as the current study. Other literature in high-income countries (Ee & Chang, 2010; Gu & Day, 2013; Klassen et al., 2011; Peixoto et al., 2020; Pendergast et al., 2011; Thieman et al., 2014) highlighted intrapersonal resources associated with teacher resilience. The established relationship between the self-efficacy and teacher efficacy can therefore enable pre-service teachers to resile despite significant risk factors (Ebersöhn et al., 2020).

#### 5.4.3 FINDINGS: PRIMARY RESEARCH QUESTION

This section addresses the primary research question by drawing conclusions from the secondary research questions and hypotheses. Therefore, the primary research question is highlighted, namely:

# How can insight into the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged education context inform knowledge on teacher resilience?

I have argued that there is a need for an evidence-based theoretical framework for teacher resilience within a challenged context given (i) the dearth of quantitative teacher resilience studies, (ii) knowledge overall on resilience in teachers, and (iii) specifically pre-service teachers in (iv) particularly developing countries. In addition, the current study generated global teacher resilience discourses from Africa to contribute to the lack of scientific knowledge from this region.

The respondents in the sample (2015–2017) were 1,193 final year pre-service teachers (i.e., 1,193 completed the FIRE Teacher Resilience Measure) at an HEI in South Africa (seen as an exemplar of a challenged context). Most respondents were female, and almost all respondents were fluent in English, while approximately half were fluent in Afrikaans and the majority were not fluent in African languages. Most respondents were enrolled in the FET and FP/ECD pre-service teacher programmes. The current study's reliability, validity, and statistical power were established to enhance the generalisation of findings to final year pre-service teachers at the University of Pretoria and contribute to the utility of an existing scale measuring intrapersonal resilience-enabling pathways to teacher resilience in a challenged context.

The findings demonstrated that pre-service teachers' self-efficacy and teacher efficacy beliefs in a challenged context are high (within-case and across case). Final year pre-service teachers in the current study were thus able to employ specific internal traits as a protective resource indicated in previous high-income countries (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018, 2020; Wosnitza et al., 2018) and in a challenged context (Bosch, 2020; Coetzee et al., 2017; Ebersöhn, 2014; Ebersöhn et al., 2020; Mansfield et al., 2018; Yokus,



2015). Therefore, intrapersonal protective factors related to resilience (Ebersöhn, 2014; Ebersöhn et al., 2020; Ee & Chang, 2010; Peixoto et al., 2018; Tschannen-Moran & Woolfolk Hoy, 2001) may enable positive adaptation assisting pre-service teachers to recover from setbacks in a challenged context. The confidence in recovery from setbacks found in pre-service teachers in the current study may counteract adversity or lead to resilience within a high-risk ecology, assisting teachers to navigate obstacles successfully when facing challenging working conditions (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018, 2020; Waddell, 2007; Wosnitza et al., 2018).

The presence of teacher efficacy beliefs as a vital protective factor for teacher resilience (Beltman et al., 2011, 2018; Day & Gu, 2013; Gu & Day, 2007; Hong, 2012; Morgan, 2011; Peixoto et al., 2018; Pendergast et al., 2011; Price et al., 2012; Sammons et al., 2007; Yada et al., 2021) specifically in a challenged context (Bosch, 2020; Ebersöhn et al., 2020; Gu & Li, 2013; Jackson & Rothmann, 2005) demonstrated pre-service teachers' confidence regarding the teaching domain (Pfitzner-Eden, 2016a; Raath & Hay, 2016), engagement with teaching tasks (Lemon & Garvis, 2016; Morgan, 2011) and influence on learning and behaviour of learners (Beltman et al., 2018; Klassen et al., 2011; Kyriakides et al., 2013; Moulding et al., 2014; Ngidi & Ngidi, 2019; O'Neill & Stephenson, 2012b; Özdemir, 2008; Swan et al., 2011). Teacher efficacy plays a major role for pre-service teachers in overcoming challenges they encounter in their work (Tournaki et al., 2009).

The significantly higher teacher efficacy (Morgan, 2011; Peixoto et al., 2018) of preservice teachers in the current study further indicated the importance of teacher efficacy for pre-service teachers' confidence in recovery from setbacks in a challenged context. As a result, pre-service teachers, in spaces of high adversity, may pursue goals as well as challenges, rebound from setbacks, persist or demonstrate grit, hardiness as well as optimism when faced with obstacles (Bandura, 1986, 1993, 1996, 1997; Fredrickson et al., 2003; Hewitt et al., 2017; Mansfield et al., 2018; Perkins-Gough, 2013). Therefore, even if there is adversity, self-efficacy and teacher efficacy can act as protective resources when pre-service teachers believe they can recover from setbacks and succeed (Hewitt et al., 2017).

Demographic statistical differences for gender provided mixed results (within-case and cross-case comparison) for the self-efficacy and teacher efficacy beliefs of pre-service teachers but compared effectively with literature regarding varied findings on pre-service teacher gender profiles in developed (Beltman et al., 2018; Fischer et al., 2018; Klassen & Durksen, 2014; O'Neill & Stephenson, 2012b; Pfitzner-Eden, 2016a) and developing (Çelik et al., 2018; Murshidi et al., 2006; Özdemir, 2008; Üstüner, 2017; Yokus, 2015) countries. On the other hand, the significant results for the enrolled pre-service teacher programme demographic underlined existing research (Bouillet et al., 2014; Perlman & Pearson, 2012; Woodcock, 2011) for higher intrapersonal resilience factors for primary phase pre-service teachers in a challenged context (Kavita & Hassan, 2018; Molina et al., 2017; Zuma et al., 2016).



Finally, the correlation analysis demonstrated a significant positive relationship between the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context. The current study's results thus signify the association of self-efficacy and teacher efficacy, which contribute to teacher resilience amongst pre-service teachers as with other studies in a challenged context (Ebersöhn et al., 2020) and high-income countries (Morgan, 2011; Peixoto et al., 2018). Therefore, based on the results from the completed FIRE Teacher Resilience Measure, pre-service teachers within a challenged context indicated self-efficacy (i.e., confidence in recovery from setbacks) and teacher efficacy beliefs (i.e., confidence in teaching and behaviour management) as significant protective internal factors for enabling teacher resilience in spaces of high social disadvantage.

The findings from the primary research question thus informed knowledge on teacher resilience in a challenged context through quantitative methodology. The current study indicated the utility of existing teacher resilience scales and provided comparable results for future studies in a challenged context worldwide. In addition, the findings provided much-needed evidence for pre-service teacher resilience in a challenged context. Furthermore, the current study generated knowledge for the Global South and African context, which lack teacher resilience research in comparison with the Global North and European countries.

#### 5.5 THEORISING TEACHER RESILIENCE FINDINGS IN A CHALLENGED CONTEXT

#### 5.5.1 INTRODUCTION

This section revisits the conceptual framework to situate the current study's findings within an evidence-based framework for teacher resilience in a challenged context. The insights for teacher resilience inform quantitative comparison of the self-efficacy and teacher efficacy beliefs of pre-service teachers in a disparate Global South place.

#### 5.5.2 SITUATING THE FINDINGS WITHIN THE CONCEPTUAL FRAMEWORK

In this section, I interpret the findings of the current study within the conceptual framework, as discussed in Chapter 2 (see Section 2.8), to generate meaning and reflect on the choice of the conceptual framework.

The findings of this study indicate the presence of *high internal protective resources* (i.e., self-efficacy and teacher efficacy beliefs) that buoyed pre-service teachers to resile despite chronic and cumulative risk factors, including an unstable education system (such as the *#*Fees-Must-Fall movement). As postulated by Bandura in the Social Cognitive Theory (SCT) (Bandura, 1986), pre-service teachers' behaviour was the result of a reciprocal caution between experiences which develops through interaction with the environment (Bandura, 1986, 1989a, 1989b; Peixoto et al., 2018; Schunk, 2008).



Based on Banduras' view of self-efficacy (Bandura, 1977, 1986, 2001), the pre-service teachers' self-perceived beliefs on how capable or prepared they feel to organise or accomplish a specific goal, action or task to manage prospective situations or obtain a valued outcome, is high. The pre-service teachers in the current study, thus seem to actively contribute to their development through utilising individual traits (e.g., self-efficacy and teacher efficacy) as health-sustaining protective resources (Block & Block, 1980; Block & Kremen, 1996; Johnson & Down, 2013; Letzring et al., 2005; Mansfield et al., 2016). Given Bandura's (1977, 1986, 2001) view on self-efficacy, it is thus likely that pre-service teachers had an enthusiastic, positive, multi-perspective, confident outlook on life with confidence in their ability to recover from setbacks and regulate emotions, thoughts and actions (Bandura, 1986, 1993, 1996, 1997; Ee & Chang, 2010; Hewitt et al., 2017; Perkins-Gough, 2013). These findings can link with pre-service teachers' grit and hardiness (Kobasa et al., 1982; Maddi et al., 2017; Perkins-Gough, 2013; Prince-Embury, 2010; Von Culin et al., 2014), sense of coherence (A. Antonovsky's [1979] salutogenic theory), positive emotions (Fredrickson et al.'s [2003] broaden-and-build theory) and their adaptive coping processes (Skinner & Zimmer-Gembeck, 2011; Willers et al., 2013).

The choice of incorporating Bandura's Social Cognitive view on self-efficacy (Bandura, 1977, 1986, 2001), instead of, for example, Rotter's theory of the Locus of Control [1966]), has been confirmed by numerous researchers (Cassidy, 2015; Chesnut, 2017; Duffin et al., 2012; Hewitt et al., 2017; Lemon & Garvis, 2016; Morgan et al., 2010; Moulding et al., 2014; O'Neill & Stephenson, 2012b; Pfitzner-Eden, 2016a, 2016b; Raath & Hay, 2016; Skaalvik & Skaalvik, 2010; Snowman & McCown, 2013). However, the theory has been criticised for ambiguous definitions, methodological deficiency and inadequate evaluation or methods to validate conclusions (Biglan, 1987; Borkovec, 1978; Eastman & Marzillier, 1984; Kazdin, 1978; Teasdale, 1978; Tryon, 1981). A valid self-efficacy appraisal (Chesnut, 2017) is also only possible if pre-service teachers accurately recognise their abilities. Furthermore, although Bandura (1977) assumed that high self-efficacy provides the ability to cope with potentially aversive events, the Social Cognitive Theory does not adhere to a placed based argument needed in a Global South place. Due to this limitation, I also incorporated Ebersöhn's (2014) *contextualised structural disparity lens* in the conceptual framework to contextualise teacher resilience and teacher efficacy beliefs within a severely challenged context.

Ebersöhn (2015) argued for place-based (e.g., Global South) research through contextual lenses to provide the necessary needed evidence for teacher resilience. Most resilience studies originate from North America or Europe without an understanding of resilience at a local contextual level (Ebersöhn, 2017; Johnson & Down, 2013; Shean, 2015). Therefore, the following section emphasises an evidence-based framework for teacher resilience in a challenged context given the empirical findings of the current study.



# 5.5.3 AN EVIDENCE-BASED FRAMEWORK FOR TEACHER RESILIENCE IN A CHALLENGED CONTEXT

Given the significant gaps in teacher resilience knowledge in a severely challenged context (Ebersöhn, 2014, 2016; Ebersöhn et al., 2020; Ebersöhn & Loots, 2017; Mansfield et al., 2018; Ngidi & Ngidi, 2019), the need for an evidence-based theoretical framework for teacher resilience in spaces of high challenge is evident. Therefore, in Figure 5.2, I present an *evidence-based framework for self-efficacy and teacher efficacy as intrapersonal resilience-enabling pathways for teacher resilience in a challenged context*.

#### Figure 5.2

An Evidence-based Framework for Self-efficacy and Teacher Efficacy as Resilience-enabling for Teacher

#### Resilience in a Challenged Context



The current study's findings provided novel insights into intrapersonal resilience-enabling pathways (i.e., self-efficacy and teacher efficacy of pre-service teachers) indicative of pre-service teachers resiling. The current study generated teacher resilience knowledge vital to discourses on protective resources (enablers) involved in processes relating to resilience while acknowledging the risk factors (constraints) (Ebersöhn, 2016).

The conceptual framework was thus adapted (see Figure 5.2) to indicate *pre-service teachers' high self-efficacy and teacher efficacy beliefs* as intrapersonal resilience-enabling pathways in spaces of structural disparity. Intrapersonal resources highlight *one pathway* to resilience in a challenged context. In spaces of severe challenge, teachers may therefore be able to *mobilise and employ* intrapersonal resources for *adaptive responses* to risks (Bosch, 2020; Ebersöhn, 2019a; Ebersöhn et al., 2020).



The current study established that, despite chronic and cumulative risk factors, the *significant relationship* between self-efficacy and teacher efficacy beliefs enables pre-service teachers to resile. Both male and female pre-service teachers are thus able to recover from setbacks in schools and feel confident in their teaching and behaviour management. The latter holds for all enrolled pre-service teacher efficacy. Pre-service teachers also demonstrated significantly higher teacher efficacy. Teacher efficacy for pre-service teachers in a challenged context can thus support confidence in recovery from setbacks (Bandura, 1986, 1993, 1996, 1997; Hewitt et al., 2017; Mansfield et al., 2018; Perkins-Gough, 2013).

Pre-service teachers' ability to resile despite the structural disparity evident in Global South spaces, can thus lead to better than expected outcomes. Such outcomes include *job satisfaction, retention, commitment, well-being* and *quality teaching* (Ainsworth & Oldfield, 2019; Arnup & Bowles, 2016; Beltman et al., 2018; Bobek, 2002; Brouskeli et al., 2018; Brunetti, 2006; Cefai & Cavioni, 2014; Day & Gu, 2010, 2013; Gu, 2018; Gu & Li, 2013; Hong, 2012; Mansfield et al., 2016; Morgan, 2011; Peixoto et al., 2018; Peters & Pearce, 2012; Wosnitza et al., 2014). These outcomes are indicators of positive adaptation (Bobek, 2002; Day, 2012; Jackson & Rothmann, 2005; Mansfield et al., 2016) for pre-service teachers.

*High levels* of self-efficacy and teacher efficacy may therefore *buffer pre-service teachers against* predicted or expected outcomes (including teacher stress, burnout and attrition [Arends, 2011; Cefai & Cavioni, 2014; Fleming et al., 2013; Jackson & Rothmann, 2005; Mansfield et al., 2018; Thieman et al., 2012; Wosnitza et al., 2014]) when facing severe structural constraints. As a result, pre-service teachers whose contexts assume maladjustment and poor outcomes can negotiate unexpected or unpredicted beneficial outcomes (Masten, 2014; Morgan, 2011).

Pre-service teachers must be prepared for adversity (see Table 1.1) and the constant adaptation required in a challenged context (Ebersöhn, 2014). Therefore, well-being agendas, emphasising the *identification and mobilisations of resilience enabling pathways* are needed for higher and basic education. Pre-service teachers should be *continuously supported to develop as teachers* (Beltman et al., 2018). Teacher education policy and initial teacher education programmes can be adapted, given the findings of the current study, to *highlight intrapersonal resources* and include teacher resilience intervention programmes to promote resilience in pre-service teachers and training institutions. Since pre-service teachers felt more confident in their teaching and behaviour management (teacher efficacy), these beliefs can further be utilised to promote their recovery from setbacks in a challenged context.

Given the extreme adversity and inequality in the Global South (including South Africa), it is crucial that such identified traits (i.e., self-efficacy and teacher efficacy) are developed, in conjunction with contextual support, during teacher education training and throughout the teaching career. Although the current study was limited to intrapersonal resources due to extant data available, other enabling pathways to resilience (such as relational resilience



(Ebersöhn, 2012; Jordan, 2006, 2013) as well as transactional-ecological process thinking (Ungar, 2011; Ungar et al., 2013) must also be incorporated to sustain resilience on all systemic levels (i.e., micro, meso, exo and macro) over time (Bronfenbrenner, 1979). Through a teacher resilience paradigm, theorising demonstrated enabling pathways and well-being adaptation of pre-service teachers rather than focusing on disparities and risks only (Ebersöhn, 2016).

### 5.6 CHALLENGES AND LIMITATIONS OF THE STUDY

#### 5.6.1 INTRODUCTION

The current study's results need to be interpreted considering the possible challenges and limitations of the research. Therefore, Section 5.6 acknowledge potential caveats.

#### 5.6.2 EXTANT CROSS-SECTIONAL DATA

An identifiable restriction of the current study relates to the nature and extent of the data. Since I relied on extant data for data analysis and no further quantitative or qualitative data was obtained, the exploration of the self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context was restricted. Since the available questionnaires (2015-2017) limited the results, the topic under investigation could not be explored further with enriched data. Consequently, further contextual information could not be incorporated into findings (Babbie, 2021; L. Cohen et al., 2018).

The current study was not engaged in the recruitment of respondents or the original data collection process (Johnston, 2017) of the FIRE project. Thus, any errors during the data collection phase and the original focus of the FIRE project restricted data analysis (Johnston, 2017; Pienta et al., 2011; Widaman et al., 2011). Purposive selection of extant data (i.e., completed FIRE Teacher Resilience Measure utilised in the FIRE project) was employed to acquire information on a particular phenomenon, namely teacher resilience. Although I used non-probability sampling that may limit the generalisability (L. Cohen et al., 2018; Maree & Pietersen, 2019a), it is reasonable to deduce that samples from one context can be representative of samples from other similar contexts (Gravetter & Forzano, 2018), leading to the assumption that generalisation of the findings may be possible to similar settings (i.e., final year pre-service teachers at the University of Pretoria).

The reliability (see Section 4.3.2), validity (see Section 4.3.3) and statistical power analysis (see Section 4.3.4) obtained may further strengthen the possibility of generalisability of the results. Nonetheless, the cross-sectional nature of the extant data did not allow inquiry of causation or a developmental trajectory since the cohorts (2015, 2016, 2017) of pre-service teachers did not represent a longitudinal sample. Therefore, longitudinal research is needed



in future studies to explore possible directionality between constructs such as self-efficacy and teacher efficacy beliefs of pre-service teachers in a challenged context. Further exploration can enable conclusions about how the individual (and contextual) level may influence teacher resilience in spaces of high structural disparity.

#### 5.6.3 SINGLE-COUNTRY AND INSTITUTIONAL STUDY

This study had a limited scope since it was a single-country or mono-national study (Esser & Vliegenthart, 2017). The current study employed an extant dataset from a single HEI teacher education programme in South Africa, possibly limiting the applicability to other programme types or institutions that are different from those of the initial teacher study programme in the current study. Furthermore, although, similar studies (Australia [Beltman et al., 2018], the Czech Republic [Wosnitza et al., 2018], Germany [Peixoto et al., 2018; Wosnitza et al., 2018], Ireland [Morgan, 2011; Peixoto et al., 2018; Wosnitza et al., 2018], Malta [Peixoto et al., 2018; Wosnitza et al., 2018] and Portugal [Peixoto et al., 2018, 2020; Wosnitza et al., 2018]) have been conducted with similar measures employed as in the current study, information about the FIRE Teacher Resilience Measure within the Global South context remain limited.

Since the current study only compared pre-service teachers' self-efficacy and teacher efficacy beliefs in one national teacher education programme, it limits the representation of diverse geographical regions or programme approaches. As a single-country study, the findings from the current study does inform knowledge on teacher resilience within a challenged context. Nonetheless, it would be beneficial to conduct similar studies across various institutions and the Global South context to build generalisability.

#### 5.6.4 TEACHER SELF-REPORT QUESTIONNAIRE DATA

A self-report is any test, measure, questionnaire or survey that relies on respondents' estimation or perception of behaviours, feelings, beliefs, experiences or attitudes. Self-report data is prone to introduce bias and validity concerns relating to under-reporting (e.g., avoiding socially undesirable responses), over-reporting (e.g., providing socially desirable answers) or alone standing attitudinal data (e.g., teachers only reporting on behaviours and attitudes without comparable achievement data) (Chesnut, 2017; L. Cohen et al., 2018; Maree & Pietersen, 2019b, 2019c). Furthermore, when self-reporting is employed, respondents must understand the questions and wording as intended by the research (L. Cohen et al., 2018).

These requirements could not be guaranteed due to the extant data used in the current study. In addition, although teacher self-report data provides a quick, cost-effective and flexible procedure to gather large scale data, it may produce inaccurate results since respondents may consciously or subconsciously display what they believe to be the desired or socially preferred response (Maree & Pietersen, 2019b, 2019c). The data in this study were based on self-



perceived reporting from the FIRE Teacher Resilience Measure (i.e., Likert scale questionnaire). Pre-service teachers in the original FIRE project may have provided answers viewed as "acceptable" (i.e., social desirability bias) since most of the responses were above the midpoint of four (neutral on a seven-point Likert scale).

Given the extant self-report data employed from pre-service teachers and the results obtained, it is furthermore important to investigate the caveat of a possible Dunning-Kruger effect as highlighted in Chapter 4. The Dunning-Kruger effect (J. Kruger & Dunning, 1999) is described as a psychological concept based on the tendency of individuals to have overtly favourable (i.e., overconfidence) or optimistic views of their abilities in social, cognitive and intellectual domains (Al Sultan et al., 2018; Dunning, 2011; J. Kruger & Dunning, 1999). The miscalibration or misestimation (J. Kruger & Dunning, 1999) may stem from the differences between the self-perception or self-assessment and actual performance (across various domains and tasks) with poor performers overestimating abilities, while top performers (i.e., higher skill levels and knowledge) tend to be better-calibrated self-estimates or underestimate themselves or their ability to complete a task (Balážiková, 2018; J. Kruger & Dunning, 1999; Pavel et al., 2012).

The miscalibration may be due to limited insight into cognitive processes, metacomprehension or self-monitoring skills (J. Kruger & Dunning, 1999) and a pervasive cognitive bias (Harvey, 1997) or meta-ignorance (i.e., ignorance of ignorance) (Dunning, 2011). Therefore, the Dunning-Kruger hypothesis postulated that the degree to which an individual can appraise their ability accurately depends, at least partially, upon possessing the ability in question (Gignac & Zajenkowski, 2020). Empirical research (Gignac & Zajenkowski, 2020) has estimated the association between self-assessed ability and objectively measured ability. Within teacher research, scholars (Al Sultan et al., 2018; Coutinho et al., 2020; Podgoršek & Lipovec, 2017; Tomczyk, 2020; Van Staden & Zimmerman, 2017) worldwide have noted the Dunning-Kruger phenomenon, also outside the Global North (Coutinho et al., 2020; Van Staden & Zimmerman, 2017). Within the national context (Van Staden & Zimmerman, 2017), the overly positive reporting of teachers on teacher questionnaire data due to possible aspects of social desirability or a lack of understanding about certain questions on a measure is also emphasised.

Van Staden and Zimmerman (2017) cautioned that teachers' data, especially within a low-performance context such as South Africa, may yield unreliable, unrealistic or invalid results since teachers may feel vulnerable and defensive during a research process. The effect has also been reported with undergraduate students (Balážiková, 2018; J. Kruger & Dunning, 1999; Mahmood, 2016) and pre-service teachers (Al Sultan et al., 2018; Podgoršek & Lipovec, 2017), including studies directly related to pre-service teacher efficacy beliefs (Al Sultan et al., 2018). This reporting may thus indicate that pre-service teachers, in this study, have a false



sense of confidence given their ability to recover from setbacks in a school setting and their confidence in teaching and behaviour management or overestimate their abilities.

As mentioned in Chapter 2, for evaluation of self-efficacy, researchers depend on the respondent (pre-service teachers' respondents in the current study) to draw upon metacognition about the self, presented task and employable strategies that might lead to positive outcomes (Bandura, 1997). However, McIntosh et al. (2019) argued that meta-cognitive differences could contribute to the Dunning-Kruger effect but are neither necessary nor sufficient for it to occur. In addition, Gignac and Zajenkowski (2020) echoed that although the Dunning-Kruger phenomenon may be plausible to some degree or for some skills, the magnitude of the effect may be smaller than reported previously. Nonetheless, given the limitation, the current study may provide added insight for teacher preparation programmes seeking to identify pre-service teachers experiencing the Dunning-Kruger effect. In addition, Dunlosky and Rawson (2012) encouraged the self-reflection of undergraduates on their current competency levels to learn a set of abilities.

#### 5.6.5 RESPONSE RATE

Given the pre-service teacher graduation data (2015, 2016, 2017), as presented in Chapter 2, and the demographic data on completed questionnaires (N), as depicted in Chapter 4, the response rate of the current study is portrayed in Table 5.3.

#### Table 5.3

Year	Ν	Graduates	Response rate (%)		
Case 1 (2015)	313	697	44.91%		
Case 2 (2016)	169	751	22.50%		
Case 3 (2017)	711	794	89.54%		

Response Rate for the Current Study

As illustrated in Table 5.3, the sample size and response rate for Case 2 (2016) was low (22.50%), possibly due to the #Fees-Must-Fall movement that emerged at South African public universities in 2016. The low response rate may have compromised data reliability, and caution should be used to draw conclusions. The response rates for Case 1 (2015) and Case 3 (2017) are higher, with L. Cohen et al. (2018) indicating that a response rate of 50% is satisfactory. The high response rate for Case 3 (2017) may enhance the reliability of the data. The rigorousness of statistical analysis was not affected due to the sample size (as reported by the power analysis in Chapter 4).



#### 5.6.6 FIRE TEACHER RESILIENCE MEASURE IN THE SOUTH AFRICAN CONTEXT

As discussed in Chapter 1, the FIRE Teacher Resilience Measure was administered in English with final year pre-service teachers. Although English is a main language of teaching and learning at the University of Pretoria, with most pre-service teachers indicating English fluency, the questionnaire was not translated into any other home language of respondents. As such, potential language difficulties in understanding the questionnaire items should be contemplated as almost one-tenth of the respondents indicated that they were not fluent in the language that the questionnaire was presented to them. Furthermore, the language demographic question on the FIRE Teacher resilience measure (i.e., "List of languages you are fluent in") was an open-ended question with no clear indication of the meaning of "fluency". The possibility remains that respondents could misinterpret such questions and only indicate their dominant (home) language or indicate a conservative answer relating to language fluency (i.e., only indicate one language). However, even if the language demographic question on the Teacher Resilience Measure was a yes or no question across languages, the measure is not robust enough to make further meaningful inferences about the possible role of multilingualism.

Terms such as "student" used in the original ENTREE questionnaire (Mansfield & Wosnitza, 2015), was also depicted in the FIRE Teacher Resilience Measure despite the convention in SA to refer to "learners" which can make questions inherently ambiguous (L. Cohen et al., 2018) to respondents. Therefore, I interpreted the data with sensitivity by considering cross-cultural and socioeconomic factors influencing the data as the measure was not standardised for the South African context. Nonetheless, the reliability analysis of the measure compares well with reliability results from other studies (Beltman et al., 2018; Morgan, 2011; Peixoto et al., 2018, 2020; Wosnitza et al., 2018). Lastly, since extant data was employed in the current study, collected through an existing measure in the FIRE project (see Section 1.2) the literature discourses presented in Chapter 2 did not inform the design of the FIRE Teacher Resilience Measure.

Due to these challenges, conclusions drawn from the current study remain tentative but provide important knowledge for additional research in teacher resilience within the Global South. The following section provides the recommendations of the current study.

# 5.7 RECOMMENDATIONS OF THE STUDY

#### 5.7.1 INTRODUCTION

The current study compared pre-service teachers' self-efficacy and teacher efficacy beliefs to inform knowledge on teacher resilience in a challenged context. Thus, it is also valuable to recognise the application of findings for the interplay between research, theory, and implications to benefit psychological science and educational practice (Creswell & Creswell,



2018). Therefore, Section 5.7 details the recommendations as consequences of the results of the current study for future research, practice and training as they relate to teacher resilience in a challenged context.

### 5.7.2 RECOMMENDATIONS FOR TRAINING AND DEVELOPMENT

The findings of this study could benefit initial teacher training programmes and pre-service teacher development in a challenged context. Teacher resilience research and interventions to promote resilience in pre-service programmes are limited (Beltman et al., 2018). The following is recommended for training and developmental purposes since initial teacher education programmes can play a vital role in novice teachers building resilience capacity (Beltman et al., 2011):

- Teacher training programmes and institutions should promote the identification and development of internal protective resources (e.g., self-efficacy and teacher efficacy) to enable teacher resilience in a challenged context.
- The training of pre-service teachers should include knowledge of possible internal protective resources and the development of resilience-related skills to identify and navigate towards resilience capacities.
- Enable pre-service teachers to examine existing beliefs regarding teacher resilience and integrating these with learning during teacher education.
- Develop teacher resilience-building modules or interventions for pre-service teachers during initial teacher education programmes for the Global South context.
- Involve pre-service teachers in developing, designing, applying, and assessing preservice teacher resilience-building programmes in a challenged context.
- Pre-service resilience-building programmes should be streamlined for specific preservice teaching programmes (i.e., FP/ECD, IP, SP and FET phase) to meet teachers' needs effectively given the statistical difference found between intrapersonal traits in different phases in the current study.
- Pre-service resilience-building programmes should be presented in the early stages of teacher training, continuing throughout the teacher training programme into the first years of teaching given the challenges faced in a Global South context.
- Pre-service teacher resilience-building programmes should include knowledge of how to support teacher resilience in schools and how teachers can facilitate this process in a challenged context.
- HEIs, including the teacher educators and cooperating or mentor teachers in placement schools, should support pre-service teachers to promote the identification and development of internal and contextual protective resources to nurture teacher resilience.



## 5.7.3 RECOMMENDATIONS FOR POLICY DEVELOPMENT AND PRACTICE

Based on the results and findings of the current study, the following is recommended for policy development and practice:

- Educational institutions (basic and higher education) may consider formulating resilience frameworks and agendas to enable teachers to thrive, not just survive in a challenged context.
- The current study urges policy, schools and school management to seek opportunities to nurture internal protective resources. As a result, teachers can be supported to maintain their teaching quality, job satisfaction and remain committed to the profession.
- Important role-players (including learners, parents, teachers, communities, organisations and institutions) should be involved in resilience-building efforts to enable adaptive outcomes to risk and barriers within a Global South setting.
- Teachers' understanding of their responsibility, within a given context, as resilience assets for learners can be highlighted.
- Continuous mentoring and support, through for example building resilience networks, to promote resilience-related experiences, competencies, traits and knowledge given challenges in the Global South. Therefore, school management training should also include ways in which school leadership may support teachers and promote the development of internal protective resources.

# 5.7.4 RECOMMENDATIONS FOR FUTURE RESEARCH

Given the findings, I suggest possible areas of future exploration for deeper insight into areas that have not been investigated in this study.

# 5.7.4.1 Teacher resilience-building programmes in a challenged context

Given that there are limited studies (Beltman et al., 2011, 2018) on the interventions to build resilience in pre-service programmes, research on the development and possible benefit of interventions or modules on teacher resilience within a Global South context seems necessary. Therefore, further research regarding the enhancement of pre-service teacher resilience through teacher resilience-building programmes is recommended.

# 5.7.4.2 Pre-service resilience studies within a Global South context

Up to date, there is seemingly no nationwide studies investigating pre-service teacher resilience informing resilience within a Global South landscape. Given the limited knowledge outside the Global North and the Global South teaching landscape challenges, it seems timely



to take a national approach to research in this field. This study could thus be expanded to other teacher preparation programmes that prepare students to teach within a Global South context. In this regard, areas of influence which I could not explore in the current study (as explained in Chapter 4) was the effect that other demographic variables (e.g., language) may have on pre-service teacher resilience. Furthermore, future researchers could seek to explore the extant data available from the additional domains (e.g. (i) teacher professionalism [TR-Prof], (ii) teacher emotion [TR-Emot], (iii) teacher motivation [TR-Mot], and (iv) teacher social capacity [TR-Soc]).

Future research on pre-service teacher resilience in a challenged context can benefit from a mixed-method design. Such a design may provide qualitative interpretations of preservice teachers' internal protective resilience capacities (e.g., high levels of self-efficacy and teacher efficacy beliefs) in a challenged context. Since the current study focused on resilience as an internal trait only, future researchers could explore contextual factors and the resilience process of pre-service teachers in challenged contexts.

#### 5.7.4.3 Longitudinal studies

Besides additional pre-service resilience studies within a Global South context to inform teacher resilience knowledge in a challenged context, longitudinal studies (or a combination of cross-section and longitudinal designs) are needed to understand pre-service teacher resilience better. A longitudinal study could benefit from multi-time measurements and multi-method approaches, including qualitative data. To administer the FIRE Resilience Measure longitudinally from the first to the fourth year for pre-service teachers, but then also during the first year of teaching and thereafter may allow the investigation of resilience's temporal and causal process dimensions (Ungar, 2011).

Inferential statistical procedures can also be performed on biographical variables such as age, which was not feasible in the current study as explained in Chapter 4. Such research may also provide the opportunity to explore the development of a pre-service teacher resilience trajectory in a challenged context which can assist in understanding transformation processes and clarifying different contexts affect results in different ways (Esser & Vliegenthart, 2017).

#### 5.7.4.4 Questionnaire development and standardisation

Given the complex construct of resilience and suitable questions in a challenged South African context (e.g., language categories), teacher resilience measures should be further developed to conduct nationwide longitudinal research. Appraising resilience presents challenges due to resilience's multidimensional and dynamic nature (Beltman et al., 2018). As discussed in Chapter 1, the FIRE Teacher Resilience Measure consisted of items that were sourced from existing scales (Coetzee, 2013; Mansfield & Wosnitza, 2015; Morgan, 2011; Peixoto et al.,



2018; Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012). The scales indicated acceptable reliability and validity in preceding studies (Mansfield & Wosnitza, 2015; Morgan, 2011; Peixoto et al., 2018, 2020; Watt & Richardson, 2007, 2008, 2012; Watt et al., 2012; Wosnitza et al., 2018). Previous studies of the measure were discussed in Chapter 3, while the reliability and validity results of the scales utilised in the current study were described in Chapter 4.

The FIRE Resilience Measure has, however, not been standardised for the South African context. Further questionnaire development in this regard is necessary to consider the instrument used to determine pre-service teacher resilience in a challenged context given cross-cultural and socioeconomic influences in culturally diverse settings. Furthermore, the FIRE Teacher Resilience measure could be translated into isiZulu (indicated as the most spoken language in South Africa in Chapter 1) to be utilised in rural areas to investigate teacher resilience. Providing additional reliability and validity evidence may strengthen the appropriateness of worldwide use and allow for cross-cultural comparisons to expand the generalisability of the constructs. Future researchers could also seek to establish the reliability and validity of the FIRE Teacher Resilience Measures in settings other than South African preservice teachers. Contextual factors influencing teacher resilience should also be incorporated into questionnaire development to provide necessary information for a challenged context.

#### 5.8 CONCLUDING REMARKS

This chapter provided an overview of the chapters in the current study and compared the results with findings from existing literature. The overview and comparison were followed by addressing the research questions and forming conclusions from the statistical hypotheses. Furthermore, the adapted conceptual framework, based on the findings in the current study, was presented. The chapter was concluded by a discussion of the limitations of the current study and recommendations for further research, practice and training.

The current study contributed to teacher resilience knowledge by comparing the selfefficacy and teacher efficacy beliefs of pre-service teachers in a challenged context. Extant questionnaire data on teacher resilience were analysed, and traits that act as protective resources were identified. I found that the self-efficacy and teacher efficacy, as intrapersonal resilience-enabling pathways to teacher resilience, is high for pre-service teachers in a challenged context (within-case and cross-case analysis). In addition, the teacher efficacy beliefs of pre-service teachers are statistically significantly higher than their confidence to recover from setbacks in schools (within-case and cross-case analysis). Although no statistically significant differences were found between the demographic variables (i.e., gender and enrolled pre-service teaching programme) for the within-case analysis for Case 1 and Case 2, the within-case analysis for Case 3 (2017) demonstrated a statistically significant difference with female pre-service teachers' self-efficacy and teacher efficacy beliefs higher



than their male counterparts. A statistically significant difference was also identified for enrolled pre-service teaching programmes (Case 3) with FP/ECD pre-service teachers' self-efficacy and teacher efficacy beliefs higher than FET phase pre-service teachers. IP pre-service teachers' teachers' teacher efficacy was also significantly higher than FET phase pre-service teachers.

In conclusion, the findings of the current study highlight self-efficacy and teacher efficacy as high intrapersonal resilience-enabling pathways in this group. As intrapersonal pathways, self-efficacy and teacher efficacy plausibly enable pre-service teachers to resile despite chronic and cumulative risk factors. Therefore, the findings of this study advance the understanding of teacher resilience in a challenged context. Moreover, the current study informs limited knowledge on quantitatively derived teacher resilience findings from an oftenunder-represented Global South setting. Evidence from an African perspective is presented to inform global teacher resilience discourses. The current study also contributes to teacher resilience measurement knowledge with insights on the utility of acknowledged teacher resilience scales in South Africa. The findings act as a precursor for comparative teacher resilience results worldwide.

---000----



# References

- Afdal, H. W. (2019). The promises and limitations of international comparative research on teacher education. *European Journal of Teacher Education, 42*(2), 258–275. https://doi.org/10.1080/02619768.2019.1566316
- Aickin, M., & Gensler, H. (1996). Adjusting for multiple testing when reporting research results: The Bonferroni vs Holm methods. *American Journal of Public Health*, 86(5), 726–728. https://doi.org/10.2105/AJPH.86.5.726
- Ainsworth, S., & Oldfield, J. (2019). Quantifying teacher resilience: Context matters. *Teaching and Teacher Education, 82,* 117–128. https://doi.org/10.1016/j.tate.2019.03.012
- Almedom, A. M. (2005). Resilience, hardiness, sense of coherence, and posttraumatic growth: All paths leading to "light at the end of the tunnel"? *Journal of Loss and Trauma, 10*(3), 253–265. https://doi.org/10.1080/15325020590928216
- Al Sultan, A., Henson, H., Jr., & Fadde, P. J. (2018). Pre-service elementary teachers' scientific literacy and self-efficacy in teaching science. *IAFOR Journal of Education*, 6(1), 25–41. https://files.eric.ed.gov/fulltext/EJ1172004.pdf
- American Psychological Association. (2017). *Ethical principles of psychologists and code of conduct (2003, amended effective June 1, 2010, and January 1, 2017).* https://www.apa.org/ethics/code/index.aspx
- American Psychological Association. (2020a). *APA dictionary of psychology*. https://dictionary.apa.org/intrapsychic
- American Psychological Association. (2020b). *Publication manual of the American Psychological Association* (7th ed.). https://doi.org/10.1037/0000165-000
- Ansfield, B. (2018). Unsettling "inner city": Liberal Protestantism and the postwar origins of a keyword in urban studies. *Antipode*, *50*(5), 1166-1185.
- Anthony, E. J. (1987). Risk, vulnerability, and resilience: An overview. In E. J. Anthony & B. J. Cohler (Eds.), *The invulnerable child* (pp. 3–48). The Guilford Press.
- Antonovsky, A. (1979). *Health, stress and coping: New perspectives on mental and physical well-being.* Jossey-Bass.
- Antonovsky, A., & Sourani, T. (1988). Family sense of coherence and family adaptation. *Journal of Marriage and the Family*, *50*(1), 79–92. https://doi.org/10.2307/352429



- Antonovsky, H., & Sagy, S. (1986). The development of a sense of coherence and its impact on responses to stress situations. *The Journal of Social Psychology, 126*(2), 213– 225.
- Arends, F. (2011, November). Teacher shortages? The need for more reliable information at school level. *Review of Education Skills Development and Innovation*. http://www.hsrc.ac.za/uploads/pageContent/2702/RESDI%20newsletter,%20November %202010%20issue.pdf
- Arnup, J., & Bowles, T. (2016). Should I stay or should I go? Resilience as a protective factor for teachers' intention to leave the teaching profession. *Australian Journal of Educational Research*, 60(3), 229–244. https://doi.org/10.1177/0004944116667620
- Ary, D., Jacobs, L. C., Irvine, C. K. S., & Walker, D. A. (2019). *Introduction to research in education* (10th ed.). Cengage Learning.
- Averill, R. M., & McRae, H. S. (2019). Culturally sustaining initial teacher education: Developing student teacher confidence and competence to teach indigenous learners. *The Educational Forum*, 83(3), 294–308. https://doi.org/10.1080/00131725.2019.1599657
- Babbie, E. (2021). The practice of social research (15th ed.). Cengage Learning.
- Balážiková, M. (2018). Dunning-Kruger debate and students' assessment. *Marketing Identity,* 6(1/1), 230–242.
- Bandura, A. (1977). Self-efficacy: Toward a unifying theory of behavioral change. *Psychological Review, 84*(2), 191–215. https://doi.org/10.1037/0033-295X.84.2.191
- Bandura, A. (1982). Self-efficacy mechanism in human agency. *American Psychologist, 37*(2), 122–147. https://doi.org/10.1037/0003-066X.37.2.122
- Bandura, A. (1986). Social foundations of thought and actions: A social cognitive theory. Prentice-Hall.
- Bandura, A. (1989a). Human agency in social cognitive theory. *American Psychologist,* 44(9), 1175–1184. https://doi.org/10.1037/0003-066X.44.9.1175
- Bandura, A. (1989b). Social cognitive theory. In R. Vasta (Ed.), *Annals of child development* (Vol. 6, pp. 1–60). JAI Press.
- Bandura, A. (1993). Perceived self-efficacy in cognitive development and functioning. *Educational Psychologist, 28*(2), 117–148. https://doi.org/10.1207/s15326985ep2802\_3
- Bandura, A. (1996). Ontological and epistemological terrains revisited. Journal of Behavior Therapy and Experimental Psychiatry, 27(4), 323–345. https://doi.org/10.1016/S0005-7916(96)00049-3



Bandura, A. (1997). Self-efficacy: The exercise of control. W. H. Freeman and Company.

- Bandura, A. (2001). Social cognitive theory: An agentic perspective. *Annual Review of Psychology, 52,* 1–26. https://doi.org/10.1146/annurev.psych.52.1.1
- Bandura, A. (2002). Socio cognitive theory in cultural context. *Applied Psychology*, *51*(2), 269–290. https://doi.org/10.1111/1464-0597.00092
- Bandura, A. (2006). Adolescent development from an agentic perspective. In F. Pajares & T. Urdan (Eds.), Self-efficacy beliefs of adolescents (pp. 1–43). Information Age Publishing.
- Bandura, A., & Adams, N. E. (1977). Analysis of self-efficacy theory of behavioral change. *Cognitive Therapy and Research, 1*(4), 287–310. https://doi.org/10.1007/BF01663995
- Bandura, A., Adams, N. E, & Beyer, J. (1977). Cognitive processes mediating behavioral change. *Journal of Personality and Social Psychology*, *35*(3), 125–139. https://doi.org/10.1037/0022-3514.35.3.125
- Bandura, A., Adams, N. E., Hardy, A. B., & Howells, G. N. (1980). Tests of the generality of self-efficacy theory. *Cognitive Therapy and Research*, 4(1), 39–66. https://doi.org/10.1007/BF01173354
- Bashant, J. (2014). Developing grit in our students: Why grit is such a desirable trait, and practical strategies for teachers and schools. *Journal for Leadership and Instruction*, *13*(2), 14–17. https://files.eric.ed.gov/fulltext/EJ1081394.pdf
- Begum, Z., Rahman, A., & Majoka, M. I. (2020). Contribution of professional efficacy and demographic factors towards performance of secondary schools teachers. *The Dialogue, 15*(1), 137–146. http://www.qurtuba.edu.pk/thedialogue/The%20Dialogue/15\_1/15\_1\_15.pdf
- Beltman, S. (2020). Understanding and examining teacher resilience from multiple perspectives. In C. F. Mansfield (Ed.), *Cultivating teacher resilience: International approaches, applications and impact* (pp. 11–26). Springer. https://doi.org/10.1007/978-981-15-5963-1
- Beltman, S., Mansfield, C., & Price, A. (2011). Thriving not just surviving: A review of research on teacher resilience. *Educational Research Review*, 6(3), 185–207. https://doi.org/10.1016/j.edurev.2011.09.001
- Beltman, S., & Mansfield, C. F. (2018). Resilience in education: An introduction. In M. Wosnitza, F. Peixoto, S. Beltman, & C. F. Mansfield (Eds.), *Resilience in education: Concepts, contexts and connections* (pp. 3–9). Springer International Publishing. https://doi.org/10.1007/978-3-319-76690-4


- Beltman, S., Mansfield, C. F., Wosnitza, M., Weatherby-Fell, N., & Broadley, T. (2018). Using online modules to build capacity for teacher resilience. In M. Wosnitza, F. Peixoto, S. Beltman, & C. F. Mansfield (Eds.), *Resilience in education: Concepts, contexts and connections* (pp. 237–253). Springer International Publishing. https://doi.org/10.1007/978-3-319-76690-4\_14
- Bennell, P., Hyde, K., & Swainson, N. (2002). The impact of the HIV/AIDS epidemic on the education sector in sub-Saharan Africa: A synthesis of findings and recommendations of three country studies. Centre for International Education, University of Sussex Institute of Education. https://hivhealthclearinghouse.unesco.org/sites/default/files/resources/1127\_644%20S ynthesis.pdf
- Besharati, N., & Tsotsotso, K. (2015). *In search for the education panacea: A systematic review and comparative meta-analysis of interventions to improve learner achievement in South Africa* [Unpublished research report]. University of the Witwatersrand.
- Bhana, D., Morrell, R., Epstein, D., & Moletsane, R. (2006). The hidden work of caring: Teachers and the maturing AIDS epidemic in diverse secondary schools in Durban. *Journal of Education, 38*(1), 5–23. https://journals.co.za/docserver/fulltext/joe/38/1/54.pdf?expires=1600963522&id=id&ac cname=guest&checksum=A01DF4E374A042B75FA55F4DBA87D5C3
- Bhengu, C. (2019, May 3). SA hangs its head in shame as Time magazine highlights inequality. TimesLIVE. https://www.timeslive.co.za/news/south-africa/2019-05-03-sahangs-its-head-in-shame-as-time-magazine-highlights-inequality/
- Biglan, A. (1987). A behavior-analytic critique of Bandura's self-efficacy theory. *The Behavior Analyst, 10*(1), 1–15. https://doi.org/10.1007/BF03392402
- Bird, F. (2020). A defense of objectivity in the social sciences, rightly understood. Sustainability: Science, Practice and Policy, 16(1), 83–98. https://doi.org/10.1080/15487733.2020.1785679
- Bisseker, C. (2019, March 21). *IMF report slams SA's education folly*. BusinessLIVE. https://www.businesslive.co.za/fm/features/2019-03-21-imf-report-slams-saseducation-folly/
- Black, G. L. (2015). Developing teacher candidates' self-efficacy through reflection and supervising teacher support. *In Education, 21*(1), 78–98. https://files.eric.ed.gov/fulltext/EJ1246782.pdf
- Block, J., & Kremen, A. M. (1996). IQ and ego-resiliency: Conceptual and empirical connections and separateness. *Journal of Personality and Social Psychology*, 70(2), 349–361. https://doi.org/10.1037/0022-3514.70.2.349



- Block, J. H., & Block, J. (1980). The role of ego-control and ego-resiliency in the organization of behavior. In W. A. Collins (Ed.), *Development of cognition, affect, and social relations: The Minnesota Symposia on Child Psychology* (Vol. 13, pp. 39–101). Lawrence Erlbaum Associates.
- Bobek, B. L. (2002). Teacher resiliency: A key to career longevity. *The Clearing House: A Journal of Educational Strategies, Issues and Ideas, 75*(4), 202–205. https://doi.org/10.1080/00098650209604932
- Borkovec, T. D. (1978). Self-efficacy: Cause or reflection of behavioral change? Advances in Behaviour Research and Therapy, 1(4), 163–170. https://doi.org/10.1016/0146-6402(78)90003-6
- Bos, H. (2017). Promising interventions are great, but are they enough? *Journal of Research on Educational Effectiveness, 10*(3), 541–544. https://doi.org/10.1080/19345747.2017.1335090
- Bosch, Z. (2020). Comparing teacher resilience in primary schools in challenged context [Unpublished master's thesis]. University of Pretoria.
- Bouillet, D., Pavin Ivanec, T., & Miljević-Riđički, R. (2014). Preschool teachers' resilience and their readiness for building children's resilience. *Health Education*, *114*(6), 435–450. https://doi.org/10.1108/HE-11-2013-0062
- Brandon, D. B. (2000). Self-efficacy: Gender differences of prospective primary teachers in Botswana. *Research in Education, 64*(1), 36–43. https://doi.org/10.7227/RIE.64.4
- Bronfenbrenner, U. (1979). *The ecology of human development: Experiments by nature and design*. Harvard University Press.
- Brouskeli, V., Kaltsi, V., & Loumakou, M. (2018). Resilience and occupational well-being of secondary education teachers in Greece. *Issues in Educational Research*, 28(1), 43– 60. http://www.iier.org.au/iier28/brouskeli.pdf
- Brouwers, A., & Tomic, W. (2000). A longitudinal study of teacher burnout and perceived self-efficacy in classroom management. *Teaching and Teacher Education*, 16(2), 239– 253. https://doi.org/10.1016/S0742-051X(99)00057-8
- Brown, A. L., Lee, J., & Collins, D. (2015). Does student teaching matter? Investigating preservice teachers' sense of efficacy and preparedness. *Teaching Education*, 26(1), 77– 93. https://doi.org/10.1080/10476210.2014.957666
- Brunetti, G. J. (2006). Resilience under fire: Perspectives on the work of experienced, inner city high school teachers in the United States. *Teaching and Teacher Education*, 22(7), 812–825. https://doi.org/10.1016/j.tate.2006.04.027



- Bryan, J. (2005). Fostering educational resilience and achievement in urban schools through school-family-community partnerships [Special issue]. *Professional School Counseling*, *8*(3).
- Buchanan, J., Prescott, A., Schuck, S., Aubusson, P., Burke, P., & Louviere, J. (2013). Teacher retention and attrition: Views of early career teachers. *Australian Journal of Teacher Education*, 38(3), 112–129. https://doi.org/10.14221/ajte.2013v38n3.9
- Buddelmeyer, H., & Powdthavee, N. (2016). Can having internal locus of control insure against negative shocks? Psychological evidence from panel data. *Journal of Economic Behavior & Organization, 122*, 88–109. https://doi.org/10.1016/j.jebo.2015.11.014
- Caprara, G. V., Barbaranelli, C., Steca, P., & Malone, P. S. (2006). Teachers' self-efficacy beliefs as determinants of job satisfaction and students' academic achievement: A study at the school level. *Journal of School Psychology*, *44*(6), 473–490. https://doi.org/10.1016/j.jsp.2006.09.001
- Cassidy, S. (2015). Resilience building in students: The role of academic self-efficacy. *Frontiers in Psychology, 6,* Article 1781. https://doi.org/10.3389/fpsyg.2015.01781
- Castro, A. J., Kelly, J., & Shih, M. (2010). Resilience strategies for new teachers in highneeds areas. *Teaching and Teacher Education, 26*(3), 622–629. https://doi.org/10.1016/j.tate.2009.09.010
- Cefai, C., & Cavioni, V. (2014). Social and emotional education in primary school. Springer. https://doi.org/10.1007/978-1-4614-8752-4\_8
- Çelik, I., Aka, S. T., Çalik, F., Bayraktar, G., & Bayram, M. (2018). The investigation on physical education teacher candidate's resilience, tenacity and motivation levels. *Journal of Education and e-Learning Research*, *5*(3), 174–178. https://doi.org/10.20448/journal.509.2018.53.174.178
- Chakraborti, S., & Graham, M. A. (2019). *Nonparametric statistical process control*. John Wiley & Sons.
- Chambers, F. (1997). What do we mean by fluency? *System, 25*(4), 535–544. https://doi.org/10.1016/s0346-251x(97)00046-8
- Chen, L. T., & Liu, L. (2019). Content analysis of statistical power in educational technology research: Sample size matters. *International Journal of Technology in Teaching and Learning*, 15(1), 49–75. https://sicet.org/main/wpcontent/uploads/2019/12/4\_chen\_liu.pdf
- Chesnut, S. R. (2017). On the measurement of preservice teacher commitment: Examining the relationship between four operational definitions and self-efficacy beliefs. *Teaching and Teacher Education, 68,* 170–180. https://doi.org/10.1016/j.tate.2017.09.003



- Chesnut, S. R., & Burley, H. (2015). Self-efficacy as a predictor of commitment to the teaching profession: A meta-analysis. *Educational Research Review, 15,* 1–16. https://doi.org/10.1016/j.edurev.2015.02.001
- Chisholm, L., Hoadley, U., WaKivulu, M., Brookes, H., Prinsloo, C., Kgobe, A., Mosia, D., Narsee, H., & Rule, S. (2005). *Educator workload in South Africa*. HSRC Press.
- Chong, S., Low, E. L., & Goh, K. C. (2011). Emerging professional teacher identity of preservice teachers. *Australian Journal of Teacher Education*, 36(8), 50–64. https://files.eric.ed.gov/fulltext/EJ937005.pdf
- Chun, C. A., Moos, R. H., & Cronkite, R. C. (2006). Culture: A fundamental context for the stress and coping paradigm. In P. T. P. Wong & L. C. J. Wong (Eds.), *Handbook of multicultural perspectives on stress and coping* (pp. 29–53). Springer. https://doi.org/10.1007/b137168
- Clark, S. K. (2020). Examining the development of teacher self-efficacy beliefs to teach reading and to attend to issues of diversity in elementary schools. *Teacher Development, 24*(2), 127–142. https://doi.org/10.1080/13664530.2020.1725102
- Clement, M. (2017). Why combatting teachers' stress is everyone's job. *The Clearing House: A Journal of Educational Strategies, Issues and Ideas, 90*(4), 135–138. https://doi.org/10.1080/00098655.2017.1323519
- Coetzee, S. (2013). Sustaining teacher career resilience in a resource-constrained rural education setting: A retrospective study [Doctoral thesis, University of Pretoria]. UP Repository. https://repository.up.ac.za/handle/2263/40230
- Coetzee, S., Ebersöhn, L., Ferreira, R., & Moen, M. (2017). Disquiet voices foretelling hope: Rural teachers' resilience experiences of past and present chronic adversity. *Journal of Asian and African Studies, 52*(2), 201–216. https://doi.org/10.1177/0021909615570955
- Cohen, J. (1988). *Statistical power analysis for the behavioral sciences* (2nd ed.). Lawrence Erlbaum Associates.
- Cohen, J. (1992). A power primer. *Psychological Bulletin, 112*(1), 155–159. https://doi.org/10.1037/0033-2909.112.1.155
- Cohen, L., Manion, L., & Morrison, K. (2018). *Research methods in education* (8th ed.). Routledge.
- Cohn, M. A., Fredrickson, B. L., Brown, S. L., Mikels, J. A., & Conway, A. M. (2009). Happiness unpacked: Positive emotions increase life satisfaction by building resilience. *Emotion*, 9(3), 361–368. https://doi.org/10.1037/a0015952



- Connor, K. M., & Davidson, J. R. T. (2003). Development of a new resilience scale: The Connor-Davidson resilience scale (CD-RISC). *Depression & Anxiety, 18*(2), 76–82. https://doi.org/10.1002/da.10113
- Cook, C. R., Miller, F. G., Fiat, A., Renshaw, T., Frye, M., Joseph, G., & Decano, P. (2017).
   Promoting secondary teachers' well-being and intentions to implement evidencebased practices: Randomized evaluation of the achiever resilience curriculum [Special issue]. *Psychology in the Schools, 54*(1). https://doi.org/10.1002/pits.21980
- Cooper, H. (2018). Reporting quantitative research in psychology: How to meet APA style journal article reporting standards (2nd ed.). American Psychological Association. https://doi.org/10.1037/0000103-000
- Coovadia, H., Jewkes, R., Barron, P., Sanders, D., & McIntyre, D. (2009). The health and health system of South Africa: Historical roots of current public health challenges. *The Lancet, 374*(9692), 817–834. https://doi.org/10.1016/S0140-6736(09)60951-X
- Corder, G. W., & Foreman, D. I. (2014). *Nonparametric statistics: A step-by-step approach* (2nd ed.). John Wiley & Sons.
- Coutinho, M. V. C., Thomas, J., Fredricks-Lowman, I., & Bondaruk, M. V. (2020). The Dunning-Kruger effect in Emirati college students: Evidence for generalizability across cultures. *International Journal of Psychology and Psychological Therapy*, 20(1), 29–36. https://dialnet.unirioja.es/servlet/articulo?codigo=7288616
- Cozby, P. C., & Rawn, C. D. (2016). Methods in behavioural research. McGraw-Hill Ryerson.
- Creswell, J. W. (2003). *Research design: Qualitative, quantitative, and mixed methods approaches* (2nd ed.). Sage
- Creswell, J. W. (2014). *Research design: Qualitative, quantitative, and mixed methods approaches* (4th ed.). Sage.
- Creswell, J. W., & Creswell, J. D. (2018). *Research design: Qualitative, quantitative, and mixed methods approaches* (5th ed.). Sage.
- Dados, N., & Connell, R. (2012). The Global South. *Contexts, 11*(1), 12–13. https://doi.org/10.1177/1536504212436479
- Daniels, L. M., Radil, A. I., & Goegan, L. D. (2017). Combinations of personal responsibility: Differences on pre-service and practicing teachers' efficacy, engagement, classroom goal structures and wellbeing. *Frontiers in Psychology, 8,* Article 906. https://doi.org/10.3389/fpsyg.2017.00906
- Daniilidou, A., Platsidou, M., & Gonida, S.-E. (2020). Primary school teachers' resilience: Association with teacher self-efficacy, burnout and stress. *Electronic Journal of*



Research in Education Psychology, 18(3), 549–582. http://ojs.ual.es/ojs/index.php/EJREP/article/view/3487/3974

- Day, C. (2012). New lives of teachers. *Teacher Education Quarterly, 39*(1), 7–26. https://files.eric.ed.gov/fulltext/EJ977354.pdf
- Day, C., & Gu, Q. (2007). Variations in the conditions for teachers' professional learning and development: Sustaining commitment and effectiveness over a career. Oxford Review of Education, 33(4), 423–443. https://doi.org/10.1080/03054980701450746
- Day, C., & Gu, Q. (2010). Resilience counts. In C. Day., & Q. Gu (Eds.), *The new lives of teachers* (pp. 156–176). Routledge. https://doi.org/10.4324/9780203847909
- Day, C., & Gu, Q. (2013). *Resilient teachers, resilient schools: Building and sustaining quality in testing times.* Routledge. https://doi.org/10.4324/9780203578490
- Day, C., Stobart, G., Sammons, P., Kington, A., Gu, Q., Smees, R., & Mujtaba, T. (2006). Variations in teachers' work, lives and effectiveness. DfES Publications. https://dera.ioe.ac.uk/6405/1/rr743.pdf
- De Jong, G. (2010). *Invisible enemy: The African American freedom struggle after 1965*. John Wiley & Sons.
- Delale-O'Connor, L. A., Alvarez, A. J., Murray, I. E., & Milner, H. R., IV. (2017). Self-efficacy beliefs, classroom management, and the cradle-to-prison pipeline. *Theory Into Practice*, *56*(3), 178–186. https://doi.org/10.1080/00405841.2017.1336038
- Department of Basic Education, Republic of South Africa. (2018a). *Annual report 2017/18*. https://www.gov.za/sites/default/files/gcis\_document/201810/dbe-201718-annualreport.pdf
- Department of Basic Education, Republic of South Africa. (2018b, October 1). *Basic Education on increased number of qualified teachers in education system* [Press release]. https://www.gov.za/speeches/public-education-system-1-oct-2018-0000
- Department of Basic Education, Republic of South Africa. (2018c). *Education statistics in South Africa 2016.* https://www.education.gov.za/Portals/0/Documents/Publications/Education%20Statistic %20SA%202016.pdf?ver=2018-11-01-095102-947
- Department of Basic Education, Republic of South Africa. (2019). Do you want to make a difference? Then become a teacher. https://www.education.gov.za/Informationfor/Teachers/InitialTeacherEducation.aspx
- Department of Basic Education, Republic of South Africa. (2020). 2019 school realities. https://www.education.gov.za/Portals/0/Documents/Reports/School%20Realities%2020 19%20Final%20.pdf?ver=2020-02-07-101051-330



Department of Basic Education. Republic of South Africa. (2021). National Senior Certificate: 2020 Examination report.

https://www.education.gov.za/Portals/0/Documents/Reports/2021%20NSC%20Reports/2020NSCREPORT.pdf?ver=2021-07-19-142304-897

- Department of Higher Education and Training, Republic of South Africa. (2018a). Language Policy for Higher Education. *Government Gazette*, 632(41463). https://www.dhet.gov.za/Policy%20and%20Development%20Support/Government%20 Notice%20Revised%20Language%20Policy%20for%20Higher%20Education.pdf
- Department of Higher Education and Training, Republic of South Africa. (2018b). *Statistics* on post-school education and training in South Africa: 2016. http://www.dhet.gov.za/Research%20Coordination%20Monitoring%20and%20Evaluati on/6\_DHET%20Stats%20Report\_04%20April%202018.pdf
- DeVellis, R. F. (2012). Scale development: Theory and applications (3rd ed.). Sage.
- De Vos, A. S., Strydom, H., Fouché, C. B., & Delport, C. S. L. (2011). *Research at grass roots: For the social sciences and human service professions* (4th ed.). Van Schaik.
- Diale, B., Pillay, J., & Fritz, E. (2014). Dynamics in the personal and professional development of Life-orientation teachers in South Africa, Gauteng Province. *Journal of Social Science*, *38*(1), 83–93. https://doi.org/10.1080/09718923.2014.11893239
- Diener, E., Oishi, S., & Park, J. (2014). An incomplete list of eminent psychologists of the modern era. Archives of Scientific Psychology, 2(1), 20–31. https://doi.org/10.1037/arc0000006
- Diko, N. N., & Letseka, M. (2009). Policy appropriation in teacher retention and attrition: The case of North-West Province. *Perspectives in Education*, *27*(3), 228–236.
- Dinham, J., Chalk, B., Beltman, S., Glass, C., & Nguyen, B. (2017). Pathways to resilience: How drawings reveal pre-service teachers' core narratives underpinning their future teacher-selves. *Asia-Pacific Journal of Teacher Education*, 45(2), 126–144. https://doi.org/10.1080/1359866X.2016.1204424
- Dion, D. (2003). Evidence and inference in the comparative case study. In G. Goertz & H. Starr (Eds.), *Necessary conditions: Theory, methodology, and applications* (pp. 95– 112). Rowman & Littlefield.
- Dirlik, A. (2007). Global South: Predicament and promise. *The Global South, 1*(1), 12–23. https://muse.jhu.edu/article/398223/pdf
- Donnellan, M. B., Trzesniewski, K. H., & Lucas, R. E. (2011). Introduction. In K. H. Trzesniewski, M. B. Donnellan, & R. E. Lucas (Eds.), *Secondary data analysis: An introduction for psychologists* (pp. 3–10). American Psychological Association. https://doi.org/10.1037/12350-000



- Duckworth, A. L., Peterson, C., Matthews, M. D., & Kelly, D. R. (2007). Grit: Perseverance and passion for long-term goals. *Journal of Personality and Social Psychology*, *92*(6), 1087–1101. https://doi.org/10.1037/0022-3514.92.6.1087
- Duffin, L. C., French, B. F., & Patrick, H. (2012). The Teachers' Sense of Efficacy Scale: Confirming the factor structure with beginning pre-service teachers. *Teaching and Teacher Education, 28*(6), 827–834. https://doi.org/10.1016/j.tate.2012.03.004
- Dunlosky, J., & Rawson, K. A. (2012). Overconfidence produces underachievement: Inaccurate self evaluations undermine students' learning and retention. *Learning and Instruction*, 22(4), 271–280. https://doi.org/10.1016/j.learninstruc.2011.08.003
- Dunn, H. S. (2020). Creative resilience and globalization from within: Evolving constructs for analysing culture, innovation, and enterprise in the global south. *Annals of the International Communication Association, 44*(1), 4–18. https://doi.org/10.1080/23808985.2018.1547121
- Dunning, D. (2011). The Dunning–Kruger effect: On being ignorant of one's own ignorance.
  In J. M. Olson & M. P. Zanna (Eds.), *Advances in experimental social psychology* (Vol. 44, pp. 247–296). Academic Press.
- Durrheim, K., & Painter, D. (2014). Collecting quantitative data: Sampling and measuring. In
   M. Terre Blanche, K. Durrheim, & D. Painter (Eds.). *Research in practice: Applied methods for the social sciences* (2nd ed., pp. 131–159). UCT Press.
- Easterly, R. G., III, & Myers, B. E. (2018). Personal resilience as a predictor of professional development engagement and career satisfaction of agriscience teachers. *Journal of Agricultural Education*, *59*(1), 119–134. https://doi.org/10.5032/jae.2018.01119
- Eastman, C., & Marzillier, J. S. (1984). Theoretical and methodological difficulties in Bandura's self-efficacy theory. *Cognitive Therapy and Research, 8*(3), 213–229. https://doi.org/10.1007/BF01172994
- Eaton, N. R., & Krueger, R. F. (2011). Secondary data analysis in psychopathology research. In K. H. Trzesniewski, M. B. Donnellan, & R. E. Lucas (Eds.), Secondary data analysis: An introduction for psychologists (pp. 149–162). American Psychological Association. https://doi.org/10.1037/12350-009
- Ebersöhn, L. (2007). Voicing perceptions of risk and protective factors in coping in a HIV&AIDS landscapes reflecting on capacity for adaptiveness. *Gifted Education International, 23*(2), 149–159. https://doi.org/10.1177/026142940702300205
- Ebersöhn, L. (2012). Adding 'flock' to 'fight and flight': A honeycomb of resilience where supply of relationships meets demand for support. *Journal of Psychology in Africa, 22*(1), 29–42. https://doi.org/10.1080/14330237.2012.10874518



- Ebersöhn, L. (2014). Teacher resilience: Theorizing resilience and poverty. *Teachers and Teaching: Theory and Practice, 20*(5), 568–594. https://doi.org/10.1080/13540602.2014.937960
- Ebersöhn, L. (2015). Making sense of place in school-based intervention research. *Contemporary Educational Psychology, 40,* 121–130. https://doi.org/10.1016/j.cedpsych.2014.10.004
- Ebersöhn, L. (2016). Enabling spaces in education research: An agenda for impactful, collective evidence to support all to be first among un-equals. South African Journal of *Education*, *36*(4), Article 1390. https://doi.org/10.15700/saje.v36n4a1390
- Ebersöhn, L. (2017). A resilience, health and well-being lens for education and poverty. South African Journal of Education, 37(1), Article 1392. https://doi.org/10.15700/saje.v37n1a1392
- Ebersöhn, L. (2019a). *Flocking together: An indigenous psychology theory of resilience in Southern Africa*. Springer. https://doi.org/10.1007/978-3-030-16435-5
- Ebersöhn, L. (2019b). Training educational psychology professionals for work engagement in a context of inequality and trauma in South Africa. *South African Journal of Education*, 39(Suppl. 2), Article 1716. https://doi.org/10.15700/saje.v39ns2a1716
- Ebersöhn, L., & Ferreira, R. (2012). Rurality and resilience in education: Place-based partnerships and agency to moderate time and space constraints. *Perspectives in Education, 30*(1), 30–42.
- Ebersöhn, L., Ferreira, R., Graham, M. A., Versfeld, J., Bosch, Z., Seaworyeh, I., & Tomlinson, J. (2020). *The Isithebe social connectedness intervention study* (Report prepared for The Synergos Institute). Centre for the Study of Resilience, University of Pretoria. https://www.up.ac.za/media/shared/228/Isithebe/final-synergos-report-21-feb-2020.zp195143.pdf
- Ebersöhn, L., & Loots, T. (2017). Teacher agency in challenging contexts as a consequence of social support and resource management. *International Journal of Educational Development, 53,* 80–91. https://doi.org/10.1016/j.ijedudev.2016.11.005
- Ebersöhn, L., Sefotho, M., Mampane, R., Loots, T., Omidire, F., Sherman, V., & Nxumalo-Tsebe, T. (2014). *Imbeleko Report: Cultivating resourcefulness, not dependency*.
  Better Care Network. http://www.bettercarenetwork.org/sites/default/files/Imbeleko%20Report%20-%20Cultivating%20Resourcefulness,%20Not%20Dependency.pdf
- Ee, J., & Chang, A. (2010). How resilient are our graduate trainee teachers in Singapore? *The Asia-Pacific Education Researcher, 19*(2), 321–331. https://repository.nie.edu.sg/bitstream/10497/14283/1/TAPER-19-2-321.pdf



- Eriksson, M., & Lindström, B. (2006). Antonovsky's sense of coherence scale and the relation with health: A systematic review. *Journal of Epidemiology & Community Health, 60*(5), 376–381. https://doi.org/10.1136/jech.2005.041616
- Esser, F., & Vliegenthart, R. (2017). Comparative research methods. In C. S. Davis & R. F. Potter (Eds.), *The international encyclopedia of communication research methods* (pp. 1–22). John Wiley & Sons. https://doi.org/10.1002/9781118901731.iecrm0035
- Evans-Palmer, T. (2010). The potency of humor and instructional self-efficacy on art teacher stress. *Studies in Art Education, 52*(1), 69–83. https://doi.org/10.1080/00393541.2010.11518824
- Evans-Palmer, T. (2016). Building dispositions and self-efficacy in preservice art teachers. *Studies in Art Education, 57*(3), 265–278. https://doi.org/10.1080/00393541.2016.1177364
- Fackler, S., & Malmberg, L. E. (2016). Teachers' self-efficacy in 14 OECD countries: Teacher, student group, school and leadership effects. *Teaching and Teacher Education, 56,* 185–195. https://doi.org/10.1016/j.tate.2016.03.002
- Faculty of Education, University of Pretoria. (2016). *BEd programme information*. University of Pretoria. https://www.up.ac.za/media/shared/6/admission-requirements-and-programme-information.zp88708.pdf
- Fantom, N. J., & Serajuddin, U. (2016). *The World Bank's classification of countries by income* (World Bank Policy Research Working Paper [WPS] 7528). World Bank Group. https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=2741183
- Faul, F., Erdfelder, E., Lang, A. G., & Buchner, A. (2007). G\*Power 3: A flexible statistical power analysis program for the social, behavioral, and biomedical sciences. *Behavior Research Methods*, 39(2), 175–191. https://doi.org/10.3758/BF03193146
- Ferreira, R., & Ebersöhn, L. (2012). Partnering for resilience. Van Schaik.
- Field, A. (2018). Discovering statistics using IBM SPSS statistics (5th ed.). Sage.
- Fischer, B., Bisterfeld, M., & Staab, O. (2018). Individual's patterns of commitment, resilience and subjective well-being of prospective physical education teachers. *Australian Journal of Teacher Education*, 43(2), 39–55. https://doi.org/10.14221/ajte.2018v43n2.3
- Fives, H., & Buehl, M. M. (2010). Teachers' articulation of beliefs about teaching knowledge: Conceptualizing a belief framework. In L. Bendixen & F. Feucht (Eds.), *Personal epistemology in the classroom: Theory, research, and implications for practice* (pp. 470–515). Cambridge University Press. https://doi.org/10.1017/CBO9780511691904.015.



- Flannery, R. B., Jr., & Flannery, G. J. (1990). Sense of coherence, life stress, and psychological distress: A prospective methodological inquiry. *Journal of Clinical Psychology, 46*(4), 415–420. https://doi.org/10.1002/1097-4679(199007)46:4<415::AID-JCLP2270460407>3.0.CO;2-A
- Fleming, J. L., Mackrain, M., & LeBuffe, P. A. (2013). Caring for the caregiver: Promoting the resilience of teachers. In S. Goldstein & R. Brooks (Eds.), *Handbook of resilience in children* (2nd ed., pp. 387–397). Springer. https://doi.org/10.1007/978-1-4614-3661-4\_22
- Fletcher, D., & Sarkar, M. (2013). Psychological resilience: A review and critique of definitions, concepts, and theory. *European Psychologist*, 18(1), 12–23. https://doi.org/10.1027/1016-9040/a000124
- Flores, M. A. (2018). Teacher resilience in adverse contexts: Issues of professionalism and professional identity. In M. Wosnitza, F. Peixoto, S. Beltman, & C. F. Mansfield (Eds.), *Resilience in education: Concepts, contexts and connections* (pp. 167–184). Springer International Publishing. https://doi.org/10.1007/978-3-319-76690-4\_10
- Flyvbjerg, B. (2011). Case study. In N. K. Denzin & Y. S. Lincoln (Eds.), *The Sage handbook of qualitative research* (4th ed., pp. 301–316). Sage.
- Foxcroft, C., & Roodt, G. (Eds.). (2013). *Introduction to psychological assessment in the Southern African context* (4th ed.). Oxford University Press Southern Africa.
- Fraenkel, J. R., Wallen, N. E., & Hyun, H. H. (2012). How to design and evaluate research in education (8th ed.). McGraw-Hill. https://saochhengpheng.files.wordpress.com/2017/03/jack\_fraenkel\_norman\_wallen\_h elen\_hyun-how\_to\_design\_and\_evaluate\_research\_in\_education\_8th\_edition\_mcgraw-hill\_humanities\_social\_sciences\_languages2011.pdf
- Francis, D., & Webster, E. (2019). Poverty and inequality in South Africa: Critical reflections. *Development Southern Africa*, 36(6), 788–802. https://doi.org/10.1080/0376835X.2019.1666703
- Fraser, J. D. C. (2006). Mediation of learning. In M. M. Nieman, & R. B. Monyai (Eds.), *The educator as mediator of learning* (pp. 1–21). Van Schaik Publishers.
- Fredrickson, B. L. (2001). The role of positive emotions in positive psychology: The broadenand-build theory of positive emotions. *American Psychologist*, 56(3), 218–226. https://doi.org/10.1037/0003-066X.56.3.218
- Fredrickson, B. L., Tugade, M. M., Waugh, C. E., & Larkin, G. R. (2003). What good are positive emotions in crisis? A prospective study of resilience and emotions following the terrorist attacks on the United States on September 11th, 2001. *Journal of Personality* and Social Psychology, 84(2), 365–376.



- Gardiner, M. (2008). Education in rural areas. *Issues in Education Policy, 4,* 1–32. https://www.saide.org.za/resources/Library/Gardiner,%20M%20-%20Education%20in%20Rural%20Areas.pdf
- Gavora, P. (2010). Slovak pre-service teacher self-efficacy: Theoretical and research considerations. *The New Educational Review, 21*(2), 17–30. https://pdfs.semanticscholar.org/f284/be870680436674ae2c02038c595e43fded01.pdf
- Ghaith, G., & Yaghi, H. (1997). Relationships among experience, teacher efficacy, and attitudes toward the implementation of instructional innovation. *Teaching and Teacher Education, 13*(4), 451–458. https://doi.org/10.1016/S0742-051X(96)00045-5
- Ghazali, D. (2008, April). Kesahan dan kebolehpercayaan dalam kajian kuantitatif dan kualitatif [Validity and reliability in quantitative and qualitative studies]. *Jurnal Institut Perguruan Islam*.
- Gibbons, J. D., & Chakraborti, S. (2010). *Nonparametric statistical inference* (5th ed.). Taylor and Francis.
- Gibbs, S., & Miller, A. (2014). Teachers' resilience and well-being: A role for educational psychology. *Teachers and Teaching: Theory and Practice, 20*(5), 609–621. https://doi.org/10.1080/13540602.2013.844408
- Gibson, S., & Dembo, M. H. (1984). Teacher efficacy: A construct validation. Journal of Educational Psychology, 76(4), 569–582. https://doi.org/10.1037/0022-0663.76.4.569
- Gignac, G. E., & Zajenkowski, M. (2020). The Dunning-Kruger effect is (mostly) a statistical artefact: Valid approaches to testing the hypothesis with individual differences data. *Intelligence, 80,* Article 101449. https://doi.org/10.1016/j.intell.2020.101449
- Gogolin, I. (2002). Linguistic and cultural diversity in Europe: A challenge for educational research and practice. *European Educational Research Journal, 1*(1), 123–138. https://doi.org/10.2304/eerj.2002.1.1.3

González, J. M. (Ed.). (2008). Encyclopedia of bilingual education (Vol. 2). Sage.

- Gonzalez-DeHass, A. R., & Willems, P. P. (2016). Nurturing self-regulated learners: Teacher, peer, and parental support of strategy instruction. *The Educational Forum*, *80*(3), 294–309. https://doi.org/10.1080/00131725.2016.1173751
- Gravetter, F. J., & Forzano, L. A. B. (2018). *Research methods for the behavioral sciences* (6th ed.). Cengage Learning.
- Grovogui, S. (2011). A revolution nonetheless: The Global South in international relations [Special issue]. *The Global South, 5*(1). https://doi.org/10.2979/globalsouth.5.1.175



- Gu, Q. (2014). The role of relational resilience in teachers' career-long commitment and effectiveness. *Teachers and Teaching: Theory and Practice, 20*(5), 502–529. https://doi.org/10.1080/13540602.2014.937961
- Gu, Q. (2018). (Re)conceptualising teacher resilience: A social-ecological approach to understanding teachers' professional worlds. In M. Wosnitza, F. Peixoto, S. Beltman, & C. F. Mansfield (Eds.), *Resilience in education: Concepts, contexts and connections* (pp. 13–33). Springer International Publishing. https://doi.org/10.1007/978-3-319-76690-4
- Gu, Q., & Day, C. (2007). Teachers resilience: A necessary condition for effectiveness. *Teaching and Teacher Education, 23*(8), 1302–1316. https://doi.org/10.1016/j.tate.2006.06.006
- Gu, Q., & Day, C. (2013). Challenges to teacher resilience: Conditions count. British Educational Research Journal, 39(1), 22–44. https://doi.org/10.1080/01411926.2011.623152
- Gu, Q., & Li, Q. (2013). Sustaining resilience in times of change: Stories from Chinese teachers. Asia-Pacific Journal of Teacher Education, 41(3), 288–303. https://doi.org/10.1080/1359866X.2013.809056
- Guo, Y., Justice, L. M., Sawyer, B., & Tompkins, V. (2011). Exploring factors related to preschool teachers' self-efficacy. *Teaching and Teacher Education, 27*(5), 961–968. https://doi.org/10.1016/j.tate.2011.03.008
- Guskey, T. R. (1988). Teacher efficacy, self-concept, and attitudes toward the implementation of instructional innovation. *Teaching and Teacher Education, 4*(1), 63–69. https://doi.org/10.1016/0742-051X(88)90025-X
- Gustafsson, M., & Nuga Deliwe, C. (2020). How is the COVID-19 pandemic affecting educational quality in South Africa?: Evidence to date and future risks. Stellenbosch University. https://cramsurvey.org/wp-content/uploads/2020/07/Gustafsson.-Nuga.-How-is-the-COVID-19-pandemic-affecting-educational-quality-in-South-Africa\_-1.pdf
- Hansson, K., & Cederblad, M. (2004). Sense of coherence as a meta-theory for salutogenic family therapy. *Journal of Family Psychotherapy*, *15*(1-2), 39–54. https://doi.org/10.1300/J085v15n01\_04
- Harber, C., & Mncube, V. (2011). Is schooling good for the development of society?: The case of South Africa. South African Journal of Education, 31(2), 233–245. https://doi.org/10.15700/saje.v31n2a483
- Hartman, J. L., Turner, M. G., Daigle, L. E., Exum, M. L., & Cullen, F. T. (2009). Exploring the gender differences in protective factors: Implications for understanding resiliency. *International Journal of Offender Therapy and Comparative Criminology*, 53(3), 249–277. https://doi.org/10.1177/0306624X08326910



- Harvey, N. (1997). Confidence in judgment. *Trends in Cognitive Sciences, 1*(2), 78–82. https://doi.org/10.1016/S1364-6613(97)01014-0
- Hays, D. G., & Singh, A. A. (2011). *Qualitative inquiry in clinical and educational settings*. Guilford Press.
- Heale, R., & Twycross, A. (2015). Validity and reliability in quantitative studies. *Evidence-Based Nursing*, *18*(3), 66–67. https://doi.org/10.1136/eb-2015-102129
- Heneman, H. G., III, Kimball, S., & Milanowski, A. (2006). The Teacher Sense of Efficacy Scale: Validation evidence and behavioral prediction (WCER Working Paper No. 2006-7). University of Wisconsin–Madison, Wisconsin Center for Education Research. https://files.eric.ed.gov/fulltext/ED496313.pdf
- Hewitt, S., Buxton, S., & Thomas, A. (2017). A new way of learning: How can an understanding of self-efficacy and emotional resilience be used to develop successful teaching and learning strategies for students in initial teacher education. *Teacher Education Advancement Network Journal, 9*(1), 22–35. http://ojs.cumbria.ac.uk/index.php/TEAN/article/view/345/480
- Higher Education Data Analyzer. (n.d.). https://www.heda.co.za/PowerHEDA/dashboard.aspx
- Hong, J., Greene, B., Roberson, R., Francis, D. C., & Keenan, L. R. (2018). Variations in preservice teachers' career exploration and commitment to teaching. *Teacher Development*, 22(3), 408–426. https://doi.org/10.1080/13664530.2017.1358661
- Hong, J. Y. (2012). Why do some beginning teachers leave the school, and others stay? Understanding teacher resilience through psychological lenses. *Teachers and Teaching: Theory and Practice, 18*(4), 417–440. https://doi.org/10.1080/13540602.2012.696044

Howell, K. E. (2013). An introduction to the philosophy of methodology. Sage.

- Howie, S., Van Staden, S, Tshele, M., Dowse, C., & Zimmerman, L. (2012). PIRLS 2011 Summary Report: South African children's reading literacy achievement. Centre or Evaluation and Assessment, University of Pretoria. https://repository.up.ac.za/handle/2263/65996
- Howie, S. J., Combrinck, C., Roux, K., Tshele, M., Mokoena, G. M., & McLeod Palane, N. (2016). *PIRLS literacy 2016: South African highlights report*. Centre for Evaluation and Assessment. http://www.shineliteracy.org.za/wp-content/uploads/2018/01/pirls-literacy-2016-hl-report.zp136320.pdf
- Hutchinson, J., & Pretelt, V. (2010). Building resources and resilience: Why we should think about positive emotions when working with children, their families and their schools. *Counselling Psychology Review, 25*(1), 20–27.



IBM Corp. (2020). IBM SPSS Statistics for Windows (Version 27.0) [Computer software].

- IGI Global. (2020). What is pre-service teachers. In *IGI Global.com dictionary*. Retrieved May 13, 2020, from https://www.igi-global.com/dictionary/pre-service-teachers/23201
- International Monetary Fund. (2004). *Health and development: Why investing in health is critical for achieving economic development goals.* https://www.imf.org/external/pubs/ft/health/eng/hdwi/hdwi.pdf
- Jackson, L., & Rothmann, S. (2005). Work-related well-being of educators in a district of the North-West Province. *Perspectives in Education, 23*(3), 107–122.
- Jenkins, R. (2019, October 4). Why we need more female teachers across all levels of education. *UNICEF*. https://blogs.unicef.org/blog/need-more-female-teachers-across-levels-education/
- Jennings, P. A., Frank, J. L., Snowberg, K. E., Coccia, M. A., & Greenberg, M. T. (2013). Improving classroom learning environments by Cultivating Awareness and Resilience in Education (CARE): Results of a randomized controlled trial. *School Psychology Quarterly, 28*(4), 374–390. https://doi.org/10.1037/spq0000035
- Johnson, B., & Down, B. (2013). Critically re-conceptualising early career teacher resilience. *Discourse: Studies in the Cultural Politics of Education, 34*(5), 703–715. https://doi.org/10.1080/01596306.2013.728365
- Johnson, B., & Gray, R. (2010). A history of philosophical and theoretical issues for mixed methods research. In A. Tashakkori & C. Teddlie (Eds.), Sage handbook of mixed methods in social and behavioral research (2nd ed., pp. 69–94). Sage.
- Johnston, M. P. (2017). Secondary data analysis: A method of which the time has come. *Qualitative and Quantitative Methods in Libraries, 3*(3), 619–626. http://www.qqmljournal.net/index.php/qqml/article/view/169/170
- Jordan, J. V. (2006). Relational resilience in girls. In S. Goldstein & R. B. Brooks (Eds.), Handbook of resilience in children. Springer.
- Jordan, J. V. (Ed.). (2013). *The power of connection: Recent developments in relationalcultural theory.* Taylor and Francis.
- Kang, H. (2013). The prevention and handling of the missing data. *Korean Journal of Anesthesiology, 64*(5), 402–406. https://doi.org/10.4097/kjae.2013.64.5.402
- Kavita, K., & Hassan, N. C. (2018). Work stress among teachers: A comparison between primary and secondary school teachers. *International Journal of Academic Research in Progressive Education and Development, 7*(4), 60–66. https://doi.org/10.6007/IJARPED/v7-i4/4802



- Kazdin, A. E. (1978). Conceptual and assessment issues raised by self-efficacy theory. *Advances in Behaviour Research and Therapy*, 1(4), 177–185. https://doi.org/10.1016/0146-6402(78)90005-X
- Kelly, D. R., Matthews, M. D., & Bartone, P. T. (2014). Grit and hardiness as predictors of performance among West Point cadets. *Military Psychology*, 26(4), 327–342. https://doi.org/10.1037/mil0000050
- Kidger, J., Brockman, R., Tilling, K., Campbell, R., Ford, T., Araya, R., King, M., & Gunnell, D. (2016). Teachers' wellbeing and depressive symptoms, and associated risk factors: A large cross sectional study in English secondary schools. *Journal of Affective Disorders, 192,* 76–82. https://doi.org/10.1016/j.jad.2015.11.054
- Kim, H., & Cho, Y. (2014). Pre-service teachers' motivation, sense of teaching efficacy, and expectation of reality shock. *Asia-Pacific Journal of Teacher Education*, 42(1), 67–81. https://doi.org/10.1080/1359866X.2013.855999
- Kirk, J., & Wall, C. (2010). Resilience and loss in work identities: A narrative analysis of some retired teacher's work-life histories. *British Educational Research Journal, 36*(4), 627–641. https://doi.org/10.1080/01411920903018216
- Kitching, K., Morgan, M., & O'Leary, M. (2009). It's the little things: Exploring the importance of commonplace events for early-career teachers' motivation. *Teachers and Teaching: Theory and Practice, 15*(1), 43–58. https://doi.org/10.1080/13540600802661311
- Klassen, R. M., & Durksen, T. L. (2014). Weekly self-efficacy and work stress during the teaching practicum: A mixed methods study. *Learning and Instruction*, 33, 158–169. https://doi.org/10.1016/j.learninstruc.2014.05.003
- Klassen, R. M., Tze, V. M. C., Betts, S. M., & Gordon, K. A. (2011). Teacher efficacy research 1998–2009: Signs of progress or unfulfilled promise? *Educational Psychology Review*, 23(1), 21–43. https://doi.org/10.1007/s10648-010-9141-8
- Kleinsasser, R. C. (2014). Teacher efficacy in *teaching and teacher education*. *Teaching and Teacher Education*, *44*, 168–179. https://doi.org/10.1016/j.tate.2014.07.007
- Knight, C. (2007). A resilience framework: Perspectives for educators. *Health Education*, *107*(6), 543–555. https://doi.org/10.1108/09654280710827939
- Kobasa, S. C. (1979). Stressful life events, personality, and health: An inquiry into hardiness. Journal of Personality and Social Psychology, 37(1), 1–11. https://doi.org/10.1037/0022-3514.37.1.1
- Kobasa, S. C., Maddi, S. R., & Kahn, S. (1982). Hardiness and health: A prospective study. Journal of Personality and Social Psychology, 42(1), 168–177. https://doi.org/10.1037/0022-3514.42.1.168



- Kruger, J., & Dunning D. (1999). Unskilled and unaware of it: How difficulties in recognizing one's own incompetence lead to inflated self-assessments. *Journal of Personality and Social Psychology*, 77(6), 1121–1134. https://doi.org/10.1037/0022-3514.77.6.1121
- Kruger, P., & Janeke, H. C. (2011). *Psychological research: Only study guide for PYC3704.* Unisa.
- Kruskal, W. H., & Wallis, W. A. (1952). Use of ranks in one-criterion variance analysis. Journal of the American Statistical Association, 47(260), 583–621. https://doi.org/10.1080/01621459.1952.10483441
- Kunnari, I., Ilomäki, L., & Toom, A. (2018). Successful teacher teams in change: The role of collective efficacy and resilience. *International Journal of Teaching and Learning in Higher Education*, 30(1), 111–126. https://files.eric.ed.gov/fulltext/EJ1169801.pdf
- Kuo, B. C. H. (2013). Collectivism and coping: Current theories, evidence, and measurements of collective coping. *International Journal of Psychology*, 48(3), 374– 388. https://doi.org/10.1080/00207594.2011.640681
- Kyriacou, C. (2001). Teacher stress: Directions for future research. *Educational Review*, 53(1), 27–35. https://doi.org/10.1080/00131910120033628
- Kyriakides, L., Christoforou, C., & Charalambous, C. Y. (2013). What matters for student learning outcomes: A meta-analysis of studies exploring factors of effective teaching. *Teaching and Teacher Education, 36,* 143–152. https://doi.org/10.1016/j.tate.2013.07.010
- Lan, G., Kamalski, J., Lau, G., Baas, J., Blom, A., & Adil, M. (2014). A decade of development in Sub-Saharan African science, technology, engineering and mathematics research (A report by the World Bank and Elsevier). World Bank. https://documents1.worldbank.org/curated/en/237371468204551128/pdf/910160WP0P12 6900disclose09026020140.pdf
- Le Cornu, R. (2009). Building resilience in pre-service teachers. *Teaching and Teacher Education, 25*(5), 717–723. https://doi.org/10.1016/j.tate.2008.11.016
- Le Cornu, R. (2013). Building early career teacher resilience: The role of relationships. *Australian Journal of Teacher Education, 38*(4), 1–16. https://doi.org/10.14221/ajte.2013v38n4.4
- Lee, R. E. (2018). Breaking down barriers and building bridges: Transformative practices in community- and school-based urban teacher preparation. *Journal of Teacher Education, 69*(2), 118–126. https://doi.org/10.1177/0022487117751127
- Leedy, P. D., & Ormrod, J. E. (2015). *Practical research: Planning and design* (11th ed.). Pearson Education.



- Legrain, P., Escalié, G., Lafont, L., & Chaliès, S. (2019). Cooperative learning: A relevant instructional model for physical education pre-service teacher training? *Physical Education and Sport Pedagogy, 24*(1), 73–86. https://doi.org/10.1080/17408989.2018.1561838
- Lemon, N., & Garvis, S. (2016). Pre-service teacher self-efficacy in digital technology. *Teachers and Teaching: Theory and Practice, 22*(3), 387–408. https://doi.org/10.1080/13540602.2015.1058594
- Lent, R. W., & Brown, S. D. (2006). On conceptualizing and assessing social cognitive constructs in career research: A measurement guide. *Journal of Career Assessment*, *14*(1), 12–35. https://doi.org/10.1177/1069072705281364
- Letseka, M. (2012). In defence of Ubuntu. *Studies in Philosophy and Education, 31*(1), 47–60. https://doi.org/10.1007/s11217-011-9267-2
- Letseka, M. (2013). Educating for *ubuntu/botho*: Lessons from Basotho indigenous education. *Open Journal of Philosophy*, *3*(2), 337–344. https://doi.org/10.4236/ojpp.2013.32051
- Letzring, T. D., Block, J., & Funder, D. C. (2005). Ego-control and ego-resiliency: Generalization of self-report scales based on personality descriptions from acquaintances, clinicians, and the self. *Journal of Research in Personality*, 39(4), 395– 422. https://doi.org/10.1016/j.jrp.2004.06.003
- Lieberman, A., & Darling-Hammond, L. (Eds.). (2012). *High quality teaching and learning: International perspectives on teacher education.* Routledge.
- Lincoln, Y. S., Lynham, S. A., & Guba, E. G. (2011). Paradigmatic controversies, contradictions, and emerging confluences, revisited. In N. K. Denzin & Y. S. Lincoln (Eds.), *The Sage handbook of qualitative research* (4th ed., pp. 97–128). Sage.
- Logan, T. (2020). A practical, iterative framework for secondary data analysis in educational research. *The Australian Educational Researcher, 47*(1), 129–148. https://doi.org/10.1007/s13384-019-00329-z
- Lohbeck, L. (2018). The interplay between the motivation to teach and resilience of student teachers and trainee teachers. In M. Wosnitza, F. Peixoto, S. Beltman, & C. F. Mansfield (Eds.), *Resilience in education: Concepts, contexts and connections* (pp. 93–106). Springer International Publishing. https://doi.org/10.1007/978-3-319-76690-4\_6
- López, X., Valenzuela, J., Nussbaum, M., & Tsai, C. C. (2015). Editorial: Some recommendations for the reporting of quantitative studies. *Computers & Education, 91,* 106–110. https://doi.org/10.1016/j.compedu.2015.09.010
- Lor, P. (2011). International and comparative librarianship. De Gruyter Saur.



- Lowe, G. M., & Prout, P. F. (2019). Reframing teacher in-service training in Kenya: Recommendations from the literature. *Africa Education Review, 16*(3), 54–66. https://doi.org/10.1080/18146627.2017.1340803
- Luthar, S. S., Cicchetti, D., & Becker, B. (2000). The construct of resilience: A critical evaluation and guidelines for future work. *Child Development*, *71*(3), 543–562. https://doi.org/10.1111/1467-8624.00164
- MacInnes, J. (2016). An introduction to secondary data analysis with IBM SPSS statistics. Sage.
- Mackenzie, S. (2012). I can't imagine doing anything else: Why do teachers of children with SEN remain in the profession? Resilience, rewards and realism over time. *Journal of Research in Special Educational Needs*, *12*(3), 151–161. https://doi.org/10.1111/j.1471-3802.2011.01221.x
- Macupe, B. (2018, January 26). 'Varsity doesn't prepare us to teach in rural areas'. Mail & Guardian. https://mg.co.za/article/2018-01-26-00-varsity-doesnt-prepare-us-to-teach-in-rural-areas
- Maddi, S. R., Matthews, M. D., Kelly, D. R., Villarreal, B. J., Gundersen, K. K., & Savino, S. C. (2017). The continuing role of hardiness and grit on performance and retention in West Point cadets. *Military Psychology, 29*(5), 355–358. https://doi.org/10.1037/mil0000145
- Magubane, K. (2019, June 20). SA's inequality is not just staying put, it's on the increase Moody's report. Fin24. https://www.fin24.com/Economy/South-Africa/sas-inequality-isnot-just-staying-put-its-on-the-increase-moodys-report-20190620
- Mahmood, K. (2016). Do people overestimate their information literacy skills? A systematic review of empirical evidence on the Dunning-Kruger effect. *Communications in Information Literacy, 10*(2):198–213. https://doi.org/10.15760/comminfolit.2016.10.2.24
- Makalela, L. (2015). Breaking African language boundaries: Student teachers' reflections on translanguaging practices. *Language Matters*, 46(2), 275–292. https://doi.org/10.1080/10228195.2014.986664
- Makalela, L. (Ed.). 2018. *Shifting lenses: Multilanguaging, decolonisation and education in the global South*. Centre for Advanced Studies of African Society (CASAS).
- Mammadov, R., & Çimen, İ. (2019). Optimizing teacher quality based on student performance: A data envelopment analysis on PISA and TALIS. *International Journal* of Instruction, 12(4), 767–788. https://doi.org/10.29333/iji.2019.12449a



- Mampane, R., & Bouwer, C. (2006). Identifying resilient and non-resilient middle-adolescents in a formerly black-only urban school. South African Journal of Education, 26(3), 443– 456. http://www.sajournalofeducation.co.za/index.php/saje/article/view/89
- Mann, H. B., & Whitney, D. R. (1947). On a test of whether one of two random variables is stochastically larger than the other. *The Annals of Mathematical Statistics, 18*(1), 50–60. https://doi.org/10.1214/aoms/1177730491
- Mansfield, C., Beltman, S., & Price, A. (2014). 'I'm coming back again!' The resilience process of early career teachers. *Teachers and Teaching: Theory and Practice, 20*(5), 547–567. https://doi.org/10.1080/13540602.2014.937958
- Mansfield, C., Beltman, S., Weatherby-Fell, N., & Broadley, T. (2016). Classroom ready?
  Building resilience in teacher education. In R. Brandenberg, S. McDonough, J. Burke,
  & S. White (Eds.), *Teacher education: Innovation, intervention and impact* (pp. 211–229). Springer. https://doi.org/10.1007/978-981-10-0785-9
- Mansfield, C. F., Beltman, S., Price, A., & McConney, A. (2012). "Don't sweat the small stuff:" Understanding teacher resilience at the chalkface. *Teaching and Teacher Education*, *28*(3), 357–367. https://doi.org/10.1016/j.tate.2011.11.001
- Mansfield, C. F., Ebersöhn, L., Beltman, S., & Loots, T. (2018). Great southern lands: Making space for teacher resilience in South Africa and Australia. In M. Wosnitza, F. Peixoto, S. Beltman, & C. F. Mansfield (Eds.), *Resilience in education: Concepts, contexts and connections* (pp. 53–71). Springer International Publishing. https://doi.org/10.1007/978-3-319-76690-4
- Mansfield, C. F., & Wosnitza, M. (2015). *Teacher Resilience Questionnaire*—Version 1.5. Murdoch University.
- Mapfumo, J. S., Chitsiko, N., & Chireshe, R. (2012). Teaching Practice generated stressors and coping mechanisms among student teachers in Zimbabwe. South African Journal of Education, 32(2), 155–166. https://doi.org/10.15700/saje.v32n2a601
- Maphalala, M. & Mpofu, N. (2019, November 13). South Africa must up its game and produce more teachers. The Conversation. https://theconversation.com/south-africamust-up-its-game-and-produce-more-teachers-125752
- Maree, K., & Pietersen, J. (2019a). Sampling. In K. Maree (Ed.), *First steps in research* (3rd ed., pp. 213–224). Van Schaik Publishers.
- Maree, K., & Pietersen, J. (2019b). Surveys and the use of questionnaires. In K. Maree (Ed.), *First steps in research* (3rd ed., pp. 195–212). Van Schaik Publishers.
- Maree, K., & Pietersen, J. (2019c). The quantitative research process. In K. Maree (Ed.), *First steps in research* (3rd ed., pp. 183–194). Van Schaik Publishers.



- Masten, A. S. (2001). Ordinary magic: Resilience processes in development. *American Psychologist, 56*(3), 227–238.
- Masten, A. S. (2011). Resilience in children threatened by extreme adversity: Frameworks for research, practice and translational synergy [Special section]. *Development and Psychopathology*, *23*(2), 493–506. https://doi.org/10.1017/S0954579411000198
- Masten, A. S. (2014). Ordinary magic: Resilience in development. The Guilford Press.
- Masten, A. S. (2018). Resilience theory and research on children and families: Past, present, and promise [Special issue]. *Journal of Family Theory & Review, 10*(1). https://doi.org/10.1111/jftr.12255
- Masten, A. S., Best, K. M., & Garmezy, N. (1990). Resilience and development: Contributions from the study of children who overcome adversity. *Development and Psychopathology*, 2(4), 425–444. https://doi.org/10.1017/S0954579400005812
- Masten, A. S., & Reed, M. J. (2005). Resilience in development. In C. R. Snyder & S. J. Lopez (Eds.), *Handbook of positive psychology* (2nd ed., pp. 74–88). Oxford University Press.
- Masten, A. S., & Wright, M. O. D. (2010). Resilience over the lifespan: Developmental perspectives on resistance, recovery and transformation. In J. W. Reich, A. Zautra, & J. S. Hall (Eds.), *Handbook of adult resilience* (pp. 213–236). The Guilford Press.
- Matthews, G. (2016). Distress. In G. Fink (Ed.), *Stress: Concepts, cognition, emotion, and behavior* (Vol. 1, pp. 219–226). Academic Press.
- Mavunga, G. (2019). #FeesMustFall protests in South Africa: A critical realist analysis of selected newspaper articles. *Journal of Student Affairs in Africa*, 7(1), 81–99. https://doi.org/10.24085/jsaa.v7i1.3694
- McClellan, E. (2017). A social-cognitive theoretical framework for examining music teacher identity. Action, Criticism, and Theory for Music Education, 16(2), 65–101. https://doi.org/10.22176/act16.2.65
- McFarland, J., Hussar, B., Wang, X., Zhang, J., Wang, K., Rathbun, A., Barmer, A., Forrest Cataldi, E., & Bullock Mann, F. B. (2018). *The condition of education 2018* (NCES 2018-144). National Center for Education Statistics. https://files.eric.ed.gov/fulltext/ED583502.pdf
- McIntosh, R. D., Fowler, E. A., Lyu, T., & Della Sala, S. (2019). Wise up: Clarifying the role of metacognition in the Dunning-Kruger effect. *Journal of Experimental Psychology: General, 148*(11), 1882–1897. https://doi.org/10.1037/xge0000579
- McKeever, M. (2017). Educational inequality in apartheid South Africa. *American Behavioral Scientist, 61*(1), 114–131. https://doi.org/10.1177/0002764216682988



- McKinney, C., & Guzula, X. (2019, February 5). *How language policy can improve student performance*. Daily Maverick. https://www.dailymaverick.co.za/article/2019-02-05-how-language-policy-can-improve-student-performance/
- McMillan, J. H., & Schumacher, S. (2010). *Research in education: Evidence-based inquiry* (7th ed.). Pearson.
- Meaney, K. S., Bohler, H. R., Kopf, K., Hernandez, L., & Scott, L. S. (2008). Service-learning and pre-service educators' cultural competence for teaching: An exploratory study. *Journal of Experiential Education*, *31*(2), 189–208. https://doi.org/10.1177/105382590803100206
- Mergler, A. G., & Tangen, D. (2010). Using microteaching to enhance teacher efficacy in preservice teachers. *Teaching Education, 21*(2), 199–210. https://doi.org/10.1080/10476210902998466
- Meyer, D. K., & Turner, J. C. (2006). Re-conceptualizing emotion and motivation to learn in classroom contexts. *Educational Psychology Review*, 18(4), 377–390. https://doi.org/10.1007/s10648-006-9032-1
- Mills, M., Van de Bunt, G. G., & De Bruijn, J. (2006). Comparative research: Persistent problems and promising solutions. *International Sociology*, 21(5), 619–631. https://doi.org/10.1177/0268580906067833
- Milner, K., & Khoza, H. (2008). A comparison of teacher stress and school climate across schools with different matric success rates. South African Journal of Education, 28(2), 155–173. http://www.sajournalofeducation.co.za/index.php/saje/article/view/38/102
- Mlachila, M. M., & Moeletsi, T. (2019). Struggling to make the grade: A review of the causes and consequences of the weak outcomes of South Africa's education system (IMF Working Paper No. 19/47). International Monetary Fund. https://www.imf.org/en/Publications/WP/Issues/2019/03/01/Struggling-to-Make-the-Grade-A-Review-of-the-Causes-and-Consequences-of-the-Weak-Outcomes-of-46644
- Moalosi, S. W. T. (2013). Teachers' self efficacy: Is reporting non-significant results essential? *Journal of International Education Research, 9*(4), 397–406. https://doi.org/10.19030/jier.v9i4.8266
- Moè, A., Pazzaglia, F., & Ronconi, L. (2010). When being able is not enough. The combined value of positive affect and self-efficacy for job satisfaction in teaching. *Teaching and Teacher Education*, 26(5), 1145–1153. https://doi.org/10.1016/j.tate.2010.02.010
- Molina, M. L., Fialho, A. R., Amaral, P. L., Bach, S. L., Rocha, L. M., & Souza, L. D. M. (2017). Well-being and associated factors among elementary school teachers in southern Brazil. *Revista CEFAC*, *19*(6), 812–820. https://doi.org/10.1590/1982-021620171962217



- Montiel, C. J. (2018). Peace psychologists and social transformation: A Global South perspective. *Peace and Conflict: Journal of Peace Psychology, 24*(1), 64–70. https://doi.org/10.1037/pac0000290
- Morgan, M. (2011). Resilience and recurring adverse events: Testing an assets-based model of beginning teachers' experiences. *The Irish Journal of Psychology, 32*(3-4), 92–104. https://doi.org/10.1080/03033910.2011.613189
- Morgan, M., Ludlow, L., Kitching, K., O'Leary, M., & Clarke, A. (2010). What makes teachers tick? Sustaining events in new teachers' lives. *British Educational Research Journal*, 36(2), 191–208. https://doi.org/10.1080/01411920902780972
- Moulding, L. R., Stewart, P. W., & Dunmeyer, M. L. (2014). Pre-service teachers' sense of efficacy: Relationship to academic ability, student teaching placement characteristics, and mentor support. *Teaching and Teacher Education*, *41*, 60–66. https://doi.org/10.1016/j.tate.2014.03.007
- Muller, S. M., Dodd, A., & Fiala, K. A. (2014). Comparing protective factors and resilience among classroom-based teachers and community-based educators. *Education, 134,* 548–558.
- Mullis, I. V. S., Kennedy, A. M., Martin, M. O., & Sainsbury, M. (2004). PIRLS 2006 assessment framework and specifications (2nd ed.). TIMSS and PIRLS International Study Center, Lynch School of Education, Boston College. https://timssandpirls.bc.edu/PDF/P06Framework.pdf
- Murshidi, R., Konting, M. M., Elias, H., & Fooi, F. S. (2006). Sense of efficacy among beginning teachers in Sarawak. *Teaching Education*, *17*(3), 265–275. https://doi.org/10.1080/10476210600849730
- Neiss, M. B., Sedikides, C., & Stevenson, J. (2011). Using secondary data to test questions about the genetic basis of behavior. In K. H. Trzesniewski, M. B. Donnellan, & R. E. Lucas (Eds.), Secondary data analysis: An introduction for psychologists (pp. 133–148). American Psychological Association. https://doi.org/10.1037/12350-008
- Neuman, W. L. (2014). Social research methods: Qualitative and quantitative approaches (7th ed.). Pearson.
- Ngidi, D. P., & Ngidi, S. A. (2019). Determination of factors influencing pre-service teachers' sense of self-efficacy. *South African Journal of Higher Education, 33*(5), 98–111. https://doi.org/10.20853/33-5-3598
- Ngqakamba, S. (2019, January 4). *The 'real' matric pass rate is 37.6% DA*. News24. https://www.news24.com/SouthAfrica/News/the-real-matric-pass-rate-is-376-da-20190104



- Nieto, S., & Ramos, R. (2015). Educational outcomes and socioeconomic status: A decomposition analysis for middle-income countries. *Prospects*, *45*(3), 325–343. https://doi.org/10.1007/s11125-015-9357-y
- Nieuwenhuis, J. (2016). Analysing qualitative data. In K. Maree (Ed.), *First steps in research* (2nd ed., pp. 104–131). Van Schaik.
- Nkoana, L. N. (2017). A systematic review of school-based interventions in high risk schools [Master's dissertation, University of Pretoria]. UP Repository. https://repository.up.ac.za/bitstream/handle/2263/67844/Nkoana\_Systematic\_2017.pdf ?sequence=1&isAllowed=y
- Nolan, A., Taket, A., & Stagnitti, K. (2014). Supporting resilience in early years classrooms: The role of the teacher. *Teachers and Teaching: Theory and Practice, 20*(5), 595–608. https://doi.org/10.1080/13540602.2014.937955
- O'Leary, Z. (2017). The essential guide to doing your research project (3rd ed.). Sage.
- Olsen, S. T. (2017). How can a focus on teacher well-being in pre-service training promote the resilience of primary school student teachers? *Journal of the European Teacher Education Network, 12,* 118–125.
- Omidire, M. F., & Ayob, S. (2020). The utilisation of translanguaging for learning and teaching in multilingual primary classrooms. *Multilingua*, 1–25. https://doi.org/10.1515/multi-2020-0072
- O'Neill, S., & Stephenson, J. (2012a). Does classroom management coursework influence pre-service teachers' perceived preparedness or confidence? *Teaching and Teacher Education, 28*(8), 1131–1143. https://doi.org/10.1016/j.tate.2012.06.008
- O'Neill, S., & Stephenson, J. (2012b). Exploring Australian pre-service teachers sense of efficacy, its sources, and some possible influences. *Teaching and Teacher Education*, *28*(4), 535–545. https://doi.org/10.1016/j.tate.2012.01.008
- Organisation for Economic Co-operation and Development. (2009). *Creating effective teaching and learning environments: First results from TALIS*. OECD Publishing. http://www.oecd.org/education/school/43023606.pdf
- Organisation for Economic Co-operation and Development. (2014a). *Education at a glance 2014: OECD indicators.* OECD Publishing. https://doi.org/10.1787/eag-2014-en
- Organisation for Economic Co-operation and Development. (2014b, May). *Education indicators in focus*. OECD Publishing. https://www.oecd.org/education/skills-beyondschool/EDIF\_23%20eng%20(2014)EN.pdf



- Organisation for Economic Co-operation and Development. (2017a). Do new teachers feel prepared for teaching? *Teaching in Focus, 17*. OECD Publishing. https://doi.org/10.1787/980bf07d-en
- Organisation for Economic Co-operation and Development. (2017b). *Education at a glance* 2017: OECD indicators. OECD Publishing. https://doi.org/10.1787/eag-2017-en
- Organisation for Economic Co-operation and Development. (2018). How do primary and secondary teachers compare? *Education Indicators in Focus*, 58. OECD Publishing. https://doi.org/10.1787/535e7f54-en
- Organisation for Economic Co-operation and Development. (2019a). *Education at a glance 2019: OECD indicators*. OECD Publishing.
- Organisation for Economic Co-operation and Development. (2019b). *TALIS initial teacher preparation study.* http://www.oecd.org/education/school/talis-initial-teacher-preparation-study.htm
- Organisation for Economic Co-operation and Development. (2019c). *TALIS 2018 technical report*. OECD Publishing. https://www.oecd.org/education/talis/TALIS\_2018\_Technical\_Report.pdf
- O'Sullivan, M. (2006). Professional lives of Irish physical education teachers: Stories of resilience, respect and resignation. *Physical Education and Sport Pedagogy*, *11*(3), 265–284. https://doi.org/10.1080/17408980600986314
- Özdemir, S. M. (2008). An investigation of prospective primary teachers' self-efficacy beliefs regarding teaching process in terms of certain variables. *Educational Administration: Theory and Practice, 54,* 277–306.
- Panhwar, A. H., Ansari, S., & Shah, A. A. (2017). Post-positivism: An effective paradigm for social and educational research. *International Research Journal of Arts & Humanities*, 45(45), 253–260. https://sujoold.usindh.edu.pk/index.php/IRJAH/article/view/3371/2447
- Papatraianou, L. H., & Le Cornu, R. (2014). Problematising the role of personal and professional relationships in early career teacher resilience. *Australian Journal of Teacher Education, 39*(1), 100–116. https://doi.org/10.14221/ajte.2014v39n1.7
- Pareek, S., & Rathore, N. S. (2016). Gender differences in character strengths and virtues of teachers in higher education. *Indian Journal of Positive Psychology, 7*(3), 312–317.
- Patterson, J. H., Collins, L., & Abbott, G. (2004). A study of teacher resilience in urban schools. *Journal of Instructional Psychology*, *31*(1), 3–11.



- Pavel, S. R., Robertson, M. F., & Harrison, B. T. (2012). The Dunning-Kruger effect and SIUC University's aviation students. *Journal of Aviation Technology and Engineering*, 2(1), 125–129. https://doi.org/10.5703/1288284314864
- Pearce, J., & Morrison, C. (2011). Teacher identity and early career resilience: Exploring the links. Australian Journal of Teacher Education, 36(1), 48–59. https://files.eric.ed.gov/fulltext/EJ920009.pdf
- Peglar, D. J., & Reker, G. T. (2008). Development and validation of the Multidimensional Resilience Inventory (MRI) for older adults. *Gerontologist, 48,* 19–20.
- Peixoto, F., Silva, J. C., Pipa, J., Wosnitza, M., & Mansfield, C. (2020). The Multidimensional Teachers' Resilience Scale: Validation for Portuguese teachers. *Journal of Psychoeducational Assessment*, *38*(3), 402–408. https://doi.org/10.1177/0734282919836853
- Peixoto, F., Wosnitza, M., Pipa, J., Morgan, M., & Cefai, C. (2018). A multidimensional view on pre-service teacher resilience in Germany, Ireland, Malta and Portugal. In M. Wosnitza, F. Peixoto, S. Beltman, & C. F. Mansfield (Eds.), *Resilience in education: Concepts, contexts and connections* (pp. 73–89). Springer International Publishing. https://doi.org/10.1007/978-3-319-76690-4
- Pendergast, D., Garvis, S., & Keogh, J. (2011). Pre-service student-teacher self-efficacy beliefs: An insight into the making of teachers. *Australian Journal of Teacher Education, 36*(12), 46–58. https://doi.org/10.14221/ajte.2011v36n12.6
- Perkins-Gough, D. (2013). The significance of GRIT: A conversation with Angela Lee Duckworth. *Educational Leadership*, *71*(1), 14–20.
- Perlman, D. J., & Pearson, P. (2012). A comparative analysis between primary and secondary teachers: A self-determination perspective. *The Shield: Research Journal of Physical Education and Sports Science, 7*(2012), 8–17. https://ro.uow.edu.au/cgi/viewcontent.cgi?article=1329&context=sspapers
- Peters, J., & Pearce, J. (2012). Relationships and early career teacher resilience: A role for school principals. *Teachers and Teaching*, 18(2), 249–262. https://doi.org/10.1080/13540602.2012.632266
- Petersen, N. (2014). The 'good', the 'bad' and the 'ugly'? Views on male teachers in foundation phase education. *South African Journal of Education, 34*(1), Article 772. https://doi.org/10.15700/201412120926
- Pfitzner-Eden, F. (2016a). I feel less confident so I quit? Do true changes in teacher selfefficacy predict changes in preservice teachers' intention to quit their teaching degree? *Teaching and Teacher Education, 55,* 240–254. https://doi.org/10.1016/j.tate.2016.01.018



- Pfitzner-Eden, F. (2016b). Why do I feel more confident? Bandura's sources predict preservice teachers' latent changes in teacher self-efficacy. *Frontiers in Psychology, 7,* Article 1486. https://doi.org/10.3389/fpsyg.2016.01486
- Pienta, A. M., O'Rourke, J. M., & Franks, M. M. (2011). Getting started: Working with secondary data. In K. H. Trzesniewski, M. B. Donnellan, & R. E. Lucas (Eds.), *Secondary data analysis: An introduction for psychologists* (pp. 13–25). American Psychological Association. https://doi.org/10.1037/12350-001
- Pierre, P. N., & Worrell, F. C. (2003). Differences between elementary and secondary school teachers in Trinidad and Tobago on teacher self-efficacy. *Caribbean Curriculum, 10,* 112–132. https://uwispace.sta.uwi.edu/dspace/bitstream/handle/2139/6630/Caribbean%20Curric ulum%20vol%2010%2c%202003.pdf?sequence=1&isAllowed=y
- Pietersen, J., & Maree, K. (2019a). Overview of some of the most popular statistical techniques. In K. Maree (Ed.), *First steps in research* (3rd ed., pp. 271–326). Van Schaik Publishers.
- Pietersen, J., & Maree, K. (2019b). Standardisation of a questionnaire. In K. Maree (Ed.), *First steps in research* (3rd ed., pp. 259–270). Van Schaik Publishers.
- Pietersen, J., & Maree, K. (2019c). Statistical analysis I: Descriptive statistics. In K. Maree (Ed.), *First steps in research* (3rd ed., pp. 225–239). Van Schaik Publishers.
- Pietersen, J., & Maree, K. (2019d). Statistical analysis I: Inferential statistics. In K. Maree (Ed.), *First steps in research* (3rd ed., pp. 240–258). Van Schaik Publishers.
- Plüddemann, P. (2015). Unlocking the grid: Language-in-education policy realisation in postapartheid South Africa. *Language and Education, 29*(3), 186–199. https://doi.org/10.1080/09500782.2014.994523
- Podgoršek, M., & Lipovec, A. (2017). Self-assessment ability of pre-service teachers. *The New Educational Review, 48*(2), 213–223. https://doi.org/10.15804/tner.2017.48.2.17
- Polat, D. D., & İskender, M. (2018). Exploring teachers' resilience in relation to job satisfaction, burnout, organizational commitment and perception of organizational climate. *International Journal of Psychology and Educational Studies, 5*(3), 1–13. https://doi.org/10.17220/ijpes.2018.03.001
- Price, A., Mansfield, C., & McConney, A. (2012). Considering 'teacher resilience' from critical discourse and labour process theory perspectives. *British Journal of Sociology of Education, 33*(1), 81–95. https://doi.org/10.1080/01425692.2011.614748
- Prince-Embury, S. (2010). Introduction to the special issue: Assessing resiliency in children and adolescents. *Journal of Psychoeducational Assessment, 28*(4), 287–290. https://doi.org/10.1177/0734282910366830



- Prosser, B. (2008). The role of the personal domain in middle years teachers' work. *Australian Journal of Middle Schooling, 8*(2), 11–16.
- Raath, S., & Hay, A. (2016). Self-efficacy: A South African case study on teachers' commitment to integrate climate change resilience into their teaching practices. *Cogent Education, 3*(1), Article 1264698. https://doi.org/10.1080/2331186X.2016.1264698
- Ramlo, S. E., & Newman, I. (2011). Q methodology and its position in the mixed-methods continuum. Operant Subjectivity: The International Journal of Q Methodology, 34(3), 172–191.
- Raudenbush, S. W., Rowan, B., & Cheong, Y. F. (1992). Contextual effects on the selfperceived efficacy of high school teachers. *Sociology of Education*, 65(2), 150–167. https://doi.org/10.2307/2112680
- Republic of South Africa. (1996a). Act No. 84, 1996: South African Schools Act, 1996. *Government Gazette, 377*(17579).
- Republic of South Africa. (1996b). *The Constitution of the Republic of South Africa (Act 108 of 1996)*. Government Printer.
- Republic of South Africa. (2000). Act No. 31 of 2000: South African Council of Educators Act. *Government Gazette, 422*(21431).
- Republic of South Africa. (2019). *South Africa yearbook 2018/19* (26th ed.). Government Communications (GCIS).
- Rieg, S. A., Paquette, K. R., & Chen, Y. (2007). Coping with stress: An investigation of novice teachers' stressors in the elementary classroom. *Education, 128*(2), 211–226.
- Rizeanu, S., & Vasiliu, D. (2016). Correlation between hardiness and the sense of coherence. *Romanian Journal of Experimental Applied Psychology, 7*(4), 9–15. https://doi.org/10.15303/rjeap.2016.v7i4.a2
- Roni, S. M., Merga, M. K., & Morris, J. E. (2020). Conducting quantitative research in education. Springer. https://doi.org/10.1007/978-981-13-9132-3
- Ross, J. A., Cousins, J. B., & Gadalla, T. (1996). Within-teacher predictors of teacher efficacy. *Teaching and Teacher Education*, *12*(4), 385–400. https://doi.org/10.1016/0742-051X(95)00046-M
- Roth, M., & Von Collani, G. (2007). A head-to-head comparison of big-five types and traits in the prediction of social attitudes: Further evidence for a five-cluster typology. *Journal* of *Individual Differences*, 28(3), 138–149. https://doi.org/10.1027/1614-0001.28.3.138
- Rotter, J. B. (1954). Social learning and clinical psychology. Prentice-Hall. https://doi.org/10.1037/10788-000



- Rotter, J. B. (1966). Generalized expectancies for internal versus external control of reinforcement. *Psychological Monographs: General and Applied, 80*(1), 1–28. https://doi.org/10.1037/h0092976
- Rubenstein, L. D., Ridgley, L. M., Callan, G. L., Karami, S., & Ehlinger, J. (2018). How teachers perceive factors that influence creativity development: Applying a Social Cognitive Theory perspective. *Teaching and Teacher Education*, *70*, 100–110. https://doi.org/10.1016/j.tate.2017.11.012
- Rubin, A., & Babbie, E. R. (2014). *Research methods for social work* (8th ed.). Brooks/Cole, Cengage Learning.
- Rutter, M. (1985). Resilience in the face of adversity: Protective factors and resistance to psychiatric disorder. *The British Journal of Psychiatry*, *147*(6), 598–611. https://doi.org/10.1192/bjp.147.6.598
- Rutter, M. (2006). Implication of resilience concepts for scientific understanding. *Annals of the New York Academy of Sciences, 1094*(1), 1–12. https://doi.org/10.1196/annals.1376.002
- Rutter, M. (2012). Resilience as a dynamic concept. *Development and Psychopathology,* 24(2), 335–344. https://doi.org/10.1017/S0954579412000028
- Saavedra, C. M., & Pérez, M. S. (2018). Global south approaches to bilingual and early childhood teacher education: Disrupting global north neoliberalism. *Policy Futures in Education, 16*(6), 749–763. https://doi.org/10.1177/1478210317751271
- Salifu, I., Alagbela, A. A., & Gyamfi-Ofori, C. (2018). Factors influencing teaching as a career choice (FIT-Choice) in Ghana. *Teaching Education*, 29(2), 111–134. https://doi.org/10.1080/10476210.2017.1365360
- Salkind, N. J., & Shaw, L. A. (2020). Statistics for people who (think they) hate statistics using R. Sage.
- Sammons, P., Day, C., Kington, A., Gu, Q., Stobart, G., & Smees, R. (2007). Exploring variations in teachers' work, lives and their effects on pupils: Key findings and implications from a longitudinal mixed-method study. *British Educational Research Journal, 33*(5), 681–701. https://doi.org/10.1080/01411920701582264
- Sauro, J., & Lewis, J. R. (2016). *Quantifying the user experience: Practical statistics for user research* (2nd ed.). Morgan Kaufmann.
- Sayed, Y., & Badroodien, A. (2016). Teachers and social cohesion in the Global South: expanding the notion of education quality [Editorial]. *Education as Change*, *20*(3), 1– 14. https://doi.org/10.17159/1947-9417/2016/1975



- Sayed, Y., & McDonald, Z. (2017). Motivation to become a Foundation Phase teacher in South Africa. South African Journal of Childhood Education, 7(1), Article 548. https://doi.org/10.4102/sajce.v7i1.548
- Schnyder, U., Büchi, S., Sensky, T., & Klaghofer, R. (2000). Antonovsky's sense of coherence: Trait or state? *Psychotherapy and Psychosomatics*, 69, 296–302. https://doi.org/10.1159/000012411

Schunk, D. H. (2008). Learning theories. An educational perspective (5th ed.). Pearson.

Schwartz, M., & Harris, H. (2017). Ethics in the global South (Vol. 18). Emerald Publishing.

- Scotland, J. (2012). Exploring the philosophical underpinnings of research: Relating ontology and epistemology to the methodology and methods of the scientific, interpretive, and critical research paradigms. *English Language Teaching, 5*(9), 9–16. https://doi.org/10.5539/elt.v5n9p9
- Seligman, M. E. P. (2002). Authentic happiness: Using the new positive psychology to realize your potential for lasting fulfillment. Free Press.
- Seligman, M. E. P., Ernst, R. M., Gillham, J., Reivich, K., & Linkins, M. (2009). Positive education: Positive psychology and classroom interventions. Oxford Review of Education, 35(3), 293–311. https://doi.org/10.1080/03054980902934563
- Selye, H. (1976). Stress without distress. In G. Serbam (Ed.), *Psychopathology of human adaptation* (pp. 137–146). Springer. https://doi.org/10.1007/978-1-4684-2238-2
- Shalem, Y., & Hoadley, U. (2009). The dual economy of schooling and teacher morale in South Africa. International Studies in Sociology of Education, 19(2), 119–134. https://doi.org/10.1080/09620210903257224
- Sharpe, D. (2013). Why the resistance to statistical innovations? Bridging the communication gap. *Psychological Methods, 18*(4), 572–582. https://doi.org/10.1037/a0034177
- Shean, M. (2015). Current theories relating to resilience and young people: A literature review. Victorian Health Promotion Foundation. https://evidenceforlearning.org.au/assets/Grant-Round-II-Resilience/Current-theoriesrelating-to-resilience-and-young-people.pdf
- Silva, J. C. (2013). Clima escolar, satisfação, stresse profssional e colaboração entre professores: Que relações? [School climate, job satisfaction, professional stress and collaboration between teachers: What kind of relations?] [Unpublished doctoral thesis]. Universidade Aberta.
- Simkins, C. (2015). *Technical report: Teacher supply and demand in South Africa: 2013 to 2025*. Centre for Development and Enterprise (CDE). https://www.cde.org.za/wp-



content/uploads/2018/07/Teacher-Supply-and-Demand-2013-2025-Technical-Report-March2015-CDE.pdf

- Simon, M. K. (2011). *Dissertation and scholarly research: Recipes for success*. Dissertation Success.
- Sing, N., & Maringe, F. (2020). Learner dropout in South African schools: Epistemological and management challenges. In K. Ademeyo (Ed.), *The education systems of Africa* (pp. 1–15). Springer. https://doi.org/10.1007/978-3-030-43042-9\_22-1
- Skaalvik, E. M., & Skaalvik, S. (2010). Teacher self-efficacy and teacher burnout: A study of relations. *Teaching and Teacher Education*, 26(4), 1059–1069. https://doi.org/10.1016/j.tate.2009.11.001
- Skinner, E. A., & Zimmer-Gembeck, M. J. (2007). The development of coping. Annual Review of Psychology, 58, 119–144. https://doi.org/10.1146/annurev.psych.58.110405.085705
- Skinner, E. A., & Zimmer-Gembeck, M. J. (2011). Perceived control and the development of coping. In S. Folkman (Ed.), *The Oxford handbook of stress, health, and coping* (pp. 35–59). Oxford University Press.

Snowman, J., & McCown, R. (2013). *Ed Psych*. Cengage Learning.

- Somech, A., & Drach-Zahavy, A. (2000). Understanding extra-role behavior in schools: The relationships between job satisfaction, sense of efficacy, and teachers' extra-role behavior. *Teaching and Teacher Education*, *16*(5–6), 649–659. https://doi.org/10.1016/S0742-051X(00)00012-3
- Soodak, L. C., & Podell, D. M. (1996). Teacher efficacy: Toward the understanding of a multifaceted construct. *Teaching and Teacher Education*, *12*(4), 401–411. https://doi.org/10.1016/0742-051X(95)00047-N
- Sosa, T., & Gomez, K. (2012). Connecting teacher efficacy beliefs in promoting resilience to support of Latino students. Urban Education, 47(5), 876–909. https://doi.org/10.1177/0042085912446033
- South African Council of Educators. (2017). Code of professional ethics. https://www.sace.org.za/assets/documents/uploads/sace\_24644-2017-02-08-Code%20of%20ethics%20Brochure%20FINAL%20PRINT%205%20SEPT.pdf
- South African Qualifications Authority. (2020). *Information.* https://www.saqa.org.za/about-saqa
- Spaull, N. (2011). A preliminary analysis of SACMEQ III South Africa (Stellenbosch Economic Working Papers: 11/11). Department of Economics and the Bureau for



Economic Research, University of Stellenbosch. https://resep.sun.ac.za/wp-content/uploads/2017/10/wp-11-2011.pdf

- Spaull, N. (2015). Schooling in South Africa: How low-quality education becomes a poverty trap. In A. De Lannoy, S. Swartz, L. Lake, & C. Smith C (Eds.), South African Child Gauge 2015 (pp. 34–41). Children's Institute, University of Cape Town. http://www.ci.uct.ac.za/sites/default/files/image\_tool/images/367/Child\_Gauge/South\_A frican\_Child\_Gauge\_2015/ChildGauge2015-lowres.pdf
- Spaull, N., Pretorius, E., & Mohohlwane, N. (2020). Investigating the comprehension iceberg: Developing empirical benchmarks for early-grade reading in agglutinating African languages. South African Journal of Childhood Education, 10(1), a773. https://doi.org/10.4102/sajce.v10i1.773
- Spearman C. (1904). The proof and measurement of association between two things. *The American Journal of Psychology, 15*(1), 72–101. https://doi.org/10.2307/1412159
- Stanford, B. H. (2001). Reflections of resilient, persevering urban teachers. *Teacher Education Quarterly, 28*(3), 75–87.
- Statistics South Africa. (2012a). *Census 2011 Statistical release P0301.4*. https://www.statssa.gov.za/publications/P03014/P030142011.pdf
- Statistics South Africa. (2012b). *Poverty profile of South Africa: Application of the poverty lines on the LCS 2008/2009.* http://www.statssa.gov.za/publications/Report-03-10-03/Report-03-10-032009.pdf
- Statistics South Africa. (2016). Community Survey 2016, Statistical release P0301. http://cs2016.statssa.gov.za/wp-content/uploads/2016/07/NT-30-06-2016-RELEASEfor-CS-2016-\_Statistical-releas\_1-July-2016.pdf
- Statistics South Africa. (2018). *General Household Survey 2018. Statistical release P0318.* https://www.statssa.gov.za/publications/P0318/P03182018.pdf
- Statistics South Africa. (2020a). Poverty. http://beta2.statssa.gov.za/?page\_id=739&id=1
- Statistics South Africa. (2020b, February 11). *Quarterly Labour Force Survey (QLFS)* Q4:2019. http://www.statssa.gov.za/?p=12948
- Stewart, D. (2014). Resilience: An entry point for African health promoting schools? *Health Education, 114*(3), 197–207. https://doi.org/10.1108/HE-06-2013-0023
- Strauser, D. R., & Lustig, D. C. (2003). The moderating effect of sense of coherence on work adjustment. *Journal of Employment Counseling*, 40(3), 129–140. https://doi.org/10.1002/j.2161-1920.2003.tb00863.x



- Struwig, F. W., & Stead, G. B. (2013). *Planning, designing and reporting research* (2nd ed). Pearson Education.
- Swan, B. G., Wolf, K. J., & Cano, J. (2011). Changes in teacher self-efficacy from the student teaching experience through the third year of teaching. *Journal of Agricultural Education, 52*(2), 128–139. https://doi.org/10.5032/jae.2011.02128
- Tait, M. (2008). Resilience as a contributor to novice teacher success, commitment, and retention. *Teacher Education Quarterly, 35*(4), 57–75.
- Taylor, J. L. (2013). The power of resilience: A theoretical model to empower, encourage and retain Teachers. *The Qualitative Report, 18*(35), 1–25. https://files.eric.ed.gov/fulltext/EJ1043493.pdf
- Taylor, W. (2016). Teacher education. In *Encyclopædia Britannica*. https://www.britannica.com/topic/teacher-education
- Teasdale, J. D. (1978). Self-efficacy: Toward a unifying theory of behavioral change? Advances in Behaviour Research and Therapy, 1(4), 211–215.
- Tekin, A. K., & Kotaman, H. (2013). The epistemological perspectives on action research. Journal of Educational and Social Research, 3(1), 81–91. https://doi.org/10.5901/jesr.2013.v3n1p81
- Terre Blanche, M., Kelly, K., & Durrheim, K. (2014). Why qualitative research? In M. Terre Blanche, K. Durrheim, & D. Painter (Eds.). *Research in practice: Applied methods for the social sciences* (2nd ed., pp. 271–284). UCT Press.
- The Hamilton Project. (2007). Age distribution of undergraduate students, by type of institution. https://www.hamiltonproject.org/charts/age\_distribution\_of\_undergraduate\_students\_b y\_type\_of\_institution
- The Presidency, Republic of South Africa. (2013). Act No. 4 of 2013: Protection of Personal Information Act, 2013. *Government Gazette, 581*(37067).
- Theron, L., Cameron, C. A., Didkowsky, N., Lau, C., Liebenberg, L., & Ungar, M. (2011). A "day in the lives" of four resilient youths: Cultural roots of resilience. *Youth & Society*, 43(3), 799–818. https://doi.org/10.1177/0044118X11402853
- Theron, L. C. (2011). Vulnerable, but invincible?: Ecosystemic pathways to South African youths' resilience [PDF]. https://resilienceresearch.org/files/PDF/Inaugural%20Lecture%20LC%20Theron.pdf
- Theron, L. C. (2016). Towards a culturally and contextually understanding of resilience: Privileging the voices of black, South African young people. *Journal of Adolescent Research, 31*(6), 635–670. https://doi.org/10.1177/0743558415600072



- Theron, L. C., & Malindi, M. J. (2010). Resilient street youth: A qualitative South African study. *Journal of Youth Studies*, *13*(6), 717–736. https://doi.org/10.1080/13676261003801796
- Theron, L. C., & Theron, A. M. C. (2010). A critical review of studies of South African youth resilience, 1990–2008. *South African Journal of Science, 106*(7/8), Article #252. https://doi.org/10.4102/sajs.v106i7/8.252
- Theron, L. C., & Theron, A. M. C. (2014). Meaning-making and resilience: Case studies of a multifaceted process. *Journal of Psychology in Africa, 24*(1), 24–32. https://doi.org/10.1080/14330237.2014.904099
- Thieman, E. B., Henry, A. L., & Kitchel, T. (2012). Resilient agricultural educators: Taking stress to the next level. *Journal of Agricultural Education, 53*(1), 81–94. https://doi.org/10.5032/jae.2012.01081
- Thieman, E. B., Marx, A. A., & Kitchel, T. (2014). "You've always got challenges": Resilience and the preservice teacher. *Journal of Agricultural Education*, *55*(4), 12–23. https://files.eric.ed.gov/fulltext/EJ1122816.pdf
- Thomas, G. (2011). A typology for the case study in social science following a review of definition, discourse, and structure. *Qualitative Inquiry, 17*(6), 511–521. https://doi.org/10.1177/1077800411409884
- Thomas, K. E., & Mucherah, W. M. (2016). The contextual difference: Developing preservice teacher efficacy through immersive learning experiences. *Education and Urban Society*, *48*(4), 364–383. https://doi.org/10.1177/0013124514533795
- Tikly, L. (2001). Globalisation and education in the postcolonial world: Towards a conceptual framework. *Comparative Education, 37*(2), 151–171. https://doi.org/10.1080/03050060120043394
- Tikly, L., & Bond, T. (2013). Towards a postcolonial research ethics in comparative and international education. *Compare: A Journal of Comparative and International Education, 43*(4), 422–442. https://doi.org/10.1080/03057925.2013.797721
- Tomczyk, Ł. (2020). Skills in the area of digital safety as a key component of digital literacy among teachers. *Education and Information Technologies*, 25(1), 471–486. https://doi.org/10.1007/s10639-019-09980-6
- Tournaki, N., Lyublinskaya, I., & Carolan, B. V. (2009). Pathways to teacher certification: Does it really matter when it comes to efficacy and effectiveness? *Action in Teacher Education*, 30(4), 96–109. https://doi.org/10.1080/01626620.2009.10734455
- Trafimow, D., & Marks, M. (2015). Editorial. *Basic and Applied Social Psychology, 37*(1), 1– 2. https://doi.org/10.1080/01973533.2015.1012991



- Trefzer, A., Jackson, J. T., McKee, K., & Dellinger, K. (2014). The Global South and/in the Global North: Interdisciplinary investigations. *The Global South, 8*(2), 1–15. https://doi.org/10.2979/globalsouth.8.2.1
- Trochim, W. M. K. (2006). *Convergent and discriminant validity*. Research Methods Knowledge Base. https://www.socialresearchmethods.net/kb/convdisc.php
- Tryon, W. W. (1981). A methodological critique of Bandura's self-efficacy theory of behavior change. *Journal of Behavior Therapy and Experimental Psychiatry*, *12*(2), 113–114. https://doi.org/10.1016/0005-7916(81)90003-3
- Trzesniewski, K. H., Donnellan, M. B., & Lucas, R. E. (Eds.). (2011). Secondary data analysis: An introduction for psychologists. American Psychological Association. https://doi.org/10.1037/12350-000
- Tschannen-Moran, M., & Johnson, D. (2011). Exploring literacy teachers' self-efficacy beliefs: Potential sources at play. *Teaching and Teacher Education, 27*(4), 751–761. https://doi.org/10.1016/j.tate.2010.12.005
- Tschannen-Moran, M., & Woolfolk Hoy, A. (2001). Teacher efficacy: Capturing an elusive construct. *Teaching and Teacher Education, 17*(7), 783–805. https://doi.org/10.1016/S0742-051X(01)00036-1
- Tschannen-Moran, M., & Woolfolk Hoy, A. (2007). The differential antecedents of selfefficacy beliefs of novice and experienced teachers. *Teaching and Teacher Education*, 23(6), 944–956. https://doi.org/10.1016/j.tate.2006.05.003
- Tschannen-Moran, M., Woolfolk Hoy, A., & Hoy, W. K. (1998). Teacher efficacy: Its meaning and measure. *Review of Educational Research, 68*(2), 202–248. https://doi.org/10.3102/00346543068002202
- Tsouloupas, C. N., Carson, R. L., & MacGregor, S. K. (2014). The development of high school teachers' efficacy in handling student misbehavior (TEHSM). *The Journal of Educational Research*, *107*(3), 230–240. https://doi.org/10.1080/00220671.2013.788992
- Tugade, M. M., & Fredrickson, B. L. (2004). Resilient individuals use positive emotions to bounce back from negative emotional experiences. *Journal of Personality and Social Psychology, 86*(2), 320–333. https://doi.org/10.1037/0022-3514.86.2.320
- Tugade, M. M., Fredrickson, B. L., & Feldman Barrett, L. (2004). Psychological resilience and positive emotional granularity: Examining the benefits of positive emotions on coping and health. *Journal of Personality*, 72(6), 1161–1190. https://doi.org/10.1111/j.1467-6494.2004.00294.x
- Tuli, F. (2010). The basis of distinction between qualitative and quantitative research in social science: Reflection on ontological, epistemological and methodological



perspectives. *Ethiopian Journal of Education and Sciences, 6*(1), 97–108. https://doi.org/10.4314/ejesc.v6i1.65384

- Ungar, M. (2008). Resilience across cultures. *The British Journal of Social Work, 38*(2), 218–235. https://doi.org/10.1093/bjsw/bcl343
- Ungar, M. (2010). What is resilience across cultures and contexts? Advances to the theory of positive development among individuals and families under stress. *Journal of Family Psychotherapy*, *21*(1), 1–16. https://doi.org/10.1080/08975351003618494
- Ungar, M. (2011). The social ecology of resilience: Addressing contextual and cultural ambiguity of a nascent construct. *American Journal of Orthopsychiatry, 81*(1), 1–17. https://doi.org/10.1111/j.1939-0025.2010.01067.x
- Ungar, M. (2012a). Researching and theorizing resilience across cultures and contexts. *Preventive Medicine, 55*(5), 387–389. https://doi.org/10.1016/j.ypmed.2012.07.021
- Ungar, M. (2012b). Social ecologies and their contribution to resilience. In M. Ungar (Ed.), The social ecology of resilience: A handbook of theory and practice (pp. 13–31). Springer. https://doi.org/10.1007/978-1-4614-0586-3
- Ungar, M., Ghazinour, M., & Richter, J. (2013). Annual research review: What is resilience within the social ecology of human development [Special issue]? *The Journal of Child Psychology and Psychiatry*, *54*(4). https://doi.org/10.1111/jcpp.12025
- United Nations. (2015, October). *Transforming our world: The 2030 Agenda for sustainable development.* https://www.un.org/ga/search/view\_doc.asp?symbol=A/RES/70/1&Lang=E
- United Nations Children's Fund. (2000, June). *Defining quality in education* [Paper presentation]. The International Working Group on Education Meeting, Florence, Italy. https://neqmap.bangkok.unesco.org/wp-content/uploads/2000/12/QualityEducation.pdf
- United Nations Educational, Scientific and Cultural Organization. (2019a). *Meeting commitments: Are countries on track to achieve SDG 4?* http://uis.unesco.org/sites/default/files/documents/meeting-commitments-are-countrieson-track-achieve-sdg4.pdf
- United Nations Educational, Scientific and Cultural Organization. (2019b). *World Teachers' Day 2019* [Fact sheet]. http://uis.unesco.org/sites/default/files/documents/worldteachers-day-2019-fact-sheet.pdf
- UNESCO Institute for Statistics. (2012). *International Standard Classification of Education: ISCED 2011.* http://uis.unesco.org/sites/default/files/documents/international-standardclassification-of-education-isced-2011-en.pdf


- University of Pretoria. (2015). Fourth Year Intervention in Research in Education (FIRE) project: Ethical application approval (Submission number UP 14/03/01).
- Üstüner, M. (2017). Personality and attitude towards teaching profession: Mediating role of self-efficacy. *Journal of Education and Training Studies, 5*(9), 70–82. https://doi.org/10.11114/jets.v5i9.2536
- Utts, J. M., & Heckard, R. F. (2012). *Mind on statistics* (4th ed.). Brooks/Cole, Cengage Learning.
- Van Breda, A. D. (2018). A critical review of resilience theory and its relevance for social work. *Social Work*, *54*(1), 1–18. https://doi.org/10.15270/54-1-611
- Van der Berg, S. (2015). What the Annual National Assessments can tell us about learning deficits over the education system and the school career. *South African Journal of Childhood Education, 5*(2), 28–43. http://www.scielo.org.za/pdf/sajce/v5n2/03.pdf
- Vanderbilt-Adriance, E., & Shaw, D.S. (2008). Conceptualising and re-evaluating resilience across levels of risk, time, and domains of competence. *Clinical Child and Family Psychology Review, 11*(1-2), Article 30. https://doi.org/10.1007/s10567-008-0031-2
- Van Diemen, E. (2019, September 18). *Teaching and learning international survey: Where does SA rank?* News24. https://m.news24.com/SouthAfrica/News/teaching-and-learning-international-survey-where-does-sa-rank-20190917
- Van Rensburg, G. H., Alpaslan, A. H., Du Plooy, G. M., Gelderblom, D., Van Eeden, R., & Wigston, D. J. (2010). *Research in the social sciences*. Unisa.
- Van Staden, S., & Zimmerman, L. (2017). Evidence from the Progress in International Reading Literacy Study (PIRLS) and how teachers and their practice can benefit. In V. Scherman, R. J. Bosker, & S. J. Howie (Eds.), *Monitoring the quality of education in schools: Examples of feedback into education systems from developed and emerging economies* (pp. 123–138). Sense.
- Van Tilburg, L. (2019, November 14). *Kids are being failed by the education system; a teacher crisis*. BizNews. https://www.biznews.com/thought-leaders/2019/11/14/kids-failed-education-teacher-crisis
- Ventegodt, S., Flesborg-Madsen, T., Anderson, N. J., & Merrick, J. (2005). The life mission theory VII. Theory of existential (Antonovsky) coherence: A theory of quality of life, health, and ability for use in holistic medicine. *The Scientific World Journal, 5,* 377– 389. https://doi.org/10.1100/tsw.2005.47
- Vieluf, S., Kunter, M., & Van de Vijver, F. J. R. (2013). Teacher self-efficacy in cross-national perspective. *Teaching and Teacher Education*, 35, 92–103. https://doi.org/10.1016/j.tate.2013.05.006



- Viviers, A. M., & Cilliers, F. (1999). Die verband tussen salutogenese en werkoriëntasie. *Tydskrif vir Bedryfsielkunde, 25*(1), 27–32. https://doi.org/10.4102/sajip.v25i2.676
- Von Culin, K. R., Tsukayama, E., & Duckworth, A. L. (2014). Unpacking grit: Motivational correlates of perseverance and passion for long-term goals. *The Journal of Positive Psychology*, 9(4), 306–312. https://doi.org/10.1080/17439760.2014.898320
- Wabule, A. (2020). Resilience and care: How teachers deal with situations of adversity in the teaching and learning environment. *The Independent Journal of Teaching and Learning*, 15(1), 76–90. https://journals.co.za/doi/pdf/10.10520/EJC-1d66a60c00
- Waddell, J. (2007). The time is now: The role of professional learning communities strengthening the resiliency of teachers in urban schools. In D. M. Davies (Ed.), *Resiliency reconsidered: Policy implications of the resiliency movement* (pp. 123–145). Information Age Publishing.
- Walsh, F. (2002). Bouncing forward: Resilience in the aftermath of September 11. *Family Process, 41,* 34–36.
- Walsh, F. (2007). Traumatic loss and major disasters: Strengthening family and community resilience. *Family Process, 46*(2), 207–227. https://doi.org/10.1111/j.1545-5300.2007.00205.x
- Wang, H., Hall, N. C., & Rahimi, S. (2015). Self-efficacy and causal attributions in teachers: Effects on burnout, job satisfaction, illness, and quitting intentions. *Teaching and Teacher Education*, 47, 120–130. https://doi.org/10.1016/j.tate.2014.12.005
- Watson, S., & Marschall, G. (2019). How a trainee mathematics teacher develops teacher self-efficacy. *Teacher Development*, 23(4), 469–487. https://doi.org/10.1080/13664530.2019.1633392
- Watt, H. M. G., & Richardson, P. W. (2007). Motivational factors influencing teaching as a career choice: Development and validation of the FIT-Choice scale. *The Journal of Experimental Education*, 75(3), 167–202. https://doi.org/10.3200/JEXE.75.3.167-202
- Watt, H. M. G., & Richardson, P. W. (2008). Motivations, perceptions, and aspirations concerning teaching as a career for different types of beginning teachers. *Learning and Instruction, 18*(5), 408–428. https://doi.org/10.1016/j.learninstruc.2008.06.002
- Watt, H. M. G., & Richardson, P. W. (2012). An introduction to teaching motivations in different countries: Comparisons using the FIT-Choice scale. Asia-Pacific Journal of Teacher Education, 40(3), 185–197. https://doi.org/10.1080/1359866X.2012.700049
- Watt, H. M. G., Richardson, P. W., Klusmann, U., Kunter, M., Beyer, B., Trautwein, U., & Baumert, J. (2012). Motivations for choosing teaching as a career: An international comparison using the FIT-Choice scale. *Teaching and Teacher Education, 28*(6), 791– 805. https://doi.org/10.1016/j.tate.2012.03.003



- Widaman, K. F., Little, T. D., Preacher, K. J., & Sawalani, A. M. (2011). On creating and using short forms of scales in secondary research. In K. H. Trzesniewski, M. B. Donnellan, & R. E. Lucas (Eds.), Secondary data analysis: An introduction for psychologists (pp. 39–61). American Psychological Association. https://doi.org/10.1037/12350-003
- Wilcoxon, F. (1945). Individual comparisons by ranking methods. *Biometrics Bulletin, 1*(6), 80–83. https://doi.org/10.2307/3001968
- Willers, M., Potgieter, J. C., Khumalo, I. P., Malan, L., Mentz, P. J., & Ellis, S. (2013). Coping and cultural context: Implications for psychological health and well-being. In M. P. Wissing (Ed.), *Well-being research in South Afric*a (pp. 251–272). Springer. https://doi.org/10.1007/978-94-007-6368-5
- Willms, J. D. (2011). Managing and using secondary data sets with multidisciplinary research teams. In K. H. Trzesniewski, M. B. Donnellan, & R. E. Lucas (Eds.), Secondary data analysis: An introduction for psychologists (pp. 27–38). American Psychological Association. https://doi.org/10.1037/12350-002
- Wilson, W. J. (2012). *The truly disadvantaged: The inner city, the underclass, and public policy* (2nd ed.). The University of Chicago Press.
- Windle, G. (2010). What is resilience? A review and concept analysis. *Reviews in Clinical Gerontology, 21*(2), 152–169. https://doi.org/10.1017/S0959259810000420
- Wissing, M. P., Temane, Q. M., Khumalo, I. P., Kruger, A., & Vorster, H. H. (2013).
  Psychosocial health: Disparities between urban and rural communities. In M. P.
  Wissing (Ed.), *Well-being research in South Afric*a (pp. 415–438). Springer.
  https://doi.org/10.1007/978-94-007-6368-5
- Wolters, C. A., & Daugherty, S. G. (2007). Goal structures and teachers' sense of efficacy: Their relation and association to teaching experience and academic level. *Journal of Educational Psychology*, 99(1), 181–193. https://doi.org/10.1037/0022-0663.99.1.181
- Woodcock, S. (2011). A cross sectional study of pre-service teacher efficacy throughout the training years. *Australian Journal of Teacher Education, 36*(10), 23–34. https://doi.org/10.14221/ajte.2011v36n10.1
- Woolfolk Hoy, A., Hoy, W. K., & Davis, H. A. (2009). Teachers' self-efficacy beliefs. In K.
  R. Wenzel & A. Wigfield (Eds.), *Handbook of motivation at school* (pp. 627–653).
  Routledge/Taylor & Francis Group.
- Woolfolk Hoy, A., & Spero, R. B. (2005). Changes in teacher efficacy during the early years of teaching: A comparison of four measures. *Teaching and Teacher Education*, 21(4), 343–356. https://doi.org/10.1016/j.tate.2005.01.007

World Bank. (2018). World Development Report 2018: Learning to realize education's



promise. Washington, DC: World Bank, 2018. \$60.00; \$39.95 (pbk.). *Population and Development Review, 44*(2), 404–405. https://doi.org/10.1111/padr.12164

- World Bank. (2020a). Education statistics (EdStats): International Standard Classification of Education (ISCED). https://datatopics.worldbank.org/education/wRsc/classification
- World Bank. (2020b, September). *Primary education, teacher (% female)*. https://data.worldbank.org/indicator/SE.PRM.TCHR.FE.ZS
- World Bank. (2021). World Bank country and lending groups: Country classification. https://datahelpdesk.worldbank.org/knowledgebase/articles/906519
- Wosnitza, M., Delzepich, R., Schwarze, J., O'Donnell, M., Faust, V., & Camilleri, V. (2018). Enhancing teacher resilience: From self-reflection to professional development. In M. Wosnitza, F. Peixoto, S. Beltman, & C. F. Mansfield (Eds.), *Resilience in education: Concepts, contexts and connections* (pp. 275–288). Springer International Publishing. https://doi.org/10.1007/978-3-319-76690-4\_16
- Wosnitza, M., O'Donnell, M., Morgan, M., Mansfield, C. F., & Beltman, S. (2014). *Teachers* resilience in Europe: A theoretical framework. ENTREE.
- Yada, A., Björn, P. M., Savolainen, P., Kyttälä, M., Aro, M., & Savolainen, H. (2021). Preservice teachers' self-efficacy in implementing inclusive practices and resilience in Finland. *Teaching and Teacher Education*, 105, 103398. https://doi.org/10.1016/j.tate.2021.103398
- Yin, R. K. (2018). Case study research and applications: Design and methods (6th ed.). Sage.
- Yokus, T. (2015). The relation between pre-service music teachers' psychological resilience and academic achievement levels. *Educational Research and Reviews, 10*(14), 1961– 1969.
- Yonezawa, S., Jones, M., & Singer, N. R. (2011). Teacher resilience in urban schools: The importance of technical knowledge, professional community, and leadership opportunities. *Urban Education*, *46*(5), 913–931. https://doi.org/10.1177/0042085911400341
- Yost, P. R. (2016). Resilience practices. *Industrial and Organizational Psychology*, 9(2), 475–479. https://doi.org/10.1017/iop.2016.42
- Zartman, I. W. (2012). Comparative case studies. In D. J. Christie (Ed.), *The encyclopedia of peace psychology*. Blackwell Publishing. https://doi.org/10.1002/9780470672532.wbepp043
- Zimmer-Gembeck, M. J., Skinner, E. A., Modecki, K. L., Webb, H. J., Gardner, A. A., Hawes, T., & Rapee, R. M. (2018). The self-perception of flexible coping with stress: A new



measure and relations with emotional adjustment. *Cogent Psychology, 5*(1), 1537908. https://doi.org/10.1080/23311908.2018.1537908

Zuma, K., Simbayi, L. C., Rehle, T., Mbelle, N., Zungu, N. P., Mthembu, J., North, A., Van Zyl, J., Jooste, S., Moyo, S., Wabiri, N., Maduna, V., Mabaso, M., Naidoo, I., Chasela, C., Chikovore, J., & the Educator Survey II Study Team. (2016). *The health of educators in public schools in South Africa 2016*. NACOSA. https://www.nacosa.org.za/wp-content/uploads/2017/06/the-health-of-educators-in-public-schools.pdf

---00000----



# Appendices

Appendix A: Background and overview of ENTREE

Appendix B: FIRE Teacher Resilience Measure

Appendix C: FIRE Project ethical approval and extracts

**Appendix D:** University of Pretoria BEd programme information



### APPENDIX A: BACKGROUND AND OVERVIEW OF ENTREE

### ENHANCING TEACHER RESILIENCE IN EUROPE (ENTREE): EUROPE & AUSTRALIA [2013-2015] [http://entree-project.eu/en/]

- Aim: Project aims to assist teacher face the everyday challenges in school life by enhancing their teacher resilience / To help teachers in Europe to build their capacity for resilience in the face of the increasing demands of a rapidly changing school environment.
- Resilience definition: Resilience is understood to be the process by which teacher mange the challenges of their everyday teaching practice and provide
  quality education for all students. Teacher resilience (within the ENTREE project) refers to the process of, capacity for, or outcome of positive adaptation and
  ongoing professional commitment and growth in the face of challenging circumstances. Resilience is shaped by individual, situational and broader contextual
  characteristics that interrelate in dynamic ways to provide risk (challenging) or protective (supportive) factors.
- Population: Pre-service and practicing teachers
- · Countries: EU countries Germany, Ireland, Malta, Portugal and the Czech Republic as well as Australia
- · Funded: European Commission's Lifelong Learning Programme and is supported by an international consortium of experts provide
- Measures/framework incorporated/developed: Created tools and methods to help teachers increase their capacity for resilience. Online and face-to-face training materials crated. Self-assessment tool developed and used. ENTREE teacher resilience measure
- PI: Marold Wosnitza [Project Team: Mark Morgan, Carmel Cefai, Matthias Henkel, Francisco Peixoto, Susan Beltman & Caroline Mansfield]

#### BUILDING RESILIENCE IN TEACHER EDUCATION (BRITE) [2012-2015] [https://www.brite.edu.au/who-are-we]

- Aim: Enhancing teacher resilience. Collaborative project between 3 Australian Universities that developed 5 online modules to support pre-service teachers building resilience for the profession. The aim of this program is to help pre-service teachers build their awareness of the skills and practices that will help facilitate resilience in their teaching career. Develop pre-service teachers' capacity for resilience in Australia. Highlighting and developing strategies that pre-service teachers could use to harness their unique personal resources and resources in their varied contexts.
- Population: Pre-service teachers
- Country: Australia [Three Australian Universities: Murdoch, Curtin and the University of Wollongong]
- Method: Perceptions of 146 final year pre-service teachers were gathered regarding resilience, self-efficacy, commitment and coping before completing the BRITE modules and their final professional experience school placement. Both pre- and post-school placement measures were completed by 49 participants

#### Measures/framework incorporated/developed:

- BRITE online modules (based on the Keeping Cool) developed include: "Building Resilience", "Relationships", "Wellbeing", "Teaching skills and Emotions"
- PI: Caroline Mansfield, Susan Beltman, Tania Broadley and Noelene Weathersby-Fell [Building on the success of the BRiTE project = 'Staying BRiTE https://www.stayingbrite.edu.au/]

#### KEEPING COOL: Building Teacher's Resilience Embedding resilience in the initial teacher education curriculum [2009-2012]

- · Aim: Enhancing teacher resilience. Integrating resilience into teacher education programmes. How do pre-service teachers 'view' resilience?
- Population: Pre-service teachers (98); Early career teachers (161) = N=259
- Country: Australia
- Funded: Australian Learning and Teaching Council Ltd (ALTC)
- Method: Extensive quantitative and qualitative data were gathered through the distribution of a paper-and-pen and online survey to pre-service and early career teachers in 2010 and 2011. Two versions of the survey were specifically designed by members of the project team. The first, tailored for graduate teachers assessed the following core constructs: (1) general teacher self-efficacy; (2) retrospective and prospective motivation for teaching; (3) satisfaction with teacher preparation; (4) self-perceived competency for core teaching skills; and (5) professional plans for the future. The second version, tailored for early-career teachers, measured respondents': (1) general self-efficacy; (2) motivation for teaching; (3) coping skills; (4) optimism and stress in the workplace; (5) general resilience; and (6) plans for the future.
- Measures/framework incorporated/developed
- Developed the four-dimensional (profession-related, emotional, motivational and social) framework of teacher resilience = capacities/personal resources
- PI: Caroline Mansfield & Anne Price [Project Team: Andrew McConney, Susan Beltman, Lina Pelliccione & Marold Wosnitza]
- Publications:
  - Literature review: Beltman, S., Mansfield, C., & Price, A. (2011). Thriving not just surviving: A review of research on teacher resilience. Educational research review, 6(3), 185-207.
  - Project summary: Mansfield, C.F., Beltman, S., Price, A. and McConney, A. (2012) Don't sweat the small stuff: Understanding teacher resilience at the chalkface. Teaching and Teacher Education, 28 (3). pp. 357-367.

### ENTREE TEACHER RESILIENCE MEASURE:

- Global measure of teacher resilience [Teacher Resilience Scale, Morgan 2011]
- Multidimensional Teacher Resilience Scale (MTRS) [Mansfield & Wosnitza 2015]
- Personal Life (Morgan 2011)
- Rumination Scale (Morgan 2011)
- Teacher Commitment (Morgan 2011)
- Teacher Self-Efficacy [Teacher Self-Efficacy (TSE) Scale (Morgan 2011)]
- Social context (Cefai & Cavioni 2014)
- School Support (Morgan 2011)
- Conditions Provided by Institutions for job satisfaction (CPIJS) (Silva 2013)
- Administrative and Policy Demands

### CURRENT STUDY FOCUS: Pre-service teacher self-efficacy (question 114-

122 of the FIRE Teacher Resilience Measure)

- Scale: Teacher Resilience Scale (Morgan, 2011)
- Items: 9 items
- Assess: Pre-service teachers' confidence in recovery from setbacks in school

### CURRENT STUDY FOCUS:

Pre-service teacher efficacy (questions 123-134 of the *FIRE Teacher Resilience Measure*):

- Scale: Teacher Efficacy Scale (Morgan, 2011; Peixoto et al., 2018)
- Items: 12 items
- Assess: Pre-service teachers' confidence in teaching and behaviour management



<b>SCALES UTILISED IN THE ENTREE PROJECT</b> (Peixoto, F., Wosnitza, M., Pipa, J., Morgan, M., & Cefai, C. (2018). A multidimensional view on pre-service teacher resilience in Germany, Ireland, Malta and Portugal. In M. Wosnitza, F. Peixoto, S. Beltman, & C. F. Mansfield (Eds.), <i>Resilience in education: Concepts, contexts and connections</i> (pp. 73–89). Springer International Publishing. https://doi.org/10.1007/978-3-319-76690-4)	DEVELOPED BY
Global Measure of Teacher Resilience [i.e., Teacher Resilience Scale (Morgan, 2011)](9 items)	Morgan, 2011
Multidimensional Teacher Resilience Scale (MTRS)(16 items) [developed from BRiTE Project based on the KEEPING COOL Project]	Mansfield & Wosnitza, 2015
Personal Life (4 items)	Morgan, 2011
Rumination Scale (4 items)	Morgan, 2011
Teacher Commitment (8 items)	Morgan, 2011
Teacher Self-Efficacy (added items for ENTREE) (based on) [i.e., Teacher Efficacy Scale (Morgan, 2011)] (6 items)	Morgan, 2011 Peixoto et al., 2018
Social Context (18 items)	Cefai & Cavioni, 2014
School Support (4 items)	Morgan, 2011
Conditions Provided by Institutions for Job Satisfaction (CPIJS) (12 items)	Silva, 2013
Administrative and Policy Demands (6 items)	



Г

	S	CALES UTILISED FOR THE CURRENT STUDY FRO	M THE FIRE PROJECT
ORIGINAL SCALE	ENTREE SCALE	ENTREE SYSTEM LEVEL	REFERENCES FOR SCALE DEVELOPMENT
Teacher Resilience scale (Morgan, 2011) Items: 9	Global Measure of Teacher Resilience (Morgan, 2011) Items: 9	Global measure of resilience	<ul> <li>Morgan, M. (2011). Resilience and recurring adverse events: Testing an assets-based model of beginning teachers' experiences. <i>The Irish Journal of Psychology, 32</i>(3-4), 92–104. https://doi.org/10.1080/03033910.2011.613189</li> <li>Peixoto, F., Silva, J. C., Pipa, J., Wosnitza, M., &amp; Mansfield, C. (2020). The Multidimensional Teachers' Resilience Scale: Validation for Portuguese teachers. <i>Journal of Psychoeducational Assessment, 38</i>(3), 402–408. https://doi.org/10.1177/0734282919836853</li> <li>Peixoto, F., Wosnitza, M., Pipa, J., Morgan, M., &amp; Cefai, C. (2018). A multidimensional view on pre-service teacher resilience in Germany, Ireland, Malta and Portugal. In M. Wosnitza, F. Peixoto, S. Beltman, &amp; C. F. Mansfield (Eds.), <i>Resilience in education: Concepts, contexts and connections</i> (pp. 73–89). Springer International Publishing. https://doi.org/10.1007/978-3-319-76690-4</li> </ul>
Teacher Efficacy scale (Morgan, 2011) Items: 6	Teacher Self-Efficacy scale (Morgan, 2011; Peixoto et al., 2018 (added items)) Items: 12	<ul> <li>Micro level: domain of beliefs</li> <li>Understand the teacher's beliefs on which their behaviour is built like self- efficacy:         <ul> <li>Teacher efficacy for teaching (6 items)</li> <li>Teacher efficacy for behaviour management (6 items)</li> </ul> </li> </ul>	<ul> <li>Morgan, M. (2011). Resilience and recurring adverse events: Testing an assets-based model of beginning teachers' experiences. <i>The Irish Journal of Psychology, 32</i>(3-4), 92–104. https://doi.org/10.1080/03033910.2011.613189</li> <li>Peixoto, F., Wosnitza, M., Pipa, J., Morgan, M., &amp; Cefai, C. (2018). A multidimensional view on pre-service teacher resilience in Germany, Ireland, Malta and Portugal. In M. Wosnitza, F. Peixoto, S. Beltman, &amp; C. F. Mansfield (Eds.), <i>Resilience in education: Concepts, contexts and connections</i> (pp. 73–89). Springer International Publishing. https://doi.org/10.1007/978-3-319-76690-4</li> </ul>



# APPENDIX B: FIRE TEACHER RESILIENCE MEASURE

DATE:	
NAME AND SURNAME:	
STUDENT NUMBER:	
AGE:	GENDER:
UP PROGRAMME ENROLLED FO	R:
LIST OF LANGUAGES YOU ARE F	LUENT IN:

		Source	Scale	Item	Pleas	e cho	ose y	our a	nswei	• by	
	The fo	ollowings	statements e	express different views about teaching. Read each	circli	ng the	appi	opria	te bo	x rang	ing
	senter	nce and cl	hoose the op	tion that better reflects your opinion, by checking the	from	1 = L	o not	agre	e at al	ll to 7	=
	table	below			Stron	gly ag	gree				
core	1	СМ	TR-Prof	At school I can be flexible when situations change	1	2	3	4	5	6	7
core	2	СМ	TR-Prof	I can quickly adapt to new situations at school	1	2	3	4	5	6	7
core	3	СМ	TR-Prof	I am well organised in my school work	1	2	3	4	5	6	7
core	4	СМ	TR-Prof	I reflect on my teaching and learning to make future	1	2	3	4	5	6	7
				plans							
core	5	СМ	TR-Emot	When something goes wrong at school I don't take it	1	2	3	4	5	6	7
				too personally							
core	6	СМ	TR-Emot	After reflection, I can usually find the funny side of	1	2	3	4	5	6	7
				challenging school situations					_		
core	7	СМ	TR-Emot	When I feel upset or angry at school I can manage to	1	2	3	4	5	6	7
				stay calm					_		
core	8	СМ	TR-Emot	I balance my role as a teacher with other dimensions	1	2	3	4	5	6	7
				in my life			-		_		_
core	9	CM	TR-Mot	l am generally optimistic at school	1	2	3	4	5	6	7
core	10	СМ	TR-Mot	At school I focus on building my strengths more than	1	2	3	4	5	6	7
		<u></u>		focusing on my limitations					_		_
core	11	СМ	TR-Mot	When I make mistakes at school I see these as learning	1	2	3	4	5	6	7
	10	<u>CM</u>		opportunities	1	2	2		_	6	-
core	12	СМ	TR-Mot	In my role as a teacher I set goals and work towards	1	2	3	4	5	6	7
	12	CM	TD Mat	I have realistic surrestations of musclf as a teacher	1	2	2	4	-	(	7
core	13	CM	TR-MOU	I have realistic expectations of myself as a teacher	1	2	3	4	р С	0	7
core	14	СМ	I K-MOL	can be successful	1	2	з	4	Э	0	
core	15	СМ	TR-Mot	I am good at maintaining my motivation and	1	2	3	4	5	6	7
				enthusiasm when things get challenging at school							
core	16	СМ	TR-Mot	I enjoy learning when I am at work	1	2	3	4	5	6	7
core	17	СМ	TR-Mot	I like challenges in my work	1	2	3	4	5	6	7
core	18	СМ	TR-Mot	I am persistent in my work	1	2	3	4	5	6	7
core	19	СМ	TR-Mot	I believe that I have control over my work life	1	2	3	4	5	6	7
core	20	СМ	TR-Mot	It's important to me that I put in effort to do my job well	1	2	3	4	5	6	7
core	21	СМ	TR-Soc	When I am unsure of something I seek help from	1	2	3	4	5	6	7
core	22	CM	TR-Soc	Lam good at huilding relationshins in new school	1	2	3	4	5	6	7
core	22	CIM	11-300	environments	1	2	5	Т	5	0	,
core	23	СМ	TR-Prof	In my role as a teacher, i am a good communicator	1	2	3	4	5	6	7
core	24	СМ	TR-Soc	in my work I can look at a situation a number of ways	1	2	3	4	5	6	7
				to find a solution							
core	25	CM	TR-Prof	At work I can view situations from other people's	1	2	3	4	5	6	7
				perspectives							
core	26	СМ	TR-Soc	When I am at work I can generally resolve conflicts	1	2	3	4	5	6	7
				with others							



		Source	Scale	Item	Pleas	e cho	ose y	our ai	nswei	· by	
	The fo	llowing	statements e	express different views about teaching. Read each	circli	ngthe	appi	ropria	te bo	x rang	ing
	senter	ice and cl	hoose the op	tion that better reflects your opinion, by checking the	from	1 = A	bsolu	tely n	ot Coi	ıfiden	t to
	table	Delow	D 11		7 = 50	rong	y con	fiaen		6	-
core	114	MM	Resilience	Getting over setbacks in school	1	2	3	4	5	6	7
core	115	MM	Resilience	Bouncing back, when things upset me	1	2	3	4	5	6	7
core	116	ММ	Resilience	Carrying on with my school work when things go wrong	1	2	3	4	5	6	7
core	117	MM	Resilience	Carrying on in school when events upset me	1	2	3	4	5	6	7
core	118	MM	Resilience	Feeling certain that things will come right even if there are serious problems in school.	1	2	3	4	5	6	7
core	119	MM	Resilience	Managing negative events in school when I try	1	2	3	4	5	6	7
core	120	MM	Resilience	Coping with most problems on any school day	1	2	3	4	5	6	7
core	121	MM	Resilience	Some negative things that have happened in school have made me better able to deal with problems	1	2	3	4	5	6	7
core	122	MM	Resilience	Not getting disheartend even when children's circumstances make it difficult.	1	2	3	4	5	6	7
core	123	MM	TeachEff	Teaching all the subjects on the curriculum effectively	1	2	3	4	5	6	7
core	124	ММ	TeachEff	Explaining difficult material in ways that the children will understand it	1	2	3	4	5	6	7
core	125	MM	TeachEff	Suggesting suitable examples when the children are having difficulty understanding	1	2	3	4	5	6	7
core	126	MM	TeachEff	Teaching in a way that my students will remember important information	1	2	3	4	5	6	7
core	127	MM	TeachEff	Applying the new developments in the curriculum into my teaching	1	2	3	4	5	6	7
core	128	MM	TeachEff	Helping children focus on learning tasks and avoid distractions	1	2	3	4	5	6	7
core	129	new	TeachEff	Managing inappropriate behaviour	1	2	3	4	5	6	7
core	130	new	TeachEff	Encouraging students to take responsibility for their behaviour	1	2	3	4	5	6	7
core	131	new	TeachEff	Dealing with the diverse learning needs of the students in my class.	1	2	3	4	5	6	7
core	132	new	TeachEff	Teaching students positive bahviour	1	2	3	4	5	6	7
core	133	new	TeachEff	Providing students with clear specific behaviour expectations	1	2	3	4	5	6	7
core	134	new	TeachEff	Communicating effectively with parents	1	2	3	4	5	6	7



Ple	ase rat	e how re	elevant eac	h statement below is in terms of your career as a tea	cher						
		Source	Scale	Item	Pleas	e cho	ose y	our ai	nswei	· by	
	The fo senter table	ollowing s nce and c below	tatements e hoose the op	xpress different views about teaching. Read each tion that better reflects your opinion, by checking the	circli from: <i>Stron</i>	ng the : 1 = <i>L</i> gly ag	e appi Do not gree	opria agre	te bo: e at al	x rang // to 7 :	ing =
core	1	LE	Contextual	I did not want to become a teacher.	1	2	3	4	5	6	7
core	2	LE	Contextual	I may get to love teaching in future.	1	2	3	4	5	6	7
core	3	LE	Contextual	Teachers have played a positive role in my life.	1	2	3	4	5	6	7
core	4	LE	Contextual	I want to be a teacher who instils hope in learners even in the face of many obstacles	1	2	3	4	5	6	7
core	5	LE	Contextual	As a learner, I attended a school with many challenges and few resources	1	2	3	4	5	6	7
core	6	LE	Contextual	I know how to teach in a school where there are many challenges and few resources	1	2	3	4	5	6	7
core	7	LE	Contextual	As a teacher I want to take initiative to solve problems in schools	1	2	3	4	5	6	7
core	8	LE	Contextual	As a teacher I will wait for government or officials to solve problems in schools	1	2	3	4	5	6	7
core	9	LE	Contextual	I will meet informally with other teachers to discuss ways to deal with challenges	1	2	3	4	5	6	7
core	10	LE	Contextual	I know that teachers have to teach in schools that face many, on-going challenges	1	2	3	4	5	6	7
core	11	LE	Contextual	My spirituality/religion helps me to be a teacher	1	2	3	4	5	6	7



FIT-Choice Scale (Factors Influencing Teaching Choice Scale): Copyright © HMG Watt & PW Richardson

### Please briefly state your main reason(s) for choosing to become a teacher:

# PART B - INFLUENTIAL FACTORS

For each statement below, please rate how important it was in YOUR decision to become a teacher, from 1 (not at all important in your decision) to 7 (extremely important in your decision).

Please **CIRCLE** the number that best describes the importance of each.

# "I chose to become a teacher because..."

_		not at a importar	II It					extre impor	mely tant
B1.	I am interested in teaching	1	2	3	4	5	6	7	B1.
B2.	Part-time teaching could allow more family time	1	2	3	4	5	6	7	B2.
B3.	My friends think I should become a teacher	1	2	3	4	5	6	7	B3.
B4.	As a teacher I will have lengthy holidays	1	2	3	4	5	6	7	B4.
B5.	I have the qualities of a good teacher	1	2	3	4	5	6	7	B5.
B6.	Teaching allows me to provide a service to society	1	2	3	4	5	6	7	B6.
B7.	I've always wanted to be a teacher	1	2	3	4	5	6	7	B7.
B8.	Teaching may give me the chance to work abroad	1	2	3	4	5	6	7	B8.
B9.	Teaching will allow me to shape child/adolescent values	1	2	3	4	5	6	7	B9.
B11.	I was unsure of what career I wanted	1	2	3	4	5	6	7	B11.
B12.	I like teaching	1	2	3	4	5	6	7	B12.
B13.	I want a job that involves working with children/adolescents	1	2	3	4	5	6	7	B13.
B14.	Teaching will offer a steady career path	1	2	3	4	5	6	7	B14.
B16.	Teaching hours will fit with the responsibilities of having a famil	y 1	2	3	4	5	6	7	B16.
B17.	I have had inspirational teachers	1	2	3	4	5	6	7	B17.
B18.	As a teacher I will have a short working day	1	2	3	4	5	6	7	B18.
B19.	I have good teaching skills	1	2	3	4	5	6	7	B19.
B20.	Teachers make a worthwhile social contribution	1	2	3	4	5	6	7	B20.
B22.	A teaching qualification is recognised everywhere	1	2	3	4	5	6	7	B22.
B23.	Teaching will allow me to influence the next generation	1	2	3	4	5	6	7	B23.
B24.	My family think I should become a teacher	1	2	3	4	5	6	7	B24.
B26.	I want to work in a child/adolescent-centred environment	1	2	3	4	5	6	7	B26.
B27.	Teaching will provide a reliable income	1	2	3	4	5	6	7	B27.
B29.	School holidays will fit in with family commitments	1	2	3	4	5	6	7	B29.
B30.	I have had good teachers as role-models	1	2	3	4	5	6	7	B30.



### "I chose to become a teacher because..."

10	nose to become a reacher because								
		not at a importar	ll It					extre impor	mely tant
B31.	Teaching enables me to 'give back' to society	1	2	3	4	5	6	7	B31.
B35.	I was not accepted into my first-choice career	1	2	3	4	5	6	7	B35.
B36.	Teaching will allow me to raise the ambitions of underprivileged yout	h <b>1</b>	2	3	4	5	6	7	B36.
B37.	I like working with children/adolescents	1	2	3	4	5	6	7	B37.
B38.	Teaching will be a secure job	1	2	3	4	5	6	7	B38.
B39.	I have had positive learning experiences	1	2	3	4	5	6	7	B39.
B40.	People I've worked with think I should become a teacher	1	2	3	4	5	6	7	B40.
B43.	Teaching is a career suited to my abilities	1	2	3	4	5	6	7	B43.
B45.	A teaching job will allow me to choose where I wish to live	1	2	3	4	5	6	7	B45.
B48.	I chose teaching as a last-resort career	1	2	3	4	5	6	7	B48.
B49.	Teaching will allow me to benefit the socially disadvantaged	1	2	3	4	5	6	7	B49.
B53.	Teaching will allow me to have an impact on children/adolescents	1	2	3	4	5	6	7	B53.
B54.	Teaching will allow me to work against social disadvantage	1	2	3	4	5	6	7	B54.

# PART C - BELIEFS ABOUT TEACHING

For each question below, please rate the extent to which YOU agree it is true about teaching, from 1 (not at all) to 7 (extremely). Please CIRCLE the number that best describes your agreement for each not at all extremely

								•
Do you think teaching is well paid?	1	2	3	4	5	6	7	C1.
Do you think teachers have a heavy workload?	1	2	3	4	5	6	7	C2.
Do you think teachers earn a good salary?	1	2	3	4	5	6	7	СЗ.
Do you believe teachers are perceived as professionals?	1	2	3	4	5	6	7	C4.
Do you think teachers have high morale?	1	2	3	4	5	6	7	<i>C</i> 5.
Do you think teaching is emotionally demanding?	1	2	3	4	5	6	7	C6.
Do you believe teaching is perceived as a high-status occupation?	1	2	3	4	5	6	7	C7.
Do you think teachers feel valued by society?	1	2	3	4	5	6	7	C8.
Do you think teaching requires high levels of expert knowledge?	1	2	3	4	5	6	7	C9.
Do you think teaching is hard work?	1	2	3	4	5	6	7	C10.
Do you believe teaching is a well-respected career?	1	2	3	4	5	6	7	C11.
Do you think teachers feel their occupation has high social status?	1	2	3	4	5	6	7	C12.
Do you think teachers need high levels of technical knowledge?	1	2	3	4	5	6	7	C13.
Do you think teachers need highly specialised knowledge?	1	2	3	4	5	6	7	C15.
	Do you think teaching is well paid? Do you think teachers have a heavy workload? Do you think teachers earn a good salary? Do you believe teachers are perceived as professionals? Do you think teachers have high morale? Do you think teaching is emotionally demanding? Do you think teachers feel valued by society? Do you think teaching requires high levels of expert knowledge? Do you think teaching is a well-respected career? Do you think teachers feel their occupation has high social status? Do you think teachers need high levels of technical knowledge?	Do you think teaching is well paid?1Do you think teachers have a heavy workload?1Do you think teachers earn a good salary?1Do you believe teachers are perceived as professionals?1Do you think teachers have high morale?1Do you think teaching is emotionally demanding?1Do you believe teaching is perceived as a high-status occupation?1Do you think teachers feel valued by society?1Do you think teaching requires high levels of expert knowledge?1Do you believe teaching is a well-respected career?1Do you think teachers feel their occupation has high social status?1Do you think teachers need high levels of technical knowledge?1Do you think teachers need highly specialised knowledge?1	Do you think teaching is well paid?12Do you think teachers have a heavy workload?12Do you think teachers earn a good salary?12Do you believe teachers are perceived as professionals?12Do you think teachers have high morale?12Do you think teaching is emotionally demanding?12Do you believe teaching is perceived as a high-status occupation?12Do you think teachers feel valued by society?12Do you think teaching requires high levels of expert knowledge?12Do you believe teaching is a well-respected career?12Do you think teachers feel their occupation has high social status?12Do you think teachers need high levels of technical knowledge?12Do you think teachers need highly specialised knowledge?12	Do you think teaching is well paid?123Do you think teachers have a heavy workload?123Do you think teachers earn a good salary?123Do you believe teachers are perceived as professionals?123Do you think teachers have high morale?123Do you think teachers is emotionally demanding?123Do you believe teaching is perceived as a high-status occupation?123Do you think teachers feel valued by society?123Do you think teaching requires high levels of expert knowledge?123Do you believe teaching is a well-respected career?123Do you think teachers feel their occupation has high social status?123Do you think teachers need high levels of technical knowledge?123Do you think teachers feel their occupation has high social status?123Do you think teachers need highly specialised knowledge?123	Do you think teaching is well paid?1234Do you think teachers have a heavy workload?1234Do you think teachers earn a good salary?1234Do you believe teachers are perceived as professionals?1234Do you think teachers have high morale?1234Do you think teaching is emotionally demanding?1234Do you believe teaching is perceived as a high-status occupation?1234Do you think teachers feel valued by society?1234Do you think teaching requires high levels of expert knowledge?1234Do you believe teaching is a well-respected career?1234Do you think teachers feel their occupation has high social status?1234Do you think teachers need high levels of technical knowledge?1234Do you think teachers need highly specialised knowledge?1234	Do you think teaching is well paid?12345Do you think teachers have a heavy workload?12345Do you think teachers earn a good salary?12345Do you believe teachers are perceived as professionals?12345Do you think teachers have high morale?12345Do you think teaching is emotionally demanding?12345Do you believe teaching is perceived as a high-status occupation?12345Do you think teachers feel valued by society?12345Do you think teaching is hard work?12345Do you believe teaching is a well-respected career?12345Do you think teachers feel their occupation has high social status?12345Do you think teachers need high levels of technical knowledge?12345Do you think teachers need highly specialised knowledge?12345	Do you think teaching is well paid?123456Do you think teachers have a heavy workload?123456Do you think teachers earn a good salary?123456Do you believe teachers are perceived as professionals?123456Do you think teachers have high morale?123456Do you think teaching is emotionally demanding?123456Do you believe teaching is perceived as a high-status occupation?123456Do you think teachers feel valued by society?123456Do you think teaching is hard work?123456Do you believe teaching is a well-respected career?123456Do you think teachers feel their occupation has high social status?123456Do you think teachers need high levels of technical knowledge?123456Do you think teachers need highly specialised knowledge?123456	Do you think teaching is well paid?1234567Do you think teachers have a heavy workload?1234567Do you think teachers earn a good salary?1234567Do you believe teachers are perceived as professionals?1234567Do you think teachers have high morale?1234567Do you think teaching is emotionally demanding?1234567Do you think teaching is perceived as a high-status occupation?1234567Do you think teachers feel valued by society?1234567Do you think teaching is awell-respected career?1234567Do you think teachers feel their occupation has high social status?1234567Do you think teachers feel their occupation has high social status?1234567Do you think teachers feel their occupation has high social status?1234567Do you think teachers need high levels of technical knowledge?1234567Do you think teachers need highly specialised knowledge?1234567



For	PART D - <u>YOUR</u> DECISION TO BECOME each question below, please rate the extent to which it is true	A for	TE · yo	EA OU	CH , fr	IER rom	2	(not at	t all) to
	7 (extremely). Please CIRCLE the number that best describes	you	r a	gre	err	en <sup>.</sup>	t fo	or eac	h.
	not	at al	I				(	extrem	ely
D1.	How carefully have you thought about becoming a teacher?	1	2	3	4	5	6	7	D1.
D2.	Were you encouraged to pursue careers other than teaching?	1	2	3	4	5	6	7	D2.
D3.	How satisfied are you with your choice of becoming a teacher?	1	2	3	4	5	6	7	D3.
D4.	Did others tell you teaching was not a good career choice?	1	2	3	4	5	6	7	D4.
D5.	How happy are you with your decision to become a teacher?	1	2	3	4	5	6	7	D5.
D6.	Did others influence you to consider careers other than teaching?	1	2	3	4	5	6	7	D6.

Copyright © 2004, Helen M. G. Watt & Paul W. Richardson. All rights reserved.

For information about this work, please contact Helen M. G. Watt and Paul W. Richardson.

Helen Watt, Faculty of Education, Monash University, Clayton Campus, Melbourne VIC 3800, Australia. Tel: (+61 3) 9905-3276, Email: <u>helen.watt@monash.edu</u>. Paul Richardson, Faculty of Education, Monash University, Clayton Campus, Melbourne VIC 3800, Australia. Tel: (+61 3) 9905-2771, Email: <u>paul.richardson@monash.edu</u>

Permission is hereby granted to reproduce and distribute copies of this work for nonprofit educational purposes, provided that the authors, source and copyright notice are included on each copy. This permission is in addition to rights of reproduction granted under Sections 107, 108, and other provisions of the U.S. Copyright Act. Before making any distribution of this work, please contact the authors to ascertain whether you have the current version.



### APPENDIX C: FIRE PROJECT ETHICAL APPROVAL AND EXTRACTS



UNIVERSITEIT VAN PRETORIA UNIVERSITY OF PRETORIA YUNIBESITHI YA PRETORIA

Faculty of Education Office of the Dean

### 18 March 2015

**Dear Prof Fraser** 

PERMISSION FOR RESEARCH

Your request dated 18 March 2015 regarding the FIRE 2015 research project refers.

Permission is granted to conduct the research with fourth-year students, as described in the summary of the project.

Kind regards

Irma Eloff Education Qean: cc Prof L Ebersöhn

University of Pretoria PRETORIA 0002 Republic of South Africa Tel: +27 (0)12 420 5721 Fax: +27 (0)12 420 4215

http://www.up.ac.za





Ethics Committee 4 September 2015

Dear Prof. Fraser,

REFERENCE: UP 14/03/01

Your application was considered by the Faculty of Education Ethics Committee and the final decision of the Ethics Committee is:

Your application is approved.

This letter serves as notification that you may continue with your fieldwork. Should any changes to the study occur after approval was given, it is your responsibility to notify the Ethics Committee immediately.

Please note that this is not a clearance certificate. Upon completion of your research you need to submit the following documentation to the Ethics Committee:

1. Integrated Declarations form that you adhered to conditions stipulated in this letter - Form D08

Please Note:

- <u>Any</u> amendments to this approved protocol need to be submitted to the Ethics Committee for review prior to data collection. Non-compliance implies that the Committee's approval is null and void.
- Final data collection protocols and supporting evidence (e.g.: questionnaires, interview schedules, observation schedules) have to be submitted to the Ethics Committee <u>before</u> they are used for data collection.
- Should your research be conducted in <u>schools</u>, please note that you have to submit <u>proof</u> of how you <u>adhered</u> to the Department of Basic Education (DBE) <u>policy</u> for research.
- Please note that you need to keep to the protocol you were granted approval on 
   should your
  research project be amended, you will need to submit the amendments for review.
- The Ethics Committee of the Faculty of Education does not accept any liability for research misconduct, of whatsoever nature, committed by the researcher(s) in the implementation of the approved protocol.
- On receipt of the above-mentioned documents you will be issued a clearance certificate. Please
  quote the reference number: UP 14/03/01 in any communication with the Ethics Committee.

Best wishes,

Prof Liesel Ebersöhn Chair: Ethics Committee Faculty of Education





### UNIVERSITEIT VAN PRETORIA UNIVERSITY OF PRETORIA YUNIBESITHI YA PRETORIA

Faculty of Education Department of Science, Mathematics and Technology Education William.fraser@up.ac.za 18 March 2015

Prof Irma Eloff, Dean Faculty of Education Groenkloof Campus

Dear Prof Eloff

FIRE 2015: REQUESTING PERMISSION TO USE FOURTH-YEAR STUDENTS AS RESEARCH SUBJECTS

With the new 2015-NRF grant our focus will shift to fourth-year students' experiences in the development of teacher identities during teaching practice as a community of practice. In 2014, the focus fell on the opinions of principals, parents and mentor teachers on factors impacting on effective schools and effective teachers.

- 1. We hereby ask permission to engage approximately 800 student teachers during the second and third quarters in Participatory Reflection and Action Interventions this year to achieve the following:
- (a) Establishment and identification of their (student teachers) perceptions of teacher identities. The main elements of teacher identities will be established during the first phase of the PRA.
- (b) During the second PRA session students will list factors contributing to their developing identities. The focus will be on the potential resources.
- (c) Thirdly, students will indicate which components will further their development, and
- (d) Finally list the action plans to engage the potential resources in the improvement of the teacher identities.

Prof Ferreira will lead the PRA intervention and participating researchers will be briefed by her prior to the commencement of the activities.

We also seek permission to apply a quantitative instrument to assess the development of their teacher identities by content validating the instrument against the eight educator roles.

An amendment to the 2014-ethics application will be submitted for approval to the Ethics Committee prior to the commencement of the fieldwork. At this stage we don't foresee engaging school staff members or parents in the research as had been the case in 2014.

Looking forward hearing from you soon.

Prof WJ Fraser Research Co-ordinator



e ale				
	UNIVERSITEIT	VAN PRETORIA		
	VUNIVERSITY (	OF PRETORIA	Reference:	
. Vignieran	Faculty of Education	YA PRETORIA		
PERS	ONAL DECLARAT	TION OF RESPONSIBILI	ТҮ	
Title o	f research project:			
How s effecti	student teachers a ve	and mentor teachers co	ntribute in making teac	hing more
1. I/we Edu	declare that I am/we a cation to	are cognisant of the goals of th	e Research Ethics Committ	
ا ۲	<ul> <li>develop among stu conceptualisation a</li> <li>cultivate</li> </ul>	idents and researchers a high ind conduct of educational rese	standard of ethics and eth	ical practice in the
	respondents; and promote among res	Consciousness among scho	plars especially in research	involving human
2 10.4	the research proces	s.	nan rights and dignity of hum	an respondents in
∠. //We	voluntary portion	ples of		
100 100	research at any time	on in research, implying the	at the participants might	ithdraw (
	informed consent, m	leaning that research participation	anto must i i i	andraw from the
	safety in participation	and purposes, and must give	Consent to their partial	ly informed about
	harm of any kind e.g.	put differently, that the huma	n respondents should not be	in the research.
	privacy, meaning th	at the confidentiality and		placed at risk or
	trust which implies	and an	<i>ionymity</i> of human respon-	dents should be
	betrayal in the research	hat human respondents will r ch process or its published out	ot be subjected to any acts	s of deception or
3. I/we ur	derstand what plagiari	sm entails and and		
our own own wo	ake not to make use of h. I/we also undertake rk.	another person's previous wo not to allow anyone to copy o	of the University's policy in rk without acknowledgment of ur work with the intention of	this regard. I/we or to submit it as using it as their
4. I/we und	derstand that the data	collected in the		
the University of the Universi	versity of Pretoria and for safekeeping as rea	I l/we undertake to transfer a quired by the Faculty of Educa	esearch become the instituti all raw data and documents tion.	onal property of related to our
5. I/we und Committ	derstand that <u>any ame</u> ee for review prior to da	endment to the approved pro ata collection. Non-compliance	tocol needs to be submitte	d to the Ethics
Prof WJ Fras	er	ables		null and void.
Applicant		SHANA	2014	/03/01
		Quality	Date	
		Y		
Supervisor (i	f applicable)	Signature		
		-iAnaral6	Date	
				<
		-		







The first sub-question deals with mentor teachers' perceptions of effective teachers.

1.1 What are according to mentors teachers the ideal characteristics of an effective teacher?

The second sub-question deals with mentor teachers' perceptions of an effective school.

1.2 What are mentor teachers' perceptions of an effective school?

The third sub-question requests parents to indicate what according to them should be the most ideal characteristics of an effective teacher.

1.3 What should be according to parents the characteristics of an effective teacher?

The fourth sub-question deals with mentor teachers' perceptions of the ideal characteristics of a student

1.4 What should be according to mentor teachers the ideal characteristics of a student teacher entering a

The last sub-question addresses principals' expectations of the qualities a student teacher should possess when entering the school during teaching practice for the first time.

1.5 What are according to principals the three most important qualities student teacher should possess when entering schools for the first time during teaching practice?

# 2. RESEARCH DESIGN/METHODOLOGY

2.1 Please provide a full description of the research design/methodology, and processes that will be used. Include details relating to the research sites and data collection protocols.

Both qualitative and quantitative approaches will apply to the investigation. The study will not rely on a classical integrated multiple (mixed) methods approach but more on a predominantly quantitative design, followed by a supplemental short qualitative investigation. The data collected by means of questionnaires (semantic differentials) will precede possible interviews to be conducted with the school principal.

All student teachers as part of JNM 410 received a short reading list covering the work by authors such as Bosker, Creemers and Scheerens who have worked very comprehensively on school effectiveness. These articles deal with school effectiveness from an input, process and output point of view and students were requested as part of Assignment 1, to compile a short but comprehensive synopsis of how different factors could impact on school effectiveness. The articles deal with the interpretation of school effectiveness results, a model of school effectiveness, monitoring school effectiveness in developing countries, the improvement of school effectiveness and the conceptualisation of school effectiveness.

Student teachers will be placed in 160 'controlled' Pretoria schools and in 100 other schools outside the Pretoria area during the second quarter of 2014. However these numbers are not fixed and could change.

Sampling should be regarded mainly as purposive although elements of convenience are also present as students from Block B (first quarter) and students from Block A (second quarter) will not engage with their mentor lecturers as they will not be assessed by them during the second quarter of 2014. All 680 students enrolled for JNM 410 and JNM 451 will participate in the study and the data collected, coded, transcribed and reported by the students will contribute towards the final mark for JNM 420. The JNM 451 students however will only collect information and return the questionnaires for processing as they are not further involved in JNM 420. The reader should note that for the purpose of JNM 420 the data collected will not be reported and published as Block B schools during the second quarter of 2014 will not be approached to participate in the study as many of these schools fall outside the jurisidiction of the GDE. However, permission will be sought from the GDE to conduct research in Block A schools (second quarter and third quarter) where student teachers will be controlled by mentor lecturers. Furthermore the GDE will be approach to grant permission and allow student teachers to engage with mentor teachers where mentor lecturers will participate with their students in responding to a number of questionnaires. One student

5



Because the		also conduct	
School off		and conduct one interview i	with the school principal.
managemer	e study only focu ctiveness, the nt, administrative	ses on the school effectivene content validation of the n , teaching, monitoring and ass	ss process factors and how these factors relate neasuring instruments was dealt with from sessment point of view.
During the f	irst activity of th A) to indicate wh	e study mentor teachers will at according to them are the c	use a questionnaire (semantic differential scale haracteristics of an effective teacher.
scale) (Instruschool, as we time.	econd activity of ument B & C) to ell as the charact	the study mentor teachers w indicate what according to t teristics of a student teacher e	ill again use questionnaires (semantic differentia hem should be the charateristics of an effective entering a school for teaching practice for the firs
During the thi D) to indicate	ird activity of the what according	study parents will use a ques to them should be the charact	tionnaire (semantic differential scale) (Instrument
During the for with the princ student teach (Instrument E)	urth activity of th sipal to determin er visiting a sch ).	e study one student teacher e what according to her/him ool for the first time during te	leader per school will conduct a short interview should be the three most important qualities a eaching practice should be able to de-
			e provide should be able to demonstrate
2.2 Should cli	nical data form p	art of the data assume in the	
obtaining	permission and i	nformed consent to use such	study, detail the relevant processes for
n/a			
1			
L			
22 1511			
2.3 If this is inte	ervention researc	ch describe the set	
scientific m	erit of the interve	ention you jet a nature of the	intervention and provide the
n/a			
W Please note	that you h		
Please note to your parti which degree	e that you have a n icipants when you se the procedures a	esponsibility to ensure that you d invite them to participate in your and instruments you intend to us	isclose fully the scientific status of the intervention research. Participants have the right to know to e are accepted by the scientific community.
Please note to your parti which degree	e that you have a n icipants when you se the procedures a	esponsibility to ensure that you d invite them to participate in your and instruments you intend to us	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your parti which degre	e that you have a n icipants when you se the procedures a	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA	isclose fully the scientific status of the intervention research. Participants have the right to know to e are accepted by the scientific community.
Please note to your parti which degre	e that you have a n icipants when you se the procedures a will be participation	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA	isclose fully the scientific status of the intervention research. Participants have the right to know to e are accepted by the scientific community.
Please note to your parti which degre	e that you have a n icipants when you se the procedures the procedures will be participat	esponsibility to ensure that you d invite them to participate in your and instruments you intend to us 3. HUMAN PARTICIPA ting in the study. Mention any	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your parts which degree	e that you have a n icipants when you se the procedures will be participat	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use <b>3. HUMAN PARTICIPA</b> ting in the study. Mention any	isclose fully the scientific status of the intervention research. Participants have the right to know to e are accepted by the scientific community.
Please note to your part which degree	e that you have a n icipants when you se the procedures will be participal Vulnerability Status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation	isclose fully the scientific status of the intervention research. Participants have the right to know to e are accepted by the scientific community.
Please note to your part which degree	e that you have a n icipants when you se the procedures will be participat Vulnerability Status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your part which degre	e that you have a n icipants when you se the procedures will be participat Vulnerability status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your part which degre	e that you have a n icipants when you se the procedures will be participat Vulnerability status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your part which degree	e that you have a n icipants when you se the procedures will be participat Vulnerability status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your part which degree	e that you have a n icipants when you se the procedures will be participat Vulnerability Status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to us 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your part which degree i.1 Describe who study. Role Close to 680 inal Year eaching ractice Students art of Block A econd quarter) id Block A	e that you have a n icipants when you se the procedures will be participat Vulnerability status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your part which degree to your part to Block A econd quarter hid Block B (third arter)	e that you have a n icipants when you se the procedures will be participat Vulnerability status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation University of Pretoria JNM 410 and JNM 451 Students	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your part which degree the study.	e that you have a n icipants when you se the procedures will be participat Vulnerability status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation University of Pretoria JNM 410 and JNM 451 Students	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your part which degree the study. Role Close to 680 inal Year eaching ractice Students art of Block A econd quarter) id Block B (third arter) ontrolled idents)	e that you have a n icipants when you se the procedures will be participat Vulnerability status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation University of Pretoria JNM 410 and JNM 451 Students	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your part which degree to your part to Block A econd quarter) id Block B (third arter) ontrolled udents)	e that you have a n icipants when you e the procedures will be participat Vulnerability status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation University of Pretoria JNM 410 and JNM 451 Students	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your part which degree i.1 Describe who study. Role Close to 680 inal Year eaching ractice Students art of Block A econd quarter) id Block B (third arter) ontrolled idents)	e that you have a n icipants when you e the procedures will be participat Vulnerability status	esponsibility to ensure that you d invite them to participate in your and instruments you intend to us 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation University of Pretoria JNM 410 and JNM 451 Students	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.
Please note to your part which degree	e that you have a micipants when you have a micipants when you have a micipants when you have the procedures of the proc	esponsibility to ensure that you d invite them to participate in your and instruments you intend to use 3. HUMAN PARTICIPA ting in the study. Mention any Institutional affiliation University of Pretoria JNM 410 and JNM 451 Students	isclose fully the scientific status of the intervention research. Participants have the right to know to a are accepted by the scientific community.



Mentor Teachers         Low         Geuteng Department of Gradeter Pretoria Area) and Gauteng Private School (Gradet Pretoria Area) and database Auring teaching practice         Complete questionnaires (semantic oprime of effective school (mathematice), teachers (instrument Q).           Parents         Low         Parents of learners taught teachers during teaching private school (Gradet Pretoria Area) and well as private schools in grading         Complete questionnaires (instrument Q) teaching public schools (Gradet Pretoria Area) and well as private schools in grading         Complete questionnaires (instrument Q) teaching public schools (Gradet Pretoria Area) and well as private schools in gradet pretoria Pretoria Area) and (Gradet Pretoria Area) and gradeting public schools (Gradet Pretoria Area) and assessed by memori leador in a shoolino regarding the tree assessed by memori leador in a shoolino frequenties (Gradet Pretoria Area) and (Gradet Pretoria Area) and teacher by student teachers school de by student teachers activity teachers withing practice         Will respond to the questions (Gradet Pretoria Area) and (Gradet Pretoria Area) and (Gradet Pretoria Area) and (Gradet Pretoria Area) and (Instrument E) of the student teacher leador in a shool principal regarding for or his ophilon regarding the teachers school de able to constrait when visiting actional frequenties and the able to activity and (Instrument E) with teachers and (Instrument E) with teachers and (Instrument E) with teachers and (Instrument E) of the student teacher and teachers as and (Instrument E) of the student teacher and teachers as and (Instrument E) of the student teacher (Instrument E) of the student (Instrument E) with t							
Parents         Parents         Parents         Parents         Complete questionnaires (Instrument D Gauteng public schools (Greater Pretoria Area) their percoptions of effective teachers will as private schools in the same area whom gassed by mentor         Complete questionnaires (Instrument D Complete questionnaires (Instrument D Complete questionnaires (Instrument D Based Pretoria Area) and Cauteng Department of Education Public schools (Greater Pretoria Area) attended by student teachers during teaching         Complete questionnaires (Instrument D Complete questionnaires (Instrument D Based Pretoria Area) attended by student teachers should be able to demonstrate whom visiting schools for the first time.           Student Teacher Leaders         Low         Gauteng Pretoria Area) attended by student teachers should be able to demonstrate.           Student Teacher Leaders         Low         University of Pretoria JNM 410 and JNM 451 students         Will conduct one short interview winn visiting schools for the first time.           VI Leaders         Low         University of Pretoria JNM 410 and JNM 451 students         Will conduct one short interview winn visiting schools for the first time.           VI Leaders         Low         University of Pretoria JNM 410 and JNM 451 students         Will conduct one short interview winn visiting schools for the first time.           VI Leaders         Low development of sancetersitions supervise(s) but should not be signed by the participants wite supervise(s) but should not be signed by the participants wite supervise(s) but should not be signed by the participants.           Attacher teachers         Educored que	6 1	Mentor Teacher	's Low	Gauteng Department of Education Public schools (Greater Pretoria Area) an Gauteng Private Schools (Greater Pretoria Area) attended by student teachers during teaching practice	Complete questionnaires (semantic differential scale) regading their opinions of effective schools (Instrument B), teachers (Instrument A) and student teachers (Instrument C). The focus is not on a specific school, a specific student teacher or an an		
Principals         Low         Gauteng Department of Education Public schools Greater Pretoria Area) and Gauteng Private Schools (Greater Pretoria Area) and Gauteng Private Schools (Greater Pretoria Area) attended by student teachers should be able to demonstrate when visiting schools for the first time.           Student Teacher Leaders         Low         University of Pretoria JNM 410 and JNM 451 student. 410 and JNM 451 students.         Will respond to the questions (Instrument E) with the school principal regarding her or his opinion regarding there on his opinion regarding her or his opinion regarding the respondent of the school principal regarding her or his opinion regarding the respondent of the school principal regarding her or his opinion regarding the respondent of the school principal regarding her or his opinion regarding the respondent of the school principal regarding her or his opinion regarding the first time.           Image: the school principal regarding her or his opinion regarding the respondent of the school principal regarding her or his opinion regarding the respondent of the school principal regarding her or his opinion regarding the respondent of the school principal regarding her or his opinion regarding the respondent of the school principal regarding her or his opinion regarding the respondent of the school principal students any or respondent of the supervisor(s) but stoud not be signed by the participants.           Image: the school principal respondent of the school principal supervisor(s) but school not be signed by the participants with supervisor(s) but school not be signed by the participants with supervisor(s) but school not be signed by the participant with supervisor(s) but school not be signed by the school in the Greater Principal supervisor(s) but school not be signed by the principal supervisor school by the school in th		Parents	Low	Parents of learners taught by student teachers during teaching practice in Gauteng public schools (Greater Pretoria Area) as well as private schools in the same area where student teachers will be assessed by mentor lecturers.	Complete questionnaires (Instrument D - semantic differential scale) regading their perceptions of effective teachers. The focus is not on a specific teacher but on teachers in general.		
Student Teacher Leaders       Low       University of Pretoria JNM 410 and JNM 451 students.       Will to conduct one short interview (Instrument E) with the school principal regarding her or his ophion regarding the three most important chararacteristics students teachers should be able to emonstrate when visiting schools for the first time.         Image: Content of Superior Superior Superior Superior Content of Superior Content of Superior Content of Superior Content of Superior Content of Superior Superior Superior Superior Superior Superior Superior Superior Superior Superior Superior Superior Superior Superior Superior Superior Superis Superis Superior Superior Superior Superior Superio		Principals	Low	Gauteng Department of Education Public schools (Greater Pretoria Area) and Gauteng Private Schools (Greater Pretoria Area) attended by student teachers during teaching practice	Will respond to the questions (Instrument E) of the student teacher leader in a short interview regarding her or his opinion regarding the three most important characteristics students teachers should be able to demonstrate when visiting schools for the first time		
Please ensure that you attach to this application a draft letter of invitation to participate in the research on a content of issues outlined in this application. The letter of invitation must be signed by the students and supervisor(s) but should not be signed by the participants yet.     S. 2. Describe your sampling procedure. Include how you will recruit and select participants. Attach as addenda any draft versions of adverts/letters inviting participation in your project.     Al student teachers listed as Block A (Second quarter) and Block B (Third quarter) undergoing their teaching pratice in public GDE schools and private schools in the Greater Pretoria Area will form part of the students had been done in advance to match the availability of mentor lecturers able to visit and assess them during teaching practive.     S. 2. Provide additional information regarding the criteria that will be used as the basis for inclusion/exclusion of certain participants.     Students excluded from the study will be those students who will not be joined by their mentor lecturers inviting students who will not be joined by the interview schedules for processing. They don't do JNM 410 and therefore don't have to compile a final phase (Phase 5) of the investigation. Many mentor lecturers is a final phase (Phase 5) of the investigation. Many mentor lecturers is a final phase for processing. They don't do JNM 410 and therefore don't have to compile a final phase phase.	Student Teacher Leaders		Low	University of Pretoria JNM 410 and JNM 451 students.	Will to conduct one short interview (Instrument E) with the school principal regarding her or his opinion regarding the three most important characteristics students teachers should be able to demonstrate when visiting schools for the first time.		
inclusion/exclusion of certain participants. Technically no fourth-year student will be excluded from the first and third phases of the study (observation of classroom practice and observation of mentor teacher's classroom practice). The only fourth-year students excluded from the study will be those students who will not be joined by their mentor lecturers indicated that they would wish not to form part of the investigation. Many mentor lecturers have will only be collecting data during the second quarter of 2013 and return the completed questionnaires and report.	3.2 All tea res stut the	<ul> <li>P letterhea content of is supervisor(s)</li> <li>2 Describe your sau addenda any draf student teachers in pub ieach sample. Could dents had been do m during teaching p</li> <li>Please provide add</li> </ul>	d for each group ssues outlined in but should not b mpling procedu ft versions of ad listed as Block olic GDE school d be regarded a ne in advance ractive.	re. Include how you will recruit a gaptication. The letter of in e signed by the participants yet. re. Include how you will recruit a lverts/letters inviting participation A (Second quarter) and Blo s and private schools in the Gr s purposive sampling and conv to match the availability of me	of invitation to participate in the research on a a sure that the content of the letter reflects the vitation must be signed by the student and and select participants. Attach as <u>n in your project.</u> ck B (Third quarter) undergoing their eater Pretoria Area will form part of the renient sampling as the grouping of the ntor lecturers able to visit and assess		
7	Tech of c stude (subj indica will o interv repor	inclusion/exclusion hnically no fourth-ye lassroom practice ents excluded from ject methodologists) ated that they would only be collecting da view schedules for t.	adorial information of certain partial of certain partial information observation and observation the study will during the finate wish not to for the during the set processing. The processing of the set of th	ion regarding the criteria that w cipants. be excluded from the first and it on of mentor teacher's classen be those students who will no al phase (Phase 5) of the invest m part of the investigation this accord quarter of 2013 and retu ey don't do JNM 410 and the	ill be used as the basis for third phases of the study (observation pom practice). The only fourth-year t be joined by their mentor lecturers tigation. Many mentor lecturers have year. The JNM 451 pipeline students rn the completed questionnaires and refore don't have to compile a final		
				7			



APPENDIX D: UNIVERSITY OF PRETORIA BED PROGRAMME INFORMATION

# **Faculty of Education**

Fakulteit Opvoedkunde Lefapha la Thuto



# 2016 BEd PROGRAMME INFORMATION



### GENERAL INFORMATION

### REGULATIONS AND CURRICULA

The rules for the degrees published here are subject to change and may be amended prior to the commencement of the academic year in 2016.

The General Regulations (G Regulations) apply to all faculties of the University of Pretoria. It is expected of each student to familiarise himself or herself well with these regulations. Ignorance concerning these regulations will not be accepted as an excuse for any transgression.

### 1 Admission to degree studies

### 1.1 General

- 1.1.1 To register for a first bachelor's degree at this University, a candidate must in addition to the required National Senior Certificate with admission to degree studies, comply with the specific admission requirements for particular modules and fields of study as prescribed in the admission regulations and the faculty regulations of the departments.
- 1.1.2 The following persons may also be considered for admission:
  - A candidate who is in possession of a certificate which is deemed by the University to be equivalent to the required National Senior Certificate, with admission to degree studies.
  - A candidate who is a graduate from another tertiary institution or has been granted the status of a
    graduate of such an institution.
  - A candidate who completed the Institutional Proficiency Test successfully, as prescribed by the University.
- 1.1.3 Senate may limit the number of students allowed to register for a module/ programme, in which case the Dean concerned may, at her/his own discretion, select from the students who qualify for admission those who may be admitted.
- 1.1.4 Subject to faculty regulations a candidate will only be admitted to postgraduate bachelor's degree studies, if he or she is already in possession of a recognised bachelor's degree or equivalent qualification.

### 2 Registration for a particular year of study

At the beginning of an academic year, a student registers for all the modules he or she intends taking in that particular year (whether these be first semester, second semester or year modules). A student may also only register for modules that will fit in the lecture, test and examination timetables.

### 3 Examination admission and pass requirements

A sub-minimum of 40% is required for the year and/or semester mark for admission to the examination in each module. A student who obtains a final mark of 40 – 49% in a module qualifies for a supplementary examination. If a pass mark has been obtained in a module, but the required sub-minimum of 40% has not been obtained in the examination, the student will have to write a supplementary examination. A final mark of at least 50% is required to pass a module.

### 3.1 Examinations

The examinations for first semester modules take place in May/June, while all other examinations (second semester modules and year modules) take place in October/November.

### 3.2 Ancillary examinations

After completion of an examination and before the examination results are published, the examiners may summon a student for an anciliary examination on particular aspects of the work of that module.

### 3.3 Aegrotat/extraordinary examinations

Students, who do not write their examinations on the scheduled day, may apply for an aegrotat /extraordinary examination at the Student Administration Offices. Lecturers are not allowed to grant any permission for this category of examination. It is the responsibility of the student to ascertain whether his/her request has been successful. If permission has been granted, the student must write the aegrotat/extraordinary examination during the supplementary examination. Such a student will not quality for a further supplementary examination.

Application for the above examination must be handed in at the Student Administration Offices together with a valid medical certificate not later than three working days after the module should have been written. A student who has been granted permission to write an aegrotatiextraordinary examination and who then fails to write the examination will not qualify to submit any such application at a later stage. A medical certificate will not be accepted where it states that a student appeared III or declared him-/herself unit to write the examination.



### 1.5 Degree with distinction

### 1.5.1 BEd Foundation Phase teaching

The degree is conferred with distinction to a student who obtains an overall weighted average (GPA) of 75% or higher in the BEd programme, with the condition that the degree is completed in the prescribed 4 years.

### 1.5.2 BEd Intermediate Phase teaching

The degree is conferred with distinction to a student who obtains an overall weighted average (GPA) of 75% or higher in the BEd programme, with the condition that the degree is completed in the prescribed 4 years.

### 1.5.3 BEd Senior- and Further Education and Training Phase teaching

The degree is conferred with distinction to a student who obtains an overall weighted average (GPA) of 75% or higher in the BEd programme, with the condition that the degree is completed in the prescribed 4 years.

### 2 Packages In the BEd programme

### 2.1 Foundation Phase Teaching (Code 09133011)

Package coordinators: Prof JC Joubert and Dr JC van Heerden Tel: 012 420 5636/5563, email: Ina.joubert@up.ac.za and judy.vanheerden@up.ac.za

Madula desoriation	Madula aada	Cree	Credits							
Module description	Module code		Y2	Y3	¥4					
Fundamental modules										
Academic Information management	AIM 101 or	6								
-	AIM 111, 121	8								
Literacles in education	JLZ 110, 120 or									
	JLZ 111, 121	12								
First aid	JNH 454				3					
Core modules										
Education	OPV 112, 122	24								
	OPV 212, 222		40							
	OPV 312, 322			60						
Literacies in education	JLZ 300			12						
Research project	JNM 461, 464				24					
Teaching practice	PRO 280		6							
	PRO 380			6						
	PRO 452, 453				56					
Health and safety	JGV 210		6							
Foundation phase mathematics	JGS 121	6								
	JGS 211		12							
	JGS 212		12							
Literacy practices	JGL 110	6								
	JGL 200		24							
	JGL 311			6						
	JGL 461				6					
	JGL 464				6					
Human movement studies	JMB 124	6								
Learning support	JLD 220		12							
ECD-studies	JVK 130	12								
Professional practice	JEP 111	6								
	JFP 451				12					
Arts and culture	JLK 110, 120	12								
NS and technology	JST 320			12						
Life skills programme	JLP 220		12							
Methodology of learning support	JMD 351			6						
ECD-studies or	JVK 400 or				24					
Learning support	JLD 400				24					

---00000----